CHARACTERISATION OF STRESS RESPONSES IN *LACTOBACILLUS PARACASEI* AND *BIFIDOBACTERIUM ANIMALIS* (SYN. *LACTIS*)



# SCHOOL OF MOLECULAR SCIENCES VICTORIA UNIVERSITY

FOR THE DEGREE OF DOCTOR OF PHILOSOPHY

2004

ΒY

SUSAN JESSIE PEPPER

WER THESIS 660.62 PEP 30001008546253 Pepper, Susan Jessie Characterisation of stress responses in Lactobacillus paracasei and

## DECLARATION

I certify, that unless otherwise stated, the work presented in this thesis is solely my own work, and has not been submitted previously, in whole or in part, for any other academic award. The results detailed in Chapters 3 and 4 have been previously presented in the following:

Pepper, S. J., N. Shah and M. L. Britz. 1995. A rapid and simple method for comparative identification of lactic acid bacteria using microtitre plates. Abstracts, International Dairy Lactic Acid Bacteria Conference, Palmerston North, New Zealand. Page 7.13.

Pepper, S. J., M. L. Britz and P. Chambers. 1996. Approaches to purification of acid stress proteins in *Lactobacillus acidophilus*. Abstract, Chromatography 96 Sydney.

Pepper, S. J., M. L. Britz and P. Chambers 1997. Changes in protein profiles and characterisation of up-regulated proteins of *Lactobacillus acidophilus* grown in sub-optimum conditions. Abstract, Australian Society for Microbiology, Annual Scientific Meeting, Adelaide, 1997. Page A84

S. J. Pepper. 1997. Use of fatty acid methyl esters as part of the identification of *Lactobacillus acidophilus*. Abstract, Australian Society for Microbiology, as part of workshop entitled 'Approaches to Identification and Classification of Lactic acid bacteria', Adelaide, 1997. Page A131

S. J. Pepper. 1997. SDS-PAGE fingerprinting of lactic acid bacteria. Abstract, Australian Society for Microbiology, as part of workshop entitled 'Approaches to Identification and Classification of Lactic acid bacteria', Adelaide, 1997. Page A131



### ABSTRACT

## Characterisation of stress responses in Lactobacillus paracasei and Bifidobacterium animalis (syn. Lactis)

by

#### Susan Jessie Pepper

This thesis describes a study of the nature of the responses to stress, particularly acid stress, in *Lactobacillus paracasei* and *Bifidobacterium animalis* (syn. *lactis*). The aims of the study were to validate the identification of the bacteria being studied, to determine the nature of their stress responses and compare them to known responses and to gain information regarding the proteins up-regulated during stress.

Nineteen strains of lactic acid *:* acteria were tested in terms of their fatty acid methyl ester (F sME), sodium dodecyl sulphate-poly acrylamide gel electrophoresis (SDS-PAGE) and carbohydrate utilisation profiles to provide tools to differentiate these strains. However, these tests allowed only low levels of discrimination. Molecular methods were then used to identify one of the LAB characterised above, VUP 12006, which was later used for detailed analysis of acid stress responses. On the basis of 16S rRNA gene analysis, it was concluded that VUP 12006 was a strain of *Lb. paracasei*. Strain VUP 12006 was found to have an optimum growth temperature range of 37°C ± 0.5 and an optimum pH of 6.3 ± 0.2. Strain VUP 12006 was subjected to pH and temperature stress by culture under conditions that decreased the  $\mu_{max}$  by 50% (pH 4.3/37°C and pH 6.3/45°C). Under acid stress, a 42 kDa protein, which was later found to be membrane- or cell-surface-associated, was observed to be up-regulated. Under heat stress, two proteins of mass 60 kDa and 70 kDa, which were up-regulated four-fold during growth at 45°C, were shown to cross react with antibodies made against *E. coli* HSP60 and *E. coli* HSP70. The N-terminal sequence of the 42 kDa protein was determined and found to share 100% homology with a protein sequence found in the draft genome of *Lb. casei* ATCC 334. PCR-based techniques were used to determine

III

the nucleotide sequence of the gene known to encode the 42 kDa AphA protein. The ORF was 1263 nucleotides in length with a G+C content of 47.66% and the translated amino acid sequence of AphA was shown to be rich in serine (15.4%) and alanine (13.3%). A 28 amino acid signal peptide was predicted by SPScan and the cleavage site was confirmed by the N-terminal sequence obtained for this protein. The protein was found to share homology with GSP-781, a glucan B binding protein identified in *Streptococcus mutans*. Although it has been suggested that GSP-781 may be associated with the maintenance of the stability of the cell wall or cell membrane, the biological functions of both GSP-781 and AphA are yet to be determined.

Based on a long history of use as a starter culture, strain VUP Bb 12 was selected for characterisation of stress responses. Identification of the test strain was confirmed as *B. animalis* (syn. lactis) through successful PCR amplification. Strain Bb 12 was determined to have an optimum growth temperature range of  $42^{\circ}C \pm 0.2$  and an optimum pH of  $6.3 \pm 0.2$ . The response to acid stress in strain VUP Bb 12 was studied using 1-D and 2-D SDS-PAGE. Strain VUP Bb 12 was cultured at 42°C and pH values fixed at pH 4.0, 5.0, 6.0 or 7.0. Protein profiles for growth at neutral pH (7.0) and mildly acid pH (6.0) were essentially similar. At pH 5.0 an increase was noted in the density of a band of around 33 kDa. At pH 4.0 a protein band of similar size was upregulated over 10-fold over twenty-four hours growth. This 33 kDa protein was later found to be membrane- or cell surface-associated. The N-terminal sequence of this protein could not be determined experimentally, possibly due to the presence of carbohydrate or lipid. The function of this protein also remains to be determined. Changes in protein profiles during acid stress were observed in cytosolic as well as non-cytosolic proteins. Strain VUP Bb 12 was grown at pH 6.3 and 42°C and challenged by a sudden change in pH (to pH 4.3). 2-D SDS-PAGE gels were compared to confirm the presence of proteins that were up-regulated during growth at pH 4.3. N-terminal sequence data was obtained for four of these, namely Bsb-5, Bsb-6, Bsb-7 and Bsb-8, two of which were identical (Bsb-5 and Bsb-6). A database search revealed a match to two of these proteins whose function was found to be related to ribosome metabolism. The possible function of Bsb-6 was related to elongation factor-P and Bsb-8 to a 50S ribosomal protein L7/L12. The remaining protein (Bsb-7) was a hypothetical protein of unknown function. In conclusion, both species were able to respond to acid stress with novel responses which were physiologically guite different between the species studied.

IV

# TABLE OF CONTENTS

Declaration	
Abstract	
Table of Contents	V
List of Tables and Figures	XI
List of Appendices	XVII
Acknowledgements	XVIII
Abbreviations	XIX

## CHAPTER ONE

GENERAL INTRODUCTION1	
1.1 Probiotic foods1	
1.1.1 An introduction to probiotic foods	1
1.1.2 Inhibition of food spoilage organisms	3
1.1.3 Probiotics: the concept	4
1.1.4 Probiotics: definitions of probiotic, prebiotic and synbiotic	5
1.1.5 Probiotics: criteria for selection for use	6
1.1.6 Potential benefits of the consumption of probiotic foods	7
1.2 Lactic acid bacteria11	
1.2.1 Introduction to LAB	12
1.2.2 The genus Lactobacillus	15
1.2.3 Genus and species characteristics	16
1.2.4 An early history of the identification of Lactobacillus species	18
1.2.5 Recent history of Lactobacillus identification	19
1.2.6 The genus Bifidobacterium	22
1.2.7 History the of genus Bifidobacterium	24
1.2.8 Species characteristics	24
1.3 Probiotic Lactic Acid Bacteria	
1.3.1 Survival of LAB in yoghurt	28
1.3.2 Survival of LAB in the digestive tract of humans	29

1.4 Bacterial responses to growth in stressing environments	
1.4.1 Bacterial response to growth in stressed conditions	30
1.4.2 Protein synthesis and folding in a normal cellular environment	30
1.4.3 Stress responses	33
1.4.4 Mechanisms of signalling sub-optimum growth conditions	34
1.4.5 General responses to cell stress	36
1.4.6 Chaperone families and the cellular response to heat stress	37
1.4.7 Interactions between chaperone proteins	54
1.4.8 Oxidative stress response	55
1.4.9 Mechanisms of acid tolerance and acid stress responses	57
1.4.10 Proteases	65
1.4.11 The effect of stressors on cell membrane fatty acid composition	66
1.4.12 General and specific stress response of Bacillus subtilis	68
1.5 Interaction of chaperones, regulation of the stress response and the effect of cross protection on survival	
1.5.1 Regulation of the heat shock response	69
1.5.2 Regulation of stress responses other than heat	70
1.5.3 The effect of cross protection on survival responses	72
1.6 Aims of this thesis74	

# CHAPTER TWO

2.1 Materials	5
2.1.1 Bacterial strains and plasmids	75
2.1.2 Chemicals	75
2.2 General methods7	5
2.2.1 Buffers	78
2.3 Microbiological methods7	8
2.3.1 Media	78
2.3.2 Preservation of cultures	80
2.3.3 Gram stain	81
2.3.4 Method of biomass determination	82
2.3.5 Carbohydrate utilisation test (rapid test)	82

2.3.6 Selection of acid tolerant strains	85
2.3.7 Fermenter operation	85
2.3.8 Determination of growth rates	86
2.3.9 Determination of factors that affect growth	87
2.4 Analytical methods	
2.4.1 Sample collection and treatment	89
2.4.2 Analysis of carbohydrates and metabolic products	92
2.4.3 Estimation of protein concentration (Lowry assay)	93
2.4.4 Fatty acid methyl ester (FAME) profiles	94
2.5 Protein separation methods94	
2.5.1 Polyacrylamide gel electrophoresis (PAGE)	94
2.5.2 Polyacrylamide gel electrophoresis (2-D)	98
2.5.3 Visualisation of protein bands on polyacrylamide gels	99
2.5.4 Western blotting	101
2.5.5 Isoelectric focussing (IEF) using a Rotofor cell	103
2.5.6 Continuous elution electrophoresis	104
2.5.7 N-terminal sequencing	106
2.5.8 Analysis of protein sequences	106
2.6 Molecular biology methods106	
2.6.1 Preparation of plasmid DNA from bacterial cells	106
2.6.2 Quantification of nucleic acids	107
2.6.3 Agarose gel electrophoresis	108
2.6.4 Preparation of plasmid vector for cloning	108
2.6.5 Preparation of competent bacterial cells	110
2.6.6 Construction and design of oligonucleotide probes	110
2.6.7 Construction of representative genomic DNA library	113
2.6.8 Colony blotting	114
2.6.9 Preparation of labelled oligonucleotide probes used in hybridisation procedures	115
2.6.10 Removal of unincorporated label	116
2.6.11 Hybridisation of oligonucleotide probes	116
2.6.12 DNA amplification	117
2.6.13 Sequencing of DNA	117
2.6.14 Analysis of DNA sequences	118

2.6.15 Bioinformatic analysis of proteins	118
2.7 Photography	. 119
2.7.1 Photography of SDS-PAGE and agarose gels	119
2.7.2 Still Photography	119
2.7.3 Exposure of hybridised membranes to X-ray film	119

# CHAPTER THREE

3.1 Introduction120	
3.2 Identification of test strains	
3.2.1 Gram stain and growth on MRS agar 1	21
3.2.2 Carbohydrate utilisation 1	22
3.2.3 Fatty acid methyl ester profile 1	23
3.2.4 SDS-PAGE profiles (fingerprinting of whole cell protein patterns) 1	31
3.2.5 Molecular methods of identification 1	31
3.3 Physiological characterisation of the selected strains	
3.3.1 Optimum temperature and pH for growth for strain VUP 12006 1	48
3.3.2 Growth characteristics under optimum conditions (pH controlled at pH 6.3) 1	52
3.3.3 Growth characteristics for cultures commencing at optimum pH but without continued pH control 1	56
3.3.4 Effect of lactic acid concentration on growth 1	56
3.3.5 Survival in extreme pH 1	60
3.4 Effects of elevated temperature and low pH on growth 161	
3.4.1 Growth rate at 45°C or pH 4.0 of strain VUP 12006	64
3.4.2 Changes in protein concentration of cells grown at 45°C or at pH 4.0 1	66
3.4.3 Changes in metabolite concentration of cells grown at 45°C or at pH 4.0	66
3.4.4 Effect of growth at low pH and elevated temperature on protein profiles (1-D SDS-PAGE)	69
3.4.5 Effect of growth at low pH and elevated temperature on immunoblot reactions 1	73
3.4.6 Effect of acid and temperature shift on FAME profiles 1	75
3.4.7 Effect of change in pH during growth on the expression of the 45kDa protein 1	76
3.5 Protein profiles of other Lactobacillus strains when grown at different pHs 178	
3.5.1 Growth at pH 4.0 or shift from pH 6.3 to 4.0 during growth 1	78
3.6 Partial purification and identification of acid stress protein of VUP 12006 183	

	3.6.1 Partial purification using native PAGE and IEF (Prep and Rotofor cells)	185
	3.6.2 Purification by denaturing electrophoresis and N-terminal sequencing	185
	3.6.3 Construction of representative genomic library and screening for gene encoding the 42 kDa protein	188
	3.6.4 Investigation of the <i>Lb. casei</i> ATCC 393 draft genome sequence for 42kDa acid stress protein N-terminal sequence	190
	3.6.5 PCR based approach to identification of the 45kDa protein	191
	3.6.6 Anchored PCR	198
	3.6.7 Downstream PCR	198
	3.6.8 Submission of sequence data to ANGIS for analysis	201
3.7	Discussion	
	3.7.1 Identification of test strains	215
	3.7.2 Characterisation of Lb. acidophilus strain VUP 12006	222
	3.7.3 Response of Lb. acidophilus strain VUP 12006 to challenge at lethal pH	229
	3.7.4 Comparison of test strains; the response to acid stress	230
	3.7.5 Lb. paracasei strain VUP 1200; structure and function of AphA	230
3.8	Conclusion	

# CHAPTER FOUR

4.1 Introduction	.236	
4.2 Identification of test strain	.237	
4.3 Growth kinetics, protein profiling and effect of elevated temperature, low pH and aeration on growth curves of strain VUP Bb 12	ł 242	
4.3.1 Optimum temperature and pH for growth		244
4.3.2 Observations of protein profiles of Bb 12 cultures grown between 22°C and 47°C		244
4.3.3 Observations of protein profiles of Bb 12 cultures grown between pH	4 and pH	7 246
4.3.4 Observations of protein profiles of Bb 12 cultures without pH control		252
4.3.5 Observations of protein profiles of strain VUP Bb 12 cultures grown in a fermenter sparged with instrument air		252
4.3.6 N-terminal sequencing		255
4.3.7 Effect of heat, acid and oxidative stress on growth curves of strain VU	JP Bb 12	255
4.4 Analysis of physiological and biochemical changes in <i>B. animalis</i> (syn. lactistrain VUP Bb 12 following growth in different pH conditions	is) 258	

4.4.1 Growth characteristics of strain VUP Bb 12 following sudden decrease in pH or continuous exposure to low pH	258
4.4.2 Effect of growth at low pH on protein profiles: 2-D SDS-PAGE profiles	270
4.5 Discussion	
4.5.1 Characterisation of <i>B. animalis</i> (syn. lactis) strain VUP Bb 12	310
4.5.2 Characterisation of <i>B. animalis</i> (syn. lactis) strain VUP Bb 12 during acid stress	319
4.6 Conclusion	
REFERENCES	

# LIST OF TABLES AND FIGURES

Number	Page
Tables	
Table 1.1 Examples of food, feed products and beverages produced using LAB        and the typical strains used in their manufacture	2
Table 1.2 General features of the genus Lactobacillus	17
Table 1.3 List of species of the genus Bifidobacterium and origin	23
Table 1.4 General features of the genus Bifidobacterium	25
Table 1.5 Important characteristics of B. animalis	27
Table 1.6 Stress induced protein (including chaperone families)	39
Table 2.1 Source of bacterial strains and plasmids used in this study	76
Table 2.2 Carbohydrates used in rapid testing	83
Teble 2.3 Stock solutions for SDS-PAGE	95
iable 2.4 Volume of stock reagents for SDS-PAGE	97
Table 2.5 Solutions used for Coomassie blue stain of SDS-PAGE gels	100
Table 2.6 Solutions used for silver staining of SDS-PAGE gels	100
Table 2.7 Stock solutions used in Western blotting technique	102
Table 2.8 Solutions used for electro-blotting proteins to PVDF membrane	105
Table 2.9 Reagent mixture for preparation of plasmid vector prior to cloning	109
Table 2.10 Composition of oligonucleotide probes used in the identification of LAB	111
<b>Table 2.11</b> Composition of oligonucleotide probes used for amplification of gene products or probing genomic DNA libraries	
<b>Table 3.1</b> Carbohydrate utilisation pattern in the type strain and 7 putative        strains of Lb. acidophilus determined using a rapid microtitre plate method	112 124
<b>Table 3.2</b> Fatty acid methyl esters present in the type strain and 7 presumptive <i>Lb. acidophilus</i> strains and two other lactobacillus strains ( <i>Lb. jugarati</i> and <i>Lb. casei</i> )	128

Table 3.3 Composition of the oligonucleotide primers (Genus level identification)	133
Table 3.4 Composition of the oligonucleotide primers (Species level identification)	134
Table 3.5 The best hit of sequences showing significant alignment	138
<b>Table 3.6</b> Maximum specific growth rate and pH growth range of test strains        determined from growth in microtitre tray cultures	149
Table 3.7 Percentage difference between minimum and maximum specific growth        rates achieved during growth at each different temperature between 12 and 51°C	153
<b>Table 3.8</b> Percentage difference between minimum and maximum specific growth        rates achieved during growth at different pH values between 2 and 8	153
Table 3.9 The effect of acid and temperature shift on FAME profile	176
Table 3.10 PCR primers used in round one of gene identification	191
Table 3.11 PCR primers used in round two of gene identification	195
Table 3.12 Expected and observed PCR product sizes from round two of gene identification	197
Table 3.13 PCR primers used for anchored PCR	198
Table 5 14 Composition of PCR primers used for third round of PCR	199
Table 3.15 Expected size of PCR products	199
Table 3.16 Count and percentage of amino acids in ORF sequence	207
Table 3.17 Signal sequence for proteins determined to be functionally similar to AphA	212
Table 3.18 Summary of the identification procedures used in this study	223
Table 4.1 Primers used for identification of strain VUP 12006	237
Table 4.2 Characteristics of Bifidobacterium animalis (syn. lactis) strain VUP Bb 12	243
Table 4.3 Solutions used for protein extraction from cell homogenates of strain VUP Bb12	274
<b>Table 4.4</b> Physical properties of solvents used to extract proteins from homogenates        of strain VUP Bb 12	274
Table 4.5 Molecular weight and pl of protein up-regulated during acid stress	300
<b>Table 4.6</b> Results of N-Terminal sequencing of cytosolic proteins up-regulatedduring growth of strain Bb 12 at pH 4.3	311

Table 4.7 Comparison of optimum growth temperature and pH to published values	312
Figures	
Figure 1.1 Overview of the classification of LAB	14
Figure 1.2 Model for GroEL-GroES-mediated folding reaction	44
<b>Figure 1.3</b> The proposed role of chaperone HSP70 (DnaK), co-chaperones DnaJ and GrpE and the chaperonins (HSP60 and HSP10) in protein folding in <i>E. coli</i>	49
Figure 1.4 Pathways of arginine catabolism	65
Figure 3.1 Examples of a standard profile obtained using bacterial fatty acid methyl esters (Sigma) and a FAME profile obtained during testing	126
<b>Figure 3.2</b> 1-D SDS-PAGE profiles of presumptive <i>Lb. acidophilus</i> strains using whole cells for protein preparation	132
<b>Figure 3.3</b> PCR amplification of DNA from <i>Lactobacillus</i> strain VUP 12006, <i>Lactobacillus acidophilus</i> ASC-388 (type strain) and <i>Lactobacillus casei</i> GCRL-163 using primers LbLMA1-rev and R16-1	133
<b>Figure 3.4</b> PCR amplification of DNA from <i>Lactobacillus</i> strain VUP 12006, <i>Lactobacillus acidophilus</i> ASC-388 (type strain) and <i>Lactobacillus casei</i> GCRL-163 using primers PAF and 536R	137
Figure 3.5 ClustalW alignment of 16S sequences of three strains of LAB (forward primer)	139
Figure 3.6 ClustalW alignment of 16S sequences of three strains of LAB (reverse primer)	141
Figure 3.7 Alignment of the V1 region of the 16S rRNA genes of <i>Lb. casei</i> ATCC 393 and 334, <i>Lb. paracasei</i> ATCC 25302 and strain VUP 12006	143
<b>Figure 3.8</b> PCR amplification of DNA from <i>Lactobacillus</i> strain VUP 12006, <i>Lactobacillus acidophilus</i> ASC-388 (type strain) and <i>Lactobacillus casei</i> GCRL-163 using primers pairs ACI 16S 1 and ACI 16S 2 and primers pairs PR 1 and CAS 2	145
<b>Figure 3.9</b> Computer screen captured images from the software program used to calculate specific growth rates for <i>Lb. paracasei</i> strain VUP 12006 grown in MRS broth at pH 6.3 and 37°C	150
Figure 3.10 $\mu_{max}$ plotted against temperature (A) and pH (B)	151
<b>Figure 3.11</b> Growth characteristics of <i>Lb. paracasei</i> strain VUP 12006 under optimum conditions measured by changes in OD <sub>600</sub> , biomass, glucose concentration and addition of sodium hydroxide	154

<b>Figure 3.12</b> Growth characteristics of <i>Lb. paracasei</i> strain VUP 12006 grown without pH control, starting at optimum pH measured by changes in OD <sub>600</sub>	157
Figure 3.13 The effect of lactate concentration on the growth of <i>Lb. paracasei</i> strain VUP 12006	159
Figure 3.14 The effect of exposure to very low pH on the survival of <i>Lb. paracasei</i> strain VUP 12006	162
Figure 3.15 Relationship of growth phase to acid resistance	163
<b>Figure 3.16</b> Effect of acid and high temperature on strain VUP 12006 measured by changes in optical density	165
<b>Figure 3.17</b> Effect of acid and high temperature on strain VUP 12006 measured by changes in protein concentration	167
<b>Figure 3.18</b> Effect of acid and high temperature on strain VUP 12006 measured by changes in glucose (A) and lactate concentration (B)	168
<b>Figure 3.19</b> 1-D SDS-PAGE of cellular proteins of <i>Lb. paracasei</i> strain VUP 12006 grown at pH 6.3 and 37°C	170
<b>Figure 3.20</b> 1-D SDS-PAGE of cellular proteins of <i>Lb. paracasei</i> strain VUP 12006 grown at pH 4.0 and 37°C	171
<b>Figure 3.21</b> 1-D SDS-PAGE of cellular proteins of <i>Lb. paracasei</i> strain VUP 12006 grown at pH 6.3 and 45°C	172
<b>Figure 3.22</b> Immunoblot to detect regulation of HSP60 and HSP70 analogues during growth at 37°C and 45°C	174
Figure 3.23 The effect of continuous acid shift or spike shift on the regulation of the 45kDa acid stress protein as determined by 1-D SDS-PAGE	179
<b>Figure 3.24</b> Comparison of 1D SDS-PAGE protein profiles of <i>Lactobacillus</i> strains VUP 12006, 12001 and 12007 when grown at different pH values	181
Figure 3.25 Comparison of protein banding patterns between cells prepared three ways	184
<b>Figure 3.26</b> PVDF electro-blotted membrane of 1-D SDS-PAGE showing excision of 45 kDa protein band for sequencing	187
Figure 3.27 Sau3A I digestion of DNA from Lb. paracasei strain VUP 12006	189
Figure 3.28 SPScan of Scaffold 3 of the Lb. casei draft genome	192

Figure 3.29 Lb. casei strain ATCC 334 sequence data obtained from Scaffold 3 of draft genome version '20sept02' showing location of PCR primers	194
<b>Figure 3.30</b> PCR amplification of DNA from <i>Lactobacillus</i> strain VUP 12006 using primers designed from <i>Lb. casei</i> ATCC 334 draft genome	196
<b>Figure 3.31</b> PCR amplification of DNA from <i>Lactobacillus</i> strain VUP 12006 using primers designed from <i>Lb. casei</i> ATCC 334 draft genome	200
Figure 3.32 Consensus sequence of strain VUP 12006	202
<b>Figure 3.33</b> Nucleotide sequence and amino acid sequence from the ORF determined from the consensus nucleotide sequence of <i>Lb. paracasei</i> strain VUP 12006 coding for AphA	206
<b>Figure 3.34</b> ClustalW alignment of the nucleotide and protein sequences comparing the AphA gene of strain VUP 12006 and sequence from Scaffold 3 of <i>Lb. casei</i> ATCC 334	208
<b>Figure 3.35</b> Multiple alignment of the amino acid sequence of some of the functionally similar proteins to AphA gene in strain VUP 12006	213
Figure 3.36 A Kyte and Doolittle plot of hydrophobicity	216
Figure 4.1 PCR amplification of DNA from <i>Bifidobacterium</i> strain VUP Bb 12 using primers Lm 26 and Lm3	238
Figure 4.2 PCR amultification of DNA from <i>Bifidobacterium</i> strain VUP Bb 12 using primers PAF and 536R	240
<b>Figure 4.3</b> Result of Blastn search of GenBank using sequence data from primer pair PAF/536R and DNA from strain VUP Bb 12	241
Figure 4.4 Optimum temperature and pH for growth of <i>B. animalis</i> (syn. lactis Bb 12)	245
Figure 4.5 SDS-PAGE profiles of <i>Bifidobacterium animalis (</i> syn. <i>lactis)</i> strain VUP Bb 12 grown over a range of temperatures	247
Figure 4.6 SDS-PAGE profiles of strain VUP Bb 12 grown over a range of pH levels	249
<b>Figure 4.7</b> Laser densitometry of protein profiles of strain VUP Bb 12 grown at pH 4.0 shown at selected time intervals	251
Figure 4.8 SDS-PAGE profile of strain VUP Bb 12 grown without pH control	253
Figure 4.9 Comparison of <i>Bifidobacterium animalis (</i> syn. <i>lactis)</i> strain VUP Bb 12 grown in low and high oxygen environment	254

Figure 4.10 Effect of acid and heat shift on strain VUP Bb 12 measured by changes in optical density	257
<b>Figure 4.11</b> Optical density changes seen during culture of <i>Bifidobacterium animalis</i> (syn. <i>lactis</i> ) strain VUP Bb 12 at different pH values and pH shift during growth	261
<b>Figure 4.12</b> Viable cell count changes seen during culture of <i>Bifidobacterium animalis</i> (syn. <i>lactis</i> ) strain VUP Bb 12 at different pH values and pH shift during growth	264
Figure 4.13 HPLC standards and samples	267
<b>Figure 4.14</b> Effect of a sudden decrease in pH or constant exposure to acidic pH on glucose (Figure 4.14 A) and lactate concentration (Figure 4.14 B) on <i>Bifidobacterium animalis</i> (syn. <i>lactis</i> ) strain VUP Bb 12	269
Figure 4.15 Comparison of protein banding between samples prepared three ways	272
Figure 4.16 Comparison of detergent extraction methods	275
Figure 4.17 Solvent extraction of proteins from <i>B. animalis</i> (syn. lactis) strain Bb12	278
Figure 4.18 Comparison of proteins extracted in the presence of chloroform and diethyl ether and a mixture of chloroform and diethyl ether	280
<b>Figure 4.19</b> Computer propessed image of two dimensional SDS-PAGE of <i>B. animalis</i> (syn. <i>lactis</i> ) strain VUP Bis 12 grown at pH 6.3 (cleared lysate in Tris)	282
<b>Figure 4.20</b> Computer processed image of two dimensional SDS-PAGE of <i>B. animalis</i> (syn. <i>lactis</i> ) strain VUP Bb 12 grown at pH 6.3 (solvent extracted protein)	284
<b>Figure 4.21</b> Computer processed image of two dimensional SDS-PAGE of <i>B. animalis</i> (syn. <i>lactis</i> ) strain VUP Bb 12 grown at pH 4.3 (protein extracted in the presence of Tris)	286
<b>Figure 4.22</b> Computer processed image of two dimensional SDS-PAGE of <i>B. animalis</i> (syn. <i>lactis</i> ) strain VUP Bb 12 grown at pH 4.3 (solvent extracted protein)	288
<b>Figure 4.23</b> Computer processed image of two dimensional SDS-PAGE of <i>B. animalis</i> (syn. <i>lactis</i> ) strain VUP Bb 12 (protein extracted from freeze dried cells)	292
<b>Figure 4.24</b> Computer processed image of two dimensional SDS-PAGE of <i>B. animalis</i> (syn. <i>lactis</i> ) strain VUP Bb 12 (solvent extracted protein from freeze dried cells)	294
<b>Figure 4.25</b> Computer processed image of two dimensional SDS-PAGE of <i>B. animalis</i> (syn. <i>lactis</i> ) strain VUP Bb 12 grown without pH control (cleared lysate in Tris)	296
<b>Figure 4.26</b> Computer processed image of two dimensional SDS-PAGE of <i>B. animalis</i> (syn. <i>lactis</i> ) strain VUP Bb 12 grown without pH control (solvent extracted protein)	298

<b>Figure 4.27</b> Computer processed image of one dimensional SDS-PAGE of <i>B. animalis</i> (syn. <i>lactis</i> ) strain Bb12 grown with pH control at pH 6.3 (Rotofor fraction 1-12) and 4.3 (Retofor fractions 1-12)	200
(Rotoror fractions 1-13)	302
<b>Figure 4.28</b> Computer processed image of one dimensional SDS-PAGE of <i>B. animalis</i> (syn. <i>lactis</i> ) strain Bb12 grown with pH control at pH 6.3 (Rotofor fraction 13-20) and	
4.3 (Rotofor fractions 14-20)	304
<b>Figure 4.29</b> Computer processed image of two dimensional SDS-PAGE of <i>B. animalis</i> (syn. <i>lactis</i> ) strain Bb 12 grown with pH control at pH 6.3	307
<b>Figure 4.30</b> Computer processed image of two dimensional SDS-PAGE of <i>B. animalis</i> (syn. <i>lactis</i> ) strain Bb 12 grown with pH control at pH 4.3	309

# LIST OF APPENDICES

Nu	mber	Page
A1	List of laboratory instrumentation, equipment and chemicals	358
A2	List of suppliers	365
A3	HPLC trace of a culture grown at pH 6.3 and 37°C taken two hours after ino	culation 367
A4	ORNL microbial blast results of N-terminal sequence of 45 kDa protein of	<i>Lb. paracasei</i> strain
VU	P 12006 against <i>Lb. casei</i> draft genome	368
A5	Blastn (nucleotide) and BlastX (function) searches. Genebank, Swissp	rot and spTrEMBL
sea	rch for similarity to gene encoding AphA protein	370

## ACKNOWLEDGMENTS

The author wishes to thank:

Prof. Margaret Britz for supervision of the project

Assoc. Prof. Mary Millikan

Mohamed Mohideen

Vilnis Ezernieks for technical assistance with GS-MS

Matt Knight and Karoline Tellbach for valuable technical assistance with molecular biology techniques

Oleg Yuriev for technical assistance with protein techniques

Dr Duncan Rouch

Robert Willis for technical assistance with protein techniques in Chapter Four

Bev Burns for assistance with editing the final version of the Thesis

The members of the Royal Yacht Club of Victoria in particular Terri and Colin, Linda and Dave, Bob, Richard and Wanda, for making allowances for me wanting to do something other than sailing and of course my family who can now stop asking "The Question".

This research has been facilitated by access to the Australian Proteome Analysis Facility established under the Australian Government's Major National Research Facilities Program.

# ABBREVIATIONS

ΔΨ	electrical potential
∆рН	pH gradient
AH	acid habituation
AR	acid resistance
ATP	adenosine triphosphate
ATR	acid tolerance response
bp	base pair
BSA	bovine serum albumin
CBFT	Centre for Bioprocessing and Food Technology
Ci	Curie
CIRCE	controlling inverted repeat of chaperone expression
CSIRO	Commonwealth Scientific and Industrial Research
	Organisation
DNA	deoxyribonucleic acid
DNAse	deoxyribonuclease
DNS	3' 5'-dinitrosalicylic acid
dNTP	deoxyribonucleotide
dsDNA	double stranded DNA
DTT	dithiothreitol
EDTA	Ethylenediaminetetra-acetic acid (disodium salt)
F/C	Folin/Ciocalteau
FAME	fatty acid methyl ester
g	acceleration due to gravity
h	hour
HCI	hydrochloric acid
HSP	heat shock protein
IEF	Iso electric focussing
IPG	Immobilised pH gradient
kb	kilobase
kd	kilodalton
kDa	10 <sup>3</sup> Daltons
LAB	lactic acid bacteria
L	litre
L/min	litres per minute
L-B	Luria-Bertani medium
mA	milliampere
min	minute
mL	millilitre
mm	millimetre
mol	mole
MRS	de Mann, Rogosa and Sharpe medium

MW	molecular weight
NaOH	sodium hydroxide
NDO	non-digestible oligosaccharides
OD	optical density
ORF	open reading frame
PAGE	polyacrylamide gel electrophoresis
рН	hydrogen-ion exponent
рН <sub>і</sub>	internal pH
рН <sub>о</sub>	external pH
pl	isoelectric point
PMF	proton motive force
RNA	ribonucleic acid
RNase	ribonuclease
rpm	revolutions per minute
rRNA	ribosomal RNA
RS	reducing sugar
S	second
SDS	sodium dodecyl sulphate
SOS	save our souls
T <sub>m</sub>	melting temperature
TEMED	N,N,N',N'-tetramethylethylenediamine
u.v.	ultraviolet
ν	volt
v/v	volume/volume
VUT	Victoria University of Technology
W	watts
w/v	weight/volume
w/w	weight/weight
X-gal	5-bromo-4-chlor-3-indolyl-β-D-galactopyranoside

# Chapter One

# LITERATURE REVIEW

## **GENERAL INTRODUCTION**

The objective of this project was to gain a basic understanding of how *Lactobacillus paracasei* and *Bifidobacterium animalis* (syn. *lactis*) respond when challenged with the changes in temperature, pH and oxidation levels which the organisms typically experience during the manufacture and consumption of fermented dairy foods. These organisms, which ferment milk during the production of yoghurt and cheese, are exposed to a rigorous manufacturing process as well as a hostile digestive tract before finally colonising or passing through the lower gastrointestinal tract. This involves growing in or surviving a wide range of pH values from alkaline to highly acidic, and temperature values that vary from refrigeration (2–6°C) up to 45°C. A better understanding of how the organisms respond to stress, particularly at a molecular level, will provide a sounder base for the development of starter cultures for improved manufacturing of fermented food products. This chapter therefore provides a review of the organisms and their use in production of dairy foods, including their role in conferring potential health benefits and the known responses of bacteria to stressful environments.

## **1.1 Probiotic foods**

## 1.1.1 An introduction to probiotic foods

Lactic acid bacteria (LAB) have been used to ferment a variety of foods for at least 4,000 years. In particular, *Lb. acidophilus* and *Lb. delbrueckii* subspecies *bulgaricus* have been used in the production of these traditional fermented foods.

1

Table 1.1 Examples of the food, feed products and beverages produced using LAB and the typical strains used in their manufacture (Hammes and

Vogel, 1995; Djordjevic *et al.*, 1997; Wood, 1998)

Fermented Product	Example of Product	Examples of the LAB used to ferment these products
Vegetables	Vegetables are fermented worldwide mostly on a small scale and in household amounts, but others are of significant economic importance, particularly olives, sauerkraut (produced from cabbage) and cucumbers.	Lb. bavaricus, Lb. brevis, Lb. delbrueckii, Lb. plantarum, Lactococcus lactis, Leuconostoc mesenteroides, Pediococcus pentosaceus
Silage fermentation	Silage is a form of conserved forage that has become an important source of animal feed over winter months.	Lb. plantarum, Lb. curvatus
Sourdough bread and related products	A multitude of baked goods can be made using a sourdough process, including panettone, different types of brioche and rye breads.	Lb. pontis, Lb. sanfrancisco
Alcoholic and non-alcoholic beverages	These include the production of wine, beer, saké and distilled spirit. In wine manufacture LAB are used to de-acidify the product through malolactic fermentation.	Lb. saké , Leuconostoc oenos,
Dairy foods	Fermented milk, yoghurt, cheese, kefir, buttermilk and cultured cream represent about 20% of the value of fermented food production worldwide.	Lb. acidophilus, Lb. delbrueckii, Streptococcus species, Lb. casei, Lb. kefir, Lb. helveticus, Lactococcus lactis
Preserved meat	Fermented sausage	Leuconostoc mesenteroides, Lb. casei, Lb. saké, Lb. curvatus
Regional foods	Fermented foods particularly from Africa, central Asia, the orient and the Commonwealth of Independent States (CIS) are an important contribution to regional nutrition. Japan produces many fermented products such as koikuchi shoyu and miso in which LAB contribute to the final product while kimchi is a popular and traditional way of preserving vegetables in Korea.	Principally Lactobacillus species

This practice has improved flavour and texture without adversely affecting other organoleptic properties of the food (Hull *et al.*, 1992) and it has been noted that consumption of LAB has impacted positively on human and animal health (Vandamme *et al.*, 1996). The food industry currently seeks to offer the benefit of consumption of these bacteria as part of an increasing range of functional foods. Functional foods are similar in appearance to conventional foods but when consumed as part of a normal diet demonstrate physiological benefits and/or reduce the risk of chronic disease beyond basic nutritional functions (Scott and Lee, 1996). LAB have many uses in industry and agriculture and examples of these uses are listed in Table 1.1. They contribute to the long shelf life of these products and confer beneficial health effects on consumers by inhibiting the growth of food spoilage organisms and producing probiotic substances. Many of the attributes are due to the production of lactic acid.

### 1.1.2 Inhibition of food spoilage organisms

The basis of the ancient art of food preservation is often directly related to the fermentation activity of LAB. These bacteria can inhibit the growth of food spoilage organisms by several means:

- production of hydrogen peroxide by many starter culture bacteria, including *Lactobacillus delbrueckii* ssp. *bulgaricus*, which is considered to be an important food preservative (Yi *et al.*, 1998). The amount of H<sub>2</sub>O<sub>2</sub> produced depends upon the strain and its production has been shown to inhibit the growth of *Clostridium* species, *Staphylococcus* species and some psychrotrophs (Wood, 1998);
- production of bacteriocins and bacteriocin-like substances (Tagg *et al.*, 1976; Fooks *et al.*, 1999) which are proteins that are antagonistic to a wide range of Gram-positive and some Gram-negative organisms;

- production of lactic acid and other acids (including acetic and formic acid), which reduce the extra-cellular pH below levels at which most neutrophiles can grow and survive (see section 1.1.6. for further discussion of lactic acid). This leads to the collapse of the proton motive force as un-dissociated acids transport protons across the membrane, which results in the acidification of the cytoplasm of food spoilage organisms (Kashket, 1987, Gonçalves *et al.*, 1997) and
- by competing successfully for nutrients so that the overall level of nutrients available to other bacteria is reduced, causing poor growth of food spoilage and pathogenic bacteria (Saloff-Coste, 1997).

#### 1.1.3 Probiotics: the concept

The concept of probiotics (literally meaning 'for life') is that certain bacterial groups, such as *Lactobacillus* and *Bifidobacterium* species, may have health promoting properties. These bacteria may reduce or prevent disease due to their high numbers and metabolic activities (Gibson and Roberfroid, 1995). This concept was first proposed by Elie Metchnikoff, who lived between 1845 and 1916, one of the first researchers into probiotic organisms and their affect on human health. Metchnikoff (1907) studied a group of Balkan people and concluded that their long life span was in part due to the consumption of large quantities of LAB in fermented foods. He also claimed that lactic acid and other metabolites, present in sour milk fermented by LAB, inhibited the growth and toxicity of the anaerobic, spore-forming bacteria commonly found in the bowel and suppressed "unhealthy fermentation in the intestine". Metchnikoff suggested that the microflora of the large bowel must be detrimental to the host and he attributed premature ageing to the circulation of toxic compounds produced by organisms in the intestinal tract and then absorbed into the blood stream.

these toxins. However, Metchnikoff also offered an alternative suggestion of administering lactic acid and/or lactic acid-producing bacteria to achieve the same effect. At that time, it was recognised by some scientists that lactic acid produced by LAB inhibited the growth of non-acid tolerant bacteria, thus replacing toxin-producing bacteria with beneficial bacteria (Tannock, 1995). It must be pointed out, however, that there were some commentators who were less than devoted to Metchnikoff's theories. These included Faust (1910), who found that Metchnikoff's work on LAB and lactic acid was "too heuristic and optimistic to be worth much attention".

## 1.1.4 Probiotics: definitions of probiotic, prebiotic and synbiotic

The term 'probiotic' was originally coined by Lilly and Stillwell (1965) to describe substances produced by one organism that stimulate the growth of another. However, the current use of the term is attributed to Fuller (1992) and is described as "an organism or substance, which contributes to intestinal microbial balance" (Hull et al., 1992). Other authors (Havenaar and Huis in't Veld, 1992) have expanded this description to "a mono- or mixed-culture of live organisms which, applied to man or animal, beneficially affects the host by improving the properties of the indigenous microflora". The latter definition excludes the use of non-digestible oligosaccharides (NDOs) and other food products (nutraceuticals) that have also been shown to exhibit an effect that would be labelled as 'probiotic' under Fuller's definition. More recently Salminen (1999) argued that nonviable cells and cell components should be included in any definition of the term. For the purpose of this thesis, a probiotic is defined as "a microbial cell preparation or components of microbial cells that have a beneficial effect on the health and well being of the host" (Salminen et al., 1999). Other "non-digestible food ingredients that beneficially affect the host by selectively stimulating the growth and/or activity of one or a limited number of bacteria in the colon, that have the potential to improve host health" are considered here to be prebiotics (Rowland, 1998; Sako et al., 1999). In addition to the definitions of probiotic and prebiotic, a third term is coming into common use: synbiotic, which describes the use of a mixture of probiotics and prebiotics (Fooks *et al.*, 1999). For example, synbiotic yoghurt may contain a probiotic organism like *Lb. acidophilus* or *B. bifidum* and a NDO prebiotic such as inulin. If the prebiotic cannot be fermented by the resident microflora, but is readily metabolised by the co-administered probiotic organism, the impact of the live organism is enhanced so that the effect is synergistic (Rowland, 1998). This synbiotic yoghurt would also be classified as a functional food as the sum benefits of its consumption would be greater than the simple nutritional outcome achieved by eating the food.

### 1.1.5 Probiotics: criteria for selection for use

Marks (1996), Reid (1999), Ouwehand (1999) and Kailaspathy (2000) described the following selection criteria:

- must be totally safe;
- must resist destruction by gastric and intestinal enzymes and by oxygen;
- must show reasonable persistence in the intestinal tract;
- should demonstrate antagonism to harmful bacteria;
- must possess desirable enzyme patterns in that;
  - the enzymes present in a probiotic bacteria should not act to produce toxic metabolites and
  - should be capable of hydrolysing a variety of the compounds found in the diet,
    which allows fuller use of the nutritional value of food. In particular the enzymes

present in probiotic bacteria should aid the digestion of sugars such as lactose and sucrose (Wood, 1998);

- must meet the requirements for use in large-scale industrial production of food including sensory properties, stability and viability and
- must show demonstrable positive health effects in animal studies.

Marteau and Rambaud (1996) point out that while some endogenous LAB have been responsible for rare cases of local or systemic infection, no cases of illness have yet been demonstrated after ingestion of probiotic LAB.

## 1.1.6 Potential benefits of the consumption of probiotic foods

Benefits claimed to be associated with the consumption of LAB fermented foods include: reduction in pH within the gastrointestinal tract, improved lactose utilisation, prevention of diarrhoea, competitive exclusion of intestinal pathogens, anti-tumour activity, stimulation of the immune system, and control of serum cholesterol level (Fernandes *et al.*, 1987; Hoier, 1992; Hammes and Tichaczek, 1994; Duncan and Edberg, 1995; Playne, 1995; Mattila-Sandholm *et al.*, 1999; Tannock, 1999). Each of these benefits will be discussed in detail below.

## 1.1.6.1 Consumption of lactic acid

The consumption of lactic acid, produced by LAB bacteria, is thought to influence human health positively in several ways. Oberman and Libudzisz (1998) claim the following mechanisms, in particular, to be important:

• the accumulation of lactic acid within the intestine may influence the composition of the indigenous microflora by inhibiting the growth of undesirable organisms;

- lactic acid influences the physical properties of food. This may result in easier digestion and improved peristalsis of the food. Additionally, in the elderly, where secretion of gastric juice may be reduced, lactic acid decreases stomach acidity which increases the absorption of calcium and iron and
- the L(+) form of lactic acid is capable of being absorbed through the large intestine where it is utilised as an energy source. D(-) lactic acid is excreted by humans and it is recommended that LAB used in food manufacture produce the L(+) form.

#### 1.1.6.2 Improved lactose utilisation

Lactose, the main carbohydrate in milk, must be digested by the enzyme lactase (β-galactosidase) into its constituent sugars, glucose and galactose, before these are absorbed from the gut. Some individuals have low lactase activity in the small intestine that is associated with age and racial characteristics (this metabolic deficiency is said to be present in half the adult population of the world) and therefore can digest only limited amounts of dairy products containing lactose. Failure to hydrolyse lactose leads to its fermentation in the large intestine, which may result in diarrhoea, flatulence and abdominal pain (Oberman and Libudzisz, 1998). Lactase-intolerant individuals better tolerate foods containing fermented lactose, for example yoghurt, because of the added beta galactosidase in this diet and in addition the fermentation process of yoghurt production converts some of the lactose present in milk to lactic acid.

#### 1.1.6.3 Prevention of diarrhoea

The prevention and control of viral, bacterial and antibiotic-associated diarrhoea has been attributed to the intake of probiotic bacteria, although the evidence to support the use of probiotics in the prevention of all types of diarrhoea is equivocal (Marteau and Rambaud, 1996). Some studies have shown that the duration and severity of rotavirus infection can be reduced through the administration of *Lb. rhamnosus* GG. There is evidence of protection against "antibiotic associated enteric disorders" but studies, repeated by other authors, have not always been able to replicate the demonstrated effect (Marteau and Rambaud, 1996). Variation in experimental results may be due to poor experimental protocol, the choice of strain selected for each trial and the difficulties in obtaining samples from human volunteers while adhering to ethical considerations (Playne, 1995).

### 1.1.6.4 Control of intestinal pathogens

Intestinal pathogens may be controlled by competition for nutrients present in the intestine in limited quantities, competition for attachment sites on the mucosal cells, creation of selective environmental conditions and formation of antagonistic compounds (Hull *et al.*, 1992; Fuller, 1993). These are similar to the mechanisms by which LAB inhibit food spoilage organisms, as discussed in section 1.1.2. Studies have shown that lactic acid bacteria in particular *Lb. casei* and *Lb. acidophilus* can inhibit the growth of *Helicobactor pylori*, a cause of stomach ulcers, but the role of probiotic organisms in eradicating *H. pylori* from ulcer sufferers remains to be determined (Canducci *et al.*, 2002; Wendakoon and Ozimek, 2002; Sgouras *et al.*, 2004).

#### 1.1.6.5 Anti-tumour activity and stimulation of the immune system

Claimed anti-cancer properties of probiotic organisms have been attributed to:

- conversion or elimination of procarcinogenic or carcinogenic compounds;
- reduction of the level of faecal enzymes responsible for the conversion of procarcinogens into carcinogens; and
- the stimulation of the immune system (Hammes and Tichaczek, 1994; Tannock *et al.*, 1999)

9

Faecal enzymes (for example, β-glucuronidase, β-glucosidase, urease, azoreductase and nitroreductase) have been used to monitor mucosal carcinogenesis as they convert procarcinogens to carcinogens. It is believed that higher levels of these enzymes are associated with a higher cancer risk (Oberman and Libudzisz, 1998). The consumption of LAB appears to moderate the effect of these faecal microbial enzymes, possibly by a general reduction in the numbers of organisms producing xenobiotic metabolising enzymes; LAB have low levels of xenobiotic enzymes. LAB also appear to have a role in preventing DNA damage, suppressing pre-neoplastic changes and suppressing tumours, although the mechanisms for these are unclear (Ebringer *et al.*, 1995; Rowland, 1996). Suppression of cancer activity may also be controlled indirectly through modulation of the immune system. It has been shown that whole cells and cell wall fragments of LAB activate macrophages in humans and animals. It has also been found that some probiotic bacteria (*Lb. acidophilus, Lb. delbrueckii* subsp. *bulgaricus* and bifidobacteria) influence the *in vitro* regulation of γ-interferon (Oberman and Libudzisz, 1998).

### 1.1.6.6 Control of serum cholesterol levels

A high cholesterol level is a risk factor for cardiovascular disease in humans and one study (Fernandes *et al.*, 1987) has shown a reduction in serum cholesterol associated with consumption of fermented foods. A study by Taranto *et al.* (1997) suggested that cholesterol reduction is linked to bacterial bile salts hydrolase activity, which causes the release of unconjugated bile acids resulting in cholesterol precipitation.

**1.1.6.7** Consumption of bio-active peptides produced during fermentation of milk products Milk proteins, upon ingestion, are broken down by gastrointestinal enzymes to produce a range of peptides, many of which have been identified as being bioactive. These peptides may also be formed by the action of bacterial enzymes during fermentation of milk by LAB. The biological significance of these peptides, their impact on human health and production of novel functional food ingredients containing these peptides, are currently the subject of intensive research (Playne *et al.*, 2003). For example, some bioactive peptides that demonstrated an anti-hypertensive effect have been detected in sour milk. These peptides showed an inhibitory effect on angiotensin I-converting enzyme and could only be demonstrated after fermentation of casein by lactic acid bacteria. The same peptides could not be demonstrated after hydrolysis of casein (Yamamoto and Takano, 1999). Use of these bioactive peptides for pharmaceutical or other commercial use, for example as a functional food used to treat borderline hypertension, seems increasingly likely.

#### 1.1.6.8. Treatment of irritable bowel syndrome, ulcerative colitis and Crohn's disease

The results of clinical trials, with respect to treatment of bowel disease, have been encouraging; however, only small numbers of patients have been included in each trial. These results also indicate that in the case of ulcerative colitis, acute pancreatitis and irritable bowel syndrome, available data has demonstrated benefits of probiotic therapy, but in the case of Crohn's disease, the role of probiotics is not yet clearly defined (Kirchgatterer and Knoflach, 2004).

## 1.2 Lactic acid bacteria

Since the first description of bacteria, scientists have systematically divided them into several groups to aid taxonomic classification. The end-point of classification is the species and it consists of the type strain and all other strains sufficiently similar to be included in the species with it. Unlike higher organisms, bacteria are often similar in nature, resulting in a greater need for methods that separate them into species, sub-species and strains. LAB have been a particular challenge in this regard and in recent years new methods of speciation of lactobacilli have resulted in many changes in classification. Bacterial classification is built on two major kinds of information, phenotypic and phylogenetic. Phenotypic information describes the physical appearance and other direct

observations of bacteria, including size, shape and biochemical traits. Phylogenetic information comes from the developmental history of an organism and is derived from comparison of the structure of highly conserved macromolecules within the cell (DNA and rRNA). Classification systems for LAB based on phenotypic and phylogenetic data are described in detail in the following sections.

#### 1.2.1 Introduction to LAB

LAB are Gram-positive, prokaryotic organisms that are commonly found in a variety of environmental niches; they are commensal colonisers of mammalian gastrointestinal tracts and are found in soils. LAB produce lactic acid as the common end-product of metabolism and can be broadly classified as either homo- or hetero-fermentative, based on the principal end-products made. LAB require a fermentable carbohydrate for growth and those that convert glucose predominantly (more than 90%) to lactic acid are classified as homo-fermentative. Those producing lactic acid plus one or more moles of acetic acid, ethanol and carbon dioxide are classified as hetero-fermentative. Homo-fermentative LAB convert glucose to lactate through the Embden-Meyerhof-Parnas (EMP) pathway by means of a fructose-1,6-diphosphate aldolase.

Hetero-fermentative LAB produce equimolar amounts of lactate, ethanol and carbon dioxide as major end-products through the phosphoketolase pathway by means of a xylulose-5-phosphate phosphoketolase (Saloff-Coste, 1997). There are numerous members of the LAB group including the following genera: *Streptococcus, Lactobacillus, Lactococcus, Carnobacterium, Oenococcus, Pediococcus, Tetragenococcus, Vagococcus, Enterococcus, Weissella* and *Leuconostoc*. The position of these genera in LAB classification can be seen in Figure 1.1.





Figure 1.1 Overview of the classification of LAB.

Generally, LAB can be sub-divided into three phenotypic groups according to their fermentation patterns. The groups are further subdivided into three clusters based on similarity of rRNA sequences or DNA-rRNA hybridisations. *Lb. paracasei* is a member of Cluster 2 and fermentation group I.
# 1.2.2 The genus Lactobacillus

There are currently more than 100 members of the genus *Lactobacillus* and they are genetically quite diverse (Hammes and Tichaczek, 1994; Schleifer and Ludwig, 1995; Euzeby, 2003). LAB are divided into three phenotypic groups, which may then be further sub-divided into either three or five phylogenetic clusters according to the classification systems of either Collins or Schleifer (Collins *et al.*, 1991; Hammes and Vogel, 1995; Schleifer and Ludwig, 1995). The three major phenotypic groups are:

- obligate homofermentative lactobacilli, which ferment hexoses to lactic acid;
- facultative heterofermentative lactobacilli, which ferment hexoses to lactic acid only or to lactic acid and acetic acid, ethanol and formic acid under glucose limitation and;
- obligate heterofermentative lactobacilli, which ferment hexoses to lactic acid, acetic acid, ethanol and carbon dioxide, and pentoses to lactic acid and acetic acid (Pot *et al.*, 1994).

However these phenotypic groups do not reflect the classifications based on phylogenetic relationships. The relationships, based on reverse transcriptase sequencing of 16S rRNA, are better described by the three clusters of Collins (1991). The three phylogenetic clusters are:

- Cluster 1: The *Lb. delbrueckii* group. This group, principally consisting of obligately homofermentative LAB, includes *Lb. delbrueckii* and *Lb. acidophilus*.
- Cluster 2: The Casei-pediococcus group. According to Collins (1991) this is the largest group and contains more than thirty *Lactobacillus* species. Included in this group are *Lb. casei*, *Lb. curvatus* and *Lb. fermentum*

Cluster 3: The *Leuconostoc* group. This group contains members of the genus *Leuconostoc* and obligately hetero-fermentative lactobacilli. Schleifer (1995) proposed a different classification based on an extended data set of 16S rRNA gene sequences and subdivision of Collin's so-called 'Casei-pediococcus group', however there appears to be no clear evidence that allows unambiguous allocation of bacteria to either of the classification systems proposed by either Collins and Schleifer.

The five groups proposed by Schleifer are: the *Lb. acidophilus* group, the *Lb. salivarius* group, the *Lb. reuteri* group, the *Lb. buchneri* group and the *Lb. plantarum* group.

# 1.2.3 Genus and species characteristics

The general features of the genus *Lactobacillus* are shown in Table 1.2. Historically, identification of *Lb. casei* was based upon a number of characteristics, including:

- colonial morphology;
- growth requirements;
- staining behaviour and
- carbohydrate utilisation tests.

More recently, identification based on analysis of chemical and biochemical traits have been added to these identification methods. These include the characterisation of:

- enzyme profiles such as measurement of peptidase and esterase activity (Tsakalidou *et al.*, 1994) as well as characterisation of lactic dehydrogenase (Gasser, 1970);
- whole cell protein profiles (Pot *et al.*, 1993; Patarata *et al.*, 1994; Tsakalidou *et al.*, 1994); and

Table 1.2 General features of the genus Lactobacillus (Hansen and Mocquot, 1970; Collinset al., 1991; Holt et al., 1994; Hammes and Vogel, 1995)

Characteristic	Description
Shape	Rod-shaped
Gram stain	Positive
Spores	Not present
Motility	Rare
G+C content of DNA	32-54 %
Growth environment	Facultative anaerobes
Metabolism	Fermentative and Saccharoclastic
Nitrate	Not reduced
Gelatine	Not liquefied
Catalase	Negative
Cytochrome	Negative
Optimum growth temperature	30-40°C
Normal distribution	Widely distributed including soil, mammal gastrointestinal tracts and food products

fatty acid methyl esters (FAMEs).

The relative usefulness of each of these methods varies greatly and in general around 17 phenotypic tests are required to identify a *Lactobacillus* isolate (Hammes and Vogel, 1995). Generally, these techniques only help when identifying an organism to genus level but some of the techniques are useful in separating strains. These phenotypic methods are rapidly giving way to molecular techniques, which are fast to perform and increasingly relatively cost effective. Molecular techniques for classification are described in section 1.2.5.

# 1.2.4 An early history of the identification of Lactobacillus species

Orla-Jensen (1919) divided the lactobacilli into three physiological groups: *Betabacterium*, *Streptobacterium* and *Thermobacterium*, the classification being based on physiological and nutritional characteristics as well as the method of fermentation. The identification also took into account the optimum growth temperature and the optical configuration of the lactic acid produced. One of lactobacilli Orla-Jensen classified, *Thermobacteriuim intestinale*, was subsequently noted to have all of the features corresponding to *Lb. acidophilus*. Rogosa and Sharpe (1959) extended this work by including temperature growth ranges and vitamin requirements in the identification procedure (Gasser, 1970).

Moro is believed to be the first person to describe *Lb. acidophilus*, which he did in 1900 under the name *Bacillus acidophilus* (Hansen and Mocquot, 1970). The first use of *Lactobacillus* as a genus is attributed to Beijerinck in 1901 (Vandamme *et al.*, 1996). In 1920, the organism that Moro described was included by Holland in the genus *Lactobacillus*, although it is believed that the organism that was described by Moro was accidentally substituted or destroyed before this time (Johnson *et al.*, 1980; Fujisawa *et al.*, 1992). Hansen and Mocquot were the first to publish the name *Lb. acidophilus*, based on a combination of the name *Bacillus acidophilus* and a bacterium

having the general characteristics of the genus *Lactobacillus* (Hansen and Mocquot, 1970). Hansen and Mocquot gave a complete phenotypic description of the bacterium on which to base identification. According to Hansen, the name *Lb. acidophilus* had been widely used, but no valid description to allow identification had been previously published. Following the recommendations of the International Subcommittee on the Taxonomy of the lactobacilli and closely related organisms, the name as *Lb. acidophilus* was confirmed and ATCC 4356 was designated as the type strain (Hansen and Mocquot, 1970). The identification of lactobacilli based upon differentiation into fermentation groups persisted until the late 1970s, when the first rRNA gene sequencing methods for bacterial speciation became available. This enabled phenotypic and phylogenetic data to be married.

It was also around the early twentieth century that an organism describes by Orla-Jensen as *Lactobacterium casei* was reclassified as *Streptobacterium casei*. The first use of the current name *Lb. casei* is attributed to Hansel and Lessel in 1971 and ATCC 393 was designated the type strain.

#### 1.2.5 Recent history of Lactobacillus identification

Research into the classification of all *Lactobacillus* species advanced rapidly with the availability of molecular biology tools such as polymerase chain reaction (PCR) and DNA sequencing. In fact, genus level identification relied on classical methods until 2002 when Dubernet described *Lactobacillus* sp. primers for PCR-based identification. However, molecular biology tools have allowed specific identification of species and differentiation between strains within species. During the past decade, there have been a large number of taxonomic reassignments that have been based principally on DNA-rRNA hybridisation and rRNA sequencing. These have indicated large discrepancies between classical taxonomy and molecular taxonomy (Collins *et al.*, 1991; Decallone *et al.*, 1991; Pot *et al.*, 1993; Rodtong and Tannock, 1993; Dicks *et al.*, 1995; Rementzis and Samelis, 1996).

DNA-DNA hybridisation gives a measure of relatedness across the whole genome but is not particularly useful for identification of an unknown organism, as a large nurriber of known organisms may have to be tested before a close match is found (Schleifer and Ludwig, 1995). DNA-DNA hybridisation has had limited use in differentiating LAB strains, but it has been used to differentiate *Carnobacterium* species (Dicks *et al.*, 1995) and to confirm re-assignment of lactobacilli (Fujisawa *et al.*, 1992). A refinement of this technique, DNA-rRNA hybridisation, introduced in the early 1970s, allows discrimination of organisms at family and genus level. It has been a principal tool used in the re-classification of LAB although this has now been overtaken by the use of ribotyping (Vandamme *et al.*, 1996).

Ribotyping analyses variations in the chromosomal positions or structure of rRNA (Schmidt, 1995). Ribosomal RNA has been conserved through millions of years of evolutionary divergence and most prokaryotes have three rRNAs, called 5S, 16S and 23S. The 5S is usually too small for reliable testing and only 16S and 23S are used for phylogenetic comparison. In prokaryotes, 16S rRNA is primarily used to trace evolution and determines the extent of relatedness of bacteria (Schleifer and Ludwig, 1994; Schleifer and Ludwig, 1995). Meaningful phylogenetic data has been acquired from multiple operons and multiple isolates (Amann *et al.*, 1994). Sequence information so obtained has been used to differentiate fifty-four *Lactobacillus* strains isolated from the intestinal tract of mice (Rodtong and Tannock, 1993). Within the last 20 years, the development of PCR methods has allowed rapid and certain identification of strains of LAB, although some sub-species assays still fail to distinguish between bacteria with closely related 16S sequence homology such as sub-species of *Lb. delbrueckii* (Torriani *et al.*, 1999).

PCR identification of probiotic bacteria has been used to:

- confirm subspecies identities using randomly amplified polymorphic DNA PCR (RAPD-PCR)
  based on the proline iminopeptidase gene sequence (Torriani *et al.*, 1999);
- detect specific probiotic organisms in animal feed (Castellanos *et al.*, 1996). Castellanos amplified a variable region of 16S rDNA of about 200 bp in *Lb. plantarum*, *Lb. rhamnosus* and *Lb. fermentum*; and
- identify Lactobacillus isolates from the gastrointestinal tract, silage and yoghurt using the 16S-23S intergenic spacer region (Tannock *et al.*, 1999).

The current status of *Lb. casei* has been the subject of much discussion. A number of opinions have been put forward as to whether *Lb. casei* should remain a single species made up of several sub-species (Dellaglio *et al.*, 2002) or whether the former subspecies of *Lb. casei*, except the type strain, should be included within the species *Lb. paracasei* (Collins *et al.*, 1989). The name *Lactobacillus paracasei* subsp. *paracasei* was first proposed by Collins (1989).

In 1991 it was first proposed that the *Lb. casei* type strain, ATCC 393, be reclassified as *Lb. zeae* and replaced by ATCC 334 (Dellaglio *et al.*, 1991). Although that proposal was rejected, it was resubmitted in both 1996 and 2002 (Dicks *et al.*, 1996; Dellaglio *et al.*, 2002). The basis of the proposal was the low levels of DNA homology (30-50%) of the type strain compared with other strains of *Lb. casei* and sequence similarity of a 277 bp recA gene fragment. The status of the type strain is still unresolved.

Although yet to be confirmed by the Judicial Commission and subject to International code of Nomenclature of Bacteria, the current proposed reclassification is that the species name *Lb. paracasei* is rejected and strains previously classified as *Lb. casei* be reunited within the species *Lb. casei*.

# 1.2.6 The genus Bifidobacterium

Bifidobacteria were originally classified within the general 'lactic acid bacteria' group set aside by Orla-Jensen in the 1920s. This was based on the physiological observation that bifidobacteria produce lactic acid as a major product of fermentation. However 16S rRNA profiles show distinct differences to lactobacilli and other LAB (Schleifer and Ludwig, 1995).

Thirty-one members of the genus Bifidobacterium are currently recognised with the most recent members, *B. inopinatum* and *B. denticolens* being added in 1996 (Table 1-3) (Crociani et al., 1996). They are usually isolated from human and animal sources, for example from dental caries, intestines and genital tracts (Gomez et al., 2000). However, isolates have also come from wastewater, honey bees and fermented dairy products (Pot et al., 1994). According to Tamine (1999) the species of *Bifidobacterium* found in humans are *B. adolescentis*, *B. bifidum*, *B. breve*, *B.* infantis, B. lactis and B. longum. Bifidobacteria are phylogenetically grouped in the actinomycete branch of the Gram-positive bacteria and are characterised by a high G+C content that varies from 54 to 67%. They possess the glycolytic fermentation enzyme, fructose-6-phosphate phosphoketolase, the key enzyme of the 'bifid shunt', which characterises the species. The endproducts of glucose metabolism are acetic and lactic acids, which occurs without generation of CO<sub>2</sub> (Gomez and Malcata, 1999). They are saccharolytic organisms and all *Bifidobacterium* isolated from human sources have been able to utilise glucose, galactose and lactose as carbon sources (Gomez and Malcata, 1999). Bifidobacteria are generally characterised as non-spore-forming, non-motile, and catalase-negative anaerobes. Various shapes have been noted from short curved rods to bifurcated Y-shaped rods. According to Gomez and Malacata (1999), Bifidobacterium have an optimum growth pH between 6.0 and 7.0 and virtually no growth occurs below pH 4.5-5.0 or above pH 8.0-8.5.

# Table 1.3. List of species of the genus Bifidobacterium, and origin and year species first

reported (Krzewinski, 1997; Krzewinski, 2002)

B. adolescentis	Adult faeces (1963)
B. angulatum	Adult faeces (1974)
B. animalis	Rat, rabbit and calf faeces (1969)
B. asteroides	Bees 1969)
B. bifidum	Child and adult faeces, vagina (1924)
B. boum	Rumen and piglet faeces (1979)
B. breve	Child faeces, vagina (1963)
B. catenulatum	Child and adult faeces, vagina (1974)
B. choerinum	Piglet faeces (1979)
B. coryneforme	Bees (1969)
B. cuniculi	Rabbit faeces (1979)
B. denticolens	Buccal cavity (1996)
B. dentium	Buccal cavity (1974)
B. gallicum	Adult faeces (1990)
B. gallinarum	Chicken faeces (1983)
B. globosum	(unknown)
B. indicum	(unknown)
B. infantis	Child faeces, vagina (1963)
B. inopinatum	Buccal cavity (1996)
B. lactis (subjective synonym of B. animalis)	Fermented milk (1997)
B. longum	(unknown)
B. magnum	Rabbit faeces (1974)
B. merycicum	Rumen (1991)
B. minimum	Sewage (1974)
B. pseudocatenulatum	Child faeces (1979)
B. pseudolongum	Animal faeces (1969)
B. pullorum	Chicken faeces (1974)
B. ruminantium	Rumen (1991)
B. saeculare	Rabbit faeces (1991)
B. subtile	Sewage (1974)
B. suis	Piglet faeces (1971)
B. thermophilum	Animal faeces, rumen (1969)
1	

The optimum growth temperature is 37-41°C, with a maximum temperature that permits growth around 43-45°C and virtually no growth below 25-28°C (Gomez and Malcata, 1999). The general features of the genus *Bifidobacterium* are shown in Table 1.4. Identification of *Bifidobacterium* to genus level may be unreliable due to non-specific cultural and physiological characteristics. Identification based on fermentation products is especially difficult as variable amounts of formic acid, acetic acid, succinic acid and ethanol may occur depending on growth conditions (Gomez and Malcata, 1999).

# 1.2.7 History of the genus Bifidobacterium

Bifidobacteria were first described by Tissier in 1899-1900 and the first isolation, *Bacillus bifidus communis*, was from infant faeces (Gomez and Malcata, 1999). The description of the genus came several years later by Orla-Jensen in 1924, when *Bacillus bifidus* was renamed *B. bifidum* although the organism was also referred to as *Lactobacillus bifidus*. After 1960, the genus was enlarged to 16 members and differentiation was based on fermentative and serological features (Pot *et al.*, 1994). With the introduction of identification based on DNA-DNA hybridisation and sodium dodecyl sulphate-polyacrylamide gel electrophoresis (SDS-PAGE), several more species have been described, although taxonomy based on molecular methods has conflicted with phenotypic methods (Miyake *et al.*, 1998).

#### **1.2.8 Species characteristics**

*B. animalis* (syn. *lactis*) strain VUP Bb 12, used in this study, was originally known as a strain of *B. bifidum*, but when subjected to molecular biology typing (16S rRNA) it was found that the identification did not fit into any existing species of *Bifidobacterium* and was classified as a novel species called *B. lactis* (Wind, 2001). *B. lactis* was first described as a novel oxygen tolerant species isolated from milk (Meile *et al.*, 1997).

# Table 1.4 General features of the genus Bifidobacterium (Sgorbati et al., 1995; Krzewinski,

1997; Gomez and Malcata, 1999; Tamine, 1999).

Characteristic	Description
Shape	Rod-shaped including: uniform, branched, bifurcated Y and V forms, spatulate and club shaped rods
Gram stain	Positive
Spores	Not present
Motility	Negative
G+C content of DNA	54-67 %
Growth environment	Strict anaerobes
Metabolism	Fermentative and saccharoclastic
Catalase	Negative
Optimum growth temperature	37-41°C
Normal distribution	Widely distributed including mammal gastrointestinal tracts and food products

This new species name was however subsequently rejected by The International Committee on Systematic Bacteriology (ICSB), Subcommittee on the Taxonomy of Bifidobacterium, Lactobacillus and related organisms (Klein, 2001).

Unpublished work by Chr. Hansen on 16S rRNA and other genetic sequences of *B. lactis* strain Bb 12, shows significant differences between *B. animalis* and *B. lactis* strain Bb 12 (Wind, 2001). Wind suggests, however, that these differences are not sufficient to place strain Bb 12 in a novel group but further research is required to give a definitive identification of the organism. This view is supported by Cai *et al.* (2000), who suggest that *B. lactis* should be considered a junior subjective synonym of *B. animalis*.

Chr. Hansen refer to strain Bb 12 as *B. animalis* (syn. *lactis*) Bb-12, which is line with current accepted nomenclature of ICSB. Researchers from the Nestlé Research Centre analysed the 16S-23S internally transcribed spacer region of strains of *B. lactis* and *B. animalis* (Ventura *et al.*, 2001). Based on that analysis, they concluded that *B. lactis* should not be considered a junior subjective synonym of *B. lactis* but rather be reclassified as a subspecies of *B. animalis*. This proposed reclassification remains unresolved at present. Table 1.5 lists the classical features on which an identification of *B. animalis* (syn. *lactis*) is based. *B. animalis* (syn. *lactis*) strain VUP Bb 12 has been used by the starter culture manufacturer Chr. Hansen since 1986 and is included in several product lines including ABT cultures, single cultures and anti-diarrhoea tablets. ABT cultures contain *Lb. acidophilus*, *S. thermophilus* and *B. lactis* strain Bb 12 (Wind, 2001).

# **1.3 Probiotic Lactic Acid Bacteria**

The minimum basis for selection of probiotic bacteria for human use includes the following:

• that the bacteria produces lactic acid;

# Table 1.5 Important characteristics of *B. animalis (Sgorbati et al., 1995; Krzewinski, 1997;*

Gomez and Malcata, 1999; Tamine, 1999).

Characteristic (Gro	wth factor or vitamin	Usual Finding	
requirement)			
Riboflavin		+	
Pantothenate		_	
Nicotinic acid		+	
Pyridoxine		_	
Thiamine		-	
Folic acid		+	
Tween 80		+	
Carbohydrates fermented *			
Glucose	+	Fructose	+
Galactose	+	Lactose	+
Maltose	_	Mannose	_
Mannitol	_	Salicin	_
Melibiose	d	Sucrose	d
Trehalose	_	Xylose	_

\* Symbols: +, >90% positive

d, 11-89% positive

-, < 10% positive

- is generally recognised as safe (GRAS); and
- has the ability to colonise/adhere to the gut wall (Playne, 1995).

Kailasapathy (2000) suggests additional factors that are important, including that the organism should survive gastric acidic conditions and be resistant to bile salts and digestive enzymes. These are discussed in sections 1.3.1 and 1.3.2. To be of any value as a probiotic, a microbe must survive two different sets of conditions prior to colonising the intestine: firstly, it must survive (possibly up to several weeks) in the food in which it is contained, then it must survive passage through the upper part of the digestive system, including the stomach, which secretes hydrochloric acid and digestive enzymes.

### 1.3.1 Survival of LAB in yoghurt

Consumers have indicated their preference for probiotic-containing foods and 20% of yoghurts now sold in Australia contain *Lb. acidophilus* and *B. bifidum* (Playne, 1995). However, the number of live organisms surviving in probiotic foods depends on the manufacturer and length of time since production (Shah *et al.*, 1995; Lankaputhra *et al.*, 1996; Kailaspathy and Rybka, 1997). In particular the survival of *Lb. acidophilus* in yoghurt is extremely variable, with numbers of viable organisms falling quickly after manufacture (Hull *et al.*, 1984; Kailaspathy and Rybka, 1997). A number of causes have been proposed to account for the loss of viability of these organisms, including the effect of hydrogen peroxide produced by starter culture organisms, low pH of storage, and oxygen penetration of storage containers. Condon (1987) claimed that in lactobacilli, toxicity of oxygen exposure varies with the strain, and loss of viability is not significant compared to reduction in viability in the strictly anaerobic *Bifidobacterium* species upon exposure to oxygen. Gilliland and Speck (1977) described the marked effect that hydrogen peroxide had on the survival of *Lb. acidophilus* in yoghurt. Laroia and Martin (1991) found that in frozen dairy desserts with low pHs,

no viable *B. bifidum* could be detected after manufacture. However, increasing the pH of the product resulted in increased survival of the bifidobacteria. Strain selection appears to be important. Clark *et al.* (1993) for example, commented on the variation of survival for each of four bifidobacteria strains (*B. infantis*, *B. longum*, *B. bifidum* and *B. adolescentis*), which were tested for the ability to survive simulated stomach pH conditions. In particular they found that *B. bifidum* did not survive brief exposure to low pH (<3) well and counts dropped several log cycles after only a few minutes. Some recent attempts at improving the viability of *Bifidobacterium* in yoghurt have concentrated on micro-encapsulation of the live bacteria (Adhikari *et al.*, 2000). The study demonstrated some improvement in viability of strains encapsulated in κ-carrageenan, but increased acid content in the product adversely affected sensory characteristics.

# 1.3.2 Survival of LAB in the digestive tract of humans

Early studies by Draser *et al.* (1969) indicated that the levels and types of intestinal bacteria change constantly. In particular the numbers of lactobacilli and bifidobacteria occasionally diminished to undetectable levels in particular parts of the intestine at certain times during the day. This is probably due to intestinal motility following the stimulation of a meal. They also found that bacteria consumed during meals are able to survive passage through the stomach because food partly neutralises acid levels. Acid-tolerant organisms have an even greater chance of surviving stomach acidity and subsequently surviving in the more alkaline conditions of the small intestine, and thus are more likely to colonise the lower gastrointestinal tract. In order to survive and colonise the GI tract, probiotic organisms must also have a high rate of epithelial attachment, a high growth rate (to prevent being washed out by peristalsis,) and resistance to organic acids and bile salts (Canganella *et al.*, 1996).

### 1.4 Bacterial responses to growth in stressing environments

# 1.4.1 Bacterial response to growth in stressed conditions

Stress has been variously described as "a disturbance of the normal functioning of a biological system caused by an environmental factor which is detected by a departure from steady state" and as "an agent which places an organism at a disadvantage as increased energy expenditure is required reducing the chances of its survival" (Hoffmann and Parsons, 1994). In this thesis, a slightly broader definition will be used in which the term 'stress' encompasses an environmental factor (a physical condition) that causes a change that is potentially injurious in a biological system. (Hoffmann and Parsons, 1994). Within this thesis, these environmental factors will be referred to as environmental stressors and any organism experiencing these stressors will be considered to be stressed or under stress. Some of the changes that may happen because of exposure to stress include changes in induction and repression of the synthesis of proteins (enzymes, isoenzymes, structural proteins and heat-shock proteins (HSP)).

#### 1.4.2 Protein synthesis and folding in a normal cellular environment

The purpose of protein synthesis is to create a polypeptide and in all cells this activity occurs at the ribosome. Once formed, the polypeptide must adopt a specific, folded three-dimensional tertiary or native structure in order to become a biologically active protein (Ramakrishnan and White, 1998). Research performed by Anfinsen and Haber in the 1960s found that the amino acid sequence of a protein contained all the information required for an amino acid chain to properly fold into an active protein molecule (Gething and Sambrook, 1992; Ruddon and Bedows, 1997). These and other observations have led to some broad rules for protein folding. The first is that the folding of a polypeptide chain is determined by the amino acid sequence in a given environment. The second is that the structure is thermodynamically controlled and its stability is not very high. The third states that the protein-folding pathway probably involves intermediates, which may evolve into either properly folded proteins (Creighton, 1995). The amino acid sequence for each protein

specifies only the primary structure, yet the process of folding occurs rapidly after biosynthesis of the polypeptide chain and can even begin before the polypeptide chain is complete. Proteins are compact structures that consist of segments of secondary structure packed tightly together. When they are made up of a large number of residues (>200), they generally contain two or more independent structural units that are referred to as domains. These domains may or may not be identical. Over the last 30 years, a number of theories that model how proteins obtain a native configuration have been put forward although none of them adequately covers all the experimental observations to date. For example Baldwin (1989) described three folding models: (1) the hydrophobic collapse model, (2) the formation of secondary structure model and (3) the specific interaction model (Baldwin, 1989).

1. The hydrophobic collapse model.

This model proposes that the collapse of hydrophobic bonds results in a molten globule state from which the folded protein arises. The molten globule state is one where a protein is neither folded nor unfolded. In this state, the molecule is generally compact and may contain considerable secondary structure but is disordered compared to a fully folded protein.

2. The formation of secondary structure model.

This model suggests that a tertiary structure can originate from the formation of stable secondary structures. The appearance of stable secondary structures produced by random folding may serve as a template for folding the remaining polypeptide chain.

3. The specific interaction model.

This model proposes the formation of specific structures from stabilising interactions. An example is the formation of one correct disulphide bond that leads to the formation of other correct disulphide

bonds (Nilsson, 1991). Creighton (1997) reviewed the significance of protein folding models and highlighted the drawbacks of each. From work on  $\alpha$ -lactalbumin, he argued that the molten globule state and other stable intermediates might simply be energetically preferred, but non-productive, forms of the protein. In addition, with regard to the specific interaction model, Creighton pointed out that disulphide bonds probably stabilise proteins only after they have reached a fully-folded conformation and therefore may not contribute greatly to directing correctly folded intermediates to a final conformation. It was suggested at the time that stabilisation of all the weak individual interactions directs the final conformation of the native protein (Creighton, 1997).

Regardless of the exact mechanism, it is suggested that protein folding is enhanced by macromolecular crowding (Ellis, 2001). Macromolecular crowding refers to "the concentration of macromolecules within the cell which is so high that a significant proportion of the cell is physically occupied and hence unavailable to other molecules" (Ellis, 2001). It is considered that molecular crowding has significant effects on reaction rates and equilibria of macromolecular reactions, particularly during the initial collapse of newly synthesised polypeptide chains and also by the association of partly folded chains into non-functional aggregates (Ellis, 2001).

Notwithstanding how proteins are folded, it has been observed that while most small proteins need little assistance to reach a stable, active structure, larger proteins, without assistance, often produce inferior conformations. These are often inactive intermediates or aggregated proteins that must be removed or refolded. It has been seen in denaturation-renaturation experiments that many polypeptides cannot reform native state without the presence of additional proteins (Ruddon and Bedows, 1997). These additional proteins have been termed molecular chaperones. Chaperones derive their name from their function: they prevent illicit interactions between proteins, particularly aggregation, by interacting with them from the time of synthesis on the ribosome until they are properly folded and in their final location within the cell. The principal functions of all molecular

chaperones are: to facilitate folding of nascent proteins to their final native state; to hold substrates in an unstructured form to allow cross membrane transport; and to prevent aggregation of unfolded proteins and allow for their renaturation. During periods of cellular stress, it is the prevention of aggregation and the regeneration of proteins that are particularly important (Boston et al., 1996). Hendrick and Hartl (1993) suggested that there is an additional requirement for classifying proteins as chaperones: they should not be part of the final functional protein structure. According to Jakob and Buchner (1994) there are several properties that characterise proteins as chaperones: "(1) suppression of aggregation during protein unfolding; (2) (it has an influence) on the yield and kinetics of folding; and (3) effects are exerted at near stoichiometric levels". Hendrick (1993) proposed a comprehensive definition of a molecular chaperone. His definition that a chaperone is "a protein that binds to and stabilises an otherwise unstable conformer of another protein and by controlled binding and release of the substrate protein, facilitates its correct fate in vivo: be it folding, oligometric assembly, transport to a particular subcellular compartment, or controlled switching between active/inactive conformations" covers all current observations regarding characterisation of chaperones. These chaperones and another group of proteins called foldases mediate in vivo protein folding in both prokaryotic and eukaryotic cells. According to Boston et al. (1996), foldases are catalysts that increase the rate of protein folding. It has been proposed that all cellular proteins interact with these proteins at some time during cell life (Gething and Sambrook, 1992). It is not surprising then that chaperones and foldases have been found throughout all cellular compartments, because the need for protein folding which results in a functioning protein is universal.

# 1.4.3 Stress responses

Bacteria must cope with a wide range of environmental stressors: extremes and sudden changes in pH and temperature, wide variation in nutrient and metabolite concentrations, and the presence of

toxic chemicals such as cadmium, lead and other heavy metals. To enhance the probability of survival, bacteria must possess many ways of rapidly detecting then adjusting to or compensating for these altered conditions. A number of systems have been identified which allow bacterial cells to sense or detect changes in the external medium and subsequently activate regulatory systems (Stock *et al.*, 1989; Tiwari *et al.*, 1996). Furthermore, within the cell, there are a number of mechanisms that protect the cellular contents from lethal stress injuries. These include activation of HSPs which aid in the proper folding and refolding of proteins damaged by growth at temperatures above growth optimum, and induction of enzyme systems that are protective against oxidative, starvation and acid stress (Craig *et al.*, 1993; Parsell and Lindquist, 1993; Boston *et al.*, 1996). HSPs are involved in the production of proteins from the synthesis of nascent chains to the folding of protein complexes; it is for this reason they have also been designated chaperones. However, their role in the cell is far more complex than this simple description implies. These concepts are discussed in further detail in the following sections.

#### 1.4.4 Mechanisms of signalling sub-optimum growth conditions

A number of systems are present within bacterial cells that allow them to detect and cope with environmental changes. These systems also operate during periods of normal cell growth and their function is either enhanced or up-regulated when the cell is stressed. These systems commonly interact with other stress responses through one or more mechanisms that control gene expression.

#### 1.4.4.1 Bacterial signalling systems: the phospho-relay system

One example of bacterial signalling systems is the system of protein pairs in prokaryotic cells, a mechanism that allows cells to signal changes within their cytoplasm and external environment to achieve a rapid adaptive response. The protein pairs form part of a two-component regulatory system, although more than two proteins may be involved (Tiwari *et al.*, 1996), and in some cases

there is more than one kinase or response regulator, or both (Stock *et al.*, 1989). Specific changes that this system may detect include:

- detection and invasion of host cells including situations that lead to symbiosis (Finlay and Falkow, 1997);
- chemotaxis (Stock et al., 1989);
- detection of environmental changes leading to spore formation or production of fruiting bodies (Albright *et al.*, 1989);
- metabolite utilisation (Parkinson and Kofoid, 1992); and
- stress responses to the presence of antibiotics and heavy metals (Parkinson and Kofoid, 1992).

A typical two-component system usually consists of a sensor protein, (e.g. a histidine protein kinase) and a cytoplasmic response regulator (Albright *et al.*, 1989; Stock *et al.*, 1989). The sensor proteins have been found to have similar C-domains while the regulator proteins share homology in their N-domains (Parkinson and Kofoid, 1992). Activation of a signal is based on phosphorylation of the N-domain of the regulatory protein, which often functions as a transcription factor, by the C-domain of the sensor protein. After it has been stimulated by a specific environmental change, the sensor protein transmits a signal to the carboxyl terminal histidine kinase domain resulting in a level of activation of the kinase corresponding to the level of its phosphorylation. Through ATP utilisation and further phosphorylation steps of the receiver domains, gene regulation and expression can occur: this is referred to as the phospho-relay (Wurgler-Murphy and Saito, 1997). Within lactic acid bacteria, several of these two-component signalling systems have been identified, all of which deal

with histidine protein kinases (O'Connell-Motherway *et al.*, 1997). However, a signalling system activating acid tolerance in *Rhizobium meliloti* has been proposed (Tiwari *et al.*, 1996), which suggests that a similar signalling system in other bacterial types may occur.

#### 1.4.4.2 Other signalling systems

The work of Ananthan *et al.* (1985) indicated that the presence of abnormal (denatured) proteins causes increased expression of the heat shock genes in eukaryotic cells. When denatured proteins were injected into *Xenopis laevis* oocytes, activation of an HSP70- $\beta$ -galactosidase hybrid gene followed. This resulted in an increased ability to degrade these abnormal proteins. With further work on the HSP70 promoter, these authors were able to determine that both heat shock and accumulation of abnormal proteins resulted in activation of HSP genes, which occurred through a common mechanism.

#### 1.4.5 General responses to cell stress

Cells may respond in either specific and/or non-specific ways to stress. In *E. coli* global, non-specific responses to change in the environment have been shown to include the stringent, stationary phase and the SOS responses.

The stringent response is caused by a lack of aminoacetylated tRNAs for protein synthesis (Spira *et al.*, 1995). Protein synthesis requires the presence of all 20 amino acids. When one or more is absent in the growth medium or cannot be made by the cell or when the codon for the missing amino acid is encountered, the process of synthesis stops. Although the missing amino acid is not required for rRNA or tRNA synthesis, their production also stops which conserves cellular energy. The coupling of the synthesis of proteins and RNA is called the stringent response. Microorganisms commonly encounter nutrient limitation during the growth cycle and respond by developing a state that allows the cell to survive long starvation periods, typically as it enters stationary phase. Part of the response may include: changes in cell size and shape, changes in

fatty acid profiles, decrease in protein synthesis and production of distinct sets of proteins. These proteins may also be produced during heat, ethanol, acid, osmotic and oxidative stress (Hartke et al., 1994). The SOS response, one of many global cellular responses, is a mechanism that saves E. coli cells from potentially lethal damage. The response was named after the telegraph signal given in the Morse alphabet when a ship is in deadly danger (Save Our Souls) (Janion, 2001). It is specifically triggered by severe damage to DNA or inhibition of DNA replication. A regulatory network that controls approximately twenty genes (the so-called SOS genes) at the transcription initiation level controls the response. The SOS genes are usually repressed by a represser (LexA) protein which binds to operators upstream of the SOS genes, and this prevents their transcription during normal cell growth. When DNA is damaged, single stranded DNA is released which results in a series of binding and cleaving steps. The DNA binds to two proteins in turn, RecA and LexA, causing autocleavage of the represser protein, LexA, which then allows transcription of the SOS genes. This allows DNA repair and other SOS functions to occur until there is a drop in the level of activated RecA. LexA will then accumulate, repressing transcription of the SOS genes until the next episode of DNA damage (Little, 1996). SOS-induced proteins, along with heat stress and stringent response proteins, have been found to be induced after exposure to cadmium stress (Ferianc et al., 1998).

#### 1.4.6 Chaperone families and the cellular response to heat stress

Originally identified by their increased abundance following heat shock (hence the name HSPs) chaperone proteins, in general, recognise exposed hydrophobic surfaces of non-native proteins that are normally within the correctly folded protein (Bukau and Horwich, 1998).

Chaperones were later found to be induced after exposure to a wide range of environmental stresses and are often detectable under normal growth conditions, where they play a role in normal protein folding. The protective nature of HSPs was demonstrated in early thermotolerance

experiments, where cells of *E. coli* that were exposed to a mild heat pre-treatment that induced HSPs, were killed more slowly when shifted to the higher temperature that was lethal to normal cells (Lindquist and Craig, 1988). Some stress proteins, in particular homologues of one particular protein (HSP60), have been identified in bacteria, fungi, plants and mammals (Zeilstra-Ryalls *et al.*, 1991). Some of the stress proteins belong to a group of structurally unrelated proteins called the molecular chaperones as well as a group of ATP-dependent proteases. It is possible for a chaperone to be an ATP-dependent protease and a heat stress protein, but not all stress proteins are chaperones and not all proteases are stress proteins (Boston *et al.*, 1996).

Molecular chaperones are pivotal in mediating protein folding in cells: specifically, these chaperones increase the yield of correctly folded protein by preventing aberrant structure formation and by degradation of mis-folded proteins (Lund, 1995). Degradation of proteins is important for several reasons, particularly because it helps to maintain homeostasis while cell structure is continually rebuilt and because mis-folded proteins are prone to aggregation (Baumeister *et al.*, 1998). The chaperones have been grouped in families according to molecular weight and function: 100, 90, 70, 60 and small HSPs. However, chaperones cannot fully function without co-chaperones and foldases and the function of chaperones, particularly where related to cell stress, are discussed below. Table 1.6 summarises the proteins for which there is evidence of induction during stress.

### 1.4.6.1 HSP60 and co-chaperone HSP10 (the chaperonins)

HSP60 and HSP10, or more properly chaperonin 60 and chaperonin 10 have been extensively studied in *E. coli* where the best evidence of their function has been gathered. When first isolated from *E. coli*, the two proteins were termed GroEL (HSP60) and GroES (HSP10).

Prot sion #					
Swiss! access	P15716 P03815 P33138 Q06716 P31539	P10413 P02829 P14625	P04475 P10591 P42368	P06139 P12612 P11983 P37282	P05380 P37283
References	(Gottesman and Maurizi, 1992; Squires and Squires, 1992; Suzuki <i>et al.</i> , 1997)	(Lindquist and Craig, 1988; Becker and Craig, 1994; Craig, 1995; Boston <i>et al.</i> , 1996)	(Zeilstra-Ryalls <i>et al.</i> , 1991; Ellis, 1993; Becker and Craig, 1994; Craig, 1995; Boston <i>et al.</i> , 1996; Morimoto <i>et al.</i> , 1997)	(Ellis, 1993; Becker and Craig, 1994; Craig, 1995; Boston <i>et al.</i> , 1996; Morimoto <i>et al.</i> , 1997)	(Craig <i>et al.</i> , 1993; Ellis, 1993; Boston <i>et al.</i> , 1996; Hartke <i>et al.</i> , 1997)
Intracellular location	nuclear and cytoplasmic	cytoplasmic cytoplasmic ER	nuclear/ cytoplasmic / mitochondrial	cytoplasmic	cytoplasmic
<u>Molecular</u> weight (kDa)	84.2 95.5 46.0 86.0 102.0	71.4 81.4 92.4	68.9 64.9 64.9	57.1 60.5 60.4 57.2	10.4 10.2
Cell type	E. coli E. coli E. coli E. coli L. lactis S. cerevisiae	E. coli S. cerevisiae Human	E. coli S. cerevisiae L. lactis	E. coli S. cerevisiae Mouse L. lactis	E. coli L. lactis
Typical Class members	CIP A or LOPD CIP B or HTPM CIP X CLPL HSP104	НрG НSР 82 Gp 94	Dnak Ssa1 HSP70	GroEL Top1 TCP1 cpn60	GroES 10 kDa -chaperonin
Sub-Class (Alternate names)	CLP proteins (HSP100(110))	06dSH	HSP70	HSP60 (Chaperonins)	HSP10
<u>Stress</u> Protein Class	Chaperones				

Table 1.6. Stress induced proteins (including chaperone families)

Stress	Sub-Class (Alternate	Typical Class	Cell type	Molecular	Intracellular	References	SwissProt
<u>Protein</u> <u>Class</u>	names)	members		weight (kDa)	location		accession #
	Small HSPs (12-30 kDa.)	18PA or Hpn 18PB or HpnE HSP20	E. coli E. coli S. cerevisiae	15.8 16.0 23.7	cytoplasmic	(Lindquist and Craig, 1988)	P29209 P29210 P15992
Co- Chaperones	DNAJ	DnaJ DnaJ protein	E. coli L. lactis	41.1 40.8	cytoplasmic	(Becker and Craig, 1994)	P08622 P35514
	GRPE	GrpE GrpE protein	E. coli L. lactis	21.8 20.6	cytoplasmic	(Becker and Craig, 1994)	P03372 P42369
Foldases	Protein disulphide isomerase	POI	Maize	57.1	endoplasmic reticulum lumen	(Boston <i>et al.</i> , 1996)	P52588
	Peptidyl prolyl isomerase	Idd	S. cerevisiae	17.2	cytoplasmic	(Boston <i>et al.</i> , 1996)	P14832
Other stress related proteins: <i>Porter/</i> <i>Antiporter</i> <i>systems</i>	Calcium/Proton antiporter	СНАА	E. coli	365	membrane	(Speelmans <i>et al.</i> , 1995)	P31801
	Sodium/Proton antiporter	NHAA Of ANT	E. coli	41.4	inner membrane	(Speelmans <i>et al.</i> , 1995)	P13738
	Cadaverine / lysine antiporter	CADB	E. coli	46.7	inner membrane	(Bearson <i>et al.</i> , 1997)	P13738
cold shock	cold shock proteins	CSPA	E. coli	7.2	cytoplasmic	(Lelivelt and Kawala, 1995; Salotra <i>et al.</i> , 1995; Kim and Dunn, 1997)	P15277

<u>Stress</u> Protein Class	Sub-Class (Alternate names)	Typical Class members	Cell type	<u>Molecular</u> weight (kDa)	Intracellular location	References	<u>SwissProt</u> accession #
	5						
Eukaryotic only system	Ubiquitin	UBB UBB UBC	Mammals	8.5 5.5	nuclear and cytoplasmic	(Varshavsky, 1997)	P02248
Bacillus related proteins	General stress protein	CIO	B. subtilis	21.9	?Ribosomal	(Hecker <i>et al.</i> , 1996)	P14194
Structural system	Surface layer proteins	SLPA	Lb. acidophilus	46.57	cell wall	(Boot <i>et al.</i> , 1993; Boot <i>et al.</i> , 1995; Boot <i>et al.</i> , 1996)	P35829
acid stress	Acid shock protein	ASR	E. coli	18.7	? cytoplasmic	(Heyde and Portalier, 1990; Farber and Pagotto, 1992; Foster, 1995; Garren et al., 1998)	P36560
	Arginine decarboxylase	ADI	E. coli	84.4	cytoplasmic	(Melnykovych and Snell, 1958; Auger et al., 1989)	P28629
	Arginine deiminase	ADI	M. arginini	46.4	? cytoplasmic	(Casiano-Colón <i>et al.</i> , 1988, Curran <i>et al.</i> , 1995)	P23793
	ClpP or peptidase family s14	CLP	E. œli	23.2		(Gottesman and Maurizi, 1992; Squires and Squires, 1992; Suzuki et al., 1997)	P19245

GroEL was initially characterised as a host factor required for bacteriophage assembly (Zeilstra-Ryalls *et al.*, 1991). The *groE* genes were identified through mutation studies with *E. coli*, where mutants of *E. coli* were isolated that could grow in the presence of the bacteriophage that would normally kill the cells. Inside the cells, virus coat particles were produced but they were unable to be assembled into new virus particles and became aggregated. Bacteriophage growth was blocked in *gro*EL mutants at the post-adsorption or injection stages. HSP60 was one of the first proteins to be described as having a chaperone function. Many homologues of HSP60 exist and have been identified in numerous bacteria including, *Streptococcus suis* (Benkirane *et al.*, 1997), *Bacillus subtilis* (Yuan and Wong, 1995), *Lactococcus lactis* (Whitaker and Batt, 1991) and *S. typhimurium* (Foster and Spector, 1995). Many hundreds of HSP60 genes in other species have been cloned and sequenced (Ellis *et al.*, 1996).

The chaperonins have been described as a family of sequence-related molecular chaperones containing two distinctive subclasses. The first is the GroE subclass found in chloroplasts, mitochondria and eubacteria while the second, TCP-1 subclass, is found in archaebacteria and the eukaryotic cytosol (Boston *et al.*, 1996). The TCP-1 class, which does not appear to be heat shock inducible (Lund, 1995) acts without a GroES co-chaperone and indeed may have the GroES function contained within the domain structure (Netzer and Hartl, 1998).

In *E. coli*, the GroEL subclass is encoded on an operon that includes the gene for GroES (Hendrick and Hartl, 1993). The *groE* operon is composed of two genes, *groES* and *groEL*, in that order, and expression of functional proteins of both are required for growth of *E. coli* at all temperatures (Fayet *et al.*, 1989). The genes for the two proteins are transcribed as a single mRNA and are under the control of two promoters. These promoters control expression of the operons  $E\sigma^{32}$  and  $E\sigma^{70}$  with the  $E\sigma^{32}$  controlling expression at higher temperatures (Lund, 1995). Sigma factors are discussed in section 1.5.3.

Biochemical and crystallographic evidence has shown that the denatured molecular weight of GroEL is a 57-1 kDa protein and the active unit is made up of 14 identical sub-units, see Figure 1.2. The sub-units combine to form two heptameric rings with an internal pocket (Hartl, 1994); this complex has a molecular weight of around a million Daltons (Bukau and Horwich, 1998). The two rings stack back-to-back and enclose two non-contiguous cavities (Saibil, 2000). Evidence indicates that opening of the cavities, also referred to as Anfinson's cage (Ellis, 1999) is associated with the binding of ATP and GroES (Bukau and Horwich, 1998). This binding causes movement of the apical domains that results in formation of the cavity, which can hold unfolded proteins (Lorimer, 1997).

The cavity of the ring enlarges around two-fold in volume and the dome shaped GroES ring covers the open end (Bukau and Horwich, 1998). It is postulated that the pocket can hold unfolded proteins of up to 57 kDa (Ellis, 2000), with most substrates being between 10 and 55 kDa (Netzer and Hartl, 1998). Later binding of ATP to the GroEL ring causes ejection of the substrate protein regardless of whether the protein has folded or not (Lorimer, 1997). Evidence suggests that only one of the two GroEL rings can be open at any one time in the 'folding–active state', that is the state in which GroES binds to the GroEL ring in the presence of ATP (Bukau and Horwich, 1998). The asymmetry of the rings appears necessary to allow HSP60 to progress through its reaction cycle. GroES is a single heptameric ring structure of 10 kDa that can form a cap that holds the protein-folding intermediates within the GroEL pocket. Timing for the reaction is set by the ATP hydrolysis cycle, which is 15 seconds at 23°C (Ellis, 1999). This time is said to be insufficient for slow-folding proteins to complete a folding cycle and it is proposed that macromolecular crowding prevents the release of the partly-folded polypeptide chain.



KEY

GroEL and GroES in polypeptide acceptor state Native protein or protein committed to fold ere ere Unfolded protein ج لووں Folding Intermediate 

Figure 1.2 Model for GroEL-GroES-mediated folding reaction (after Bukau et al., 1998).

The asymmetric GroEL-GroES complex (Box A) is the most likely state to accept unfolded polypeptides (U) or unstable folding intermediates (IntU) to form a trans ternary complex (Box B). Two possible pathways that lead to formation of the cis complex are shown in Box C. When GroES binds to the ring containing the polypeptide in the presence of ATP (forming the folding active cis-intermediate seen in Box D) major conformational changes occur in the GroEL bond while folding continues (Box E). Binding of ATP to the trans ring evicts GroES from the cis ring and the peptide is released in one of three cis GroEL ring. This results in release of the polypeptide from the apical binding sites and folding commences. ATP hydrolysis weakens the GroESforms; a native protein (N), a protein committed to fold to a native state (IntC) or an unstable folding intermediate (IntU) (Box F). This allows rebinding of the polypeptide to the same GroEL oligomer and prevention of exposure of aggregation-sensitive chains in the cytosolic medium (Ellis, 1999). Pulse-chase experiments have shown that larger substrates may remain in contact with the chaperone for an extended time, possibly up to 150 seconds (Lorimer, 1997; Ruddon and Bedows, 1997). Based on the abundance of GroEL, compared to the rate of protein synthesis, it is estimated that GroEL can only be responsible for folding 2-7% of all newly synthesised proteins. However, after a brief heat shock, the fraction of GroEL/peptide interactions can increase to 30% (Netzer and Hartl, 1998). GroES inhibits the ATP activity of GroEL in the absence of unfolded proteins (Bukau and Horwich, 1998).

#### 1.4.6.2 HSP70 (stress-70)

In *E. coli*, HSP70 was originally defined as DnaK and was the product of a host gene required for bacteriophage  $\lambda$  DNA replication (Gething and Sambrook, 1992). The primary action of DnaK appears to be binding of nascent polypeptides as they proceed from the ribosome thus preventing improper folding before synthesis of polypeptides is complete (Ruddon and Bedows, 1997). Secondary actions include transportation of unfolded peptides across membranes, involvement in the re-arrangement of oligomers, and the resolution of protein aggregates. It is believed that all of these general functions are enhanced during periods of cell stress and that protein expression is highly regulated according to current cell conditions (Craig and Gross, 1991).

As a group, heat shock protein 70, HSP70 or 'stress–70' are amongst the most highly conserved proteins known and are found in most cellular compartments (Lindquist and Craig, 1988; Craig *et al.*, 1993). They have a molecular mass of around 70 kDa and have a number of essential roles in protein metabolism including protein folding of newly translated proteins, membrane translocation, degradation of proteins and regulatory roles (Hartl, 1996). Historically HSP70 proteins were identified by induction under conditions of stress. During stress the role of HSP70 is to prevent aggregation and assist refolding of mis-folded proteins (Bukau and Horwich, 1998).

HSP70 proteins have two domains, an amino-terminal ATPase domain of approximately 45 kDa and a carboxyl-terminal peptide-binding domain of approximately 25 kDa. The amino-terminal domain is the most conserved. The most variable region is adjacent to the carboxyl-terminal region and it has been proposed that this may dictate substrate specificity (Boston *et al.*, 1996). The carboxylterminal end has been shown to form a substrate binding pocket which is a channel ~5 X 7 Å. In this pocket, DnaK (the name given to *E. coli* HSP70) appears to bind with great affinity to short hydrophobic segments in extended conformation, meaning that the polypeptide must be substantially unfolded (Bukau and Horwich, 1998). It is suggested that the short hydrophobic segments of the un-folded protein, bind approximately seven residues of DnaK; a process which temporarily shields them and reduces their tendency to form aggregates (Ellis, 2000).

For HSP70 proteins, substrate binding is controlled by an ATPase cycle and this represents a switching between the two active states. The first is an ATP-bound state where there is low affinity and fast exchange of substrates and the binding pocket is closed. The second is an ADP-bound state where there is high affinity and slow exchange of substrates and the binding pocket is closed (Bukau and Horwich, 1998). The cycle probably starts by association of DnaJ (a co-chaperone) with a peptide substrate, followed by transfer of the substrate to DnaK-ATP. This transfer is coupled to the coupling of the substrate in the binding pocket of DnaK, which is caused by ATP hydrolysis. DnaJ then leaves the complex to be replaced by GrpE (another co-chaperone), which triggers ADP release from DnaK. ATP may then be bound again releasing GrpE and the substrate from DnaK. This interaction is shown in Figure 1.3 and discussed further in section 1.5 (Interaction of chaperones).

Figure 1.3 The proposed role of chaperone HSP70 (DnaK), co-chaperones DnaJ and GrpE and the chaperonins (HSP60 and HSP10) in protein folding in <i>E. coli</i> (Johnson and Craig, 1997; Ruddon and Bedows, 1997).	(a) DnaK and DnaJ recognise and bind hydrophobic regions of short peptides in an ATP-utilising step. The binding of DnaK prevents premature folding and aggregation of proteins.	(b) GrpE increases the rate of dissociation of the bound nucleotide to DnaK in a nucleotide exchange step. (c) Most partially folded proteins complete folding independently of any further chaperone activity, but some are passed to the chaperonin complex for	further folding steps. (d) The chaperonin complex allows completion of folding of those proteins requiring additional assistance to reach a native state.	(e) Proteins may be folded to a fully functional protein after one round of assisted folding by HSP60 and HSP10. Other proteins are ejected by the complex but retain some secondary structure. These proteins may be rebound by the complex or allowed to form native structure in the cytosol without any further	chaperone intervention.	(
--	--	--	--	---	-------------------------	---





The rate at which the chaperone activity of HSP70 is performed is extremely slow, even in the presence of substrates that have been shown to stimulate ATPase activity. A regulatory mechanism has been shown to increase ATP turnover (Bukau and Horwich, 1998). Principally, members of the DnaJ co-chaperone family regulate ATP hydrolysis while the GrpE co-chaperone family accomplishes release of ADP. See section 1.4.6.6.

Although a major function of DnaK is to protect large, slow-folding, thermolabile proteins from aggregating, a condition more likely to occur during heat stress, there is little evidence offered in the literature to support the theory that HSP70 is specifically protective to cellular functions under stress conditions. However, in *in vitro* studies in bacteria, HSP70 has been shown to solubilise aggregated proteins and in eukaryotic cells HSP70 may co-operate with HSP90 to prevent heat-induced protein denaturation (Hartl, 1996).

#### 1.4.6.3 HSP90 (stress-90)

HSP90, a member of the 'stress-90' family, is another highly conserved member of the chaperone proteins. It is an abundant protein under all growth conditions, found primarily within the cytoplasm, but expression levels can be increased by heat shock and other cellular stresses (Lindquist and Craig, 1988; Gething and Sambrook, 1992). HSP90 proteins share a molecular weight of around 90 kDa, with the exception of *E. coli* HtpG (68 kDa), which lacks a highly charged region of around 50 amino acids that is characteristic of other HSP90 homologues (Jakob and Buchner, 1994). Members of this group include HSP82 in yeast and grp 94 in humans, and this group has an affinity for a number of proteins, particularly signal transduction proteins, steroid hormone receptors, actin and tubulin. It is this broad substrate affinity that characterises stress-90 proteins as a molecular chaperones (Hendrick and Hartl, 1993). Whilst the exact function of the HSP90 group is not fully elucidated, it is thought to be related to modulation of protein activity through completion of protein folding, blocking activation and stabilising inactive proteins until they are localised in the correct
cellular compartment. Craig (1993) suggested that stress-90 is essential for correct folding and functional activation of proteins in *S. cerevisiae*. Evidence is increasing that HSP90 plays an important role in receptor regulation (signal transduction) in higher cells, through its effect in modulating protein conformation, for example steroid hormone receptor function (Boston *et al.*, 1996; Morimoto *et al.*, 1997).

## 1.4.6.4 Clp family including HSP100 (HSP 110) group of proteins

The HSP100/Clp group of proteins is part of a larger group of homologous proteins, the AAA ATPase superfamily, that have diverse roles in biological systems, some of which demonstrate stress tolerance functions but all of which promote dissolution of protein aggregates (Schirmer *et al.*, 1996). In this regard, the function of this family seems to be different from other chaperones. Their principal stress tolerance function is one of dissaggregation plus protein degradation of unfolded or abnormal proteins rather than promotion of protein folding (Boston *et al.*, 1996). They may even cause unfolding of a stably folded protein if it is tagged with a specific recognition sequence (Saibil, 2000).

Some members of the Clp (Caseinolytic protease) proteins found in *E. coli* and named for the ability to promote hydrolysis of casein, have been added recently to the chaperone classes. Originally, the Clp proteins were thought to consist of only three related proteins, Clp A, B and C, but recently the number of members has expanded (Schirmer *et al.*, 1996). The Clp proteins now include eight sub-families, Clp A, B, C, D, M, N, X and Y, and are divided into two major classes, whereas ClpP is a member of the non-homologous s14 family of peptidases (Squires and Squires, 1992; Suzuki *et al.*, 1997). ClpP is discussed separately in section 1.4.10.1.

Because of a connection to heat shock response, four proteins, Clp A, B, C and X, are of particular interest. Clp A, B and C are ATPases and members of Class 1, the class that contains two

nucleotide-binding domains. ClpX belongs to Class 2; this class is shorter and contains only a single nucleotide-binding site. HSP100 chaperones are often found in association with a protease ring: ClpA (a chaperone) is often found in association with ClpP (a non-chaperone protease) (Saibil, 2000). The complex they create is a stack of four rings, with a double-ring protease in the rniddle, flanked by an ATPase ring at either end that thereby forms a continuous cavity (Saibil, 2000). Proteolysis occurs inside the protease cavity although the cavity is only large enough to accept an unfolded protein, suggesting that the chaperone's task is to unfold proteins and pass them onto the protease (Vale, 2000). This theory was supported by research performed using green fluorescent protein (GFP) tagged with a ClpA recognition sequence at its carboxyl terminus. Not only was the stable protein conformation unfolded but the GFP was also found to be directly inserted into the protease cavity (Langer, 2000; Vale, 2000).

It would appear that HSP100/Clp chaperones are not essential for growth at any specific temperature (low, normal or high) but are highly induced at higher than normal temperatures. They are also induced by other stresses such as alcohol, suggesting that these proteins provide protection of vital cellular components from these stresses (Sanchez *et al.*, 1992). In experiments with *S. cerevisiae* it has been shown that HSP104 mutant cells, in contrast to wild type cells, were not able to clear aggregates of protein accumulated during heat stress (Sanchez *et al.*, 1992). Antibodies against HSP104 from yeast react with both *E. coli* ClpB and mammalian HSP110 suggesting a relationship between them, that is, they all belong to the Clp group of proteins (Squires and Squires, 1992).

## 1.4.6.5 Small heat-shock proteins

A group of low molecular weight proteins (approx. 16 to 30 kDa) make up the chaperone group called the small HSPs. Often not detected during periods of normal growth, they are significantly induced during periods of heat stress (Boston *et al.*, 1996). The small HSPs are an abundant group

of proteins found in the cytosol of prokaryotic cells and organelles of eukaryotes. Small HSPs have only a low level of shared homology (Jakob and Buchner, 1994) and although their functions have not been fully determined, aside from stress tolerance, they may possibly have a role in cell growth and differentiation (Morimoto et al., 1997). Small HSPs can "selectively recognise and bind to nonnative proteins, they suppress non-specific aggregation during folding and unfolding and promote functional folding in an ATP independent manner" (Jakob and Buchner, 1994; Norris et al., 1997). Experiments have shown that over-expression of the small HSPs provided increased thermotolerance to plant cells, but the mechanisms involved are yet to be elucidated (Boston et al., 1996). It is proposed however that proteins bound to small HSPs during cellular stress are released to other chaperone systems when conditions are mitigated (Saibil, 2000). Functions including protection of mRNA during heat shock, overexpression of the protein in differentiating cells and interaction with non-native proteins have also been demonstrated to be associated with small HSPs, but again the controlling mechanisms are unclear (Jakob and Buchner, 1994). However, it is their role in protein folding and unfolding reactions that clearly identifies these proteins as chaperones (Boston et al., 1996).

## 1.4.6.6 Co-chaperones (DnaJ and GrpE)

Chaperones of the HSP70 class interact with another group of proteins referred to as cochaperones. Co-chaperones DnaJ and GrpE moderate the activity of DnaK in *E. coli*. Structurally the DnaJ protein consists of an N-terminal conserved domain (called the 'J' domain) of approximately 70 amino acids. There is also a glycine-rich domain and a central domain, which are referred to as the 'G' domain and 'CRR' domain respectively and a C-terminal end.

The function of DnaJ is to stimulate the rate of hydrolysis of DnaK, perhaps by increasing the ATPase activity of DnaK, (the *E. coli* HSP70 homologue) (Kelley, 1998). GrpE appears to increase the rate of dissociation of the bound nucleotide to DnaK although there are observations that it also

increased the binding ability of DnaK. In the presence of both DnaJ and GrpE, DnaK activity can be increased markedly (up to fifty-fold) (Becker and Craig, 1994). No proteins homologous to GrpE have been found in eukaryotic cells with the exception of some similar proteins found in yeast. The effects of DnaJ and GrpE have to be balanced to optimise the equilibrium between the substrate binding and release. Within the cell this is achieved by co-regulation and expression of the genes (Bukau and Horwich, 1998).

#### 1.4.6.7 Foldases

Two classes of foldases have been identified: protein disulphide isomerases (PDI) and peptidylprolyl isomerases (PPI), which have a role in protein folding in eukaryotic cells. Although homologues have been identified in prokaryotic cells, PDI is found principally in eukaryotic cells within the lumen of the endoplasmic reticulum and its function is to catalyse the production of disulphide bonds. PDI is important for translocation and folding of proteins within mammalian and yeast systems but the role of the enzyme is yet to be fully elucidated (Boston *et al.*, 1996). Another major group of protein foldases, the PPIases, catalyse isomerisation of peptide bonds around proline residues. This reaction is normally slow in protein folding. These enzymes are found within the cytoplasm of prokaryotic and eukaryotic cells (Boston *et al.*, 1996).

## 1.4.7 Interactions between chaperone proteins

Individual families of chaperone proteins do not act without collaboration with the other chaperone proteins (Johnson and Craig, 1997). It has been established that the chaperones 'co-operate' in *de novo* protein folding using sequential actions, although it is not obligatory to follow a co-operative pathway (Lund, 1995). The model for co-operation is based on the differential specificity of the chaperones for peptides; for example, DnaK has a specificity for nascent polypeptides while GroEL cross-links to full length proteins. The model is shown in Figure 1.3. A number of observations of chaperone function during protein folding in *E. coli* have led to the formulation of a hypothesis that

DnaK/DnaJ/GrpE and GroEL/GroES cooperate in folding newly synthesised proteins (Hendrick and Hartl, 1993; Frydman and Hartl, 1996; Frydman and Höhfeld, 1997). The model proposes that DnaK and/or DnaJ recognise extended segments of nascent polypeptides, binding to them and preventing their aggregation. As the chain length increases, the polypeptide collapses to a more compact state requiring fewer stabilising molecules until the full-length chain is released, which requires the action of GrpE. While some proteins may have reached a native state, others are transferred to other chaperones for further folding rounds until the native state is reached. It has been suggested that this co-operative chaperone pathway may be bi-directional with proteins being passed back and forth between the chaperonins and HSP70 (Johnson and Craig, 1997). The interactions of chaperone activities and defining the interactions with those chaperones already detected.

## 1.4.8 Oxidative stress response

Reactive forms of oxygen can also cause stress responses in bacteria. Under optimum conditions of aerobic growth, cells can convert oxygen to a harmless end-product (water) by transfer of four electrons through the electron transport chain. However, sequential reduction to form active oxygen intermediates is possible and these products can have varying toxicity. These toxic compounds can include superoxide anion radicals, hydrogen peroxide and hydroxyl radicals (Demple, 1991). Denitrification within some bacteria produces nitric oxide, which is an end-product of a particular form of respiration, while a hypohalous acid such as HOCl is generated by the interaction of hydrogen peroxide with phagocyte peroxidases that have been shown to contribute to bio-toxicity (Miller and Britigan, 1997). Furthermore, these intermediates can react in a number of ways. For example, in low pH environments, superoxide can become protonated to form HO<sub>2</sub> that can react with itself to form hydrogen peroxide and, due to its neutral charge, is more membrane permeable

(Miller and Britigan, 1997). Aerobic bacteria, which utilise oxygen as a terminal electron acceptor, must deal with the toxic by-products that are produced during oxidative phosphorylation. Facultative and strictly anaerobic bacteria must also have mechanisms to counter the effects of oxygen exposure during normal growth that arise from transient exposure to air or hydrogen peroxide produced by other organisms. Anaerobic bacteria can be exposed to these compounds when aerobes and anaerobes grow in the same environment. Anaerobes are particularly susceptible to cellular damage because they have fewer mechanisms to cope with the presence of toxic oxidants (Miller and Britigan, 1997). For pathogenic bacteria, phagocytes are often the primary source of oxidants. Phagocytes surround the pathogenic organisms and engulf them with a portion of their cell membrane releasing superoxide and hydrogen peroxide.

Damage resulting from cellular interactions with toxic oxygen compounds can affect all cellular components. Lipid membranes are particularly susceptible and oxidative damage can result in loss of the proton motive force that drives ATP formation and nutrient uptake. Extensive membrane damage can lead to cell lysis (Miller and Britigan, 1997).

The cellular response to oxidative stress consists of two parts. The first involves avoiding contact with oxidants and the second involves preventing lethal injury once oxidants are encountered (Condon, 1987; Demple, 1991; Duwat *et al.*, 1995; Rocha *et al.*, 1996; Miller and Britigan, 1997; Mukhopadhyay and Schellhorn, 1997). Mechanisms for avoidance of phagocytic encounters have been published by Miller (1997) and include the use of a signalling system by *Yersinia* species that interferes with the phagocyte response. The ability of cells to cope with oxidants once they have been encountered, particularly the prevention of lethal injury in lactobacilli due to the exposure to hydrogen peroxide produced by starter bacteria in fermented foods, determines the survival of these cells.

Demple (1991) described three responses to oxidants: hydrogen peroxide response, the superoxide response and a stationary phase response that is protective against oxidative stress injury. These responses are associated with induction of catalase and superoxide dismutase as well as other proteins that are induced when a cell enters the stationary growth phase and during heat shock (Dernple, 1991). The broad range of proteins induced to oxidative stress indicates that, unlike heat stress, there are no specific stress response mechanisms to oxidative stress.

## 1.4.9 Mechanisms of acid tolerance and acid stress responses

Bacteria, because of their uni-cellular nature, can experience significant fluctuations in pH during a normal growth cycle. The response to exposure to acid environments has been principally studied in enteric bacteria, predominantly *E. coli* and *Salmonella* species. This is due to the increase in outbreaks of food poisoning and the link between environmental conditions such as pH and expression of virulence genes (Hill *et al.*, 1995). However, acid stress responses have been studied in both pathogens and non-pathogens, including the following organisms:

- Aeromonas hydrophila (Karem et al., 1994);
- *E. coli* (Goodson and Rowbury, 1989; Heyde and Portalier, 1990; Hickey and Hirshfield, 1990; Rowbury *et al.*, 1993; Ohara and Glenn, 1994; Small *et al.*, 1994; Leyer *et al.*, 1995; Lin *et al.*, 1995; Tsai and Ingham, 1997; Rowbury and Goodson, 1999);
- Lactobacillus species (Hood and Zottola, 1988; McDonald et al., 1990; Lorca et al., 1998; De Angelis et al., 2001);
- L. lactis ssp. lactis (Cook and Russell, 1994; Hartke et al., 1995; Hartke et al., 1996; Rallu et al., 1996);

- Leuconostoc mesenteroides (McDonald et al., 1990);
- Listeria monocytogenes (Farber and Pagotto, 1992; Kroll and Patchett, 1992; Gahan et al., 1996; O'Driscoll et al., 1996; O'Driscoll et al., 1997);
- Oral streptococci (*S. sanguis, S. gordonii, S. mutans,* and *S. oralis*) and other streptococci (*S. bovis*) (Belli and Marquis., 1991; Cook and Russell, 1994; Curran *et al.*, 1995; Quivey *et al.*, 1995; Smith *et al.*, 1996);
- S. enteritidis (Humphrey et al., 1993; Humphrey et al., 1993);
- S. typhimurium (Foster and Hall, 1991; Foster, 1993; Garcio-Del Portillo et al., 1993; Foster and Bearson, 1994; Foster et al., 1994; Lee et al., 1994; Foster, 1995; Lee et al., 1995; Baik et al., 1996; Leyer and Johnson, 1997); and
- Root nodule bacteria (*Rhizobium loti* and *Rhizobium meliloti*) (Goss *et al.*, 1990; Tiwari *et al.*, 1992; Tiwari *et al.*, 1996; Tiwari *et al.*, 1996; Correa and Barneix, 1997).

Survival at low pH depends on a number of determinants including: growth phase (Gorden and Small, 1993), adaptation response (Foster, 1995), ability to switch to less pH-sensitive metabolic pathways (Miwa *et al.*, 1997), alkalisation of the internal pH (Marshall *et al.*, 1990), as well as expression of sigma factors and pH-dependent genes (Foster and Bearson, 1994). These determinants and the role in pH homeostasis are discussed in the following sections.

## 1.4.9.1 pH homeostasis

The proton gradient is generally accepted to be inter-convertible with ATP because of a proton ATPase (H+-ATPase) and, in the absence of a specific pH sensing regulator, is probably the major method of internal pH (pHi) control (Kashket, 1987). Miwa *et al.* (1997) indicated that not only do

acid tolerant bacteria have higher base levels of H+-ATPase but they may also have a higher capacity to enhance its production, particularly during the specific conditions of a low pH in the growth medium and a high cellular concentration of intracellular ATP.

Regulation of pH<sub>i</sub> in a bacterial cell is related to the permeability of cellular membranes to protons. This is because of a passive transfer of protons across membranes due to an inwardly directed force termed the proton motive force (PMF) (Axelsson, 1993). A PMF is composed of two contributing components, the first being an electrical potential ( $\Delta \Psi$ ) where the inside of the cell is negative. The second is a pH gradient ( $\Delta pH$ ) where internal pH is more alkaline. It is generally considered that biological membranes exhibit low proton permeability because of the impermeability of the lipid bilayer (Booth, 1985; Hutkins and Nannen, 1993). Control of pHi is more likely to be controlled by an active mechanism, a membrane-bound proton pump. Lactococcus and Streptococcus species maintain an internal pH that is more alkaline than the external medium but the pH<sub>i</sub> decreases as the extracellular pH (pH<sub>o</sub>) is acidified during growth and fermentation (Kashket, 1987). Kasket also found that for the particular strain of *Lb. acidophilus* studied (3532), the limiting level of cytoplasmic pH was 4.4, when the external pH was 3.5. Most bacterial cells quickly lose the ability to grow when they are unable to maintain a near neutral pH<sub>i</sub> (McDonald *et al.*, 1990). Padan et al. (1981) noted that "cherniosmotic processes can adapt so as to maintain a constant pH<sub>i</sub> of between 6.5 and 9.5, which appears to be the range permitted by cell physiology". However, as an exception to this, each type of LAB has their own, varying, threshold of pHi below which all cellular functions are inhibited (Kashket, 1987). When fermentation end-products, such as lactic acid, are in high concentrations, they act as protonophores that perturb membrane phospholipids and increase the inward leak of protons. Weak acids in their uncharged form can diffuse easily across the membrane and dissociate within the cell, lowering the pHi. The cell cannot maintain the pH gradient against this influx and cytoplasmic re-alkalisation fails, resulting in cell death (Kashket, 1987). Cook and Russell (1994) have found that *L. lactis* and *Strep. bovis* must decrease their intracellular pH and maintain a low  $\Delta$ pH to prevent accumulation of fermentation acid anions, which are toxic to the cell. This decrease in intracellular pH potentially restricts growth due to the reduction of other activities, such as biosynthetic reactions, but the organism continues to grow below the limits of other neutrophiles.

Belli and Marquis (1991) demonstrated that when cultures of oral streptococci became acidified the organisms were able to increase the level of ATPase activity. This resulted in removal of protons from the cytoplasm and subsequent protection of pH-sensitive glycolytic enzymes. Lactobacilli in particular seem capable of tolerating a wide range of pH values (Axelsson, 1993) and it has been suggested that the lactate fermentation pathway is less sensitive to low pH than other metabolic pathways (Miwa *et al.*, 1997).

### 1.4.9.2 Porter- antiporter systems

Cation antiports play a part in the acidification of the cytoplasm during growth in alkaline medium where the antiport exchanges internal cations (Na<sup>+</sup> or K<sup>+</sup>) for external protons. A Na<sup>+</sup>/H<sup>+</sup> antiporter may serve an important role in control mechanisms and homeostasis by lowering the intracellular Na<sup>+</sup> concentration that prevents Na<sup>+</sup> toxicity and by regulating internal pH. Regulation of cation antiports is probably directly related to internal pH (Speelmans *et al.*, 1995).

## 1.4.9.3 Growth phase and the relationship to acid resistance (AR)

Acid resistance (AR) is defined as the ability of a bacterial cell to survive a two-hour exposure to pH 2.5. In bacteria AR has been found to be highly dependent on growth phase (Foster and Hall, 1990; Gorden and Small, 1993). Work on AR in enteric bacteria has shown that the acid resistance of a number of *Shigella* and *Salmonella* species, including *S. sonnei* (3421), *S. flexneri* (1a-625), *S. boydii* (14), *S. dysenteriae* (12-31), *S. enteritidis* (3407), *S. typhimurium* (79-80) and *S. typhi* (1231)

was growth phase dependent (Gorden and Small, 1993). This is possibly related to the physiological changes that occur as a cell enters stationary phase, including global regulatory changes that protect the cell from decreased availability of nutrients. Lee *et al.* (1995) found that the stationary phase sigma factor (RpoS) was required for protection of *S. typhimurium* cells from acid stress using a derivative strain of LT2.

## 1.4.9.4 The role of adaptive acid tolerance (ATR)

Foster and others (Foster and Hall, 1990; Foster, 1991; Foster and Hall, 1991; Farber and Pagotto, 1992; Foster, 1993; Foster and Bearson, 1994; Foster et al., 1994; Foster, 1995; Foster and Spector, 1995) have performed detailed research on the acid tolerance response in *S. typhimurium*. This organism, which is often implicated in cases of food poisoning, may be exposed to a wide range of environmental conditions during its life cycle, including the strongly acid conditions of the stomach. Their work has concluded that ATR in *S. typhimurium* involves a two-stage process. The two stages, designated pre-acid shock and post-acid shock, are an overlapping acid protection system triggered at different levels of acidity. Pre-acid shock is induced at an external pH of approximately 5.8 while the post-acid shock is induced at or below an external pH of 4.0. During post-acid shock a number of acid shock proteins (ASPs) are induced which are important for the subsequent survival of the organism. When a cell is transferred directly from pH 7.0 to 3.3, these ASPs are not synthesised and the cell suffers lethal injury and dies. However, when cells are prestressed at pH 5.8 then subsequently transferred to pH 3.3, cells are able to maintain pH homeostasis long enough to synthesise post-shock ASPs. Both stages are required to provide cells with maximum protection against exposure to low pH; however, the pre-shock can be substituted by a brief shock at pH 4.4 for 15 minutes. During this time, a group of ASPs are transcribed, but if cells are held beyond 15 minutes these proteins are no longer detectable and the cells are unable to withstand a subsequent low pH challenge (Foster, 1993). Approximately 50 ASPs are induced

during pre- and post-acid shock and their function is believed to be related to repair or prevention of macromolecular damage. Included in these proteins are several inducible amino acid decarboxylases that include the lysine decarboxylase (CadA) and lysine-cadaverine decarboxylase systems (CadB) (Bearson *et al.*, 1997).

Goodson and Rowbury (Goodson and Rowbury, 1989; Rowbury *et al.*, 1993; Rowbury *et al.*, 1993) found the same effect in *E. coli* (strain 1829 ColV I-K94, a derivative strain of K12), which was exposed to a number of different organic acids. When the test strain was grown at an intermediate pH (5.0) before subsequent exposure to pH 3.5, it gained a survival advantage over the control strain which was grown at pH 7-0 prior to exposure to pH 3.5. However, the extent of growth was also dependent upon the type of acid used. The work of Leyer and Johnson (1993) showed that prior acid adaptation of *S. typhimurium* (LT2) also caused cross protection against other stresses, such as heat and salt. Heyde and Portelier (1990) demonstrated that eight novel proteins were induced when a derivative strain of *E. coli* K12 was shifted from pH 6.9 to pH 4.3. These included heat shock proteins as well as a subset of unknown proteins they called acid shock proteins.

## 1.4.9.5 The role of arginine utilisation in acid tolerance

As discussed in section 1.4.9.3, the AR of an organism has been found to be dependent on growth phase, however AR also appears to be dependent upon the growth medium. Three complexmedium related AR systems are present in bacteria and their activity depends on whether the cells utilise oxidative or fermentative metabolism. Two of the three systems involve amino acid (arginine and glutamate) utilisation, while the third is dependent upon  $\sigma^s$  and is discussed in section 1.5.2.

The arginine deiminase system (ADS) provides a source of ATP derived from catabolism of arginine to ornithine, CO<sub>2</sub> and NH<sub>3</sub> (Casiano-Colón *et al.*, 1988, Mobley *et al.*, 1989). Marquis (1987) claimed that "the ADS can play an important role in the acid-base physiology of each of the bacteria

(they) studied in that it allowed for the recovery from acid stresses sufficiently severe to stop growth and glycolysis". Arginine catabolism occurs in a variety of bacteria including LAB, *Helicobacter*, *Salmonella* and *Shigella*. Figure 1.4 shows distribution pathways of arginine catabolism.

Marshal *et al.* (1990) have claimed that sufficient ammonia is generated from arginine catabolism to create a micro-environment in which *H. pylori* can survive the normally lethal pH of the human stomach.

## 1.4.9.6 Microbial ureases

Urease is an enzyme that catalyses the hydrolysis of urea to yield ammonia and carbamate, which spontaneously hydrolyses to form carbonic acid and a second molecule of ammonia resulting in a net increase in pH (Mobley and Hausinger, 1989).

Urease activity has been observed in a wide variety of prokaryotic and eukaryotic organisms with particular interest being paid to pathogens like *H. pylori* and *Y. enterocolitica* which are able to live in environments of extremely low pH (<3). This extreme acid tolerance has been attributed to the ability of these organisms to utilise urea to produce large enough quantities of ammonia to maintain an internal pH compatible with survival (Marshall *et al.*, 1990).

It has been demonstrated that acid resistance was also dependent on the growth phase as well as the presence of urea in the growth medium (de Koning-Ward and Robins-Browne, 1995). LAB appear not to use a urease pathway: evidence suggests that when de-acidifying wine, the arginine deiminase pathway, not an arginine-urease pathway, was used for arginine degradation (Liu *et al.*, 1996). This pathway has also been demonstrated in *L. oenos*, *M. arginine* and *S. lactis* (Poolman *et al.*, 1987; Sugimura *et al.*, 1990; Liu *et al.*, 1996).

Pathway	Organisms
Arginase:	Saccharomyces sp. Bacillus sp.
urea→CO <sub>2</sub> + NH <sub>3</sub>	Proteus sp.
Arginine	
R	
ornithine →glutamate	
Arginase transamidase:	<i>Klebsiella</i> sp.
Arginine→ornithine→glutamate	
Arginine deiminase:	Streptococci sp., Pseudomonas sp.,
Arginine	Lactobacillus sp., Clostridium sp.
↓ 	
NH <sub>3</sub> + citrulline	
$\downarrow$	
carbamylphosphate + ornithine	
ATP + NH3 + CO2 giulamate	
Arginine decarboxylase:	E. coli, Salmonella sp., Shigella sp.
Arginine	
NH3 + Agmatine $\rightarrow$ Urea $\rightarrow$ Putrecine	

Figure 1.4 Pathways of arginine catabolism (Abdelal, 1979; Poolman et al., 1987)

## 1.4.10 Proteases

When stressors affect a cell, which results in proteins becoming damaged, there are two possible outcomes. The first is that the molecular chaperones will aid the protein to refold into its correct conformation and the second is that the protein will undergo proteolysis and subsequently be removed. Molecular chaperones may also be involved in this protein degradation and where a protein cannot be refolded they may accelerate its destruction. The ClpP, Lon and the eukaryotic Ubiquitin system are examples of molecular chaperones acting as a protease and they are discussed in the following sections.

## 1.4.10.1 ClpP

Clp is an endopeptidase of 750 kDa that cleaves peptides in a process requiring ATP hydrolysis; its role in conjunction with the Clp proteins has been mentioned previously in section 1.4.6.1. It is a dimeric protein: ClpP, the proteolytic component, consisting of twelve sub-units. ClpP can cleave peptides when acting alone but this occurs at a slower rate and acts on shorter residue lengths (Gottesman and Maurizi, 1992). The function of ClpP may be that of a master protease which is attracted to different proteins according to the specificity of the ATP regulator, which is usually a chaperone (Gottesman and Maurizi, 1992; Suzuki *et al.*, 1997). The ClpP proteins can be found across prokaryotic and eukaryotic cells.

## 1.4.10.2 The Lon protease

The Lon (La) ATP-dependent protease is an 89 kDa protein, which was described in 1964 in *E. coli*, making it one of the first HSPs to be identified (Gottesman and Maurizi, 1992). Lon functions as an endo-protease cleaving damaged proteins into peptides of 5-20 amino acids and its association with DnaK accelerates the rate of proteolysis (Parsell and Lindquist, 1993; Missiakas *et al.*, 1996; Suzuki *et al.*, 1997).

## 1.4.10.3 The ubiquitin (Ub) system

First described in 1975, ubiquitin is a 76-residue protein that is found only in eukaryotic cells. It may be found as either free protein or bound to other cellular proteins. Ubiquitin is an abundant protein, hence the name, that is involved in cell differentiation, signal transduction, DNA repair, vesicular transport and stress responses (Varshavsky, 1997). Damaged or otherwise abnormal proteins can be recognised by the Ub system and are subsequently degraded. A multi-Ub chain becomes linked to the protein after it has recognised a degradation signal. The protein-Ub complex is then degraded by the 26S proteasome. This is possibly because the signals that start the degradation process are usually buried within a normal folded protein are now uncovered in a damaged protein (Suzuki *et al.*, 1997). There is no known prokaryotic equivalent to this protein.

## 1.4.11 The effect of stressors on cell membrane fatty acid composition

Apart from induction of chaperones, a number of other mechanisms have been proposed which may allow cells to respond to unfavourable changes in the growth medium. These are often associated with the cell wall and/or membrane and include changes to the fluidity of the lipid membrane and a build-up of stress damaged proteins. When challenged by a change in temperature, the major response in membrane composition is an alteration to the fatty acid profile. Adaptive changes in membrane lipid and fatty acyl compositions, that enable bacteria to cope with a wide range of growth temperatures, have been studied in a number of food-poisoning bacteria including clostridia (*Clostridium botulinum* NCIB 4270 and *C. butyricum ATCC 6015*) and *Listeria* species. When *C. botulinum* was subjected to a decrease in growth temperature, from 37°C to 8°C, a number of changes were noted. While the major fatty acids at each growth temperature were 14:0, 16:0 and 16:1, it was found that the proportion of each varied according to the growth temperature. The proportion of 14:0 and 16:1 increased from 16.4 to 37.5% and 10.5 to 22.5% respectively, while the proportion of 16:0 dropped from 40.3 to 19.2%. It was also noted that there

was a decrease in the proportion of cyclopropane fatty acids (15:0cyc and 17:0cyc) from 7.3 to 0.5% (Evans *et al.*, 1998). Similarly, when *L. monocytogenes* was challenged by lowering the growth temperature from 30°C to 7°C, two general changes were noted. Firstly, lipids with a shorter chain length became more abundant and secondly there was a small increase in unsaturated lipids (Russell *et al.*, 1995). These changes make the membrane a less rigid structure at the lower temperature and alteration to the fatty acid profile help the cells maintain both the proton-motive force and uptake of nutrients (Russell *et al.*, 1995).

Apart from changes in temperature, acid habituation has been demonstrated to have an affect on fatty acid profiles. Acid habituation is the phenotypic response that refers to the increased resistance to extreme pH conditions after adaptation to sub-lethal acidic environments. Acid habituation (AH) occurs when *E. coli* cells are grown at pH 5.0. Once habituated they are able to stand a low pH challenge better than a control grown at pH 7.0 (Goodson and Rowbury, 1989; Bearson et al., 1997). According to Bearson (1997) AH, which is a reversible effect, probably involves a number of protein synthesis-dependent and -independent steps as well as DNA repair and the phosphate specific porin, PhoE. The effect of AH on five strains of E. coli (MJR, M23, R172, R91 and 0157:H) that exhibited wide variability in their intrinsic tolerance to exposure to low pH was examined by Brown (1997). A marked shift in fatty acid composition was demonstrated after AH. Significant percentages of mono-unsaturated fatty acids were found to be converted to cyclopropane fatty acids or were replaced by saturated fatty acids. It was also found that in a nonhabituated, intrinsically acid tolerant strain of E. coli, 0157:H a food poisoning strain, had higher levels of cyclopropane fatty acids compared to strains with a low level in intrinsic acid tolerance (Brown et al., 1997). It was concluded that acid tolerance in E. coli correlated with the content of cyclopropane fatty acids in the membrane. Cyclopropane fatty acids (CFAs) are found in around thirty-five genera of bacteria including Bifidobacterium, Lactobacillus, Salmonella, Rhizobium and

Helicobacter species (Grogan and Cronan, 1997). Three CFAs predominate in bacterial membrane lipids:

- cis-9,10-methylene hexadecanoic acid (cyclic 17:0);
- cis-11,12-methylene octadecanoic acid (lactobacillic acid or cyclic 19:0 11c); and
- cis-9,10-methylene octadecanoic acid (dihydrosterulic acid or cyclic 19:0 9c) (Grogan and Cronan, 1997).

Lactobacillic acid was first demonstrated in 1950 in a culture of *Lb. arabinosus* and is a 19-carbon cyclopropane analog of *cis*-vaccenic acid that has been demonstrated in several *Lactobacillus* and *Lactococcus* species (Johnsson *et al.*, 1995).

CFAs have a number of physiological roles within the cell membrane including membrane fluidity and chemical stability; CFAs also appear to be less reactive to low pH and oxidation than other fatty acids (Grogan and Cronan, 1997). CFAs may reduce the effects of temperature on membrane fluidity by increasing the melting temperature of the membranes and disrupting the acyl chain packing that ensures fluidity of the membrane. This change in rigidity appears to reduce permeability of the membrane to protons (Brown *et al.*, 1997).

## 1.4.12 General and specific stress response of *Bacillus subtilis*

In *B. subtilis* it has been noted that a variety of stressors, including heat, salt, ethanol and starvation stress, can induce the same set of proteins, designated the general stress proteins or GSPs. In the work of Hecker *et al.* (1996) it was proposed that this general response serves a non-specific protective function allowing adaptation to stress and starvation. In addition to the general response, it was found that each single type of stress also induced a specific set of proteins called the specific stress proteins or SSPs. These proteins were induced by and provided a specific protective

function against a unique stressor such as heat. This protective function is well characterised in *E. coli.* 

# 1.5 Interaction of chaperones, regulation of the stress response and the effect of cross protection on survival

Molecular chaperones of all cells have been conserved through evolution and although there is a great variation between organisms, the way they respond to stress is very similar. However, the methods of regulation of stress responses is not as conserved and it may be concluded that each organism must detect and regulate stress responses according to its own particular environment. Special mention will be made of an inverted repeat, named CIRCE, found in the regulatory regions of many genes coding for heat shock proteins involved in regulation of shock responses at both DNA and mRNA levels.

## 1.5.1 Regulation of the heat shock response

The heat shock response is characterised by a rapid, stress induced, transcription of genes encoding HSPs after exposure to elevated temperatures. The appearance of misfolded proteins is thought to be the specific stress signal that triggers the transcriptional response (Morimoto *et al.*, 1997).

During heat shock in *E. coli* two regulons are expressed (Missiakas *et al.*, 1996; Missiakas and Raina, 1997). The first, referred to as the 'classical heat shock regulon' is negatively regulated by the presence HSPs, particularly DnaK and is under the control of the  $\sigma^{32}$  transcription factor. The  $\sigma^{32}$  transcription factor is an RNA polymerase sub-unit that encodes products that deal with removal and refolding of damaged proteins. The second encodes proteins that take care of damage in the periplasmic space and is controlled by the  $\sigma^{E}$  transcription factor. However, in Gram positive bacteria, which do appear to contain the Gram negative heat shock specific regulators, the question

of heat shock regulation remains to be fully determined. In *L. lactis* several heat shock promoters are found adjacent to a palindromic sequence called the CIRCE element (Rallu *et al.*, 1996). CIRCE stands for controlling inverted repeat of chaperone expression (Zuber and Schumann, 1994) and consists of 9 bp separated by a 9 bp spacer (TTAGCACTC-N<sub>9</sub>-GAGTGCTAA) and is highly conserved within the groEL, dnaK and dnaJ operons (Segal and Ron, 1996). The probable function of CIRCE is as an operator site to which a repressor binds. The protein coded by *orf39*, the first gene in the dnaK operon of *B. subtilis*, was found to bind the inverted repeat at the DNA level and to serve as the repressor in *B. subtilis* (Yuan and Wong, 1995). *Orf39* was recently renamed *hrcA* and homologues are found in several other bacteria (Baldini *et al.*, 1998).

## 1.5.2 Regulation of stress responses other than heat

### 1.5.2.1 Regulation of acid stress responses

Expression of a number of bacterial genes, including stress response genes, have been associated with changes in pH. Iriarte *et al* (1995) cloned the *rpoS* gene from *Y. enterocolitica* and demonstrated that the production of an enterotoxin by this organism is partly regulated by pH. The *rpoS* gene is involved in expression of stationary phase proteins in other bacteria and is associated with stress response proteins in *Salmonella* species and acid tolerance in *Shigella* species (Lee *et al.*, 1994; Lee *et al.*, 1995; Waterman and Small, 1996). The connection to acid tolerance was noted in a comparison of virulent strains of *S. typhimurium* (UK1, SL 1344 and SL 14028s) to an avirulent strain of *S. typhimurium* (LT2). It was found the virulent strains were more acid tolerant than the avirulent strain. This was traced to a mutation in the rpoS allele of LT2 and when rpoS alleles were swapped between organisms, virulence was also swapped (Lee *et al.*, 1995). RpoS also controls the synthesis of a set of proteins up-regulated during oxidative stress (Hochman, 1997). In studies of mutants of *E. coli* K12 and *S. typhimurium* LT2, the *cadBA* operon was found to be regulated by external pH and the gene products of this operon increase levels of amino acid

decarboxylases. These products in turn help control internal pH (Auger *et al.*, 1989; Neely *et al.*, 1994; Bearson *et al.*, 1997). Coll *et al* (1994) found in *E. coli* K-12 a gene (*nmpC*), encoding an outer membrane protein that acts as a porin, that was dependent on external pH, while Slonczewski *et al* (1987) reported two genes, which were induced by decreasing internal pH. The relationship between these pH-dependent genes to those responsible for stress protein production in LAB is very unclear, and studies such as those described above are yet to be carried out in LAB. However, it has been suggested that in Gram positive bacteria, particularly the non-pathogenic LAB, regulation AT may be connected to ppGpp, the guanosine tetraphosphate responsible for the stringent response (Grogan and Cronan, 1997).

## 1.5.2.2 Regulation of stationary phase and starvation response

*E. coli* and *Salmonella* species have a sigma factor  $\sigma^s$ , a general stress response mediator encoded by the stationary phase gene, *rpoS*. It is induced during stationary phase and nutrient starvation and regulates virulence factors as well as regulating the acid tolerance response, which was discussed in the previous section (Finlay and Falkow, 1997). These different stressors signal their presence to the cell in different ways and differentially affect *rpoS* translation and/or stability. The cellular level of  $\sigma^s$  increases in response to a wide range of stress signals including heat stress and during normal growth is highly unstable and rapidly degraded (Muffler *et al.*, 1997).

CFA formation has been shown to be induced at the same time as the induction of the stringent response, although CFA formation has also been associated with transition to stationary phase. Transcription rates of CFA genes have been found to increase with the accumulation of stationary phase proteins, the products of the *rpoS* gene (Grogan and Cronan, 1997). Synthesis and accumulation of RpoS proteins are determined in part by ppGpp, the guanosine tetraphosphate responsible for the stringent response (Grogan and Cronan, 1997).

## 1.5.3 The effect of cross protection on survival responses

Induction of stress proteins by one form of stress can protect against other forms of stress. There are many examples of this in the literature: (Farber *et al.*, 1992, Leyer *et al.*, 1993, Hartke *et al.*, 1994, Hartke *et al.*, 1995, Jørgensen *et al.*, 1995, Quivey *et al.*, 1995, Baik *et al.*, 1996, Flahaut *et al.*, 1996, Flahaut *et al.*, 1997, Bearson *et al.*, 1998). A good illustration of this phenomenon can be seen in the results of work performed on *L. lactis* cells. Actively growing cells of strain IL1403 were forced into stationary phase; they were starved by growing the cells in the presence of limited glucose. The cells became significantly more resistant to heat, ethanol, acid and osmotic stress than a control culture exposed to the same stress while in exponential phase (Hartke *et al.*, 1994; Hartke *et al.*, 1995). It should be noted however, that cross-protection to stress varies markedly between LAB species (van de Guchte *et al.*, 2002).

Leyer and Johnson demonstrated that acid adaptation induces cross protection against a wide range of stressors promoting survival of *S. typhimurium* LT2 and *E. coli* 0157:H7 (Leyer and Johnson, 1992; Leyer and Johnson, 1993; Leyer *et al.*, 1995; Leyer and Johnson, 1997). Kim *et al.* (2001) have also demonstrated a variable response in log-phase cultures of *Lb. acidophilus* pre-exposed to heat, NaCl and bile stress. Their work demonstrated significant improvement in survival of cultures pre-exposed to bile stress when subsequently heat stressed but not salt stressed, and pre-exposure to heat stress did not protect against bile or salt stress.

There have been a number of outbreaks of food poisoning involving *E. coli* 0157:H7 where this preadaptation may have led to increased viability of pathogens in under-cooked burgers, cider and salami (Garren *et al.*, 1997; Semanchek and Golden, 1998).

In a study of stress responses in *L. œnos* (now renamed *Oenococcus œnos*) Garbay and Lonvaud-Funel (1996) found that exposure to one stress, either heat, acidity, ethanol or toxic fatty acids, caused a rapid loss of viability. However, the surviving cells were more able than a control to survive in wine and induce malolactic fermentation. These results suggest that any type of stress will adapt this organism for one specific stress, the exposure to the harsh conditions of wine. Additionally it appeared that the profile of proteins up-regulated during each type of stress was similar, regardless of the type of stress (Garbay and Lonvaud-Funel, 1996).

However not all studies support the view that induction of a basic set of stress proteins (preadaptation) will protect the cell against all stressors. The reaction of Enterococcus faecalis to bile salts, acid and heat shock has been studied by Flahaut (1996; 1998). It was found that exposure to bile salts or heat resulted in cross protection against heat or bile salts but when heat adapted cells were acid stressed no cross protection was found to have occurred. Indeed it was noted that bile salt stressed cells were more sensitive when later exposed to acid stress (Flahaut et al., 1996). In a later study of hydrogen peroxide stress response, it was found that the type of pre-treatment (acid, salt or increased temperature) used to stress Ent. faecalis ATCC 19433 resulted in a variation in the strength of the response. Acid pre-stress resulted in the greatest response with 11.8% surviving minute hydrogen peroxide challenge compared to 0.2% survival of the control cells. The least response came from a thermal pre-treatment where only 4.0% survived compared to 1.2% of the control cells. Ethanol and alkaline pre-stress appeared to have no influence on the outcome of the hydrogen peroxide tolerance (Flahaut et al., 1998). Adaptation to stresses other than acid has also not typically induced a high level of acid tolerance. Bacteria, therefore, may recognise some stressors as specific, for example heat stress, invoking induction of a specific set of heat stress proteins while others, such as acid stress, may be recognised as general stress signals (Bearson et al., 1997).

## 1.6 Aims of this thesis

Previously, the study of stress responses in bacteria has been confined principally to Gramnegative bacteria such as *E. coli* and *S. typhimurium*. While these investigations are far from complete, the directions of these studies have been towards gaining an understanding of the basic biological mechanisms involved in stress tolerance as well as obtaining an insight into the processes that pathogenic bacteria use to invade a host organism. This thesis attempts to increase the understanding of the response to various stressors in a group of non-pathogenic bacteria that are in common use in the dairy industry, by demonstrating that a number of proteins are upregulated during acid and heat stress.

Specifically the aims of the project were:

- to characterise strains at a biochemical and molecular level to validate taxonomy;
- to determine the nature of the stress response in *Lb. paracasei* and *B. animalis* syn. *lactis* following exposure to heat stress and acid stress;
- to compare these responses to known responses in other bacterial species;
- to characterise the proteins produced by their physical properties; and
- to gain information regarding the proteins in terms of sequence and possible related function.

# Chapter Two

## MATERIALS AND METHODS

## 2.1 Materials

#### 2.1.1 Bacterial strains and plasmids

Refer to Table 2.1. for a list of bacterial strains and plasmids used in this work. Strains originally obtained from the Commonwealth Scientific and Industrial Research Organisation (CSIRO) were a gift from Dr Martin Playne. The Australian Starter Culture Centre provided *Lb. acidophilus* type strain ASC 388. Dr Duncan Rouch from The University of Melbourne, Gilbert Chandler Campus, provided *Lb. casei* strain GCRL-163. *Bifidobacterium animals* (syn. *lactis*) strain Bb-12 was a gift from the Chr. Hansen Group, Bayswater, Australia. *E. coli* strains were obtained from commercial sources.

## 2.1.2 Chemicals

All chemicals used were analytical reagent (AR) grade and supplied by either Crown Scientific Pty Ltd (Australia) or Sigma-Aldrich (Australia) except where another supplier is specified. The sources are shown in Appendix 1. Abbreviations used for these chemicals are listed on page xix. All water used was distilled or deionised. Distilled water was prepared using Milli-RO Water purification system or deionised water using Milli-Q Ultrapure Water System (Millipore).

## 2.2 General Methods

Unless otherwise stated bacterial cultures and protein extracts were centrifuged at 4°C in a Beckman J2-HS high-speed centrifuge. Other solutions were centrifuged in an Econospin H1323, FSE Du Pont, Australia. Eppendorf tubes were centrifuged in an Eppendorf centrifuge Model 54515C, supplied by Crown Scientific Pty Ltd.

Table 2.1 Source of bacterial strains and plasmids used in this study.

Strain/Plasmid or Victoria University culture collection number	Relevant characteristics	Source/reference.
E. coli		
TG2 (TG2 is stored at VUT as VUN 0105)	SupE hsdΔ5 thi Δ(lac-proAB) Δ(srl- recA)306::Tn10(tet <sup>RI</sup> ) F'[traD36 proAB <sup>+</sup> lacl <sup>q</sup> lacZΔM15	(Sambrook <i>et al.</i> , 1989)
Lactobacillus strains		
C1 VUP12000 Lb. acidophilus	Type strain	CSIRO 2400, ATCC 4356
C2 VUP12001 Lb. acidophilus		CSIRO 2401 Originally obtained from Chr. Hansen Group.
C3 VUP 153-2345-2		Victoria University culture collection
C4		CSIRO 2403
C5 VUP12002 Lb. acidophilus		CSIRO 2404 Mutant of 2401
C6 VUP12003 Lb. acidophilus		CSIRO 2405 Mutant of 2401
C7		CSIRO 2406
C8 VUP 153-2457		Victoria University culture collection
C9 VUP 152-2352		Victoria University culture collection
C10 VUP12004 Lb. acidophilus		CSIRO 2409 Mutant of 2401
C11 VUP 152-2313-2		Victoria University culture collection
C12 VUP 153-2415		Victoria University culture collection
C13 VUP 12005 Lb. acidophilus		CSIRO 2412/21 Originally obtained from Chr. Hansen Group
C14 VUP 153-2345-3		Victoria University culture collection

.

C15		CSIRO 2422
C16 VUP12006 Lb. paracasei		CSIRO 2413/46 Originally obtained from Chr. Hansen Group
C17 VUP12007 Lb. acidophilus		CSIRO 2415
C18		Bornhofen 200g Acidophilus yoghurt
C19		Yoplait YoPlus yoghurt 'A+B' plain yoghurt
C20		Bornhofen 200g Acidophilus yoghurt
C21		Bornhofen 200g Acidophilus yoghurt
C22		Yoplait YoPlus yoghurt 'A+B' plain yoghurt
C23		Yoplait YoPlus yoghurt 'A+B' plain yoghurt
C24 VUP 152-2313-3		Victoria University culture collection
C25		CSIRO 20097
VUP12012 Lb. jugurati		Victoria University culture collection
ASC 388 Lb. acidophilus	Type strain alternately sourced compared to VUP 12000	Australian Starter Culture Research Centre
GCRL-163 <i>Lb. casei</i>		The University of Melbourne, Gilbert Chandler Research Laboratories,
Bifidobacterium strains		
Bifidobacterium animalis (syn. lactis) strain Bb-12		Originally obtained from Chr. Hansen Group
Plasmids		
pBluescript SK+	2958-bp cloning vector derived from pUC 19. Ampicillin resistant	Stratagene
pGEM T-easy		Promega

Optical densities were measured using an Ultraspec III UV/Vis spectrophotometer (Pharmacia). pH was measured using a model PHM220 pH meter, supplied by Radiometer Pacific. Microtiter trays (96 well) were scanned using a Multiscan 8 channel photometer (ELIZA plate reader) supplied by Flow Laboratories, Titertek.

## 2.2.1 Buffers

Buffers were prepared according to the methods described in Biochemistry LabFax (Chambers, 1993). All other buffers, not described here, are described in the relevant sections.

## 2.2.1.1 Tris buffer pH 7

Tris(hydroxymethyl)aminomethane (Tris-HCL) buffer solution was prepared from appropriate concentrations of Tris base. The pH was adjusted to 7.0 using HCl.

## 2.2.1.2 Citrate-phosphate buffer

Stock solutions of citrate phosphate buffer were prepared from 0.2 M citric acid ( $C_6H_8O_7$ ) and 0.2 M sodium dihydrogen phosphate (NaH<sub>2</sub>PO<sub>4</sub>).

## 2.3 Microbiological methods

## 2.3.1 Media

Unless otherwise stated all media was supplied by Oxoid Australia and sterilised by autoclaving at 121°C, 103 kilopascals for 15 minutes. The pH of each batch was checked and if necessary, adjusted immediately before autoclaving.

## 2.3.1.1 de Man, Rogosa and Sharpe (MRS) broth and agar

MRS broth, developed by de Man, Rogosa and Sharpe to enhance the growth of lactic acid bacteria, was used to support the growth of bacteria used in this study (De Man *et al.*, 1960). MRS agar plates were prepared by the addition of 1% agar to MRS broth prior to autoclaving. Unless otherwise stated, incubation of agar plates was carried out at 37°C under anaerobic conditions.

Anaerobic conditions were obtained using a two litre anaerobic jar with palladium catalyst anaerobic system (Oxoid).

The formula of MRS per litre was: 10g peptone, 10g 'Lab-Lemco' powder, 5g yeast extract, 20g glucose, 1mL Tween 80, 5g sodium acetate, 2g tri-ammonium citrate, 0.05g magnesium sulphate (MgSO<sub>4</sub>.7H<sub>2</sub>0), 0.25g manganese sulphate (MnSO<sub>4</sub>.4H<sub>2</sub>O), 2g dipotassium hydrogen phosphate. Water was added to bring the volume to 1 litre. MRS agar was made by addition of 1.5% agar to MRS broth prior to sterilisation. When MRS broth was used to culture *B. animalis* (syn. *lactis*), 0.05% cysteine hydrochloride was added prior to autoclaving. MRS broth with citrate-phosphate buffer was titrated with 0.2 M citric acid (to adjust to pH 4, 4.3 or 5.3) or 0.2 M sodium dihydrogen phosphate (to adjust to pH 7), according to the final pH required, in double strength MRS broth prior to autoclaving. The pH of the media was confirmed during and after the growth cycle by measuring the pH of aliquots removed from the culture vessel.

## 2.3.1.2 Lactobacilli fermentation broth (LFB)

LFB was made up from MRS base without Lab-Lemco powder or glucose. The pH was adjusted to 6.3 before the broth was bottled and autoclaved (115°C for 15 minutes). To each 100 mL, final volume, 10 mL of 10% filter sterilised carbohydrate and 2 mL of 0.2 % chlorophenol red was added.

## 2.3.1.3 Luria-Bertani (L-B) broth

L-B broth was prepared from 10g Tryptone, 5g Yeast extract, 5g NaCl, distilled water to 1 litre. L-B agar was made by addition of 1.5% agar to L-B broth prior to sterilisation.

## 2.3.1.4 Double strength YT broth

Double strength YT broth contained 3.2 g Tryptone, 2g yeast extract, 1g NaCl made up to 200 mL in distilled water.

## 2.3.1.5 Biolog Lactic Acid Suspension (BLA™) broth

BLA broth was prepared from 5.6 g BLA broth, 0.16 mL Tween 80 and made up to one litre with distilled water. The formula of BLA broth was not made available by the manufacturer (SPL Diagnostics). BLA was used to study the ability of lactobacilli to utilise various carbohydrate sources.

## 2.3.1.6 Ampicillin plates

One litre of L-B agar was prepared as described above. After sterilisation, the agar was cooled to 55°C, and then 5 mL of ampicillin (20 mg/mL) was added prior to pouring plates. Ampicillin (20 mg/mL) was prepared from 98% Ampicillin-Sodium salt (Aminobenzylpenicillin) (Sigma) and sterilized by filtration through a 0.22 µm filter, Millipore.

## 2.3.1.7 Colour selection plates

One litre of L-B agar was allowed to cool to 55°C then 5 mL of ampicillin (20 mg/mL), 2.5 mL of X-gal (20 mg/mL) and 1 mL IPTG (200 mg/mL) were added prior to pouring plates. IPTG was prepared from 99% isopropyl-D-thiogalactopyranoside (Sigma). X-GAL was prepared from 95% 5-bromo-4-chloro-3-indolyl N-acetyl-D-galactopyranoside (Sigma).

## 2.3.1.8 Nutrient broth

Nutrient broth was prepared from Lab-Lemco powder (1.0g), yeast extract (2.0g), peptone (5g), and sodium chloride (5g) made up to 1 L in deionised water. Nutrient agar was made by the addition of 15g of agar to 1 L of nutrient broth.

## 2.3.2 Preservation of cultures

## 2.3.2.1 Frozen stocks in glycerol

Stock cultures of LAB were made from the overnight growth on MRS agar plates. Using cotton tipped swabs the organisms were suspended in 2 mL of preserving solution (50% glycerol:50% double strength MRS broth) by gently rotating the swab in the medium. The vials were stored at

 $-20^{\circ}$ C for short-term storage, or  $-70^{\circ}$ C for long-term storage. When the culture was required the vial was removed from the freezer and cotton tipped swabs used to transfer approximately 200  $\mu$ L of the contents to MRS broth or agar for overnight culture.

### 2.3.2.2 Frozen stocks using ceramic beads

The 'protect' bead system (Protect Bacterial Preservers) utilises sterile vials that contain a number of porous beads and a cryo-preservative fluid (Technical Service Consultants, Ltd). When microorganisms are inoculated into this fluid they are bound to the porous beads. The following procedure, provided by the manufacturer, was used to prepare frozen culture stocks for use in fermentation experiments.

Overnight growth of LAB were removed from an MRS agar plate using a cotton tipped swab then rotated gently in a 'protect' vial to resuspend the bacteria in the preservative solution and coat the beads with bacteria. The vial was tightly closed and inverted 4-5 times before the preservative solution was removed as completely as possible. The vials were then stored at -20°C for short-term storage or -70°C for long-term storage. When the culture was required the vial was removed from the freezer and held on ice while two beads were removed and transferred to MRS broth for ovemight incubation. *E. coli* were preserved using a similar method; however, nutrient broth and agar was substituted for MRS broth and agar.

## 2.3.3 Gram stain

Glass slides coated with heat fixed bacteria were flooded with crystal violet (10g/L) for one minute. The stain was washed from the slide with gently running tap water then an iodine solution was added for one minute. The iodine solution was prepared from 4 g NaOH dissolved in 25 mL distilled water to which 20 g iodine and 1 g potassium iodide were added and dissolved. This solution was mixed and made up to 1 litre. After washing in tap water, the slide was briefly decolourised with

acetone-alcohol prepared by adding 300 mL of acetone added to 700 mL 95% ethanol. The slide was washed again with water and counterstained with carbol fuchsin (20 g/L), again for one minute. After washing in tap water the slide was air-dried.

## 2.3.4 Method of biomass determination

Biomass was determined by optical density (OD) and dry biomass. OD was determined by measuring absorbance at 600 nm ( $A_{600}$ ). Samples were diluted to keep the OD measurements in the linear range between 0.1 and 0.8.

To determine any biomass, ten rnL polypropylene screw-capped centrifuge tubes (ProScience) were dried overnight at 55°C then weighed to four decimal places. To each tube exactly 10 mL of test sample was added and then the tubes were centrifuged at 16,000 x g for ten minutes to obtain a firm pellet. The liquid supernatant was decanted and the tube briefly turned upside down on absorbent paper to drain the remaining supernatant. Each tube was then placed upright in a 55°C drying oven overnight. The tubes were examined for complete dryness and then re-capped with its own separately dried cap before being re-weighed to four decimal places. The weight difference was determined against a media only control and this value is referred to as the dry biomass. Biomass studies were performed in replicate.

## 2.3.5 Carbohydrate utilisation test (rapid test)

This method was developed, as part of the work described in this thesis, to identify *Lb. acidophilus* presumptively and provide a 'metabolic fingerprint' for each of the strains being tested using this method. It was designed to be simple, cheap and rapid. Identification was based upon the ability of organisms to utilise fifteen different carbohydrates (2% w/v) supplied in BLA broth, see Table 2.2. Change in absorbance was determined over six hours in 96-well, flat bottomed, microtitre plates using a microtitre plate reader (Titertek).

## Table 2.2 Carbohydrates used in rapid testing.

All carbohydrates were prepared 2% w/v in BLA broth. Carbohydrates that were filter-sterilised rather than autoclaved are marked with an asterisk.

Arabinose	Lactose	Raffinose
Cellibiose	Maltose *	Rhamnose
Fructose	Mannitol	Sorbitol
Galactose	Mannose *	Sucrose
Glucose	Melizitose	Xylose *

## 2.3.5.1 Carbohydrate substrates

The carbohydrate medium consisted of BLA broth plus one of the fifteen different carbohydrates. Carbohydrates were weighed and added to broth before autoclaving, however some could not be autoclaved and were filter-sterilised using 0.22  $\mu$ M filters. These are marked with an asterisk in Table 2.2. The media was dispensed in 200  $\mu$ L aliquots into each well of a 96-well rnicrotitre plate. Only one carbohydrate was tested per plate.

## 2.3.5.2 Preparation of cultures and testing

Unknown test or known control strains, were grown from a single colony plated on MRS agar. After overnight incubation, a cotton-tipped swab was used to gather 2-3 isolated colonies, which were then added to 2 mL of BLA broth lacking any carbohydrate source. This broth was diluted to achieve an optical density of 0.8 to 1.0 at 600 nm. Using a pre-determined plan of well positions as a guide, the unknown test strains and control strains (50  $\mu$ L) were used to inoculate a microtitre plate containing the substrate being tested. Samples, standards and controls were run in duplicate.

Controls in each tray included:

- blanks (BLA broth plus inoculum but no carbohydrate source); and
- BLA broth, with carbohydrate but without inoculum, which was used as a sterility control

The tray was placed into a microtitre reader with a 450 nm filter and the instrument was blanked against BLA broth, without carbohydrate source but with inoculum added. The absorbance (A<sub>450</sub>) of each of the well positions for every tray was then read. A lid was placed over the top of the tray before being wrapped in cling film, this reduced evaporation from the wells. Wrapped plates were placed into an incubator, set at 37°C, for six hours. The plates were passed through the microtitre reader a second time and the change in absorbance calculated. These changes were graded using

a threshold change of 100%. A positive result was a change in absorbance of at least double the starting absorbance, while a negative result was a change in absorbance less than double the starting absorbance. Strain identification was based on comparison of these results to published profiles (Hansen and Mocquot, 1970; Holt *et al.*, 1994; Hammes and Vogel, 1995; Vandamme *et al.*, 1996).

## 2.3.6 Selection of acid tolerant strains

A microtitre plate method was used to measure the changes in absorbance read at 450 nm (A<sub>450</sub>) of cultures grown at eight randomly selected pH values (1.6, 2.4, 3.0, 3.8, 4.8, 5.6, 6.3, 7.8 and 8.1). Fifty  $\mu$ L inoculums, standardised to an initial absorbance value of 0.8 at 600 nm (OD<sub>600</sub>), were added to 200  $\mu$ L of pH adjusted MRS broth. Growth temperature for all pH values was 37°C. Absorbance was measured at zero, 6 and 24 hours using a microtitre plate reader. The extent of growth was determined by calculating specific growth rates in each time interval, using the formulae described in section 2.3.8.1.

## 2.3.7 Fermenter operation

One- and two-litre bioreactors (fermenters) manufactured by Braun and Applikon or an in-house system 'Bio-Reactor Under Computer Evaluation' (BRUCE) developed by Dr T. Ueno at Victoria University, were used as a tool to study the growth of *Lb. acidophilus* and *B. animalis* (syn. *lactis*) under optimal and sub-optimal conditions. All the fermenters consisted of a supported glass reaction vessel, either one- or two-litre, which was fitted with a stainless-steel head plate. The head plate had several connections including a pre-mixed gas inlet/outlet, water inlet/outlet, sample collection port, liquid entry port, fixed thermometer pocket and stirrer assembly. Additionally there were access ports for a pH sensor, level control, foam control and a dissolved oxygen sensor. The gas inlet port was used as the inlet through which dissolved oxygen within the fermenter broth was controlled. It was also used to aid in mixing and aeration. The water inlet/outlet, which is a closed

system, was used for heat exchange. The fermenter vessel was filled with culture medium and autoclaved prior to use for 30 minutes at 121°C. When being prepared for operation the fermenter was connected to all cables and tubing and the set points (temperature, pH, dissolved oxygen etc.) adjusted. Water flow, acid/base pumps and gas flow were then switched on and when all parameters had reached the desired set points, the fermenter was inoculated. Standard set points for all fermenter runs included the stirrer rate (150 rpm) and gas flow rate (0.1 L/min).

Innocula were prepared as follows. Two 'Protect' beads coated with the test organism were transferred to 20 mL of MRS broth for 6-8 hours at 37° C. After this time 10 mL of the above culture was transferred to 200 mL MRS broth and incubated overnight at 37°C without mixing. The contents were then centrifuged (10 min at 10,000 x g) and the supernatant decanted leaving approximately 20 mL of pellet and supernatant. During the centrifuge time the fermenter system was flushed with nitrogen and pH-adjusted with either 2M HCl or 2M NaOH to the desired starting set point. Sufficient uninoculated broth was withdrawn and refrigerated to use as a blank during spectrophotometer readings. Using a syringe, the pellet was aseptically transferred through an appropriate inlet port on top of the fermenter reaction vessel so that the contents reached the desired starting optical density ( $OD_{600}$  of 0.13 to 0.23).

#### 2.3.8 Determination of growth rates

#### 2.3.8.1 Manual method for the determination of optimum temperature and pH

Absorbance values were used to construct growth curves for *Lb. acidophilus* and *B. animalis* (syn. *lactis*). Data, from duplicate runs, were collected hourly over a minimum of six hours, from fermenters held at constant temperature (37°C) and pH (2, 3, 4, 5, 6, 7, or 8). Data was also collected from fermenters that were held at constant pH (6.3) and temperature (12, 22, 27, 32, 37, 42, 45, 47, or 51°C). These values were graphed as Ln (absorbance) versus time. The slope of the tangent to the steepest part of the growth curve was used to determine the number of
generations per hour. Specific growth rates ( $\mu$ ) were calculated over two time periods using the formulae below, where X<sub>2</sub> was OD<sub>600</sub> of sample collected at time (t<sub>2</sub>) and X<sub>1</sub> was OD<sub>600</sub> of sample collected at time (t<sub>1</sub>).

The generation time was calculated from the plot of Ln  $OD_{600}$  against time and was the time taken for the absorbance to double (McMeekin *et al.*, 1993). The average of the duplicate results was then shown graphically (1/rate versus pH or temperature) and used to determine the optimum growth conditions. The optimum growth conditions were at the peak of the resulting curve.

#### 2.3.8.2 Software based method for the determination of growth rates

A modified computer software program, developed by Viega and Gutiérrez (1991) was used to graphically display growth curves of each batch fermenter run. The program automatically calculated specific growth rates between each time point as well as the maximum specific growth-rate ( $\mu_{max}$ ). These were compared to results obtained using the the above method to determine specific growth rates.

#### 2.3.9 Determination of factors that affect growth

#### 2.3.9.1 Determination of the effect of lactate on the ability to initiate growth

Sodium lactate (CH<sub>3</sub>CH(OH)COONa) was used to prepare a 10 g/L stock solution of lactate in MRS broth. The stock solution was further diluted in MRS broth to the following concentrations: 5, 2.5, 1.25, 0.63 and 0.31 g/L. MRS broth without added lactate was used as a control. Fermenters inoculated with *Lb. acidophilus* were sampled every hour for at least eight hours and again at 24 hours. Maximum specific growth rates were calculated for each lactate concentration. The graph of  $\mu_{max}$  versus lactate concentration was used to determine the effect of lactate on the ability of *Lb*.

*acidophilus* to initiate growth. The pH was controlled at 6.3 during growth and temperature set at 37°C.

#### 2.3.9.2 Determination of acid resistance by challenge at lethal pH values

The ability of *Lb. acidophilus* strains to survive at two lethal pH values was tested using a modification of a method used by Gorden and Small (1993). In the modified method MRS broth was substituted for L-B broth. Acid resistance was determined from strain survival, where if greater than or equal to ten percent of challenged bacteria survived for 2 hours at pH 2.5 the strain was considered acid resistant. As indicated by Gorden and Small "assay conditions were based upon those of a normal fasting stomach, i.e. a pH less than 3.0 and a gastric emptying time of less than two hours".

Acid resistance was determined by taking 50  $\mu$ L of a twenty-four hour culture of the test strain, which was then added to 10 mL of MRS broth pre-warmed to 37°C. The pH of the test broth was adjusted to either pH 2.0 or 2.5 before the addition of the 50  $\mu$ L sample, while a control broth was set at pH 6.3. Duplicate 10  $\mu$ L samples, were taken from each broth at time zero and every 30 minutes for six hours and again at 24 and 48 hours if required. Lawn cultures were prepared from 100  $\mu$ L taken from serial dilutions of the sample and incubated for 48 hours at 37°C. The colonies on each plate were counted and the percentage of survivors calculated relative to the time zero.

#### 2.3.9.3 Impact of growth phase on pH resistance

Acid resistance in relation to growth phase was determined by preparing an actively growing broth culture from which an aliquot was removed hourly and challenged at pH 2.5 for two hours. Two mL of an overnight culture was used to inoculate 100 mL of MRS broth at 37°C and pH 6.3. Fifty  $\mu$ L of this culture was then added to 10 mL of MRS broth (pH 2.5) pre-warmed to 37°C. Duplicate 10  $\mu$ L samples, were taken from each broth at time zero and every 30 minutes for six hours and again at

24 and 48 hours if required. Lawn cultures were prepared from 100  $\mu$ L taken from serial dilutions of the sample and incubated for 48 hours at 37°C. The colonies on each plate were then counted and the percentage of survivors calculated. The process of sampling from the growing culture was repeated hourly for eight hours. The results were compared to determine which phase of growth produced the highest number of survivors.

#### 2.3.9.4 Viable cell counts

Viable cell counts of cultures were performed by preparing serial dilutions of the culture in MRS broth. Aliquots of the dilutions were spread plated onto MRS agar and incubated for 48 hours at 37°C under anaerobic conditions. The colonies on each plate were then counted and plates with between 50 and 500 colonies used to calculate total viable cell counts after correction for dilution factors.

#### 2.4 Analytical methods

#### 2.4.1 Sample collection and treatment

#### 2.4.1.1 Sample collection for optical density, carbohydrate estimation and SDS-PAGE

At specified times 50 mL of fermenter contents was removed from the fermenter and transferred to a screw-capped 50 mL plastic centrifuge tube (Falcon). The mix rate of the fermenters was set to a higher setting for a short time before sampling (300 rpm for 30 seconds). Two mL of this sample was used for estimation of optical density read at 600 nm, while the rest of the sample was immediately placed into an ice filled cooler to stop any further metabolic reactions. When cooled, the tube was centrifuged until a soft pellet was formed (8,000 x g for 4-7 minutes). The supernatant was carefully removed for estimation of carbohydrates and an aliquot was transferred to the freezer (-20°C) until required.

According to the optical density of each sample a calculated volume of Tris buffer (pH 7.0) was added to the pellet to make each sample pellet equivalent to an optical density of 15.0. This pellet was treated as described in section 2.5.1 and frozen for later testing.

#### 2.4.1.2 Collection of samples for analysis of fatty acid methyl esters

This method is based on that used by Decallone *et al.* (1991) where the cultures used in the evaluation of fatty acid methyl esters are in the stationary phase of growth. Unless otherwise stated, ten mL of culture, grown for twenty-four hours, was removed from the fermenter and centrifuged (ten min at 8,000 x g). The pellet was resuspended in 1 mL of 1.2 N NaOH in water:methanol (1:1, v/v) and vortexed before the tubes were placed in a boiling-water bath for thirty minutes. The tubes were then cooled to room temperature before 2 mL of 6 N HCl in methanol was added (325:275, w/v). The tubes were incubated at 80°C for ten minutes then cooled to room temperature before 1.25 mL of ether-hexane (1:1, v/v) was added to each tube. After continuous rotation for ten minutes the tubes were centrifuged (2 min at 8,000 x g) then the upper phase was removed and placed in a second tube. Three mL of NaOH (0.3 N) was added to the second tube and centrifuged (10 min at 8,000 x g) before the organic phase was separated and placed into gas chromatograph sample vials. The vials were frozen (-20° C) until required.

#### 2.4.1.3 Collection and preparation of samples for 2-D SDS-PAGE

Cells from a sample of the fermenter contents were removed, centrifuged and concentrated as described in section 2.4.1.1. This cell suspension was placed, together with an equal volume of glass beads (0.10 to 0.11 mm diameter), into a stainless steel homogeniser bottle (B. Braun) then secured in the homogeniser (B. Braun MSK Cell Homogeniser) and homogenised for thirty seconds. This was repeated after the bottle had been cooled on wet ice for 2 minutes. The temperature of the contents of the bottle was not allowed to rise above 4°C. The homogenate was then centrifuged (10 minutes at 20,000 x g in a refrigerated centrifuge) to obtain a protein rich, cell

free extract, which was then dialysed to remove salts. The samples were concentrated by centrifugation in a 5 kDa molecular weight cut-off filter (Sartorius) then frozen (-20°C) until required.

For later work, this method was modified for use in a Mini-BeadBeater-8<sup>™</sup> (BioSpec Products supplied by Daintree Scientific). Up to eight, 2 ml screw-capped tubes replaced the single stainless steel cylinder used in the method described above. Using a volume ratio of 1:1:2 (0.5 g wet weight cell biomass:40mM Tris buffer pH 7:0.10 mm glass beads) each tube was loaded up to 2/3 capacity. The Mini-beadbeater-8 was run according to manufacturers instructions for 2 bursts of one minute. The tubes were cooled on wet ice between bursts. Following treatment the tubes were centrifuged (16,000 x g) and the supernatant removed and frozen until required.

To enhance the extraction of hydrophobic proteins, solvent and detergent assisted extraction of proteins was also used. Tris buffer (200  $\mu$ L of 40mM, pH 7) was added to each tube containing 0.5 g wet weight cells. The tube was mixed briefly, and then 500  $\mu$ L of solvent or detergent was added to the tube, which was placed at 4°C for 5 minutes. The contents were homogenised in the Minibeadbeater-8 as described above. After centrifugation at 16,000 x g, both the aqueous and solvent layers (if present) were carefully removed. The solvent layer was dried down under nitrogen then made up in an equivalent volume of 40mM Tris buffer pH 7. Both the aqueous and solvent extracted samples were frozen for further studies. Solvents tested included: octanol, acetone and chloroform. Detergents tested included CHAPS (4% w/v in 40 mM Tris pH 7) and Triton X-100 (2% w/v in 40 mM Tris pH 7).

#### 2.4.1.4 Dialysis and desalting

Dialysis tubing was prepared by boiling 10 to 15 cm strips for 10 minutes in 3 L of 2% (w/v) NaHCO<sub>3</sub> and 1 mM EDTA (di-sodium salts), pH 8.0, then rinsed thoroughly in distilled water and then the strips were boiled again for ten minutes in 1 mM EDTA. The tubing was rinsed inside and

out in distilled water, then stored in distilled water at 4°C. Protein samples were dialysed against two changes of 100 volumes of distilled water at 4°C.

De-salting of small volumes (less than 100  $\mu$ l) was performed in Micro Bio-Spin P-6 columns (BioRad) according to manufacturer's instructions. These tubes had stated salt retention rates of 99% and a size exclusion limit of 6,000 Daltons.

#### 2.4.1.5 Concentration of samples

Sartorius centrifuge tubes (SM13229) containing a membrane with a 5,000 Dalton molecular weight (MW) cut off were used for concentrating extracts. Two mL samples were transferred to the tube and centrifuged according to the supplier's instructions. The concentrated protein sample was collected after 20 minutes centrifugation at 8,000 x g and 4°C.

#### 2.4.1.6 Samples for protein estimation

One mL of bacterial suspension was removed from the fermenter and centrifuged to obtain a pellet. The pellet was resuspended in an Eppendorf tube in 1.0 mL of 4.6M NaOH and boiled for ten minutes. The suspension was kept at room temperature prior to immediate analysis.

#### 2.4.2 Analysis of carbohydrates and metabolic products

#### 2.4.2.1 Reducing sugar estimation

Reducing sugar (RS) estimation was performed by the dinitrosalicylic acid (DNS) method (Miller, 1959). The DNS reagent contained 3' 5'-dinitrosalicylic acid (10g), phenol (2g), sodium sulphite (0.5g) and sodium hydroxide (10g) made up to 1 L in distilled water. Samples previously collected from a fermenter run were defrosted and diluted appropriately. Sample, water blank or standard (1 mL) was added to 1 mL of DNS reagent, mixed and placed in a boiling water bath for exactly five minutes. The tubes were then placed in a cold water bath and 5 mL of distilled water added. The

absorbance at 570 nm was measured using the zero standard as blank. Standards were prepared from a standard solution (2.0 g/L) and contained from zero to 2.0 g/L of glucose.

# 2.4.2.2 High performance liquid chromatography (HPLC) estimation of sugar and metabolic products

HPLC was used to determine the extent of glucose utilisation as well as concentration of the metabolites produced during bacterial fermentation. The concentration of the following compounds was measured: glucose, lactate, acetate, maltose, formate and ethanol using a refractive index detector on a Varian HPLC. The system was fitted with an organic acid analysis column (Aminex HPX-87H ion exclusion column 300 x 7.8 mm) supplied by BioRad. The mobile phase was 5 mM  $H_2SO_4$  with a flow rate of 0.60 mL/min and the run time was 24 minutes. Lactate standards (0.1 g/L to 0.5 g/L) were prepared from a stock solution (0.5 g/L) prepared using sodium lactate. Standards (0.2 g/L to 1.0 g/L) for the remaining compounds were prepared from stock solutions of 1.0 g/L.

#### 2.4.3 Estimation of protein concentration (Lowry assay)

This is a modification of the protein assay of Lowry *et al.* (1951). Previously prepared samples of cells were diluted 1/20 (25 $\mu$ L sample, 25  $\mu$ L 4.6M NaOH, 450 $\mu$ L 0.45M NaH<sub>2</sub>PO<sub>4</sub>). To each 0.5 mL of diluted sample or standard, 2.5 mL of Lowry reagent was added and mixed by vortex. Lowry reagent was prepared immediately before use by adding 50 mL of 2% Na<sub>2</sub>CO<sub>3</sub> in 0.1M NaOH to 1.0 mL 0.5 % CuSO<sub>4</sub>.5H<sub>2</sub>O in 1% Na-K-Tartrate pH 9. After exactly 10 minutes, 0.25 mL of dilute Folin/Ciocalteau (F/C) reagent (2 mL concentrated F/C plus 3 mL deionised water) was added and the solution immediately vortexed before being allowed to stand at room temperature for 30 minutes. The absorbance was measured at 750 nm. Standards containing zero to 40  $\mu$ g of bovine serum albumin (BSA) were prepared from a 100  $\mu$ g/mL standard solution.

#### 2.4.4 Fatty acid methyl ester (FAME) profiles

FAME profiles were obtained using a Varian Gas Chromatograph model 3400, which included a Cx auto-sampler and Star chromatography software, version 4.01. The column was a BPX-70 (25.0 metres x 250 microns supplied by SGE). One μL of stored sample was injected by auto-sampler. GLC reference standards (Sigma) were used to identify unknown peaks by comparison to the retention time of standard peaks. The following fatty acids were used as standards: 8:0, 10:0, 12:0, 14:0, 14:1, 16:0, 16:1, 17:0, 18:0, 18:1, 18:2, 18:3, 19:0, 20:0, 20:1, 20:4, 21:0, 22:0 (FAME standard, Sigma). The run time was 20 minutes. The carrier gas was helium with a flow rate of 1 mL/min. An FID detector was used to detect peaks. Column temperature was ramped from 150°C to 200°C at 2°C/min.

#### 2.5 Protein separation methods

Laemmli was the first to describe an improved method of electrophoresis in 1970 using a detergent (Sodium Dodecyl Sulphate, SDS) and mercaptoethanol, which was the forerunner of the methods now in common use (Laemmli, 1970). Primarily the method described below was used to demonstrate proteins that were up-regulated during cell stress; however, the technique was also used to compare whole cell protein patterns, as studies have shown a high correlation between protein profiles and DNA-DNA hybridisation (Hertel *et al.*, 1993). Samples for identification were harvested at stationary phase after growth in MRS broth at 37°C.

#### 2.5.1 Polyacrylamide gel electrophoresis (PAGE)

One dimension SDS-PAGE of proteins was performed according to the method of Laemmli (1970) using a vertical slab system (BioRad's Modular Mini-PROTEAN II electrophoresis system and PROTEAN II SLAB SYSTEM or C.B.S. Scientific Triple wide system, Silenus Labs). The solutions described in Table 2.3 were prepared as stock solutions.

# Table 2.3 Stock solutions for SDS-PAGE.

Stock Solution	Method of preparation
Acrylamide stock solution (29.1% w/v Acrylamide, 0.9% w/v NN'- Methylenebis-acrylarnide)	29.1 g Acrylamide 0.9 g NN'-Methylenebis-acrylamide dissolved in 60 mL distilled water then made up to 100 mL with distilled water The solution was stored in an amber coloured bottle at 4°C for up to two weeks.
1.875 M Tris-HCl Buffer pH 8.8	56.8 g Tris dissolved in 150 mL distilled water then carefully adjusted to pH 8.8 with 5N HCl then made up to 250 mL with distilled water. Solution was stored at 4°C for up to four weeks.
1.25 M Tris-HCI Buffer pH 6.8	37.8 g Tris dissolved in 150 mL distilled water then carefully adjusted to pH 6.8 with 5N HCl then made up to 250 mL with distilled water. The solution was stored at 4°C for up to four weeks.
10% Sodium Dodecyl Sulphate (SDS) solution	10.0 g SDS dissolved in 85 mL distilled water then made up to 100 mL with distilled water. The solution was stored at room temperature for up to four months.
10% Ammonium Persulphate solution	Approximately 0.05 g ammonium persulphate made up in ten times volume of distilled water. The solution was made up daily as required.
10X Electrode Buffer Tris-Glycine	144.2 g Glycine 30.3 g Tris 10.0 g SDS dissolved in 800 mL distilled water then made up to one litre with distilled water. Solution was stored at room temperature for up to two months. Single strength buffer was made by diluting 10X buffer one in ten with distilled water.
TEMED	N,N,N',N'-tetramethylethylenediamine 99% pure

Homogenous polyacrylamide gel sandwiches, 12% acrylamide unless otherwise stated, were poured using a clamp assembly held in place on a casting stand. Discontinuous gels were prepared using a stacking and resolving gel. The volume of each of the stock reagents used to prepare minigels is described in Table 2.4. Larger gels were prepared by adjusting volumes according to the system size. The first three solutions used to prepare the resolving gel, (acrylamide, water and pH 8.8 buffer) were gently mixed and degassed under vacuum for 15 minutes. The remaining solutions, SDS, TEMED and ammonium persulphate were added with gentle swirling between additions. The gel was cast immediately into a vertical slab glass sandwich, then layered with butanol water (butan-2-ol: water, 1:1) and left from 30 minutes to 12 hours to polymerise at room temperature. The stacking gel was prepared in the same manner as the resolving gel and cast over the resolving gel after the removal of the butanol water. A 0.75 mm teflon comb was inserted into the acrylamide mixture to form wells (1, 10, 15 or 25 wells) that later held the samples being analysed. The stacking gel was allowed to polymerise for one hour before the gel was fitted into the buffer tank containing single strength electrode buffer and loaded with samples and standards.

#### 2.5.1.1 Sample preparation

Samples for 1D SDS-PAGE were boiled for five minutes with an equal volume of double strength loading buffer. Double strength loading buffer was prepared from 2.5 mL of 1.25 M Tris-HCl pH 6.8 buffer, 1.0 g SDS, 2.5 mL 2-mercaptoethanol, 5.8 mL glycerol (87%) and 5 mg Bromophenol Blue made up to 25 mL in distilled water. The buffer was aliquoted into Eppendorf tubes in one mL amounts and stored frozen for up to 6 months. After boiling, the samples were centrifuged (ten minutes at 16,000 x g) and the protein rich supernatant was then removed and either frozen until required (-20°C) or loaded immediately onto the gel.

Table 2.4 Volume of stock reagents for SDS-PAG
--

Reagent	Resolving gel (volume required)	Stacking gel (volume required)
Acrylamide stock solution	3.2 mL	0.8 mL
Distilled water	3.09 mL	3.6 mL
Tris-HCL pH 8.8	1.6 mL	
Tris-HCL pH 6.8		0.5 mL
10 % SDS	80 μL	50 μL
TEMED	4 μL	9 μL
10% Ammonium Persulphate	27 μL	22 μL

Low weight molecular weight standards, prepared at the same time, were boiled in loading buffer for 45 secondss, according to the manufacturer's (BioRad) instructions. Electrophoresis was performed at 150 V (constant voltage) until the bromophenol blue dye front reached the bottom of the gel sandwich, then the gel was stained by either Coomassie blue or Silver stain as described in section 2.5.3.

#### 2.5.2 Polyacrylamide gel electrophoresis (2-D)

A pre-formed immobilised pH gradient on an 11 cm plastic support strip (Immobiline DryStrip pH 4-7 or pH 3-10, Amrad Pharmacia Biotech) was rehydrated overnight by placing the strip in a sterile screw-capped test tube containing 10 mL of rehydration solution. Rehydration solution was prepared fresh from the following reagents: 24.0 g of urea, 0.25 mL of Triton X-100, 75 mg of dithiothretol (DTT) and 0.10 mL of 1.0 mol/L acetic acid, made up to a final volume of 50 mL of water. The next day the strip was removed from the test tube and the prepared sample applied to the strip. Samples were prepared by defrosting, measuring the protein concentration and diluting to obtain 4-10 µg of protein in sample buffer (1:5). Sample buffer was prepared from 24 g of urea, 1.0 mL of 2-mercaptoethanol, 1.0 mL of pharmalyte 3-10 (Amrad Pharmacia Biotech), 0.25 mL of Triton X-100 and a few grains of bromophenol blue made up to a final volume of 50 mL. The strip was focused in three stages using a Multiphor II electrophoresis unit with 2-D application kit according to the manufacturer's instructions (Amrad Pharmacia Biotech). After focussing, the strip was removed and equilibrated in solutions A and B (ten minutes in each solution). The two equilibration solutions, A and B, were prepared from Tris-HCL stock (Tris 61 g, 300 mL distilled water, adjusted to pH 6.8 with 1.0 mol/L HCl, made up to 1 L in distilled water). Both solutions contained; Tris-HCl stock 20 mL, urea 72 g, glycerol 60 mL, SDS 2 g and were made up to 200 mL with distilled water. Additionally, equilibration solution A contained DTT (25 mg /10 mL) and equilibration solution B contained iodoacetamide (0.45 g/10 mL). Later 2-D SDS-PAGE studies also used both 7 cm and

18 cm strips. These strips were re-hydrated with the sample added to the re-hydration buffer, which improved resolution of the protein spots. Focussing times varied according to the size of the strip, but in general times were set at the high end of the range of times suggested by the manufacturer (Pharmacia). The second dimension of separation was performed by placing the equilibrated strip onto either a precast 8-18% gradient gel (excel gel, Amrad Pharmacia Biotech), or a 12% vertical gel sandwich. Protean II or triple wide gels were prepared as described in 2.5.1, with the exception that volumes were increased for use in the larger systems. The second dimension was run for 120 volt hours (600 V, 40 mA, 10 W) at constant current before the plastic support strip was removed. The gel was then allowed to run until the bromophenol blue dye front had reached the lower edge of the acrylamide gel.

#### 2.5.3 Visualisation of protein bands on polyacrylamide gels

At the completion of 1-D and 2-D electrophoresis, gels were placed in fixing solution (80 mL ethanol, 20 mL acetic acid, 100 mL distilled water) for at least 30 minutes. Depending on the protein concentration of the band or spot of interest, Coomassie blue (milligram quantities of total sample) or silver staining (microgram quantities of total sample) was performed.

#### 2.5.3.1 Coomassie blue stain

Solutions used for Coomassie staining are described in Table 2.5. After fixing, the fixing solution (water:95% ethanol:glacial acetic acid, 5:4:1 v:v) was replaced with staining solution for at least 40 minutes. The gel background was then de-stained in de-staining solution until the bands were clearly visible against the blue background (around one hour). The gel was then transferred to distilled water to complete the destain (12-24 hours). The gel was placed in preserving solution for 20 minutes and then photographed and dried between cellophane sheets using the BioRad gel drying system according to the manufacturer's instructions.

# Table 2.5 Solutions for Coomassie blue stain of SDS-PAGE gels.

Solution	Method of preparation
Staining solution	1.5 g Coomassie blue R250
	105 mL Ethanol
	30 mL Acetic Acid
	made up to 300 mL with distilled water and mixed overnight
De-staining solution	210 mL Ethanol
	60 mL Acetic Acid
	made up to 600 mL with distilled water
Preserving solution	15 mL Glycerol
	53 mL Ethanol
	15 mL Acetic Acid
	made up to 150 mL with distilled water

### Table 2.6 Solution used for silver staining of SDS-PAGE gels.

Solution	Method of preparation		
Incubation solution	60 mL Ethanol		
	8.2 g anhydrous Na acetate		
	1.04 mL 25%w/v Glutaraldehyde $^{\alpha}$		
	0.4 g Na thiosulphate		
	made up to 200 mL with distilled water		
Silver solution	0.2 g Silver nitrate		
	40 $\mu$ L Formaldehyde $^{\alpha}$		
	made up to 200 ml with distilled water		
Developing solution	5.0 g Sodium carbonate		
	20 $\mu$ L Formaldehyde $^{\alpha}$		
·	made up to 200 mL with distilled water		
Stop solution	2.92 g Sodium EDTA made up to 200 ml with distilled		
	water		
Preserving solution	20 mL Glycerol made up to 200 mL with distilled water		

 $^{\mbox{\tiny \alpha}}\mbox{these}$  reagents were added immediately before use.

#### 2.5.3.2 Silver staining

After fixing for at least 30 minutes, the fixing solution was replaced with incubation solution for one hour. Solutions for silver staining are described in Table 2.6. The gel was then washed (three washes of twenty minutes) in distilled, deionised water before being placed in silver solution for 40 minutes. The silver solution was then removed and replaced with developing solution. The bands were allowed to develop (usually 2-10 minutes dependent on the protein concentration of the bands of interest). Further development of the bands was stopped by the addition of a stop solution for 5-10 minutes. The gel was washed (2 x 5 minutes) in distilled water then placed in preserving solution for 20 minutes before being photographed and dried between cellophane sheets using the BioRad gel-drying system according to the manufacturer's instructions.

#### 2.5.3.3 Determination of molecular weight and protein band densities

SDS-PAGE gels and Western immunoblots were scanned using an enhanced laser densitometer (UltroscanXL, Pharmacia LKB) or a Gel-Pro analyzer (Gel-Pro version 3.0, Media Cybernetics) to determine the molecular weight of specific protein bands and their density relative to the other bands present.

#### 2.5.4 Western blotting

Solutions used in Western blotting are described in Table 2.7. Detection of specific GroE and DnaK proteins was made using the Western blotting method. After SDS-PAGE, proteins were transferred by electroblot to a nitrocellulose membrane (BioRad) using the following method. The SDS-PAGE gel to be blotted was sandwiched with the nitrocellulose membrane between 6 sheets of 3MM chromatography paper (Whatman). The gel/membrane sandwich was then placed into a transblot transfer cell (BioRad) and filled with transfer buffer. The proteins were transferred to the membrane using the following conditions: 15V/100 mA at 15°C overnight. Following transfer, the gel was stained with Coomassie blue.

# Table 2.7 Stock solutions used in Western blotting technique.

Solution	Method of preparation
Tris buffered saline (TBS)	0.05 M Tris pH adjusted to pH 7.5
Tris buffered saline with Tween 20 (TBST)	TBS with 0.02% (v/v) Tween 20
Blocking solution	TBST with 1% (w/v) gelatine
Transfer buffer	39 mM glycine, 48 mM Tris base, 0.037% SDS, 20% (v/v) methanol in 1000 mL water

An absence of bands confirmed the completeness of the protein shift to the membrane. The membrane was then blocked for one hour in blocking solution then washed twice in TBS-T before being incubated for one hour in either a  $10^{-5}$  dilution of a rabbit anti-*E. coli* GroEL (Sigma Immuno Chemicals) or a  $10^{-5}$  dilution of a rabbit anti-*E. coli* Dna-K (DAKO). After two further washes in TBS-T the membrane was incubated for one hour in  $10^{-4}$  diluted goat anti-rabbit antibody (BioRad). Heat shocked *E. coli* cells were used as a positive control. *E. coli* cells were heat-shocked by growth at  $42^{\circ}$ C for 30 minutes. The blot was visualized using a solution containing 10 mL of methanol, 30 mg of 4-chloro-1-napthol (Sigma Immuno Chemicals) and 35 µL of hydrogen peroxide in 50 mL of TBS according to the supplier's instructions and scanned using an enhanced laser densitometer (UltroscanXL, Pharmacia LKB).

#### 2.5.5 Isoelectric focussing (IEF) using a Rotofor cell

The Rotofor preparative IEF cell was prepared and run according to the manufacturer's instructions (BioRad). The anode was filled with 25 mL acidic electrolyte (0.1 M H<sub>3</sub>PO<sub>4</sub>) and the cathode was filled with basic electrolyte (0.1 M NaOH). The system was temperature controlled at 4°C during each run by running an external coolant through a central cooling finger.

Crude protein mixtures prepared by homogenisation or bead-beating (section 2.4.1.3) were dialysed against distilled water then mixed with carrier ampholytes to a final concentration of 2% v/v to make a final volume of 55 mL. The protein-ampholyte solution was injected into the sample chamber and large air bubbles removed. Constant power (12 W) was applied to the Rotofor until the current had stabilised for one hour. Each run took approximately four hours. A harvesting box was then connected to the system and the fractions in each of the 20 sample lanes quickly and simultaneously removed by vacuum into 12 x 75 mm glass tubes. The pH of each of the fractions

was determined and then the fractions were analysed by 1-D SDS-PAGE for the presence of the proteins of interest.

#### 2.5.6 Continuous elution electrophoresis

The model 491 Prep Cell (BioRad) was used to purify crude protein mixtures prepared by homogenisation (section 2.4.1.3). Protein samples were electrophoresed through a cylindrical Omstein-Davis (Tris/chloride/glycine) discontinuous buffer system for native-PAGE where the mobility of proteins depends on both size and charge. As proteins pass through the gel they separate into bands and are eluted from the gel and passed into a collection chamber where they are mixed with an elution buffer and drawn away by the action of a peristaltic pump. The separated proteins then pass into a fraction collector in 1 mL aliquots. A 30% acrylamide/bis stock solution was used to prepare a 10 cm long cylindrical tube of 10% acrylamide. Each 10 mL contained: 4.15 mL deionised water, 2.5 mL resolving gel buffer (1.5 M Tris-HCL 8.8) and 3.33 mL acrylamide/bis 30% stock solution and was polymerised with 50 µL APS and 5 µL TEMED using the method described for preparation of 1-D SDS-PAGE. The resolving gel was topped with a 1cm, 4% stacking gel made up with 6.15 mL deionised water, 2.5 mL resolving gel buffer (0.5 M Tris-HCL 6.8) and 1.33 rnL acrylamide/bis 30% stock solution per 10 mL. The gel was polymerised with 50  $\mu$ L APS and 10  $\mu$ L TEMED using the method described for preparation of 1-D SDS-PAGE. Samples were prepared in sample buffer (0.0625 M Tris-Cl, pH 6.8, 25% glycerol, 0.012% bromophenol blue). The electrode running buffer, pH 8.3, was made up of 30.3 g Tris base and 144 g glycine in 1 L deionised water. The preparative cell was run at 10 watt constant power (40 mA and 300 volts) at 4°C. The elution flow rate was set at 0.75 to 100  $\mu$ l per minute and fractions were collected following the appearance of the dye front. Run time was around 7 hours.

# Table 2.8 Solutions used for electro-blotting proteins to PVDF membrane.

Solution	Method of preparation
CAPS stock solution	22.13 g of 3-[cyclohexylamino]-1-propanesulfonic acid in 900 mL water
CAPS transfer buffer for electro-blot cell	200 mL CAPS stock solution added to 200 mL Methanol in 1600 mL water
Amido Black 10B stain	Amido Black 10B 1g Isopropanol 25 mL Acetic acid 10mL in 65 mL water

#### 2.5.7 N-terminal sequencing

Solutions for preparing proteins for sequencing are described in Table 2.8. Protein samples were electrophoresed on 1-D or 2-D SDS-PAGE. Protein bands were electro-blotted using the technique described for Western blotting (2.5.4) to poly vinylidene diflouride (PVDF) membrane using a 3-[cyclohexylamino]-1-propanesulfonic acid (CAPS) buffer and visualised using an Amido Black 10B stain. Bands of interest were excised and sent to either The Australian Proteome Analysis Facility at Macquarie University or LaTrobe University, Department of Biochemistry.

#### 2.5.8 Analysis of protein sequences

Protein sequences were accessed and analysed through the Australian National Genomic Information Service (ANGIS) using the Internet site <u>www.angis.su.org.au</u> and through The Protein Information Resource (PIR) using the Internet site <u>http://www-nbrf.georgetown.edu/pirwww/.</u>

#### 2.6 Molecular Biology methods

Unless otherwise stated molecular biology methods were performed essentially as described in Molecular cloning: a laboratory manual (Sambrook *et al.*, 1989).

#### 2.6.1 Preparation of plasmid DNA from bacterial cells

This method was used for the isolation of plasmid DNA from *E. coli* using an alkaline lysis method. An overnight bacterial culture of plasmid-bearing bacteria, which had been grown overnight in L-B broth containing ampicillin was gently swirled and 1.5 mL transferred into an Eppendorf tube and centrifuged for 2 minutes at 16,000 x g. The supernatant was discarded and another 1.5 mL of the culture poured into the same tube and centrifuged again. The supernatant was discarded. Eighty  $\mu$ L of solution 1 (50 mM glucose, 25 mM Tris-HCI (pH 8.0), 10 mM EDTA (pH 8.0)) was added into each tube and the pellet resuspended by vortexing for a few seconds. Twenty  $\mu$ L of lysozyme solution 1 containing 10 mg/mL lysozyme) was added then mixed by gentle inversion. Two hundred  $\mu$ L of solution 2 (0.2 M NaOH, 1% SDS) was added to the tube. The contents were briefly mixed by inversion and stored on ice for 5 minutes. Solution 3 (150  $\mu$ L of 3M sodium acetate, pH 5·2) was added and the tube mixed by gentle inversion before being placed on ice for 5 minutes. The tube was centrifuged for 5 minutes at 16,000 x g and the supernatant (approximately 400  $\mu$ L) transferred to a fresh Eppendorf tube. DNAase-free RNAase A solution (5  $\mu$ L of a 10 mg/mLsolution) was then added to the supernatant and incubated at 37°C for 15 minutes. The tube was transferred to ice and 450  $\mu$ L of solution 4 (Tris-equilibrated phenol:chloroform:isoamyl alcohol mixture, 25:24:1) which was then mixed by gentle inversion. The tube was centrifuged for 5 minutes at 16,000 x g and the supernatant transferred to another tube leaving the protein interphase behind.

Two volumes of chilled 100% ethanol (stored at -20°C prior to use) was added to the supernatant and gently mixed by inversion then placed at -70°C for ten minutes. The tube was centrifuged for ten minutes at 16,000 x g then the supernatant was decanted and the tube inverted, allowing the tubes to dry over paper towel for a few minutes. The pellet was dissolved in 50  $\mu$ L of sterile distilled deionised water and stored at -20°C until required.

#### 2.6.2 Quantification of nucleic acids

This technique was used to determine the concentration and purity of nucleic acids. The nucleic acid sample being tested was diluted 1/50 with sterile distilled, deionised water. The sample was gently mixed before being transferred to one of two optically matched quartz cuvettes. The second quartz cuvette was filled with distilled water and the absorbance at 260 nm (A<sub>260</sub>) set to zero. The absorbance of the sample was then measured. The DNA concentration of the sample was then calculated using the following formula:  $\mu$ g DNA/mL = A<sub>260</sub> multiplied by 50 multiplied by the dilution factor. 1 A<sub>260</sub> unit = 50  $\mu$ g DNA/mL

#### 2.6.3 Agarose gel electrophoresis

Agarose gel electrophoresis was used to separate DNA fragments according to size. Unless otherwise stated 1% agarose gels for electrophoresis were prepared in Tris-acetate buffer (TAE). A stock solution (50 X) of TAE was prepared by mixing 242 g Tris base, 57.1 mL glacial acetic acid, 100 mL 0.5 M EDTA (pH 8.0) and making up the solution to 1 L with water. A working TAE solution was prepared by diluting 1 part in 50 of water.

Agarose particles were melted by heating on 100% power in a microwave oven (Sharp) until the solution just boiled and when cooled (50°C), 3  $\mu$ L of ethidium bromide solution (10 mg/mL) was added per 50 mL of solution. A gel was formed by pouring the agarose solution into a tray, using an eight-place comb to prepare wells for holding samples (BioRad Miniwide system). After the gel had set (twenty minutes), the comb was removed and samples loaded under running buffer. TBE running buffer was prepared from 10 x TBE stock which contained 108 g Tris Base, 55 g boric acid, 8.3 g EDTA per litre (pH 8.3). The gels were electrophoresed at 110V. Standards run at the same time were selected according to the expected size of DNA bands and supplied by Progen and MBI Fermentas.

#### 2.6.4 Preparation of plasmid vector for cloning

DNA digested by the method described in section 2.6.7.2 was ligated to the plasmid vector pBluescript SK+ using the following method. The plasmid vector was prepared by digestion with the enzyme *BamH1* (MBI Fermentas). The solutions, listed in Table 2.9, were placed into an Eppendorf tube and incubated at 37°C for one hour. After digestion, an equal volume of phenol:chloroform:isoamyl-alcohol (25:24:1) was added to the tube, mixed and then the tube was spun for 5 minutes at 14,000 x g.

# Table 2.9 Reagent mixture for preparation of plasmid vector prior to cloning.

Solution	Volume
plasmid DNA	25 $\mu$ L containing 10 $\mu$ g of DNA
Distilled water	56.5
10x buffer for <i>BamH1</i> enzyme	10 µL
Alkaline phosphatase (1U/ $\mu$ L)	1 μL
BamH1 (10 U/µL)	7.5 μL
Bovine serum albumin 100 mg/mL	7.5 μL

The supernatant was then transferred to another tube and 200  $\mu$ L of 100% ethanol and 10  $\mu$ L of sodium acetate (3M at pH 5.2) was then added to precipitate the DNA. The tube was then certrifuged and the DNA pellet dissolved in 15  $\mu$ L of sterile distilled water.

#### 2.6.5 Preparation of competent bacterial cells

*E. coli* TG2 cells were treated to induce a 'competence' phase that allowed the cells to take up DNA from the external medium. Two beads of TG2 cells (frozen stock) were transferred to 20 mL of L-B broth and incubated overnight at 37°C. One mL of the overnight culture was transferred into 100 mL of L-B broth and incubated at 37°C on a shaking incubator (200 rpm) until the OD<sub>600</sub> was 0.40 to 0.45. The culture was centrifuged in sterile 50 mL Falcon tubes for five minutes (8,000 x g at 4°C) and the supernatant discarded. The cells were then resuspended in 25 mL of 100 mM MgCl<sub>2</sub> using gentle pipetting then placed on ice for twenty minutes. The cells were centrifuged and resuspended as before in 100 mM CaCl<sub>2</sub>, then left on ice for one hour. The cells were then ready for immediate use in transformation reactions.

#### 2.6.6 Construction and design of oligonucleotide probes

Oligonucleotide probes were manufactured using an ABI-Perkin Elmer oligonucleotide synthesizer by a service provided by VU or were obtained from Sigma-Genosys (Sigma Aldrich Pty Ltd). Probes were designed using Prime GCG (Angis Biomanager) using the Internet site at <u>www.angis.su.org.au</u>. Netprimer, (Premier Biosoft International), was used to predict the properties of primers designed using Prime GCG. Tables 2.10 and 2.11 show the composition of primers used in this study.

Name	Oligonucleotide	Tm	Length	Reference
ACI 16S 1	AGCTGAACCAACAGATTCAC	59.3	20	(Walter et al., 2001)
ACI 16S 2	ACTACCAGGGTATCTAATCC	53.6	20	(Walter et al., 2001)
PR 1	CAGACTGAAAGTCTGACGG	58.5	19	(Walter et al., 2001)
CAS 2	GCGATGCGAATTTCTTTTC	63.4	20	(Walter et al., 2001)
Lm3	CGGGTGCTNCCCACTTTCATG	63.4	21	(Kaufmann <i>et al.</i> , 1997)
Lm26	GATTCTGGCTCAGGATGAACG	54.6	21	(Kaufmann et al., 1997)
LbLMA1-rev	CTCAAAACTAAACAAAGTTTC	47.1	21	(Dubernet et al., 2002)
R16-1	CTTGTACACACCGCCCGTCA	62.4	20	(Dubernet et al., 2002)
Ban 2	CATATTGGATCACGGTCC	50.5	18	(Ventura <i>et al.</i> , 2001)
23Si	CATTCGGACACCCTGGGATC	62.0	20	(Ventura <i>et al.</i> , 2001)
PAF	AGAGTTTGATCCTGGCTCAG	54.0	20	(Yeung <i>et al.</i> , 2002)
536R	GTATTACCGCGGCTGCTG	57.5	18	(Yeung <i>et al.</i> , 2002)

# Table 2.10 Composition of oligonucleotide probes used in the identification of LAB.

Table 2.11 Composition of oligonucleotide probes used for amplification of gene products orprobing genomic DNA libraries.

Name	Oligonucleotide	Tm	Length	Reference
SP45ASP	GAAACNAA(T/C)GCN(C/T)TN(C/T)C	60.5	17	This study
SP1S	AGATAGCCGAACACGATGCC	66.2	20	This study
SP1AS	CACGACGACAGCAAAGAAAAC	64.6	21	This study
SP2S	AACACAATGCTCTCACGAC	58.8	19	This study
SP2AS	ACGACGACAGCAAAGAAAAC	61.8	20	This study
SP3S	ATGGTGCGGGCAATTAAG	63.4	18	This study
SP3AS	TGAAGATGCGAACACAGAAG	61.6	20	This study
2SP1S	GTCCTTACACGCAGGGTTCT	63.0	20	This study
2SP1AS	CGAACACCCAAACCATCTCT	63.8	20	This study
2SP2AS	CGGCTCAATCATCGTCTTC	63.4	19	This study
2SP3S	CCAGAGATGGTTTGGGTGTT	63.7	20	This study
2SP4S	CGATGGTGCTGGTTGAAGT	63.8	19	This study
2SP4AS	TGAAGACGGAACAGGAAAATG	63.8	21	This study
2SP5S	GCATTTTCCTGTTCCGTCTT	62.9	20	This study
2SP5AS	AAAAGGCTGGCAACTCTGTC	63.3	20	This study
pUC/M13-F	CGCCAGGGTTTTCCCAGTCACGAC	72.5	24	www.promega.com
pUC/M13-R	TCACACAGGAAACAGCTATGAC	55.2	22	www.promega.com
3SP1S	AGACAGCAGCAGAGATAAACAAG	52.9	23	This study
3SP2S	ACAGACAGCAGCAGAGATAAAC	52.6	22	This study
3SP3S	TGCCAAAAAAAGAAGCCACAAG	52.9	22	This study
3SP1AS	CACCATCGCCAACAACACAG	54.4	20	This study
3SP4S	GCCAAATAGCCAAAGCCATAC	52.1	21	This study
3SP2AS	ATCACATCACGATCCCAACC	51.9	20	This study

#### 2.6.7 Construction of representative genomic DNA library

#### 2.6.7.1 Extraction and purification of DNA

This method of extracting DNA from *lactobacilli sp.* was taken from that published by Johnson (1980). The culture to be extracted was grown to stationary phase, and then 1 mL was removed to an Eppendorf tube and spun for two minutes (14,000 x g).

The supernatant was discarded and the pellet resuspended in 100  $\mu$ l of 50 mM Tris /HCL, 50 mM sodium chloride and 10mM EDTA pH 8. Fifteen  $\mu$ l of 15mg ml<sup>-1</sup> lysozyme solution was added to the pellet and incubated for 20 minutes at 37°C. Forty  $\mu$ l of 25% (w/v) SDS was then added to the tube and heated to 65°C for 5 minutes before vortexing for 5 seconds. Thirty-three  $\mu$ l of sodium chloride (5 M) was added before the tube was left on ice for one hour.

After centrifugation at high speed (14,000 x g) for fifteen minutes the supernatant fluid was treated with equal volumes of chloroform-isoamyl alcohol (24:1, v/v). The supernatant was re-centrifuged for 5 minutes before the upper phase was taken and the DNA precipitated with ice cold 70% alcohol.

The DNA was resuspended in 150  $\mu$ l of TE buffer (10mM Tris/HCl, 1mM EDTA, pH 8) and treated with DNAase-free RNAase to a final concentration of 10  $\mu$ g ml<sup>-1</sup> by incubation at 37°C for 30 minutes. The DNA was then stored at -20°C until required.

#### 2.6.7.2 Digestion of DNA

Extracted DNA used for preparation of the genomic DNA library was digested with the enzyme *Sau3A 1* and the reaction stopped by a phenol extraction of the enzyme. Sixty  $\mu$ L of a DNA solution containing 100  $\mu$ g of DNA prepared in section 2.5.8.1 was mixed with 84  $\mu$ L of enzyme buffer (Promega), 1  $\mu$ L of bovine serum albumin (Promega) and 0.4  $\mu$ L (or 1.2 units) of Sau 3A 1

enzyme (Promega). After digestion for 30 minutes at 37°C, an equal volume of phenol:chloroform:isoamyl alcohol (25:24:1) was added to the tube, mixed and then the tube was spun for 5 minutes at 14,000 x g. The supernatant was then transferred to another tube and 600  $\mu$ L of 100% ethanol and 100  $\mu$ L of sodium acetate (3M at pH 5.2) was then added to precipitate the DNA. The tube was then centrifuged and the DNA pellet dissolved in 50  $\mu$ L of water.

#### 2.6.7.3 Ligation of insert and vector

One  $\mu$ L of the vector DNA prepared in section 2.6.4 was added to 10  $\mu$ L of genomic DNA prepared in section 2.6.7.1, 1  $\mu$ L distilled water 1.5  $\mu$ L of ligase buffer (Promega) and, 1.5  $\mu$ L T4 DNA ligase buffer (Promega). The tube was incubated at 20°C for 2 hours then the DNA was transformed into competent cells.

#### 2.6.7.4 Transformation of bacterial cells with plasmids

All of the reaction mixture prepared in 2.6.7.3 was incubated with 200 µL of competent cells prepared as per section 2.6.5. The mixture was left on ice for one hour then incubated at 42°C for 3 minutes. Five hundred µL of double strength YT medium was added to the tube and further incubated for one hour at 37°C before being plated out on colour selection plates. Positive and negative growth controls were prepared from competent cells plated onto both L-B and ampicillin plates. Competent cells with plasmid preparation but no insert were used to check transformation efficiency. Transformed TG2 cells were indicated by white colonies on colour selection plates. These cells were transferred to microtitre plates containing double strength L-B broth and gycerol (1:1) and frozen for later use. TG2 cells without inserts were blue.

## 2.6.8 Colony blotting

This technique was used to lift colonies containing fragments of the *Lb. acidophilus* genomic DNA sequence so that they could be probed for a particular gene of interest. Circles of Hybond N<sup>+</sup>

membrane (Pharmacia) were cut to fit the Petri plates on which transformed TG2 cells, prepared as per section 2.5.7.4, had been grown. Each Petri plate held 47 transformants and one position marker. The transformants had been grown from frozen cultures prepared as described in section 2.5.7.4 and were transferred to a set location on the plate with sterile toothpicks; orientation of the membrane was marked on both plate and membrane relative to the position marker. The membrane was then placed onto the agar surface for one minute before being transferred (colony side up) onto an absorbent filter pad soaked in denaturing solution (1.5 M NaCl, 0.5 M NaOH) for 7 minutes. Next, the membrane was placed onto a pad of absorbent paper soaked in neutralising solution (1.5 M NaCl, 0.5 M Tris-HCl pH7.2, 0.001 M EDTA) for three minutes. This step was repeated with a fresh pad soaked in the same solution.

The membrane was then washed in double strength SSC solution (0.3 M NaCl, 0.03 M Na<sub>3</sub> citrate) before being air-dried colony side up for one hour. After the membrane had dried it was wrapped in plastic wrap and exposed to UV light for three minutes to cross-link the DNA to the membrane. The membrane was then stored at 4°C until required.

**2.6.9 Preparation of labelled oligonucleotide probes used in hybridisation procedures** The following reagents were combined in an Eppendorf tube and incubated at 37°C for 30-60 minutes. The oligonucleotide probe was prepared as described in section 2.6.

- 65 μl oligonucleotide probe (100 ng),
- (γ<sup>32</sup> P) ATP (3000 Ci/mmole @ 10 Ci/mL (volume varied according to time of manufacture),
- 1 μl 10 X T4 polynucleotide kinase (PNK) buffer (Promega) and
- 1 μl T4 PNK (10 units/μl) (Promega) and sterile water added, as required, to bring the final volume to 10 μl.

The reaction was stopped with one  $\mu$ I of 0.5M EDTA. Unincorporated probe was then removed from this reaction mix.

#### 2.6.10 Removal of unincorporated label

A QUIquick<sup>™</sup> nucleotide removal kit was used to remove unincorporated radioactive label (QIAGEN). The kit was used according to the manufacturer's instructions. Ten volumes of PN buffer\* were added to 1 volume of the reaction rrix, prepared in method 2.6.9, then mixed. A QUIquick<sup>™</sup> spin column was placed in a 2-mL collection tube (provided in kit) and the sample applied to the column and centrifuged. (One minute at 4,000 x g). The column was washed by adding 500 µl of PE buffer and centrifuged as before. PE and PN buffers were supplied as part of the kit. The eluent was discarded as radioactive waste and the column washed again with another 500 µl of PE buffer. The flow through was discarded as radioactive waste and the column was placed in a clean 1.5 mL centrifuge tube. DNA was eluted by adding 100-200 µl sterile distilled deionised H<sub>2</sub>O to the centre of the QIAquick<sup>™</sup> column. The column was allowed to stand one minute then centrifuged (one minute at 16,000 x g). The radioactive label was then placed at 65°C for 10 minutes before proceeding to hybridisation.

#### 2.6.11 Hybridisation of oligonucleotide probes

The purpose of the method was to hybridise a radio-labelled probe to specific genomic DNA crosslinked to a membrane prepared according to the method described in section 2.6.8 and the attachment visualised using auto-radiographic techniques. ExpressHyb hybridisation solution was used according to the manufacturer's instructions (ClonTech). The ExpressHyb solution was warmed to 68°C and stirred well to completely dissolve any precipitate. The solution was then equilibrated at 37°C. The colony lift was pre-hybridised in a thick-walled plastic bag (blender bag) containing a minimum of 15 mL of ExpressHyb solution (37°C for 30 minutes with continuous shaking). Radio-labelled probe was added to 5 mL of fresh ExpressHyb, well mixed, then the ExpressHyb pre-hybridisation solution was replaced with the fresh solution containing the radio-labelled oligonucleotide probe. All air bubbles were removed from the bag and care was taken to distribute the solution evenly. The bag was then incubated with continuous shaking at 37°C for 60 minutes. The blots were then rinsed in wash solution 1 (2 X SSC and 0.05% SDS at room temperature) using several changes of wash solution 2 (1 X SSC and 0.1% SDS at room temperature) for 40 minutes with one change of liquid. The blots were removed with forceps, excess solution was removed by shaking and the membranes were immediately sealed in plastic blender bags then exposed to X-ray film at -70°C for 1-2 days. Positive reactions as indicated by bright spots on the X-ray film were used to locate colonies containing transformed cells with plasmid inserts that hybridised with the probe, see section 2.6.3. Plasmid preparations were performed on the positive clones and the sequences of the inserts were determined according to section 2.6.14.

#### 2.6.12 DNA amplification

PCR was used to amplify fragments of genes of interest in this study. An MJ Research thermocycler model MPTC-100-96VHB supplied by Geneworks was used according to the manufacturer's instructions. Reaction mixtures and thermal cycling parameters are described in the relevant results sections.

#### 2.6.13 Sequencing of DNA

Preparation of templates for automated sequencing was performed using either a Dye Termination Cycle Sequencing Ready Reaction Kit (Perkin Elmer) or the Beckman CEQ 2000 Dye terminator sequencing protocol and performed according to manufacturer's instructions. Samples were submitted for automated sequencing to a joint VU/Monash University facility at the Department of Microbiology, Monash University or to the sequencing facility at The University of Melbourne, Gilbert Chandler Research Laboratories.

### 2.6.14 Analysis of DNA sequences

DNA sequences were analysed using the database similarity search program (BLASTn). BLASTn was accessed through ANGIS and the Computational Biology Program of the Life Sciences Division of Oak Ridge National Laboratory (ORNL) using the internet site at <u>http://genome.ornl.gov</u>/. Other DNA analysis programs accessed through ANGIS included:

- Flip 6 Frames, a Unix C program used to find and translate open reading frames (orf's);
- ClustalW. ClustalW was used to align sets of nucleotide sequences; and
- TACG. TACG was used for the restriction enzyme analysis of DNA.

### 2.6.15 Bioinformatic analysis of proteins

Bioinformatic analysis of proteins was performed through ANGIS using the following programs:

- BlastP was used to search protein databases;
- SPScan was used to scan protein sequences for the presence of secretory signal peptides;
- Secondary structure was predicted using the Garnier program;
- Isoelectric (GCG) was used to plot charge as a function of pH to determine the iso-electric point and predict molecular weight of a peptide sequence;
- Grease, was used to plot the hydrophobicity profile of a protein using the Kyte-Doolittle hydropathy plot; and

• PrettyBox (GCG) was used to prepare postscript formats of aligned protein sequences.

#### 2.7 Photography

#### 2.7.1 Photography of SDS-PAGE and agarose gels

Photographs of SDS-PAGE and agarose gels were taken with a Polaroid MP4 Land Camera with Black and White positive film (Polaroid Polaplan 667). Agarose gels were placed on a transilluminator (302 nm) (LKB2001 Macrovue) and photographs were taken using an orange filter in the dark. The size of any restriction bands present was determined from comparison with molecular size markers. SDS-PAGE gels were placed on a white light box and photographed without a filter in the dark. Size of any protein bands present was determined according to section 2.5.3.2. 2-D SDS-PAGE gels were scanned on a flat bed scanner (Hewlett Packard) and images were cropped and rotated in the photoeditor software package (Microsoft).

#### 2.7.2 Still Photography

Still photography was performed with a single lens reflex camera (Olympus OM-10) and the film was developed and printed by Prosper Photographics.

#### 2.7.3 Exposure of hybridised membranes to X-ray film

Membranes were exposed to X-ray film (Kodak X AR 5) (Integrated Sciences) in a cassette with intensifying screen (Kodak Bio-Max) between 24 and 72 hours at -70°C. X-ray films were developed in AGFA Gevaert G150 manual X-ray developer (1/5 dilution in water) for 3-5 minutes. Films were then rinsed in running water for one minute followed by fixing (AGFA Gevaert G150 fixer) and rinsing in Kodak Photoflow 600 solution (1/600 dilution in water). A final rinse in running water was performed before films were hung up to dry.

### Chapter Three

# STRESS RESPONSES IN LACTOBACILLUS PARACASEI STRAIN VUP 12006

#### 3.1 Introduction

The direction of the research undertaken in this thesis was, in part, determined from previous projects undertaken in CBFT at VUT investigating the role of stress responses in environmental and industrial bacteria. These research projects had concentrated on characterising changes in protein production by bacteria exposed to harsh industrial-manufacturing processes or environmental pollution, particularly involving heavy metals, such as conditions encountered in fermentations in the food industry and exposure to heavy metals. The proteins of interest in these projects were principally HSP60 and HSP70 and novel stress induced proteins. This research had shown that selected strains of LAB were able to survive in fermented dairy products, such as yoghurt, while others died shortly after manufacture (Shah et al., 1995). The current research was to play a part in a coordinated study to improve the survival of LAB in manufacturing processes and aimed, principally, to determine how probiotic LAB respond to low pH environments. Other changes in the physical environment that may be experienced during dairy manufacturing processes and could affect the survival of bacteria, specifically changes in temperature and gaseous environment, were also to be examined. While a number of methods could have been selected to characterise the response to low pH environments, the principal method chosen was investigation of changes in protein production.

Due to the recognised problems in identification of bacteria used in the dairy industry, the strains selected for examination were initially characterised by phenotypic and phylogenetic characteristics

to confirm their identification. The difficulties in identification were confirmed during this work and a number of tools, including PCR amplification of 16S rRNA genes, had to be used to confirm results. Although initially the research had intended to study only strains of *Lb. acidophilus* these identification methods confirmed misidentification of one of the original strains provided by CSIRO. This strain was subsequently identified as *Lb. paracasei* and was the principal bacterium investigated and reported on in this chapter.

#### 3.2 Identification of test strains

#### 3.2.1 Gram stain and growth on MRS agar

Twenty-five bacterial strains reported to be *Lb. acidophilus*, including type strain ATCC 4356, were obtained from culture collections and commercially available dairy starter cultures. They were given interim identifications of C1 to C25. C1 to C17 were obtained from the CSIRO culture collection, C18 to C23 were isolated from commercially available yoghurts (Bornhofen Acidophilus yoghurt and Yoplait Yoplus A+B brand) and C24 and C25 were obtained from the Victoria University Culture Collection. The bacteria were grown overnight under anaerobic conditions on MRS agar at 37°C then examined for typical colony morphology and Gram stain. Typical colony morphology of lactobacilli was demonstrated by the appearance of creamy-white colonies approximately 2-3 mm in diameter. In a Gram stain the cells appeared as Gram positive bacilli with rounded ends, either as single cells or in short chains. Of the original twenty-five strains only nineteen strains, including the type strain, showed the basic morphology of lactobacilli and these were placed into frozen storage, using ceramic beads and glycerol/MRS broth as described in section 2.3.2. Two cultures failed to grow on MRS agar (C3, C24) and three strains were found to be Gram positive cocci (C7, C21, C23). These frozen stocks were used for additional phenotypic characterisation. Specifically these were: carbohydrate utilisation studies, FAME fingerprinting and protein profiling using SDS-PAGE and the results of these studies are presented in the following sections.

#### 3.2.2 Carbohydrate utilisation

Patterns of carbohydrate utilisation were determined initially for a control strain VUP 12000 (*Lb. acidophilus* strain ATTC 4,356) using a traditional method that utilised lactobacilli fermentation broth in single vials (see section 2.3.1.2). This method did confirm the biochemistry of the test strain, however it was time consuming, expensive (due to the volume of reagent, cost of some of the carbohydrates and the number of vials used) and a quicker technique for examining LAB strains was sought.

Three methods were developed to solve the problem of cheap, rapid and multiple identifications. Initial trials focused on using various combinations of agar bases with 2% of selected carbohydrates. A number of different bases were examined: BLA was eventually chosen because it supported good growth when carbohydrate sources were added but growth was restricted in the absence of carbohydrate. A standard spot inoculation technique was used to transfer 35 standardised concentrations of bacterial suspensions to the surface of each plate containing a single carbohydrate source. Growth characteristics were satisfactory, however there was sufficient nutrient carryover in the inoculum to allow significant growth in the control, which made interpretation of the tests difficult.

The second method sought to compare carbohydrate utilisation using scanning densitometry. Growth on the agar plates, from the first method, was photographed using positive/negative Polaroid film. The negative was scanned using a laser densitometer and an attempt was made to quantitate the growth. The cut-off point for utilisation of a specific carbohydrate was determined for control organisms. Control plates with no carbohydrate were scanned and the difference between the control and the test plate represented the growth on the particular carbohydrate. Although this method was suitable for determining if an organism can or cannot grow on a specific substrate it is not feasible because each negative took around one hour to scan and interpret. In addition the cost
in time and money was excessive. The third method used an ELISA plate reader to quantitate growth in a liquid culture in a microtitre tray. This rapid microtitre plate method was developed to differentiate and presumptively identify LAB strains reliably, based on the results of fermentation tests. This novel method used a microtitre plate reader to quantify growth after six hours in liquid culture in a microtitre tray, which allowed the results to be obtained in one working day. A number of trials were set up to determine the best type of broth, optimal volumes, inoculum densities and incubation times and the final experimental conditions are described in section 2.3.5. These results were presented at the International Dairy Lactic Acid Bacteria Conference (Pepper *et al.*, 1995).

The nineteen stored strains that showed characteristics of being presumptive *Lb. acidophilus* and the type strain were tested for the ability to utilise fifteen different carbohydrate sources. The tests were performed in duplicate within experiments and two separate trials were performed. The seven strains that most closely resembled the typical, published results for the *Lb. acidophilus* type and results for the type strain are listed in Table 3.1. These eight strains designated VUP12000 through VUP 12007, were chosen to undergo acid tolerance screening tests. The type strain was numbered VUP 12000. All eight strains were presumptively identified as *Lb. acidophilus* based on the tests performed, although two strains were unable to utilise galactose, and the majority of *Lb. acidophilus* would normally be expected to utilise galactose. The results for sorbitol utilisation, a test that is normally inconsistent within this species, were variable and it was noted that it could not be utilised by five strains (Hammes *et al.*, 1995).

#### 3.2.3 Fatty acid methyl ester profile

This method was designed to give a fingerprint or profile of the fatty acids produced when the eight strains presumptively identified as *Lb. acidophilus* were grown under a specific set of conditions (see section 2.4.1.2).

Table 3.1 Carbohydrate utilisation pattern in the type strain and 7 putative strains of Lb. acidophilus determined using a rapid microtitre plate

## method.

Ability to utilise a particular carbohydrate was determined from changes in optical density values for growth in 15 different substrates over six hours at 37°C in

MRS broth, see section 2.3.5 for experimental details.

	-			<u>т</u>	· · · · · ·	· · · · · ·	_ <u> </u>		
	C25			,		+	+	+	
	C22	_			+		+	+	
l.	C20		<u> </u>	,	+		+	+	
ation)	C19			3		+	+	+	  • _ §
entific	C18		,		+	+	+	+	
ain id	C15			• •	+	+	+	+	
JP str	C14		,	,		,	+	+	
n VL	C12				+		+	+	
s give	CII		,		,	,		+	
n wa	ଞ				+		+	+	
strai	පී			+		+	+	+	+
ates	2				•		۰ ۲		•
et indic	C17	(12007)		+	+	+	+	+	+
brack	C16	12006)				-			
er in		)2)		-			<b>•</b>	+	
numt	C13	(1200	•	+	+	•	+	+	+
cation (	C10	(12004)		+	+	+	+	+	+
identific	C6	(12003)		+ +	+		+	+	+
Strain	 C5	(12002)							
						+	+	+	+
	3	(1200		+	+	+	+	+	+
	5	(12000)		+	+	+	+	+	+
	Expected results <sup>a</sup>		<u>م</u>	+ c	+	+	+	+	+
			Arabinose	Cellibiose	Fructose	Galactose	Glucose	Lactose	Mattose

.       .	eso	+	+	+	+	+	+	+	+	+			+	+	+	+	+	+	+	+	+	
																				<u></u>		
												,					•		-	•	•	<u> </u>
.       .       .       .       .       .         .       .       .       .       .       .         .       .       .       .       .       .         .       .       .       .       .       .         .       .       .       .       .       .       .         .       .       .       .       .       .       .       .         .       .       .       .       .       .       .       .       .         .						,				,	,	1		,				,	•	•	•	
.       .       .       .         .       <		q						,	,				,				1	•	, , ,	•	•	
.       .       .         .       .       .     <	e e	,	1		,						+						•	,	•	•	·	
<ul> <li>.</li> <li>.&lt;</li></ul>		σ	+		,		+	+				•	+		+	+	+	,	+	+	1	
		+	+	+	+	+	+	+	+	+		+	+			, T	+		•	•	1	
			•		•												•	·		,		
	ydrate		•	,		,	,					1					•	•	, 	1	•	

al., 1995, Schleifer et al., 1995). <sup>b</sup> Negative <sup>c</sup> Positive <sup>d</sup> 11-89 % positive

Long chain fatty acids present in the type strain, 7 presumptive Lb. acidophilus strains and 2 other lactobacillus strains (Lb. jugurati and Lb. casei) were converted into their corresponding methyl esters and and separated using a Varian gas chromatograph with a BPX-70 column. Examples of a standard profile obtained using bacterial fatty acid methyl esters (Sigma) and a FAME profile obtained during testing are shown in Figure 3.1. Only twelve major peaks (1-26% relative abundance) and approximately fifteen minor peaks of less than one percent relative abundance were found to be present using this method. The presence of *cis*-11,12-Methylene-octadecanoic acid (lactobacillic acid) supported the species identification of the strains being studied. The method was not used to differentiate all the fatty acids present completely and as such, the method did not give clear baseline separation for all peaks (section 2.4.4). In particular the peaks with a retention time of 10.4 and 10.6 minutes could not always be separated at the baseline for some tests and samples due to their high relative proportions present (Figure 3.1). Data for these two peaks are presented in Table 3.2 as a combined percentage, relative to the total area, of two fatty acid methyl esters (18:1(9) and 18:3), which is used when using FAME profiles as a form of fingerprint identification (Decallone et al., 1991). The standard run length was 20 minutes; when runs were extended to forty minutes, no further peaks were detected.

Three peaks, designated X1, X2, and X3, could not be identified by comparison to known standards. Table 3.2 shows the relative abundance for the major peaks calculated from data of a typical run (average results of duplicates with standard deviation less than 7%). The results appeared to show that rather than a single profile or fingerprint for all the strains, there were two profiles, designated type 1, (demonstrated by strains VUP 12000 and 12003) and type 2 (demonstrated by strains VUP 12001,12002,12004,12005,12006 and 12007). Differentiation of the two types was based on levels of C<sub>10:0</sub> and the unidentified fatty acids X1 and X2.





#### (Sigma) and a FAME profile obtained during testing.

Figure A shows standard profile prepared from mixture of Sigma fatty acid standards (189-8, 189-5 and 189-1). Figure B show FAME profile of strain VUP 12006 obtained during testing. Cells of a stationary phase culture were harvested and the long chain fatty acids extracted from these cells were converted into the corresponding methyl ester according to the method described by Decallone *et al.*, (1991) (section 2.4.12). Fame profiles were obtained using a Varian Gas Chromatograph model 3400 using a BPX-70 column (section 2.4.4) and one  $\mu$ L sample injections.

В

Α

12001     0     10     2     5     4     1     23     2     5     19	C 19:00 19 8	<sup>b</sup> C 18:1 (9) & C 18:3 10.2 21 22	C 18:00 5 12 5	C 16:01 6.8 2 2	23 8 <u>6</u> C <sup>16:0</sup>	- 8 <del>.</del>	C <sup>14:0</sup>	0.40	8. 4 0	3.5	2.05 6 6	Retention time (min) Strain No. 12001
	23	22	Q	2	26	0	e	4	0	2	0	12002
			1								1	
	ω	21	12	2	<u>ත</u>	ω	ო	0	4	0	Q	12000
12000 6 0 4 0 3 8 9 2 12 21 8												Strain No.
Strain No.         6         0         4         0         3         8         9         2         12         21         8           12000         6         0         4         0         3         8         9         2         12         21         8	12.3	10.2	9.6	6.8	6.3	5.9	4.2	4.0	3.8	3.5	2.05	Retention time (min)
Retention time (min)       2.05       3.5       3.8       4.0       4.2       5.9       6.3       6.8       9.6       10.2       12.3         Strain No.       6       0       4       0       3       8       9       5.1       12.3         12000       6       0       4       0       3       8       9       2       12       21       8		C <sub>18:3</sub>										
Retention time (min)       2.05       3.5       3.8       4.0       4.2       5.9       6.3       6.8       9.6       10.2       12.3         Strain No.       12000       6       0       4       0       3       8       9       21       8	C 19:00	bC 18:1 (9) &	C 18:00	C 16:01	C 16:0	ex	C 14:0					
C 100         C 120         X1         X2         C 14.0         X3         C 16.01         C 18.19) & C 18.119) & C 18.00         bC 18.119) & C 18.00           Retention time (min)         2.05         3.5         3.8         4.0         4.2         5.9         6.3         6.8         9.6         10.2         12.3           Retention time (min)         2.05         3.5         3.8         4.0         4.2         5.9         6.3         6.8         9.6         10.2         12.3           Strain No.         12000         6         0         4         0         3         8         9         2         12         8					< 			X	X	C <sub>12:0</sub>	C 10:0	
* Methyl ester of fatty acids (%):         C 10:0         C 12:0         X1         X2         C 14:0         X3         C 16:01         C 18:00         bC 18:1(9) & C 19:00           Retention time (min)         2.05         3.5         3.8         4.0         4.2         5.9         6.3         6.8         9.6         10.2         12.3           Strain No.         12000         6         0         4         0         3         8         9         2         12         8					(	•		X2	X	C12:0	acids (%): C <sub>10:0</sub>	<sup>a</sup> Methyl ester of fatty
te means of two independent experiments. Results of interest are highlighted, these are discussed in section 3.2.3. <sup>a</sup> Methyl ester of fatty acids (%): <sup>a</sup> Methyl ester of fatty acids (%): <sup>b</sup> C <sup>100</sup> C <sup>120</sup> X1 X2 C <sup>140</sup> X3 C <sup>160</sup> C <sup>1600</sup> b C <sup>181</sup> (9) & C <sup>183</sup> <sup>c</sup> C <sup>183</sup> C <sup></sup>				a specific suc	ssed in section	ider all pear	ine area ur lighted, thes	divided by est are high	Deciric peak	a under a si ments. Res C12:0	from the are ndent experi acids (%): C 10:0	esults were calculated re means of two indepe <sup>a</sup> Methyl ester of fatty
onverted into their corresponding methyl esters and and separated using a Varian gas chromatograph with a BPX-70 column. Twelve major peaks were under to be present. Data below is presented as a percentage of the specific fatty acids present in a particular strain, values for minor peaks are not included.         evaluation to be present. Data below is presented as a percentage of the specific fatty acids present in a particular strain, values for minor peaks are not included.         esults were calculated from the area under a specific peak divided by the area under all peaks present in a specific strain and multiplied by 100. The data e means of two independent experiments. Results of interest are highlighted, these are discussed in section 3.2.3.         • Methyl ester of fatty acids (%):       C 100       C 160       C 1601       C 1601       C 1601       C 1601       C 1601         • Methyl ester of fatty acids (%):       C 100       C 1601	eaks were it included. The data	velve major p peaks are no iplied by 100	column. Tv es for minor ain and mult	ith a BPX-70 ar strain, valu a specific stra on 3.2.3.	natograph w It in a particu ks present in ssed in secti	n gas chror Icids presen Ider all peal se are discu	ing a Varia ecific fatty a the area ur lighted, thes	eparated us ge of the sp divided by est are high	and and se a percentaç oecific peak ults of intere	ethyl esters esented as a under a s ments. Res C12:0	esponding m a below is pl from the are ndent experi acids (%): C 10:0	onverted into their corra und to be present. Da esults were calculated e means of two indepe <sup>a</sup> Methyl ester of fatty
and tatly acids present in the type strain. 7 presumptive Lb. acidophilus strains and 2 other lactobacillus strains (Lb. <i>jugurati</i> ) and Lb. <i>casel</i> ) were marked into their corresponding methyl esters and and separated using a Varian gas chromatograph with a BPX-70 column. Twelve major peaks were und to be present. Data below is presented as a percentage of the specific fatty acids present in a particular strain, values for minor peaks are not included. esults were calculated from the area under a specific peak divided by the area under all peaks present in a particular strain, values for minor peaks are not included.         exerts of the specific peak divided by the area under all peaks present in a particular strain, values for minor peaks are not included.         esults were calculated from the area under a specific peak divided by the area under all peaks present in a specific strain and multiplied by 100. The data e means of two independent experiments. Results of interest are highlighted, these are discussed in section 3.2.3.         • Methyl ester of fatty acids (%):	asei) were eaks were it included. The data	<i>ati</i> and <i>Lb. c</i> velve major p peaks are nc iplied by 100	s ( <i>Lb. jugur</i> column. Tw es for minor ain and mult	bacillus strain tth a BPX-70 ar strain, valu a specific stra on 3.2.3.	2 other lacto matograph w it in a particu ks present in ssed in secti	trains and n gas chror toids presen toids all peal se are discu	<i>cidophilus</i> s sing a Varia ecific fatty a ecific fatty a the area ur lighted, thes	ptive <i>Lb.</i> a eparated us ge of the sp divided by est are high	n, 7 presum and and se a percentaç pecific peak ults of intere	type strair ethyl esters esented as a under a s ments. Res C12:0	resent in the esponding m a below is pl from the are ndent experi acids (%): C 10:0	ong chain fatty acids p inverted into their corri und to be present. Da esults were calculated e means of two indepe <sup>a</sup> Methyl ester of fatty
gurati and Lb. case).         ong chain fatty acids present in the type strain, 7 presumptive Lb. acidophilus strains and 2 other lactobacillus strains (Lb. jugurafi and Lb. case) were anverted into their corresponding methyl estens and and separated using a Varian gas chromatograph with a BPX-70 column. Twelve major peaks were und to be present. Data below is presented as a percentage of the specific fatty acids present in a particular strain, values for minor peaks are not included.         sulf to be present. Data below is presented as a percentage of the specific fatty acids present in a particular strain, values for minor peaks are not included.         sulf to be present. Data below is presented as a percentage of the specific fatty acids present in a particular strain, values for minor peaks are not included.         sulf to be present. Data below is presented as a percentage of the specific fatty acids present in a particular strain, values for minor peaks are not included.         sulf to be present. Data below is presented as a percentage of the specific fatty acids present in a particular strain, values for minor peaks are not included.         eneans of two independent experiments. Results of interest are highlighted, these are discussed in section 3.2.3.         •Methyl ester of fatty acids (%):         •Methyl ester of fatty acids (%):         •Methyl ester of and         0       0         •Stain No.         6       0         12000       6         12000       6	<i>asei)</i> were eaks were it included. The data	<i>ati</i> and <i>Lb. c</i> velve major p peaks are nc iplied by 100	s ( <i>Lb. jugur</i> column. Tv cs for minor ain and mult	bacillus strain ith a BPX-70 ar strain, valu a specific stra on 3.2.3.	2 other lacto matograph w it in a particu ks present in ssed in secti	trains and n gas chror icids presen ider all peal be are discu	<i>cidophilus</i> s sing a Varia ecific fatty a the area ur lighted, thes	ptive <i>Lb. a</i> eparated us ge of the sp divided by est are high	n, 7 presum and and se a percentaç becific peak ults of intere	e type strair ethyl esters esented as a under a s ments. Res	resent in the esponding m a below is pl from the are ndent experi acids (%): C 10:0	guratí and Lb. casel). ong chain fatty acids p nverted into their corr und to be present. Da esults were calculated e means of two indepe a Methyl ester of fatty

		1			1	
11	50	16	26	20	0	11
23	21	20	26	22	۲	16
ດ	4	4 ,	S	9	10	ഹ
m	7	ო	7	7	<b>~</b>	7
10	23	10	17	53	თ	24
e	0	<del>~</del>	7	0	<b>~</b>	<b>~</b>
<b>~</b>	ۍ	4	ۍ	2	~	4
0	2J	ω	4	2	14	4
0	0	0	0	0	0	0
0	15	16	ω	<u>ත</u>	40	10
9	0	0	0	0	0	0
12003	12004	12005	12006	12007	Lb. jugurati	Lb. casei

<sup>a</sup> Relative proportion of fatty acid methyl ester based on percentage of total area tor the peaks detected.

<sup>b</sup> Peaks for fatty acid methyl esters 18:3 and 18:1(9) could not be separated completely at the base line during all runs using the method described in section 2.4.4 and are presented as a combined percentage.

<u>KEY:</u> Methyl decanoate (c100), Methyl dodecanoate (C120), Methyl tetradecanoate (C140), Methyl hexadecanoate (C160), Methyl cis-9-hexadecanoate (C161), Methyl

octadecanoate acid (C18:0), Methyl cis-9-octadecanoate acid (C18:1(9)), cis-11,12-Methylene-octadecanoic acid (lactobacillic acid) (C19:0)

The three differences noted between the two strains characterised as type 1 and the six strains characterised as type 2 were:

- type 1 strains showed C<sub>10:0</sub> levels of between 4 and 6 % relative abundance; this fatty acid was not detected in type 2 strains;
- the unidentified fatty acid X1 was not detected in type 1 strains but present between 7 and 16
   % of type 2 strains; and
- the unidentified fatty acid X2 was not detected in type 1 strains but present between 4 and 8 % of type 2 strains.

These data suggested that strains VUP12000 and VUP12003 may be different to VUP strains 12001, 12002, 12004, 12005, 12006 and 12007.

Two strains of other *Lactobacillus sp.* were run as a comparison to the presumptive strains of *Lb. acidophilus* tested. The first, *Lb. jugurati*, demonstrated a clearly different FAME GC pattern compared to the presumptive strains of *Lb. acidophilus*. In particular:

- C<sub>12:0</sub> was present at more than twice the level demonstrated in any of the strains of the presumptive strains of *Lb. acidophilus*;
- the relative abundance of the C<sub>18:1</sub> and C<sub>18:1(9)</sub> combined peak was 5%. This was less than
  a quarter of that demonstrated in the presumptive strains of *Lb. acidophilus* and
- C<sub>19:0</sub> appeared to be absent in this strain. In the presumptive strains of *Lb. acidophilus* tested, C<sub>19:0</sub> varied between 8 and 26%.

The second control strain, *Lb. casei*, demonstrated a FAME pattern very similar to the presumptive strains of *Lb. acidophilus* with the exception that the C<sub>18:1</sub> and C<sub>18:1</sub> (9) combined peak was 16%

relative abundance, 25 % lower than shown in any of the presumptive strains of *Lb. acidophilus* tested.

#### 3.2.4 SDS-PAGE profiles (fingerprinting of whole cell protein patterns)

SDS-PAGE analysis of whole cell proteins was evaluated to determine the utility of this approach for classifying LAB strains. The same growth conditions described for obtaining FAME profiles were used to obtain SDS-PAGE profiles. Visual comparison was made of the banding pattern produced by SDS-PAGE of whole cell proteins (see Figure 3.2). Data presented in a later section (3.4.4) shows that protein profiles taken hourly during growth at pH 6.3 and 37°C show significant changes during the growth cycle. Based on this observation, the test cultures were grown through two complete growth cycles at 37°C in MRS broth without pH control and the cells were collected in stationary phase and treated as described in section 2.4.1.

The protein profiles of all the strains tested, particularly VUP 12006, were dissimilar to the type strain (VUP 12000). In particular, strain VUP 12006 demonstrated fewer major bands and they were of higher density.

The major differences between strains were observed in the protein bands of less than 31 kDa with the exception of strains VUP 12001 and 12002 where no differences were observed. VUP 12002 is a mutant of 12001. This feature allowed differentiation of each of the strains, except VUP 12001 and VUP 12002, based on pattern matching. This method of identification was not pursued further due to the low level of discrimination obtained between strains.

#### 3.2.5 Molecular methods of identification

The initial characterisation of the *Lactobacillus* strains clearly indicated that there were no phenotypic traits that could satisfactorily confirm identification to species level. In fact the classical methods described in previous sections failed to validate the genus of the species being tested.



### Figure 3.2 1-D SDS-PAGE profiles of presumptive *Lb. acidophilus* strains using whole cells for protein preparation.

Protein samples were prepared by boiling equal volumes of whole bacterial cells with double strength loading buffer as described in section 2.4.1.1 and 2.5.1.1. Approximately 1µg of protein was applied to each well of the SDS-PAGE gel and protein bands were visualised by silver staining.

Lane 1 MW Standard Lane 2 Lb. acidophilus type strain VUP 12000 Lane 3 Strain VUP 12001 Lane 4 Strain VUP 12002 Lane 5 Strain VUP 12003 Lane 6 Strain VUP 12004 Lane 7 Strain VUP 12005 Lane 8 Strain VUP 12006 Lane 9 Strain VUP 12007 Lane 10 Molecular weight standards

MW Standards (MWS): 1. Phosphorylase B-97.4 kDa.

- 3. Ovalbumin-45 kDa
- 5. Trypsin inhibitor 21.5 kDa
- 2. Bovine serum albumin-66.2 kDa
- 4. Carbonic anhydrase-31 kDa
- 6. Lysozyme 14.4 kDa

It was therefore decided to use molecular biological techniques to identify the genus and species of strain VUP 12006, which was selected for characterisation of acid tolerance so that confirmation of identification was required. PCR-based methods were used to determine the identification of strain VUP 12006.

#### 3.2.5.1 Genus level identification using PCR

The method described by Dubernet (2002) was used to identify strain VUP 12006 to the genus level. Total genomic DNA used for all PCR reactions was extracted from stationary phase cells grown overnight in MRS broth at 37°C as described in the method in section 2.6.7.1. Briefly, lysozyme solution was used to treat the cells prior to cell lysis in SDS, which was followed by a chloroform-isoamyl alcohol treatment. The DNA was precipitated following this treatment in ice cold 70% alcohol and made up in TE buffer. The oligonucleotide primer pair, designated R16-1 and LbLMA1-rev, was used to amplify a region between a flanking terminal sequence of the 16S rRNA gene and a lactobacilli-specific intergenic spacer region of the 16S/23S genes. In the study by Dubernet (2002) *Lactobacillus* species generated PCR fragments of approximately 250 bp. The composition of the oligonucleotide primers is described in Table 3.3.

Name	Sequence	Melting temperature
R16-1	5' CTTGTACACACCGCCCGTCA 3'	62.4
LbLMA1-rev	5' CTCAAAACTAAACAAAGTTTC 3'	47.1

The reaction mixture (50µL) contained 30 pmol of each primer, 200µM each of dATP, dTTP, dCTP and dGTP, approximately 100ng of bacterial DNA, 2.5 U of *Taq* DNA polymerase (Sigma) and PCR buffer with 1.5 mM MgCl<sub>2</sub>. DNA was amplified as follows: initial denaturation at 95°C for 5 minutes, followed by 30 cycles consisting of denaturation at 95°C for 30 seconds, annealing at 55°C for 30

seconds, extension at 72°C for 30 seconds and a 7 minute final extension step at 72°C. The products were stored at 4°C until required. Aliquots of the reaction mixture (10µL) were separated on a 0.8% agarose gel, stained with ethidium bromide and visualised under UV light. The results are shown in Figure 3.3. Genomic DNA from strain VUP 12006, amplified using R16-1 and LbLMA1-rev, yielded a PCR product of approximately 240 bp. Due to the specificity of the primers for *Lactobacillus* DNA (Dubernet *et al.*, 2002) and the size of the fragments resulting from PCR of DNA extracted from strain VUP 12006, these results were consistent with strain VUP 12006 being placed in the genus *Lactobacillus*.

Two control strains, *Lb. acidophilus* type strain VUP12000 and *Lb. casei* GCRL-163 were run at the same time. It was observed that the fragment sizes of *Lb. casei* GCRL-163 and strain VUP 12006 were identical ( $\cong$ 240 bp) but smaller that obtained for *Lb. acidophilus* type strain VUP12000 ( $\cong$ 250 bp).

#### 3.2.5.2 Species level identification using PCR

The species designation of strain VUP 12006 was determined using information based on PCR products from 3 different primer sets and sequence data from pair 1. The composition of the oligonucleotide primers is shown below in Table 3.4.

Table 3.4	Composition	of the	oligonucleotide	primers
-----------	-------------	--------	-----------------	---------

Name	Sequence	Melting temperature
PAF	5' AGAGTTTGATCCTGGCTCAG 3'	54.0
536R	5' GTATTACCGCGGCTGCTG 3'	57.5
ACI 16S 1	5' AGCTGAACCAACAGATTCAC 3'	59.3
ACI 16S 2	5' ACTACCAGGGTATCTAATCC 3'	53.6
PR 1	5' CAGACTGAAAGTCTGACGG 3'	58.5
CAS 2	5' GCGATGCGAATTTCTTTTC 3'	63.4



Figure 3.3 PCR amplification of DNA from *Lactobacillus* strain VUP 12006, *Lactobacillus acidophilus* VUP12000 (type strain) and *Lactobacillus casei* GCRL-163 using primers LbLMA1-rev and R16-1.

Lane 1 is marker (EcoRI/Hind III of  $\lambda$  DNA Progen). Lanes 2-4 are PCR amplification products of DNA using primers LbLMA1-rev and R16-1. Products were obtained from the DNA of strain VUP 12006 (lane 2), *Lb. acidophilus* type strain VUP12000 (lane 3) and *Lb. casei* GCRL-163 (lane 4). Products varied in size but were approximately 240-250 bp.

The reaction mixture (50 $\mu$ L) for the first primer pair, PAF/536R contained 30 pmol of each primer, 200 $\mu$ M each of dATP, dTTP, dCTP and dGTP, approximately 100ng of bacterial DNA, 2.5 U of *Taq* DNA polymerase (Sigma) and PCR buffer without MgCl<sub>2</sub>.

DNA was amplified as follows: initial denaturation at 94°C for 2 minutes, followed by 30 cycles consisting of denaturation at 94°C for 45 seconds, annealing at 55°C for 45 seconds, extension at 72°C for 60 seconds and a 7 minute final extension step at 72°C. The products were stored at 4°C until required. Aliquots of the reaction mixture (10µL) were separated on a 0.8% agarose gel, stained with ethidium bromide and visualised under UV light. The results are shown in Figure 3.4.

Genomic DNA from strain VUP 12006 amplified using species-specific primers PAF/536R yielded a PCR product of approximately 500 bp. Sequences of the PCR products were determined using a CEQ 2000 (Beckman Coulter) according to the manufacturers instructions. The sequences in both the forward and reverse direction were edited in the software package 'bio-edit' to about the first 500 bases and submitted to ANGIS for a BLASTn search (2.6.14). Sequences showing statistically significant alignment are shown in Table 3.5.

In addition the ends of all sequences (forward and reverse separately) were trimmed to the same length and aligned by ClustalW (ANGIS). The alignments are shown in Figure 3.5 (sequence based on forward primer) and Figure 3.6 (sequence based on reverse primer). It can be seen in the alignment that there is significant sequence homology between strain VUP 12006 and *Lb. casei* strain GCRL 163. This data suggested that strain VUP 12006 was more closely related to the *Lb. casei* group and not the *Lb. acidophilus* group, which was originally suggested by carbohydrate utilisation pattern. Ward and Timmins (1999) had suggested that the V1 region of the 16S rRNA genes of *Lb. casei* and *Lb. paracasei*, was important in confirming speciation so alignment of the 50 bases in the V1 region was also performed (Figure 3.7).



Figure 3.4 PCR amplification of DNA from *Lactobacillus* strain VUP 12006, *Lactobacillus acidophilus* VUP12000 (type strain) and *Lactobacillus casei* GCRL-163 using species-specific primers PAF and 536R.

Lane 1 is marker (generuler, MBI Fermentas). Lanes 2-4 are PCR amplification products of DNA using primers LbLMA1-rev and R16-1. Products were obtained from the DNA of *Lactobacillus* strain VUP 12006 (lane 2), *Lb. acidophilus* type strain VUP12000 (lane 3) and *Lb. casei* GCRL-163 (lane 4). Products were approximately 500 bp.

#### Table 3.5 Sequences showing significant alignment.

A BLASTn search of databases for strains VUP 12006, VUP12000 and GCRL-163 was performed through ANGIS. Sequences showing significant alignment are shown below.

Strain	Sequence	<b>BLASTn Best Hit</b>	Description of Best Hit
	direction		
VUP 12006	Forward	LBA16SRRNA	Lb. paracasei gene for 16S rRNA, partial
			sequence 6/2002
VUP 12006	Reverse	AY091596.1	Lb. casei strain KH-1
VUP12000	Forward	LBARR16SA	Lb. acidophilus
VUP12000	Reverse	LBARR16SA	Lb. acidophilus
GCRL-163	Forward	LBA16SRRNA	Lb. paracasei
GCRL-163	Reverse	AY091596	Lb. casei strain KH-1

#### Figure 3.5 ClustalW alignment of 16S sequences of three strains of LAB (forward primer).

16S sequences of strain VUP12006 (46f), *Lb. acidophilus* strain ASC 388 (Asc388f) and *Lb. casei* strain GCRL-163 (LbcF) were aligned using ClustalW, a multiple alignment program accessed through ANGIS. Sequences were obtained from PCR fragments and amplified using the primer pair PAF and 536R. The ends of all sequences were trimmed to the same length. This sequence is based on data using the forward primer. Nucleotides that match are shaded in black.

		10	20	30	40	50
VUP 12006	GCCGCCG	TGCCTAAI	ACATGCAAG	II TCGAACGA T'	ICTCGINNNG	AT GAT
GCRL 163 ASC 388	GECGECG GECGGCG	TGCCTAAT	ACATGCAAG ACATGCAAG	TCGAACGAGT' TCGA <mark>C</mark> CGAG	CTCGTING A CINGAACOAA	AT GAT A <mark>C</mark> AGAT
	1	60	70	80	90	100
VUP 12006 GCRL 163 ASC 388	CGGTGCT CGGTGCT TCACTTC	TGCACCGA TGCACCGA CG <mark>TGAT</mark> GA	GATTCAACA GATTCAACA CCTTCCCCAA	TGGAACGAGT TGGAACGAGT C <mark>G CGAC</mark> C	GGCGGACGGG GGCGGACGGG GGCGGA <mark>T</mark> GGG1	IGAGTA IGAGTA IGAGTA
		110	120	130	140	150
VUP 12006 GCRL 163 ASC 388	ACACGTG ACACGTG ACACGTG	GGTAACCI GGTAACCI GG <mark>G</mark> AACCI	CGCCCTTAAG CGCCCTTAAG CGCCC <mark>CAT</mark> AG	TGGGGGGATAA( TGGGGGGATAA) T <mark>CT</mark> GGGATA <mark>C</mark> (	CATTTGGAAAO CATTTGGAAAO CA <mark>C</mark> TTGGAAAO	CAGATG CAGATG CAG <mark>E</mark> TG
	I	160	170	180	190	200
VUP 12006 GCRL 163 ASC 388	CTAATAC CTAATAC CTAATAC	CGCATAGA CGCATAGA CG <mark>C</mark> ATA <mark>AG</mark>	ATCCAAGAAC ATCCAAGAAC AAAAGCAGAT	CGCATGGTTC CGCATGGTTC CGCATG <mark>A</mark> T <mark>CA</mark>	ITGGCTGAAA( ITGGCTGAAA( GCTTA <mark>TA</mark> AAA(	GATGGC GATGGC G <mark>GC</mark> GGC
		210	220	230	240	250
VUP 12006 GCRL 163 ASC 388	GTAAGCT GTAAGCT GTAAGCT	ATCGCTTI ATCGCTTI <mark>C</mark> TCGCT <mark>A</mark> I	TGGATGGAC TGGATGGAC <mark>G</mark> GGATGG <mark>C</mark> C	CCGCGGCGTA CCGCGGGCGTA CCGCGG <mark>T</mark> G <mark>C</mark> A	ITAGCTAGTTO ITAGCTAGTTO ITAGCTAGTTO	GGTGAG GGTGAG GGT <mark>AC</mark> G
	I	260	270	280	290	300
VUP 12006 GCRL 163 ASC 388	GTAATGG GTAATGG GTAA <mark>C</mark> GG	CTCACCAA CTCACCAA C <mark>CT</mark> ACCAA	IGGCGATGAT IGGCGATGAT IGGC <mark>A</mark> ATGAT	ACGTAGCCGA ACGTAGCCGA <mark>CCA</mark> TAGCCGA	ACTGAGAGGT ACTGAGAGGT <mark>ST</mark> TGAGAG <mark>AC</mark>	IGATCG IGATCG IGATCG
		310	320	330	340	350
VUP 12006 GCRL 163 ASC 388	GCCACAT GCCACAT GCCACAT	TGGGACTG TGGGACTG TGGGACTG	AGACACGGC AGACACGGC AGACACGGC	CCAAACTCCT CCAAACTCCT CCAAACTCCT	ACGGGAGGCAC ACGGGAGGCAC ACGGGAGGCAC	GCAGTA GCAGTA GCAGTA
	h	360	370	380	390	400
VUP 12006 GCRL 163 ASC 388	GGGAATC GGGAATC GGGAATC	TTCCACAA TTCCACAA TTCCACAA	TGGACGCAA TGGACGCAA TGGACG <mark>A</mark> AA	GTCTGATGGA GTCTGATGGA GTCTGATGGA	GCAACGCCGCC GCAACGCCGCC GCAACGCCGCC	GTGAGT GTGAGT GTGAGT
		410	420	430	440	450
VUP 12006 GCRL 163 ASC 388	GAAGAAG GAAGAAG GAAGAAG	GCTTTCGG GCTTTCGG G <mark>T</mark> TTTCGG	GTCGTAAAA GTCGTAAAA ATCGTAAA <mark>C</mark>	CTCTGTTGTT CTCTGTTGTT CTCTGTTGTT CTCTGTTGTT	GGAGAAGAATO GGAGAAGAATO GG <mark>H</mark> GAAGAA <mark>C</mark> O	GTCGG GTCGC ATACA
VUP 12006 GCRL 163	CA <mark>AC</mark> GTA CA <mark>C</mark> AGTA	460 ••• CACTG ACTG				

ASC 388

**GGT**AGTACACTG

#### Figure 3.6 ClustalW alignment of 16S sequences of three strains of LAB (reverse primer).

16S sequences of strain VUP12006 (46r), *Lb. acidophilus* strain ASC 388 (Asc388r) and *Lb. casei* strain GCRL-163 (LbcR) were aligned using ClustalW, a multiple alignment program accessed through ANGIS. Sequences were obtained from PCR fragments and amplified using the primer pair PAF and 536R. The ends of all sequences were trimmed to the same length. This sequence is based on data using the reverse primer. Nucleotides that match are shaded in black.

	1	10	20	30	40	50	60
VUP 12006	CCGTGG		GTTGGATACCG	TCACGCCGA	CAACAGTTACI	CTGCCGACCA	TTCTT
ASC 388	CCGTG <mark>A</mark> I	NCTTTCTG	GTTC <mark>AT</mark> TACCG	TCAAATAAA	GCC CAGTTACT	ACCTCTATCC	TTCTT
		70 • • •   • • • •	80	90 	100	110	120 • • • •
VUP 12006 GCRL 163 ASC 388	CTCCAA CTCCAA C <mark>A</mark> CCAA	CAACAGAG CAACAGAG CAACAGAG	ITTTACGACC ITTTACGNACC TTTACGA <mark>T</mark> C	CGAAAGCCT CGAAAGCCT CGAAA <mark>A</mark> CCT	ICTTCACTCAC ICTTCACTCAC ICTTCACTCAC	CCGGCGTTGC CCGGCGTTGC CCGGCGTTGC	TCCAT TCCAT TCCAT
		130	140	150 	160	170	180
VUP 12006 GCRL 163 ASC 388	CAGACT CAGACT CAGACT	IGCGTCCA IGCGTCCA I <mark>T</mark> CGTCCA	ITGTGGAAGAT ITGTGGAAGAT ITGTGGAAGAT	TCCCTACTG TCCCTACTG TCCCTACTG	CTGCCTCCCGI CTGCCTCCCGI CTGCCTCCCGI	AGGAGTTTGG AGGAGTTTGG AGGAGTTTGG	GCCGT GCCGT GCCGT
	1 .	190	200	210	220	230	240
VUP 12006 GCRL 163 ASC 388	GTCTCA GTCTCA GTCTCA	GTCCCAAT( GTCCCAAT( GTCCCAAT(	GTGGCCGATCA GTGGCCGATCA GTGGCCGATCA	ACCTCTCAG ACCTCTCAG <mark>GT</mark> CTCTCA <mark>A</mark> (	ITCGGCTACGI ITCGGCTACGI ICCGCTA <mark>I</mark> G	ATCATCGCCI ATCATCGCCI ATCAT <mark>T</mark> GCCI	TGGT <mark>G</mark> TGGTG TGGT <mark>A</mark>
		250 	260	270 • • • •   • • • •	280	290 • • • •   • • • •	300 • • • •
VUP 12006 GCRL 163 ASC 388	AGCCAT' AGCCAT' <mark>C</mark> GCC <mark>C</mark> T'	TACCTCACO TACCTCACO TACC <mark>CT</mark> ACO	СААСТАGСТАА СААСТАGСТАА СААСТАGСТАА	TACGCCGCG( TACGCCGCG( T <mark>C</mark> C <mark>A</mark> CCGCG(	GGTCCATCCAA GGTCCATCCAA GG <mark>C</mark> CCATCC <mark>C</mark> A	AAGCGATAGC AAGCGATAGC AAGCGA <mark>C</mark> AGC	TTACG TTACG TTACG
	1 .	310 • • •   • • • •	320	330	340	350	360 • • • •
VUP 12006 GCRL 163 ASC 388	CCATCT CCATCT CC <mark>GC</mark> CT	TTCAGCCA/ TTCAGCCA/ TT <mark>T</mark> A <mark>TAAGO</mark>	AGAACCATGCG AGAACCATGCG CTC <mark>AT</mark> CATGCG	GTTCTTGGA' GTTCTTGGA' A <mark>TCTGCTTT</mark> (	ICTATGCGGTA ICTATGCGGTA CTTAT <mark>C</mark> CGGTA	ATTAGCATCTG ATTAGCATCTG ATTAGCA <mark>C</mark> CTG	TTTCC TTTCC TTTCC
	1	370	380	390	400	410	420
VUP 12006 GCRL 163 ASC 388	AAATGT AAATGT AA <mark>C</mark> TG <mark>C</mark>	TATCCCCC TATCCCCCZ TATCCC <mark>AC</mark> Z	ACTTAAGGGCA ACTTAAGGGCA ACT <mark>ATG</mark> GGGCA	GGTTACCCA GGTTACCCA GGTT <mark>C</mark> CCCA	CGTGTTACTCA CGTGTTACTCA CGTGTTACTCA	ACCCGTCCGCC ACCCGTCCGCC ACCC <mark>A</mark> TCCGCC	ACTCG ACTCG CTCG
		430	440	450	460	470	480
VUP 12006 GCRL 163 ASC 388	TTCCATO TTCCATO C <mark>C</mark> T	GTTGAATC GTTGAATC ICCCAA <mark>CG</mark>	ICGGTGCAAGC ICGGTGCAAGC IC <mark>ATCA</mark> C <mark>CGAA</mark>	ACCGATCAT ACCGATCAT GTGAATC	CAACGAGAACI CAACGAGAACI TTCGTTC <mark>AC</mark> CI	CGTTCGACTI CGTTCGACTI CG <mark>C</mark> TCGACTI	'GCATG 'GCATG 'GCATG
		490	500	510	520		
VUP 12006 GCRL 163 ASC 388	TATTAGO TATTAGO TATTAGO	GCACGCCGC GCACGCCGC GCACGCCGC	CAGCGTTCAT CCAGCGTTCAT CCAGCGTTC <b>E</b> T	CCTGAGCCA CCTGAGCCA CCTGAGCCA	GGATCAAAC GGATCAAAC GGATCAAAC		

	60 7	0 8	0 9	0 10	00
Lb. casei ATCC 393	cgatgaacgg	tgcttgcact	gagattcgac	ttaaaacgag	g tggcggacgg
Lb. casei ATCC 334	tgatgatcgg	tgcttgcacc	gagattcaac	atggaacgag	g tggcggacgg
<i>Lb. paracasei</i> ATCC 25302	tgatgatcgg	tgcttgcacc	gagattcaac	atggaacgac	g tggcggacgg

Strain VUP12006 tgatgatcgg tgcttgcacc gagattcaac atggaacgag tggcggacgg

### Figure 3.7 Alignment of the V1 region of the 16S rRNA genes of *Lb. casei* ATCC 393 and 334, *Lb. paracasei* ATCC 25302 and strain VUP 12006.

Partial sequence data from *Lb. casei* type strain ATCC 393 was extracted from the 16S ribosomal RNA gene (gi | 28194135 | gb | AF469172.1 |) between positions 62 and 111. Positions 62 to 111 are incorporated in the V1 (variable) region of the gene. Sequence data in this variable region has been shown to assist identification of members of the *casei* group. This data was aligned to similar sequence data from *Lb. casei* strain ATCC 334, *Lb. paracasei* strain ATCC 25302 and strain VUP12006.

Examination of these alignments appears to support the identification of strain VUP 12006 as *Lb. paracasei*. Additionally, when the V1 variable region sequence for strains VUP 12006, *Lb. paracasei* type strain ATCC 25302 and the *Lb. casei* type strain ATCC 393 was submitted to the draft genome of *Lb. casei* strain ATCC 334, it was found that ATCC 334 shared 100% homology with *Lb. paracasei* type strain ATCC 25302 and strain VUP 12006 and only 87% with *Lb. casei* type strain 393.

As strain VUP 12006 was originally identified by CSIRO as *Lb. acidophilus*, two further primer pairs were used to confirm the identification. Using two primer pairs described by Walter (2001), which were shown to be specific for either *Lb. acidophilus* or *Lb. casei*, PCR was performed to support the identification of strain VUP 12006. The reaction mixture (50µL) for the *Lb. acidophilus* primer pair, ACI 16S 1/ ACI 16S 2, contained 30 pmol of each primer, 200µM each of dATP, dTTP, dCTP and dGTP, approximately 500 ng of bacterial DNA, 2.5 U of *Taq* DNA polymerase (Sigma) and PCR buffer with 1.5MgCl<sub>2</sub> MgCl<sub>2</sub> and 50mM KCI.

DNA was amplified as follows: initial denaturation at 95°C for 2 minutes, followed by 30 cycles consisting of denaturation at 92°C for 30 seconds, annealing at 62°C for 30 seconds, extension at 72°C for 30 seconds and a 1 minute final extension step at 72°C. Aliquots of the reaction mixture (10μL) were separated on a 0.8% agarose gel, stained with ethidium bromide and visualised under UV light. The results are shown in Figure 3.8.

Genornic DNA from strain VUP 12006 amplified using ACI 16S 1/ ACI 16S 2 failed to produce products while DNA from *Lb. acidophilus* type strain ASC 388 produced an amplicon of approximately 850 bp. *Lb. casei* strain GCRL-163 also failed to produce PCR products.

Figure 3.8 PCR amplification of DNA from Lactobacillus strain VUP 12006, *Lactobacillus acidophilus* VUP12000 (type strain) and *Lactobacillus casei* GCRL-163 using primers pairs ACI 16S 1 and ACI 16S 2 and primers pairs PR 1 and CAS 2.

Lane 1 is marker (generuler, Progen). Lanes 2, 4 and 6 are PCR amplification products of DNA using primers ACI 16S 1 and ACI 16S 2. Products of approximately 850 bp were only obtained from the DNA of *Lb. acidophilus* type strain VUP12000 (lane 6). These are indicated by pointed arrow. Some non-specific PCR products were also observed at approximately 600 bp. No products were detected from DNA of *Lactobacillus* strain VUP 12006 (lane 2) and *Lb. casei* GCRL-163 (lane 4).

Lanes 3, 5 and 7 are PCR amplification products of DNA using primers PR 1 and CAS 2, which were specific for the *casei* group. Products of approximately 450 bp were only obtained from the DNA of *Lactobacillus* strain VUP 12006 (lane 3) and *Lb. casei* GCRL-163 (lane 5). These are indicated by diamond headed arrow. No products were detected from DNA of *Lb. acidophilus* type strain VUP12000 (lane 7).



The reaction mixture (50µL) for the *Lb. casei* primer pair, PR 1/ CAS 2, contained 30 pmol of each primer, 200µM each of dATP, dTTP, dCTP and dGTP, approximately 500ng of bacterial DNA, 2.5 U of *Taq* DNA polymerase (Sigma) and PCR buffer with 1.5MgCl<sub>2</sub> MgCl<sub>2</sub> and 50mM KCl. DNA was amplified as follows: initial denaturation at 92°C for 2 minutes, followed by 30 cycles consisting of denaturation at 95°C for 30 seconds, annealing at 56°C for 30 seconds, extension at 72°C for 30 seconds and a 1 minute final extension step at 72°C. The results are shown in Figure 3.8.

Genomic DNA from strain VUP 12006 amplified using PR 1/ CAS 2 yielded a PCR product of approximately 500 bp. A similar sized product was obtained when DNA from *Lb. casei* strain GCRL-163 was used as the template DNA. *Lb. acidophilus* type strain ASC 388 failed to produce PCR products. After a review of these results, particularly sequence alignments of 16S rRNA, it was determined that the data supported an identification of strain VUP 12006 as *Lb. paracasei*.

#### 3.3 Physiological characterisation of the selected strains

A screening process was used to identify acid tolerant strains for further characterisation as it was anticipated that an acid tolerant strain would exhibit physiological characteristics that contributed towards acid tolerance. Microtitre plates were used to measure changes in  $OD_{600}$  in the eight cultures selected for further testing following the previously described carbohydrate utilisation studies. They were grown in MRS broth (37°C, in room air) at eight randomly selected pH values between 1.6 and 8.2 (see section 2.3.6). From the results of this screening test, it was possible to broadly compare general trends in growth rates calculated from the optical density values taken during growth at each pH as described in section 2.3.8. This data was used to determine the strain that had the maximum specific growth rate,  $\mu_{max}$ , for several pH values.

These growth rates were also used to determine an approximate range over which each of the organisms could initiate and continue growth. Growth rates were calculated from the plot of log 10

 $OD_{600}$  against time (section 2.3.8). Results are shown in Table 3.6. *Lb. paracasei* strain VUP 12006 demonstrated the highest  $\mu_{max}$  value of the eight strains tested (0.23 h<sup>-1</sup>) and the widest apparent growth range (pH 3.1 to 8.2) and this strain was selected for further characterisation. Using one- or two-litre fermenter systems, *Lb. paracasei* (strain 12006) was characterised by growth rate, production of metabolic products, protein profiles (1-D and 2-D) and fatty acid methyl esters profiles at different pHs and temperature, as described in the following sections.

#### 3.3.1 Optimum temperature and pH for growth for strain VUP 12006

For fermenter operation, see section 2.3.7. The optimum conditions for growth of *Lb. paracasei* (strain 12006) in MRS broth were determined by calculating  $\mu_{max}$  values for duplicate experiments. Optical density values were measured during growth over a range of temperatures (12 to 51°C) and pH values (2 to 8). The pH was fixed at pH 6.3 and the temperature varied between 12 and 51°C and the temperature was fixed at 37°C and the pH varied between 2 to 8. Optical density values were validated as being related to growth by performing dry biomass determinations on growing cultures as described in section 2.3.4.

The maximum specific growth rate during each growth cycle was plotted against either pH or temperature and used to determine optimum growth conditions according to the method described in section 2.3.8. The growth rates were calculated using a computer program for the graphic representation of microbial growth curves and batch fermentations. The program, written in BASIC, graphically displayed results from each run (Veiga *et al.*, 1991). A typical example of the results of a single experiment is shown in the screen capture images in Figure 3.9. The results for determination of temperature optimum are presented in Figure 3.10.

Table 3.6 Maximum specific growth rate and pH growth range of test strains determinedfrom growth in microtitre tray cultures.

Cultures were grown in MRS broth at 37°C in room air in microtitre trays as described in section 2.3.6. Eight randomly selected pH values (1.6, 2.4, 3.0, 3.8, 4.8, 5.6, 6.3, 7.8 and 8.1) were used to screen bacteria for the ability to grow over a wide pH range and demonstrate satisfactory growth over that range. The pH was adjusted with HCl.

Strain	$\mu_{max}$ (h <sup>-1</sup> ) when grown at	Highest pH in which the	Lowest pH in which the strain was able to initiate		
Identification	apparent optimum pH	strain was able to initiate			
		growth	growth		
12000	0.12 (4.9ª)	7.8	3.1		
12001	0.04 (5.6)	8.2	4.9		
12002	0.06 (5.6)	7.8	3.8		
12003	0.12 (6.3)	8.2	4.9		
12004	0.18 (6.3)	8.2	3.8		
12005	0.12 (5.6)	8.2	4.9		
12006	0.23 (5.6)	8.2	3.1		
12007	0.12 (5.6)	8.2	3.8		

<sup>a</sup> apparent optimum pH is shown in brackets



(A)



#### (B)

Figure 3.9 Computer screen captured images from the software program used to calculate specific growth rates for *Lb. paracasei* strain VUP 12006 grown in MRS broth at pH 6.3 and 37°C.

Figure 3.9 (A) shows a plot of optical density values over time. For this experiment generation time was calculated as 1.16 hours with a  $\mu_{max}$  of 0.60 h(-1). Figure 3.9 (B) shows growth rates calculated for each time interval with the maximum specific growth rate occurring at the peak of the plot.



Figure 3.10  $\mu_{max}$  plotted against temperature (A) and pH (B)

Maximum specific growth rate ( $\mu_{max}$ ) was calculated at different temperatures and pHs. Lactobacillus paracasei strain VUP 12006 was grown in one litre of MRS broth in two-litre fermenters at temperatures between 12° and 51°C with the pH set at 6.3 (A) or at 37°C at pHs set between 2 and 8 (B). Nitrogen gas was sparged through the fermenter at 0.1 L/min. Data from replicates was  $\pm$  7%. Strain VUP 12006 grew over the range of temperatures used (12-51°C) and showed a distinct optimum temperature for growth at 37°C  $\pm$ 0.5°C.

The results for determination of pH optimum for strain VUP 12006 are presented in Figure 3.10, which shows the maximum specific growth rate plotted against pH (pH range was 2 to 8). Growth was detected over the pH range tested and there was a sharp optimum pH at  $6.3 \pm 0.2$ .

The percentage change in the minimum to maximum specific growth rates (the slowest and fastest specific growth rates calculated from growth curves during growth at a specific temperature) was, in general, least in optimally grown cultures and highest at the extremes of growth (Table 3.7). The extent of growth after 24 hours was also greatly reduced at the higher temperatures. The optical density values dropped below the starting value after five hours and the calculated specific growth rate became negative after three hours for a culture grown at 51° C, indicating cell lysis. Specific growth rates were also affected by pH. The percentage difference in the minimum to maximum specific growth rates was lowest in optimally grown cultures and highest at the lowest pH, see Table 3.8. At pH 2.0 the calculated growth rate became zero after one hour.

#### 3.3.2 Growth characteristics under optimum conditions (pH controlled at pH 6.3)

Once the optimum growth temperature and pH were determined, *Lb. paracasei* strain VUP 12006 was examined for basic growth characteristics at optimum temperature and pH. These characteristics included changes in optical density during the growth cycle, biomass production, glucose utilisation and the amount of NaOH required to neutralise metabolic acids produced during growth. Firstly biomass and optical density values (Figure 3.11) were compared using the method described in section 2.2.3.

Table 3.7 Percentage difference between minimum and maximum specific growth rates achieved during growth at each different temperature between 12 and 51°C and extent of growth.

T°C	12	22	27	32	37	42	45	47	51
% Difference	85	39	45	28	20	71	51	100	155
Extent of	0.319	0.570	1.200	2.406	3.265	0.940	0.54	0.295	0.150
growth <sup>a</sup>									

<sup>a</sup> Maximum OD<sub>600</sub> reached after 6 hours in a typical experiment

Table 3.8 Percentage difference between minimum and maximum specific growth rates achieved during growth at different pH values between 2 and 8 and extent of growth.

<u> </u>	_					-			
рН		2	3	4	5	6	6.3	7	8
-									
% Difference		160	91	47	40	44	20	42	30
Extent	of	0.195	0.380	0.770	1.9565	2.500	3.265	2.550	0.690
growth <sup>a</sup>									

<sup>a</sup> Maximum OD<sub>600</sub> reached after 6 hours in a typical experiment

Figure 3.11 Growth characteristics of *Lb. paracasei* strain VUP 12006 under optimum conditions measured by changes in OD<sub>600</sub>, biomass, glucose concentration and addition of sodium hydroxide.

Some basic characteristics of a culture growing in MRS broth at optimum temperature (37°) and pH (6.3) were determined by measuring changes in a number of parameters. These included: optical density measured at 600nm ( $\Box$ ), glucose utilisation ( $\diamond$ ), biomass production ( $\bullet$ ) and the amount of 2 M NaOH (mL) required to be added to the growing culture to control the pH at the predetermined optimum ( $\blacktriangle$ ). Cultures were grown in 1 L of MRS broth in a 2L fermenter. Nitrogen gas was sparged through the fermenter at 0.1 L/min. The inoculum was prepared from an overnight culture of strain VUP 12006 grown in MRS broth without pH control. The culture was centrifuged (10 min at 4000 rpm) and the cells collected and resuspended in MRS broth pre-warmed to 37°C. The fermenters were inoculated to give a starting OD of between 0.13 and 0.23 then growth was followed for 24 hours.



The results indicate that biomass and optical density changed proportionally during growth and that optical density could be considered to reflect the extent of growth of the culture. It can also be seen that the culture reached exponential phase within one hour and that stationary phase was reached after about fifteen hours, using this particular inoculation procedure. Glucose utilisation was also examined (Figure 3.11). The rate of glucose utilisation was initially relatively slow and corresponded to the low biomass in the fermenter. As the biomass increased, the rate of glucose utilisation rose until the cells entered stationary phase and growth ceased when glucose was totally consumed. The sensitivity of the glucose assay was 0.2 g/L. Figure 3.11 shows the volume of NaOH added to the culture, noting that acids produced during growth were neutralised by 2 M NaOH in order for the culture to stay at pH 6.3. The utilisation of glucose is clearly mirrored by production of acids, as indicated by the total amount of NaOH added.

### 3.3.3 Growth characteristics for cultures commencing at optimum pH but without continued pH control

A control culture was grown at constant pH 6.3 while the second started at optimal pH (6.3) but the pH was not controlled. Figure 3.12 shows optical density values over 21 hours. The culture in which the pH was not controlled showed a slightly lower  $\mu_{max}$  value (0.51) than the control (0.60), but the extent of growth as measured by optical density after twenty-one hours, was essentially the same. Figure 3.12 also shows the pH values decreasing during growth in the uncontrolled culture. The final pH reached by these cultures was generally around pH 3.0, which was consistent with the screening tests.

#### 3.3.4 Effect of lactic acid concentration on growth

Potential inhibition by lactate on growth was examined by growing strain VUP 12006 in MRS broth containing, 5, 2.5, 1.25, 0.63, or 0.31 g/L of added lactate (Figure 3.13).

# Figure 3.12 Growth characteristics of *Lb. paracasei* strain VUP 12006 grown without pH control, starting at optimum pH, measured by changes in OD<sub>600</sub>.

A control culture of *Lb. paracasei* was grown under optimal conditions (37°C, pH 6.3 in 1 L of MRS broth) in a 2L fermenter and compared to an identical culture grown under the same conditions with the exception that the pH was not controlled at 6.3. Nitrogen gas was sparged through the fermenter at 0.1 L/min. The inoculum preparation and inoculation were as described in Figure 3.11.

Optical density ( $\triangle$ ) and pH ( $\blacksquare$ ) for growth controlled at pH 6.3/37°C and optical density ( $\Box$ ) and pH ( $\bullet$ ) for growth at 37°C (pH not controlled).




# Figure 3.13 The effect of lactate concentration on the growth of *Lb. paracasei* strain VUP 12006.

Lactate concentrations between 0.31 and 10 g/L in MRS broth at 37°C were used to determine how well *Lb. paracasei* strain VUP 12006 could initiate growth in the presence of a metabolic end product. The maximum specific growth rate ( $\mu_{max}$ ) for each concentration was calculated over six hours of growth from samples collected hourly. Maximum specific growth rate (•) was then plotted against lactate concentration. A control culture of *Lb. paracasei* was grown under optimal conditions (37°C, pH 6.3 in MRS broth) in a 1L fermenter without added lactate and had a  $\mu_{max}$  of 0.63. Nitrogen gas was sparged through all the fermenters at 0.1 L/min.

To differentiate between inhibition caused by altered pH and lactate ions, pH during growth at 37°C was maintained at a set point of 6.3. The inoculum was prepared by culturing strain VUP 12006 overnight in MRS broth.

Growth was severely affected by the presence of a high concentration of lactate in the growth medium despite the pH being held at 6.3. At a level of 10 g/L growth was essentially so slow it was equivalent to the growth rate observed in a culture growing at pH 3.0. At a lactate concentration of 2.5 g/L, the maximum specific growth rate of the challenged culture was slightly more than half that achieved by a culture that was grown in the absence of lactate. A control culture of *Lb. paracasei* was grown under optimal conditions (37°C, pH 6.3 in MRS broth) in a 1L fermenter without added lactate had a  $\mu_{max}$  of 0.63. This can be compared to the level of lactate in the growth medium achieved after six hours of growth at optimum temperature and pH by the same strain.

#### 3.3.5 Survival in extreme pH

Previous sections showed VUP 12006 grew poorly at very low pH. The following experiments were preformed to determine whether VUP 12006 could survive exposure to very low pH. Utilising the method of Gorden and Small (1993), *Lb. paracasei* strain VUP 12006 was subjected to extreme (lethal) acid conditions to determine the level of acid resistance. If greater than ten percent of challenged bacteria survived pH 2.5 for 2 hours they were considered acid resistant (section 2.3.9.2). This method was used to determine the ability of strain VUP 12006 to survive exposure to very low pH rather than being able to initiate growth in media at low pH.

## 3.3.5.1 Viability of strain VUP 12006 in MRS broth at pH 2 and 2.5

An ovemight culture of VUP 12006, grown in MRS broth at 37°C, was used to seed two fresh MRS broths at pH 2.0 and pH 2.5 respectively. The incubation temperature for this experiment was set at 37°C.

Samples were collected from these flasks every 30 minutes and viable cell counts performed (section 2.3.9.2). At pH 2.0, the percentage of colony forming units, as compared to the viable cell count at time zero, fell rapidly and only forty percent viability was detected after 30 minutes and after one hour it was not possible to detect any survivors (Figure 3.14). At pH 2.5 10% of the inoculum survived at 3.5 hours and 0.1% survived for 6 hours.

The rate of death was quicker for the first 3 hours and slowed during the second three hours. *Lb. paracasei* strain VUP 12000 was tested at the same time. This strain did not possess acid tolerance when tested according to the described protocol: at pH 2.0, after 2 hours, only 0.28% of starting numbers survived.

## 3.3.5.2 Relationship of growth phase to acid resistance

Strain VUP 12006 was challenged at different stages of the growth cycle as described in section 2.3.9.3. An overnight culture, grown in MRS broth at 37°C without pH control, was used to inoculate another flask run under the same conditions. Samples of the inoculum were removed from the fermenter every hour and inoculated into MRS broth at pH 2.5/37°C for 2 hours. The number of survivors was then determined (Figure 3.15). The results of this challenge indicate that the growth phase had a clear affect on the survival of this strain.

## 3.4 Effects of elevated temperature and low pH on growth

Strain VUP 12006 had clear optima with regard to temperature and pH and growth higher than 37°C or much below pH 6.3 may place the organism under stress. It was therefore proposed to observe changes that occurred in response to the strain growing outside optimum growth conditions.



Figure 3.14 The effect of exposure to very low pH on the survival of *Lb. paracasei* strain VUP 12006.

An overnight culture of VUP 12006, grown in MRS broth at 37°C, was used to seed two fresh MRS broths at pH 2.0 ( $\blacksquare$ ) and pH 2.5 () respectively, as described in section 2.3.9.2. VUP 12000 was used as a control {pH 2.0 ( $\blacksquare$ ) and pH 2.5 ( $\bullet$ )}. The incubation temperature for this experiment was set at 37°C. Samples were collected from these flasks every 30 minutes and viable cell counts performed (section 2.3.9.4). The percentage of survivors was calculated from the viable cell counts relative to counts at time zero.



## Figure 3.15 Relationship of growth phase to acid resistance for VUP 12006.

*Lb. paracasei* strain VUP 12006 was grown in a fermenter at 37°C in MRS broth without pH control. Every hour a sample of the culture was inoculated into MRS broth at pH 2.5. After 2 hours the percentage of remaining survivors was calculated. The graph shows % survivors (•) and optical density of the parent culture (o) plotted against time. Data collected previously (Figure 3.10) showed  $\mu_{max}$  values calculated at different temperatures and pHs. From this it was determined that VUP 12006 could be grown at pH 4.3/37°C and 45°C/pH6.3 to determine physiological changes. These conditions were chosen as they resulted in a decrease by 50% of the  $\mu_{max}$  compared to optimum conditions and the conditions are similar to those encountered during yoghurt production.

Culture VUP 12006 was grown at 45°C/pH 6.3 or pH 4·0/37°C and the effect on growth rate (from OD<sub>600</sub> changes), protein concentration, carbohydrate metabolism, protein profiles and FAME profiles determined. A control culture was grown at pH 6.3/37°C. These cultures were prepared from strain VUP 12006, which was sub-cultured from stationary phase after overnight growth in MRS broth at 37°C and in an initial pH of 6.3. The pH was not controlled during growth. The inoculum was further prepared by centrifugation (10 min at 4000 rpm) and the cells collected and resuspended in MRS broth pre-warmed to 37°C as described in section 2.3.7.1. This starter was inoculated to give a starting OD of 0.2 in three 1L fermenters (at 37°C/pH 6.3, 45°C/pH 6.3 or pH 4·0/37°C). The pH and temperature were maintained at the initial set point during growth.

### 3.4.1 Growth rate at 45°C or pH 4.0 of strain VUP 12006

The effect of elevated temperature and low pH on the extent of growth was measured from optical density (OD<sub>600</sub>) readings (Figure 3.16). Elevated temperature and low pH had a marked effect on the extent and rate of growth. As determined from the optical density measured after eight hours, the control culture achieved approximately six times the extent of growth of the heat-stressed culture and four times the extent of growth of an acid-stressed culture.



# Figure 3.16 Effect of acid and high temperature on strain VUP 12006 measured by changes

## in optical density.

Cultures of strain VUP 12006 were grown in one litre of MRS broth in one-litre fermenters under controlled temperature and pH conditions. The inoculum was prepared from an overnight culture of strain VUP 12006 grown in MRS broth without pH control. The culture was centrifuged (10 min at 4000 rpm) and the cells collected and resuspended in MRS broth pre-warmed to  $37^{\circ}$ C, as described in section 2.2.6.1. These starter cells were used to inoculate 3 fermenters to an initial OD of 0.15 (pH 6.3 and  $37^{\circ}$ C) then growth was followed for eight hours. The conditions in two fermenters were immediately changed to  $45^{\circ}$ C and pH 6.3 ( ) and  $37^{\circ}$ C and pH 4.0 ( ) and the control was cultured at pH 6.3 and  $37^{\circ}$ C (). Samples were taken hourly over eight hours and the optical density determined at 600nm. Nitrogen gas was sparged through the fermenters at 0.1 L/min.

Initially the heat-stressed culture grew at the same rate as the acid stressed culture but the growth rate had declined by the second hour.

#### 3.4.2 Changes in protein concentration of cells grown at 45°C or at pH 4.0

Protein concentration of cells harvested periodically during growth at 45°C/pH 6.3, pH 4·0/37°C and pH 6.3/37°C was measured by the Lowry assay (section 2.4.3) (Figure 3.17). In the culture grown at 45°C the protein concentration fell below the starting value at the first hour and did not regain that value until after the fourth hour of culture, then tripled between hours five and six. The protein concentration of the culture grown at pH 4.0 remained stable for the first three hours and only began to rise at the fourth hour. This increase reflected the parallel rise in the control but was at a lesser rate. After six hours, protein levels in both test cultures were half that observed in the control. Measurement of protein was able to detect a lag period that was not obvious from OD<sub>600</sub> readings but otherwise confirmed the lower growth rate and smaller biomass achieved at pH 4.0 or 45°C.

#### 3.4.3 Changes in metabolite concentration of cells grown at 45°C or at pH 4.0

The glucose concentration started at 20 g/L in all cultures. After 6 hours growth, the glucose concentration in the control fell to 12.5 g/L (Figure 3.18). The culture grown at pH 4.0 mirrored the change in the glucose level until the end of the fifth hour. During the sixth hour in the culture grown at pH 4.0, the glucose level fell from 17.6 to 14.1 g/L while the control went from 18.0 to 12.5 g/L. The culture grown at 45°C showed a marked drop in glucose levels after the first hour. After five hours the glucose level in the control was still high at 18.0 g/L while the culture grown at 45°C had already dropped to 14.9 g/L. The culture grown at 45°C did not continue to demonstrate this high rate of glucose use and during the sixth hour the glucose level only decreased another 1.1 g/L compared to the control, which decreased 5.5 g/L.

166



## Figure 3.17 Effect of acid and high temperature on strain VUP 12006 measured by changes

## in protein concentration.

Cultures of strain VUP 12006 were grown in one litre of MRS broth in one-litre fermenters under controlled temperature and pH conditions. The inoculum was prepared from an overnight culture of strain VUP 12006 grown in MRS broth without pH control. The culture was centrifuged (10 min at 4000 rpm) and the cells collected and resuspended in MRS broth pre-warmed to  $37^{\circ}$ C, as described in section 2.2.6.1. These starter cells were used to inoculate 3 fermenters to an initial OD of 0.15 (pH 6.3 and  $37^{\circ}$ C) then growth was followed for six hours. The conditions in 2 fermenters were immediately changed to  $45^{\circ}$ C and pH 6.3 ( ) and  $37^{\circ}$ C and pH 4.0 ( ) and the control was cultured at pH 6.3 and  $37^{\circ}$ C (). One mL samples were taken hourly over six hours and centrifuged to obtain a pellet which was resuspended in NaOH and boiled to disrupt the cells, (section 2.4.1.6). The protein concentration was determined from treated cells using the Lowry assay. Nitrogen gas was sparged through the fermenters at 0.1 L/min.



## Figure 3.18 Effect of acid and high temperature on strain VUP 12006 measured by changes

## in glucose (A) and lactate concentration (B).

Cultures of strain VUP 12006 were grown as described in Figure 3.17. These starter cells were used to inoculate 3 fermenters to an initial OD of 0.15 (pH 6.3 and 37°C) then growth was followed for six hours. The conditions in 2 fermenters were immediately changed to 45°C and pH 6.3 ( ) and 37°C and pH 4.0 (  $\blacktriangle$  ) and the control was cultured at pH 6.3 and 37°C ( $\blacksquare$ ). Samples were taken hourly over six hours and the level of glucose and lactate determined by HPLC, (section 2.4.2.2). Nitrogen gas was sparged through the fermenters at 0.1 L/min.

Lactate levels were not detected in the control until the fourth hour but rose rapidly between hours four and six (Figure 3.18). The limit of lactate sensitivity was 0.1 g/L. However, in both the cultures grown at 45°C and pH 4.0, lactate levels increased rapidly for the first two hours in contrast to the relatively slow growth rate seen. Levels continued to rise, but at a decreased rate, between hours two and five then rose again rapidly in the sixth hour. A HPLC trace of a culture grown at pH 6.3 and 37°C taken two hours after inoculation is shown in Appendix 3.

# 3.4.4 Effect of growth at low pH and elevated temperature on protein profiles (1-D SDS-PAGE)

SDS-PAGE was run according to the method described in section 2.5.1. Estimation of molecular weight and guantification of the relative amount of protein present in each band was determined using an Ultrascan XL laser densitometer (Pharmacia) (section 2.5.3.3). In protein profiles of cultures grown under optimum conditions (pH 6.3, 37°C) approximately 30 major bands with molecular weights between 5 and 115 kDa were observed, (Figure 3.19). The banding pattern of proteins from the control culture appeared not to change significantly after the fifth hour of culture although the relative proportions of some proteins (for example, in the molecular weight range greater than 97.4 kDa) did vary at different stages of growth. The initial sample showed a banding pattern more similar to the overnight culture. However, growth at low pH and elevated temperature caused significant alteration to protein profiles compared to the control, see Figures 3.20 and 3.21. Growth at low pH in particular resulted in a significant change to the SDS-PAGE profile: late in the growth cycle the profile was made up of five major and several minor bands. Among the five major bands (25, 35, 45, 50 and 75 kDa), one band of approximately 42 kDa was particularly prominent. When the gel was scanned using laser densitometry, the density of this band corresponded to 24% of the total density of all proteins detected by silver staining in the twenty-four hour sample. This protein increased from being undetected in the first sample.



Figure 3.19 1-D SDS-PAGE of cellular proteins of *Lb. paracasei* strain VUP 12006 grown at pH 6.3 and 37°C.

Culture samples corresponding to lanes 2-13 were taken hourly after inoculation and cells were collected, concentrated, re-suspended in double strength loading buffer and boiled before analysis by 1-D SDS-PAGE (sections 2.4.1.1 and 2.5.1.1). The sample in lane 2 was collected immediately after inoculation and the sample in lane 14 was collected after 24 hours of culture. The dashed line indicates where two separately run gels have been joined. The gel was silver stained to visualise the protein bands.

Lane 1: Molecular Weight Standards Lanes 2 to 13: Samples collected hourly for 12 hours Lane 14: Sample collected after 24 hours growth

Molecular weight standards:

1.	97.4 kDa	2.	66.2 kDa
3.	45 kDa	4.	31 kDa
5.	21.5 kDa		



## Figure 3.20 1-D SDS-PAGE of cellular proteins of Lb. paracasei strain VUP 12006 grown at

## pH 4.0 and 37°C.

Culture samples corresponding to lanes 2-14 were taken hourly after inoculation and cells were collected, concentrated, re-suspended in double strength loading buffer and boiled before analysis by 1-D SDS-PAGE (sections 2.4.1.1 and 2.5.1.1). The sample in lane 1 was collected immediately after inoculation and the sample in lane 14 was collected after 24 hours of culture. The dashed line indicates where two separately run gels have been joined. The gel was silver stained to visualise the protein bands. Protein banding pattern was produced during growth at low pH (pH 4.0 and 37°C). The arrow indicates protein band of approximately 42 kDa that was up-regulated during growth under acid stress.

Lanes 1 and 15 MW Standards Lanes 2 to 13 Samples collected hourly for 12 hours Lane 14 Sample collected after 24 hours growth

Molecular Weight Standards

1.	97.4 kDa	2.	66.2 kDa
3.	45 kDa	4.	31 kDa
5.	21.5 kDa		



## Figure 3.21 1-D SDS-PAGE of cellular proteins of Lb. paracasei strain VUP 12006 grown at

## pH 6.3 and 45°C.

Culture samples corresponding to lanes 2-13 were taken hourly after inoculation and cells were collected, concentrated, re-suspended in double strength loading buffer and boiled before analysis by 1-D SDS-PAGE (sections 2.4.1.1 and 2.5.1.1). The sample in lane 1 was collected immediately after inoculation and the sample in lane 14 was collected after 24 hours of culture. The dashed line indicates where two separately run gels have been joined. The gel was silver stained to visualise the protein bands. Protein banding pattern was produced during growth at pH 6.3 and 45°C. The arrow indicates the area between 50 and 90 kDa, where a number of minor bands that are up-regulated during growth at increased temperature, were detected by laser densitometry.

Lane 1 MW Standards Lanes 2 to 13 Samples collected hourly for 12 hours

Molecular Weight Standards

1.	97.4 kDa	2.	66.2 kDa
3.	45 kDa	4.	31 kDa
5.	21.5 kDa		

A protein of molecular weight around 42 kDa made up less than two percent of the proteins present in an optimally grown culture at any time throughout the growth cycle and in a culture grown at 45°C a protein of sirnilar size constituted only four to eight percent of the protein profile. During growth at 45°C only minor bands between around 50 to 90 kDa were up-regulated and they could only be detected by laser densitometry.

One of the other proteins that appeared during acid stress, a 50 kDa protein, also increased linearly throughout the culture time, as determined by laser densitometry and constituted between zero and ten percent of all proteins present. A protein of similar molecular weight was seen in the initial sample collected during growth at pH 6.3 and 37°C. The 35 and 75 kDa proteins were not present in significant amounts until the culture was exposed to acid stress for an extended time, greater than 12 hours, and may also be associated with stationary phase responses. The protein profile of heat stressed cultures displayed a general down-regulation of a number of proteins indicated by the decrease in the number of protein bands present in the profile. Increased expression of two proteins of about 60 and 70 kDa was detected by laser densitometry after a culture was grown at 45°C. An increase in expression of these proteins was not detected in control cultures.

#### 3.4.5 Effect of growth at low pH and elevated temperature on immunoblot reactions

Following SDS-PAGE, proteins were transferred by electroblot to nitrocellulose membrane and detection of specific proteins was performed using a Western blot technique (section 2.5.4) (Figure 3.22). For convenience the immunoblot shown resulted from the combination of two antibodies in the reaction mix, however when only one antibody was used, only single bands of the appropriate size were observed. Bands visualised using this technique were quantified by laser densitiometry (section 2.5.3.3).

173



Figure 3.22 Immunoblot to detect regulation of HSP60 and HSP70 analogues during growth at 37°C and 45°C.

Culture samples grown at 37°C/pH 6.3, corresponding to lanes 1-4 and 45°C/pH 6.3, corresponding to lanes 6-9, were taken every two hours after inoculation and cells were collected, concentrated, re-suspended in double strength loading buffer and boiled before analysis by 1-D SDS-PAGE. The 1-D SDS-PAGE was electroblotted to nitrocellulose membrane and detection of specific proteins was performed using a Western Blot technique (2.5.4). Lane 10 is a positive control prepared from heat stressed *E. coli* cells grown at 42°C. Lane 5 is BIO-RAD pre-stained standards Myosin 202 kDa,  $\beta$  galactosidase 133kDa, Bovine serum albumin 71 kDa, Carbonic anhydrase 41.8 kDa, Soybean trypsin inhibitor 30.6 kDa and Lysozyme 17.8 kDa.

(approximately 70 kDa ◀── , approximately 60 kDa <── )

Using this technique, the 60 and 70 kDa proteins up-regulated during growth at elevated temperature were shown to cross-react with antibodies made against *E. coli* hsp60 and *E. coli* hsp70. Positive controls prepared from heat stressed *E. coli* cells (grown at 45°C) were run on each gel. The production of the 60 kDa protein was found to increase four-fold during growth at 45°C but the same protein remained constant during growth at 37°C and pH 6.3.

Similarly, the 70kDa protein increased approximately four-fold while the level remained constant in the control. Cultures grown at low pH showed a similar response to the control culture in that protein bands at approximately 60 and 70 kDa cross-reacted with hsp60 and hsp70 antibodies but the levels remained unchanged during growth at low pH. It appears that growth at low pH did not cause up-regulation of analogues of HSP60 and HSP70.

#### 3.4.6 Effect of acid and temperature shift on FAME profiles

FAME analysis was performed as described in section 2.2.6.3 except that samples for comparison were taken from the fermenters at the mid-exponential phase of growth. Compared to growth at pH 6.3, growth at pH 4.0 resulted in a reduction of the percentage of the  $C_{18:1(9)}$  and  $C_{18:3}$  combined peak from 26% to 11% and the  $C_{19:0}$  was also reduced from 26% to 16%, (Table 3.9). A significant increase in relative abundance in the  $C_{16:1}$  peak was observed at the same time, with an increase from 2% to 10% observed. Increasing the growth temperature resulted in the decrease of the relative abundance of  $C_{19:0}$  from 26% to 12% while the  $C_{18:0}$  peak increased from 5% to 14%. No other significant changes were seen.

### 3.4.7 The effect of change in pH during growth on the expression of the 45kDa protein

In previous sections it was shown that continuous growth at pH 4.0 resulted in the over-expression of a 42 kDa protein, which was not up-regulated during growth at elevated temperature (45°C) or during growth at optimal temperature and pH (37°C and pH 6.3).

175

## Table 3.9 The effect of acid and temperature shift on FAME profile.

Long chain fatty acids present in *Lb. paracasei* strain VUP 12006 were converted into their corresponding methyl esters and and separated using a Varian gas chromatograph with a BPX-70 column. Twelve major peaks were found to be present. Data below is presented as a percentage of the specific fatty acids present during growth at low pH (pH 4.0 and 37 °C), high temperature (45°C and pH 6.3) and at 37°C and pH 6.3. Results were calculated from the area under a specific peak divided by the area under all peaks present in a specific strain and multiplied by 100. The data are means of two independent experiments.

	a Methyl ester of fatty acids (%):										
FAME	C10:00	X1	X2	X3	C14:00	X4	C <sub>16:00</sub>	C <sub>16:1</sub>	C <sub>18:0</sub>	C <sub>18:3</sub> &	C <sub>19:00</sub>
Retention time (min)										C <sub>18:1(9)</sub> <sup>b</sup>	
	2.05	3.5	3.8	4	4.2	5.9	6.3	6.8	9.6	10.2	12.3
Growth conditions		<u> </u>	I	I	<u></u>	I	I	<u> </u>	<u>                                     </u>		1
Optimum	0	8	0	4	2	2	17	2	5	26	26
Low pH (pH 4)	1	0	1	6	0	7	15	10	11	11	16
High temperature (45° C)	0	0	2	2	0	9	17	2	14	21	12

<sup>a</sup> Relative proportion of fatty acid methyl ester based on percentage of total area for the peaks detected. <sup>b</sup> Peaks for fatty acid methyl esters 18:3 and 18:1(9) could not be separated completely at the base line during all runs using the method described in section 2.4.4 and are presented as a combined percentage.

To determine whether this protein was induced in response to acid stress, a series of fermentations were performed were the pH was shifted during growth from optimal to pH 4, then held at this pH. Alternatively the pH was shifted from pH 6.3 to 4.0 then returned to pH 6.3 to produce a spiked change in pH. The fermenters were set up, in duplicate, as follows:

Fermenter	Growth conditions
Fermenter one (F1)	Optimum growth conditions (37°C and pH 6.3) for 10 hours. Addition
	of NaOH was discontinued after 10 hours, pH was not controlled
	overnight.
Earmontor two (E2)	07°C/continuous growth at all 4.0
rennenter two (rz)	37°C/continuous growth at pH 4.0.
Fermenter three (F3)	37°C/pH shifted to pH 4 during exponential phase of growth (two
	hours after start of experiment). Low pH conditions were continued to
	the end of the experiment.
Fermenter four (F4)	37°C/pH not controlled. Growth commenced at pH 6.3. The pH
	dropped below 4 by the end of each experiment.
Fermenter five (F5)	37°C and pH 6.3, spiked to pH 4.0 then returned immediately to pH
	6.3. The spike of acidic conditions was given at early exponential
	phase, two hours after start of experiment.

In F3 the pH took approximately ten minutes to reach pH 4.0 and in F5 the spike took nearly twenty minutes to cycle from pH 6.3 to 4.0 and back to 6.3. Cells were collected hourly and extracts prepared as described in sections 2.4.1.1. and 2.5.1.1. Protein profiles were determined by 1-D SDS-PAGE. Figure 3.23 shows the profiles seen for the five fermentation conditions, concentrating on the molecular weight range between approximately 30-70 kDa. Consistent with previous

experiments, the protein profiles of samples collected hourly from F1 (pH was controlled at 6.3), showed no apparent change in the regulation of the 42 kDa protein. When a spiked decrease in pH occurred during exponential growth at pH 6.3 (F5) there was no apparent change in the protein profile. However there was no attempt made to optimise the experimental protocol to detect induction of pH stress related proteins using this approach. When cells were cultured at pH 6.3 the pH decreased to pH 4.0 during exponential growth (F3) and there was a change in the protein profile so that it then resembled the one seen for continuous growth at pH 4.0 (F2). This showed the 42 kDa protein dominating during growth through to stationary phase. These data clearly show that for strain VUP 12006 the 42 kDa is linked to exposure to low pH.

## 3.5 Protein profiles of other Lactobacillus strains when grown at different pHs

Studies of 1-D SDS-PAGE profiles were performed on the other six strains of *Lactobacillus* described in section 3.2 to determine whether they showed similar responses to growth at low or high pH. Findings are presented in the following sections.

#### 3.5.1 Growth at pH 4.0 or shift from pH 6.3 to 4.0 during growth

*Lactobacillus* strains, VUP 12001 and 12007, were grown either at pH 4.0, pH 6.3 or at pH 6.3 and shifted to pH 4.0 at early exponential phase (after 2 hours growth). These conditions are the same as those in which *Lb. paracasei* strain VUP 12006 was grown for the experiment described in section 3.4.13. SDS-PAGE was performed on protein samples taken hourly during growth in 1L fermenters in MRS broth at 37°C. The SDS-PAGE gels for strains VUP 12001 and 12007 are shown in Figure 3.24. These data show that during growth at pH 4.0 the 42 kDa protein is present at constant levels in VUP 12001 (gel B) and VUP 12007 (gel C) but not up-regulated as for strain VUP 12006 (gel A).

Figure 3.23 The effect of continuous acid shift or spike shift on the regulation of the 42 kDa acid stress protein as determined by 1-D SDS-PAGE for VUP 12006.

*Lb. paracasei* strain VUP 12006 was grown in a set of fermenters at 37°C in MRS broth. The pH in each of the fermenters varied according to the conditions described in section 3.4.8. Each gel is shown in close-up between approximately 30 and 70 kDa, as indicated on F1. An arrow beside each gel indicates the level of migration of a 45 kDa standard. Culture samples corresponding to lanes 1-11 were taken hourly after inoculation and cells were collected, concentrated, then resuspended in double strength loading buffer and boiled before analysis by 1-D SDS-PAGE (sections 2.4.1.1 and 2.5.1.1). The sample in lane 1 was collected immediately after inoculation and the sample in lane 11 was collected after 24 hours of culture. The gel was silver stained to visualise the protein bands.

Key

Fermenter	Growth conditions (All cultures were grown at 37°C in MRS broth)
F1	Optimum growth conditions (pH 6.3) for 10 hours after which pH was not controlled.
F2	Continuous growth at pH 4.0
F3	pH shifted to pH 4 during exponential phase of growth (2 hours after start of experiment).
F4	pH not controlled. Growth commenced at pH 6.3.
F5	Commenced at pH 6.3, spiked briefly to pH 4.0 after 2 hours, then returned to pH 6.3.





Figure 3.24 Comparison of 1-D SDS-PAGE protein profiles of <i>Lactobacillus</i> strains VUP 12006, 12001 and 12007 when grown at different pH values
Lactobacillus strains VUP 12006, 12001 and 12007 were grown in a set of one-litre fermenters at 37°C in MRS broth. The pH in each of the fermenters
varied according to the conditions described below. Culture samples from each fermenter were taken hourly after inoculation and cells were collected,
concentrated, re-suspended in double strength loading buffer and boiled before analysis by 1-D SDS-PAGE (sections 2.4.1.1 and 2.5.1.1).

Key

							jed to 4.3 after 2 hours	jed to 4.3 after 2 hours	jed to 4.3 after 2 hours	
Growth	4.0	4.0	4.0	6.3	6.3	6.3	6.3 char	6.3 char	6.3 char	
Strain	VUP 12006	VUP 12001	VUP 12007	VUP 12006	VUP 12001	VUP 12007	VUP 12006	VUP 12001	VUP 12007	
Gel	A	m	с U		ω	  L	U			-

Arrow(



pH 4.3

pH 6.3

pH 6.3 changed to 4.3 after 2 hours

182

During growth at pH 6.3 in VUP 12006 (gel D) this protein is difficult to detect except by laser densitometry, but a protein of approximately 42 kDa is the most predominant protein produced during growth at pH 6.3 in *Lactobacillus* strains 12001 (gel E) and 12007 (gel F). Similar results were observed in the SDS-PAGE gels where the cultures were grown initially at pH 6.3 but shifted to pH 4.0 after one hour of incubation. The 42 kDa acid stress protein, which was observed to be up-regulated in strain VUP 12006 during growth at pH 4.0 was not seen to be up-regulated during growth in similar conditions in the other strains. A protein of identical molecular weight was seen in VUP 12001 and 12007 during growth at pH 6.3.

## 3.6 Partial purification and identification of acid stress protein of VUP 12006

Several attempts were made to partially purify the 42 kDa acid stress protein seen in SDS-PAGE gels that was up-regulated during growth at pH 4.0 in strain VUP 12006, with the aim of using the purified protein for N-terminal sequencing. The difficulties encountered during purification related to the method used to prepare samples used for 1-D SDS-PAGE. Initially protein samples were prepared by boiling whole cells in double strength loading buffer. When the same cells were homogenised as part of the preparation for 2-D SDS-PAGE, prepared as described in section 2.2.6.4, the banding patterns observed by 1-D SDS-PAGE were considerably different (Figure 3.25). It was observed that the protein that was markedly up-regulated during growth at pH 4.0 did not appear to be changed when time course samples were prepared by homogenisation and centrifugation, suggesting this protein was cell-membrane or surface-located.

Electrophoretic methods that used homogenised cells as a means of partially purifying the acid responsive protein were not found to be particularly useful. The methods used to partially purify the protein are described below.

183



Figure 3.25 Comparison of protein banding patterns between cells prepared three ways.

Proteins were prepared from a single culture. Cells were resuspended in 40 mM Tris pH 7. After homogenisation the sample was centrifuged and both the clear supernatant and pellet were collected. A control sample of centrifuged whole cells was also collected. Approximately 200  $\mu$ L of each sample was added to an equal volume of double strength loading buffer and heated to 100°C for 5 minutes. The samples were centrifuged and 20  $\mu$ L was loaded onto a 1-D SDS-PAGE and visualised by Coomassie stain. Arrow marks level of 45 kDa standard.

Lane 1 Pellet of cellular debris after homogenisation

Lane 2 Whole cells

Lane 3 Supernatant removed from cellular debris

#### 3.6.1 Partial purification using native PAGE and IEF (Prep and Rotofor cells)

Continuous elution electrophoresis-native PAGE was performed using a model 491 Prep Cell (BioRad). Approximately 0.5 g of crude homogenised protein was applied to the gel column, which was run according to the manufacturer's instructions (see section 2.4.3.4). The eluate from the column was collected as one mL fractions, which were subsequently run on 1-D SDS-PAGE to confirm which fraction(s) contained the protein of interest. Approximately 52 fractions, containing eluted proteins, were collected during each run. The protein of interest appeared between fractions 37 and 47. These fractions, however, were found to have low concentrations of the protein of interest and the fractions also contained several other protein bands. Using a similar approach, preparative isoelectric focussing using a Rotofor cell (BioRad) was performed according to the method described in section 2.4.3.5. Similar difficulties were experienced using this method, in that the fraction containing the protein of interest also contained several other proteins at much higher concentrations. It was decided not to persist with either method and to pursue partial purification through denaturing electrophoresis.

#### 3.6.2 Purification by denaturing electrophoresis and N-terminal sequencing

1-D SDS-PAGE of a sample collected during the late exponential phase of growth at pH 4.0 (six hours after inoculation of the culture) was performed in a Protean II slab system (BioRad). The useable length of the resolving gel was 12 cm and using pre-stained protein standards as markers, the gel was allowed to run until the 31.6 kDa marker reached the end of the gel. The gel was electroblotted to PVDF membrane and the band visualised with Amido Black 10B. Along with the gel length, this allowed sufficient separation of protein bands for the 42 kDa protein seen in acid stressed cultures to be visualised and excised from the PVDF membrane without any apparent contamination from other bands (Figure 3.26).

The 42 kDa protein was sent for N-terminal sequencing at the Australian Proteome Analysis Facility (APAF) and the following result was obtained: Asp, Thr, Ser, Asp, Ser, Ile, Ala, Ser, Asn, Lys, (Ser/Gln/Asp), Glu, Thr, Asn, Ala, Leu, Leu, Lys, Gln, Ile. Sequencing analysis was unable to determine the amino acid in position eleven. The initial search of protein databanks revealed no direct sequence match: however the closest match with 65% similarity was to a protein superfamily group of the cauliflower mosaic virus aphid transmission protein (PIR entry S01280). This search was performed through GeneFIND (http://www-nbrf.georgetown.edu/gf-cgi/genef) and ANGIS.

Subsequent searches through the same sites, after the searchable data base size had increased, yielded the following results, none were N-terminally located. The pattern match is shown below;

	Name of sequence match	Amino Acid Sequence
N-terminal sequence of acid		DTSDSIASNKSETNALLKQI
stress protein submitted for		
pattern		
Match 1	N-glycosylation site (PS00001)	D T S D S I A S N K S (Q D) E T N A L L K Q I
Match 2	Protein kinase C phosphorylation	DTSDSIASNKS (QD) ETNALLKQI
	site (PS00005)	· · ·
Match 3	Casein kinase II phophorylation	DTSDSIASNKS (QD) ETNALLKQI
	site (PS00006)	

The sequence was resubmitted for analysis using the tBLASTn program following the release of the draft genome of *Lb. casei* version '20Sep02'. The sequence was found to match the genome between positions 9501 and 9560 of scaffold 3, as shown below.

Query: 1 DTSDSIASNKSETNALLKQI 20

Sbjct: 9560 DTSDSIASNKSETNALLKQI 9501

Further studies were then performed to identify the gene sequence of the protein and its function

during growth at low pH.



# Figure 3.26 PVDF electro blotted membrane of 1-D SDS-PAGE showing excision of 42 kDa protein band for sequencing

*Lb. paracasei* strain VUP 12006 was grown at pH 4.0 and 37° for 6 hours. The cells were harvested, concentrated, re-suspended with double strength loading buffer and boiled before analysis by 1-D SDS-PAGE. The unstained gel was electroblotted to PVDF membrane and the protein bands were visualised with Amido black 10B (section 2.5.5). The 42 kDa band, marked by the green box, was excised from the gel and submitted for sequence analysis at The Australian Proteome Analysis Facility.

# 3.6.3 Construction of representative genomic library and screening for gene encoding the

#### 42 kDa protein

Chromosomal DNA of *Lb. paracasei* strain VUP 12006 extracted according to the method described in section 2.6.7.1 was partially digested with *Sau*3A I (section 2.6.7.2). Aliquots taken at different times were run on a 0.5% agarose gel and samples showing maximum fluorescence in the 10-25 kb size range were used to clone into the *Bam*H I digested vector pBluescript SK+. See Figure 3.27. Ligation of vector and insert are described in section 2.6.7.3. *E. coli* TG2 cells were transformed with the ligation mix and colony blots containing 47 transformants and 1 position marker on each membrane were prepared as described in section 2.6.8.

Using results from the N-terminal sequence of the 42 kDa protein, a probe of 17-mer was constructed according to the method described in section 2.5.6. The probe was constructed with the following sequence:

5' GAA ACN AA(TorC) GCN (CorT)TN (CorT)C 3'

The melting temperature for the probe was calculated at  $60.52^{\circ}$ C using the following formula Melting temperature =  $81.5 + 16.6 (\log 10 [Na^+]) + 0.41 [(G + C)/n (100)] - 600/n$ 

Where n=17, [Na+]=0.5 and G+C=8.

The probe was labelled with ( $\gamma^{32}$  P) ATP and used to screen the representative genomic library by colony blot using the calculated melting temperature for the radio-labelled probe for hybridisation (section 2.6.11). Autoradiographic techniques were used to visualise positive reactions indicating which colonies contained transformed cells that hybridised with the probe.



## Figure 3.27 Sau3A I digestion of DNA from Lb. paracasei strain VUP 12006

Agarose electrophoresis gel (0.5% agarose) of *Lb. paracasei* DNA completely digested, undigested and partially digested with *Sau*3A I. The partially digested DNA was used to create a representative Genomic DNA library.

Key

Lane 1 Complete digest of Lb. paracasei DNA

Lane 2 Undigested Lb. paracasei DNA

Lane 3 Lb. paracasei DNA digested with 0.012 units of Sau3A I /µg DNA

Lane 4 SPP EcoRI marker (Progen)

SPP1/ EcoRI Fragment sizes in base pairs: 8,500, 7,350, 6,100, 4,840, 3,590, 2,810, 1,950

Over one thousand clones were screened and 19 positive *E. coli* transformants were detected. Preparations of plasmid DNA were made from these transformants as described in section 2.5.1. Size estimation of the plasmid by comparison with standards electrophoresed in 1% agarose indicated that only 5 clones had inserts over 300 bp. The inserts from the plasmids of these 5 clones were then sequenced, however, they failed to yield any significant matches when submitted to a BLASTn search. It was decided at that point to use PCR-based techniques to identify the gene associated with the sequenced amino acids, particularly as the draft genome of *Lb casei* strain ATCC 334 had recently been released to allow this approach.

# 3.6.4 Investigation of the *Lb. casei* ATCC 393 draft genome sequence for 42 kDa acid stress protein N-terminal sequence

The N-terminal amino acid sequence of the 42 kDa acid stress protein was submitted to the Oak Ridge National Laboratory (ORNL) internet site http://genome.ornl.gov/, see sections 2.6.15 and 2.6.16. A BLASTp search of the *Lactobacillus casei* draft genome (version '20sept02') was performed using that sequence data. As previously stated a sequence match with 100% homology was found between positions 9560 and 9501 in scaffold 3 of the draft genome. Scaffold 3 was one of 276 scaffolds in the draft genome and was 82,578 bp in length. Refer to Appendix 4 for a copy of the ORNL microbial BLAST results. At the same time, all known completed microbial genomes and the draft genomes of *Lb. bulgaricus, Lb. gasseri, Lc. lactis, Leuconostoc mesenteroides* and *B. longum* were searched for a match to the same N-terminal sequence but without further success.

The sequence information in Scaffold 3 was transferred to a file in ANGIS for further study. A six frame translation was performed to find and translate open reading frames (ORFs). The ORF containing the amino acid sequence of interest was found in Frame 4 (reverse) between positions 9644 and 8457. The predicted mass, length and pl of the protein were 40.802 kDa, 396 amino acids and 7.66 respectively. This information was obtained using the ANGIS Translate program.

An SPScan was used to scan the ORF for the presence of secretory signal peptides as the Nterminal sequence of the 42 kDa protein was not found at the start of the ORF. A cleavage site was predicated 28 arnino acids from the start of the gene. This information is shown in Figure 3.28. The predicted mass of the cleavage product was 3.055 kDa. The predicted mass of the protein in *Lb. casei* ATCC 393 after cleavage was 37.979 kDa.

#### 3.6.5 PCR based approach to identification of the 45 kDa protein

Based on the *Lb. casei* draft genome and using sequence information in the ORF containing the gene coding the 42 kDa protein of interest, three sets of PCR primers were designed to test for the presence of the gene in *Lb. paracasei* strain VUP 12006. The composition of the oligonucleotide primers used in this round of PCR is described in Table 3.10.

Name	Sequence	Melting
		temperature
SP1S	AGATAGCCGAACACGATGCC	66.2
SP1AS	CACGACGACAGCAAAGAAAAC	64.6
SP2S	AACACAATGCTCTCACGAC	58.8
SP2AS	ACGACGACAGCAAAGAAAAC	61.8
SP3S	ATGGTGCGGGCAATTAAG	63.4
SP3AS	TGAAGATGCGAACACAGAAG	61.6

#### Table 3.10 PCR primers used in round one of gene identification.

Figure 3.29 shows the location of these and subsequent primers based on sequence information from *Lb. casei* ATCC 334 between positions 7621 and 10381 of Scaffold 3 of the draft genome. The PCR reaction mixture (50µL) contained 30 pmol of each primer, 200µM each of dATP, dTTP, dCTP and dGTP, approximately 100ng of bacterial DNA, 2.5 U of *Taq* DNA polymerase (Sigma) and PCR buffer with 1.5 mM MgCl<sub>2</sub>. DNA was amplified as follows: initial denaturation at 95°C for 2 minutes, followed by 30 cycles consisting of denaturation at 94°C for 30 seconds, annealing at 63°C for 30 seconds, extension at 68°C for 2 minutes and a 5 minute final extension step at 68°C.

SPScan of Scaffold\_3.1

Weight matrix: GenRunData:spgpos.dat

Minimum score for SPs (threshold): 7.0

Predicted cleavage sites indicated by '^'.

> sequence: Scaffold\_3.

name: Scaffold\_3. check: 7395 from: 1 to: 396

1. 1 MKFNKVMITLVAAVTLAGSASAVTPVFA^DT 30

Score: 11.5

Probability: 3.137E-03

SP length: 28

McGeoch scan succeeded:

Charged-region statistics:

Length: 5 Charge: 2

Hydrophobic-region statistics:

Length: 8 Offset: 6 Total hydropathy: 57.4

Maximum 8-residue hydropathy: 57.4, starting at 7

## Figure 3.28 SPScan of Scaffold 3 of the *Lb. casei* draft genome.

Using sequence information obtained from Scaffold 3 of the *Lb. casei* draft genome an SPScan was performed. SPScan was used to determine the presence or absence of an amino acid signal peptide in the ORF containing the amino acid sequence of the 42 kDa acid stress protein. Information in this figure was copied from ANGIS Biomanager results file. The program detected a probable signal sequence of 28 amino acids.

7621	CGTCATTTCA	TCGGTGACTG	CTTCTACGTT	CATGTCTGCT	GGCTCCCGCG	ATGGTTGCAA
7681	GCAGAAGCAG	GACTGCCGCC	GCCAAAGCAA	ACCAAGTCCA	ACCGGCAATG	GGAACATGAG
7741	CATCTGCCAA	ATAGCCAAAG	<b>CCATAC</b> ATGA	TAACAAGGTG	GCCGCCAGCA	AGGACTAGTG
7801	CAGTGAACGT	CATGCTAAGG	CGCTGCCAGA	TTGTGACGGC	GTTGGTGAAC	CAAGCCGTGG
7861	CCATGACTAG	TAGCTGTGCC	AGCAGCAACA	AAGCCATGAG	TAGCTGTAAA	ATTGTGTTGA
7921	GATACACCAT	TTTATTGGGC	TCCTTTAAAC	GTGCTGTAAA	GGACGAATCA	ATTTTGCATT
7981	GTTGTACACG	TATCATACCA	ACAAAAATTC	AGACTTGATA	GTGTTATGCT	GCGTCTCAGT
8041	TACAGACAGC	AGCAGAGATA	AACAAGGTCG	TTTGGTGAAA	AGCCGATATC	GATTCCGTGT
8101	TGGGTATCAT	AACTGGTGAC	AATTGCGATT	CCTGTCATGT	CGCGTTTGAT	TGCTACCGGG
8161	GCGTGTCTCG	AACAAGCGAA	ATAGTTGCTG	ATAAACCTCC	TTAAG <b>TGCCA</b>	AAAAAAGAAG
8221	CCACAAGGAA	TGCGATCGAA	CATTCTTTGT	GACTTCTTT	AGATAACTCG	GTAGTTTTTA
8281	AGAAAGAGCG	TGAGCCAGAC	CGGTTAGAAA	CCGGAGTGTA	AGTGGGCCTG	AACGTGATGA
8341	CCGGGCTTTG	GTCATTGCGT	TCAGGGTCCT	TACACGCAGG	GTTCTGGTCT	GGCGAACGCG
8401	TTTA <mark>GGT1GG</mark>	GATOGTGATG	TCATCCCAAG	ACGTGATTAA	GGGTAGGTAA	AAATTACCGG
8461	TGGATATAAA	CGTAGCTGCT	TGCACC	ATGGTTTGGG	3611CCGGGCC	GGTTGGTGAG
8521	CTGAAGCCCA	TGCCGCCTTG	GCTGATCGTG	ACACTGTCAC	CGGAGACAGA	TTGGACATAA
8581	GCAACGTGAC	CATATGAACC	ATCGGCTGTC	CATTGACCGC	CAACGGATTG	ACCAGCGGCG
8641	AAGACGATGA	TTGAGCCGGC	TGCTGGCGTG	TGGTTAACAG	TGAAGCCAGC	TGCTGCGGCT
8701	GAACCGCCCC	ATTGTGCGCC	GTTGCCCCAG	CCATTACCAG	CCCATGGTGC	AACTGATTTG
8761	ACGTACCAGG	TGCACTGACC	CCAAGGATAG	GTGTTACCTG	AACCACTGAA	GTCAGCGTGG
8821	CTGCCACTGC	CTGAAG <mark>CTGT</mark>	GTTGTTGG 3	ATGGTG	GC	GCTGCTTGAA
8881	CTGCTGGTAG	TAGAAGAACT	TGCGGTTGAA	GCGGCAACGG	TCTTCAAAGC	GGCTTGGGTT
8941	GCAGCAGCTG	CTTCAGACTG	TTCTTGGGCA	AATTGGCTTT	GCAAAGCAAC	CAACTCATCT
9001	TTATGATCAG	CAATTTGTTT	ATTCAAAGCA	TCTTGTTTAT	CGTTAGCATC	TTTTTGTAGG
9061	GAAACCAACT	GTGATTTCTG	GGTTTCCAAT	GAAGCCTTAG	TTGAAACCAG	CGTCTGACGG
9121	GALENT	Carlos Greent	CAAGCCAGCT	AATTTATCCT	TGGCAACTGT	CACGGCATCA
9181	AGCGCATCTT	TGCTGGCTTG	GCTAAGTTTG	CCAACCGTCA	TGGTGCGGGC	AATTAAGTCA
9241	GACAAGTTCT	GGGAATTCAG	AACAAAGTCG	ATATAGACAT	TGCCGCT	
9301	CT	GCAAGGAGAT	CAACTGATCT	TTCAAGTTGT	TCTTGCGTGC	TGTGACATTC
9361	TTTTGAGCGG	CAACAATTTC	ACCGCTGAGT	GATGAAATCT	TGGCATCAGT	TGCGCTGATC
9421	TTAGCCGTTG	CATCGCTGAT	TTGACCATTC	TTGGTATCGA	TCTGCTTGTT	CAAATTGATA
9481	ACTTCTGTGT	TCGCATCTTC	ATTTGCTTC	AACAGTGCAT	TGGTTTCGCT	TTTGTTGGAA
9541	GCGATACTGT	CGCTTGTATC	AGCGAAAACT	GGTGTTACGG	CGCTAGCAGA	ACCTGCTAAG
9601	GTAACTGCAG	CAACCAACGT	GATCATGACT	TTATTGAATT	TCATTTTTGA	CCTCACCCTT
9661	AATTTGATTA	ACTAAAGGCT	ACTTTACTGT	GTCAATGTTA	CGTCAGCGTT	ACAGGACTGT

TAGGAGTTGG TCTCAATCTT CATTTGCGCT TCAAAATTGC GATCATGAAC GTTTTTAATC 9721 CCAATTTTTG ATGAATTTTG TATGAAAAAA GCAGCGCCGT GACGACGCTG CCTGACAAAA 9781 TTGTAGGTTA TTCGTTTGAT TTCAGGGCTT CGACCATGTC CACCCGTTGG AGTCGACGAT 9841 GCGTGAGCCA GGTGACGACG CCGGTGAAGG CCAACATGAG TAACGTGGCC GTGAGGTAGC 9901 9961 CGACAATGCT GATCGTGAGT GGAAAAACCA CGGCCTCAGT TTCGGCTTGA TACAAAATGT 10021 AGGCTGTCAG CAAATTGCCG AGCAGATAGC CGAACACGAT GCCGATCACT GTCAACACAA 10081 TGCTCTCACG ACTGATGTAC ATGGTGACCT CGCGATCGAA AAAGCCAAGC ACCTTGATTG TGGATAGTTC CCGAATGCGT TCCGAAACGT TGATGTTATT CAGATTGTAC AAGACCACGA 10141 10201 ATGACAAAAC CGCACTGAGC AGAATAAAAA TCAAAAACAAT AGGATCTAAC ATACCGCTCA 10261 TATTGCTTAA CCTTTTCTTTCCTTCCTCC-TCGAAACTAAT CCCGACAATC GCGTGATGAT 10321 TGAGCAATTG CTTAGCGAGT CGATCATTTT GTTTGCTGCT TTGGGGCTGC AAGCGGACTA

# Figure 3.29 *Lb. casei* strain ATCC 334 sequence data obtained from Scaffold 3 of draft genome version '20sept02' showing location of PCR primers.

*Lb. casei* strain ATCC 334 sequence data is shown between positions 7621 and 10381 of Scaffold 3 of the draft genome. PCR primers used to amplify DNA in this region are shown in this figure using colour coding. In general, the font is coloured for sense primers and antisense primers are highlighted. Primers were designed using PrimeGCG (ANGIS).

Name of primer	Identification	Name	Identification
SP1S	SP1S	2SP4S	
SP1AS	SP1AS	2SP4AS	2SP4AS
SP2S	SP2S	2SP5S	
SP2AS	SP2AS	2SP5AS	2SP5AS
SP3S	SP3S	3SP1S	3SP1S
SP3AS	SP3AS	3SP2S	3SP2S
2SP1S	2SP1S	3SP3S	3SP3S
2SP1AS	and the second	3SP1AS	3SP1AS
2SP2AS	2SP2AS	3SP4S	3SP4S
2SP3S	2SP3S	3SP2AS	3SP2AS
The results are shown in Figure 3.30. Genomic DNA from strain VUP 12006 amplified using SP1S/SP1AS, SP2S/SP2AS and SP3S/SP3AS were expected to yield PCR products of approximately 248, 217 and 281 bp respectively. As seen in Figure 3.30, the sizes of PCR fragments obtained were comparable to the sizes predicted from the *Lb. casei* genome. Sequence of the PCR products, in duplicate, was determined using a CEQ 2000 (Beckman Coulter) according to the manufacturer's instructions.

The sequence data for each fragment showed 100% homology to sequence found in the *Lb. casei* ATCC 334 ORF. Based on these results, further primer pairs were designed based on the sequence data available for *Lb. casei* ATCC 334 between positions 8401 and 10381 of scaffold 3. The composition of these oligonucleotide primers is described in Table 3.11.

Name	Sequence	Melting
		temperature
2SP1S	GTCCTTACACGCAGGGTTCT	63.0
2SP1AS	CGAACACCCAAACCATCTCT	63.8
2SP2AS	CGGCTCAATCATCGTCTTC	63.4
2SP3S	CCAGAGATGGTTTGGGTGTT	63.7
2SP4S	CGATGGTGCTGGTTGAAGT	63.8
2SP4AS	TGAAGACGGAACAGGAAAATG	63.8
2SP5S	GCATTTTCCTGTTCCGTCTT	62.9
2SP5AS	AAAAGGCTGGCAACTCTGTC	63.3

 Table 3.11 PCR primers used in round 2 of gene identification

The PCR reaction mixture was identical to that used for round one of PCR. The expected sizes of PCR products and those determined from the agarose gel are shown in Table 3.12. The results of the second round of PCR are shown in Figure 3.30. A slight increase in the observed size of some product pairs compared with expected sizes was observed for some primer pairs.



# Figure 3.30 PCR amplification of DNA from *Lactobacillus* strain VUP 12006 using primers designed from *Lb. casei* ATCC 334 draft genome.

Lane 1 is marker (generuler, MBI Fermentas). Lanes 2-4 are PCR amplification products of DNA using primers 2SP1S/2SP1AS {141} (Lane 2), 2SP1S/2SP2AS {294} (Lane 3), 2SP3S/2SP2AS {172} (Lane 4), No sample (Lane 5), 2SP3S/2SP4AS {656} (Lane 6), 2SP4S/2SP4AS {294} (Lane 7), 2SP5S/2SP5AS {186} (Lane 8), 2SP5S/1SP3AS {331} (Lane 9), SP3S/SP2AS {1074} (Lane 10), SP1S/SP1AS {248} (Lane 11), SP2S/SP2AS {217} (Lane 12) and SP3S/SP3AS {281} (Lane 13). Expected product sizes, based on *Lb. casei* ATCC 334 draft genome, are shown in brackets {..} after each of the primer pairs.

Table 3.12 Expected and observed PCR product sizes from round two of gene identification.

Primer pair	PCR product size estimated from	Expected size based on Lb.
	gel	<u>casei</u> draft genome
2SP1S/2SP1AS	No product	141 bp
2SP1S/2SP2AS	No product	292 bp
2SP3S/2SP2AS	≈170	172 bp
2SP3S/2SP4AS	≈700	656 bp
2SP4S/2SP4AS	≈300	294 bp
2SP5S/2SP5AS	≈190	186 bp
2SP5S/1SP3AS	≈350 bp	331 bp
2SP3S/1SP2AS	≈1200 bp	1074 bp

Genomic DNA from strain VUP 12006 amplified using combinations of primers pairs were expected to yield the following size fragments; it should be noted that some primers were used in more than one pair. It can be seen in Figure 3.30 that no products were obtained from either 2SP1S/2SP1AS or 2SP1S/2SP2AS. Sequence of the PCR products was performed at least in duplicate, from primer pairs 2SP3S/2SP2AS, 2SP3S/2SP4AS, 2SP4S/2SP4AS, 2SP3S/1SP2AS and 2SP5S/2SP5AS.

The sequence data, based on both forward and reverse primers, was aligned using the sequence consensus program in ANGIS. No stop codon was found in the sequence data obtained from *Lb. paracasei* strain VUP 12006, indicating that the gene had not yet been full sequenced: however, some differences were noted between the sequence data for two strains, notably an area of serine repeats (Figure 3.31), and further attempts were made to obtain the complete sequence of the gene. Initially anchored PCR was used in an attempt to extend known sequence data but this proved to be unsuccessful. A later trial of PCR, using primers designed from sequence data well down stream of the *Lb. casei* gene, proved more productive and the gene sequence was able to be determined. These results are presented in the following sections.

#### 3.6.6 Anchored PCR

A variation of PCR, called anchored PCR, was used to obtain the remaining gene sequence. As sufficient sequence was only available to make a single primer, a known sequence was added to the end of the DNA by ligation of a known piece of DNA. The PCR was then performed with the gene-specific primer and the anchor primer. Briefly, DNA from *Lb. paracasei* strain VUP12006 was separately digested using the restriction enzymes EcoRI and Pstl (Geneworks). EcoRI or Pstl digested DNA (8ng) was subsequently ligated to pGEM-Teasy vector (50ng) using T4 DNA ligase. Using the primer combinations 2SP2AS/M13F and 2SP2AS/M13R several attempts were made to amplify DNA fragments. M13F and M13R were primers designed to amplify DNA from the pGEM-Teasy vector (Promega). The composition of the oligonucleotide primers used for anchored PCR is shown in Table 3.13.

#### Table 3.13 PCR primers used for anchored PCR

Name	Sequence	Melting
		temperature
pUC/M13-F	CGCCAGGGTTTTCCCAGTCACGAC	72.5
pUC/M13-R	TCACACAGGAAACAGCTATGAC	55.2

Although several fragments were obtained using this method, the size of the fragments was too small (<800 bp), given that the vector was included in the product. Due to the small size of the fragments, subsequent sequencing consistently yielded only sequence data that had previously been obtained and use of this method was discontinued.

#### 3.6.7 Downstream PCR

Further primers were designed, based on the region 200-1500 bp downstream of the gene of interest in *Lb. casei* ATCC 334. The composition of these oligonucleotide primers is shown in Table 3.14.

#### Table 3.14 Composition of PCR primers used for third round of PCR

Name	Sequence	Melting
		temperature
3SP1S	AGACAGCAGCAGAGATAAACAAG	52.9
3SP2S	ACAGACAGCAGCAGAGATAAAC	52.6
3SP3S	TGCCAAAAAAAGAAGCCACAAG	52.9
3SP4S	GCCAAATAGCCAAAGCCATAC	52.1
3SP1AS	CACCATCGCCAACAACACAG	54.4
3SP2AS	ATCACATCACGATCCCAACC	51.9

The PCR reaction mixture and amplification steps were identical to that previously described for the first two rounds of PCR. Genomic DNA from strain *Lb. casei* ATCC 334 amplified using combinations of primer pairs were expected to yield the size fragments described in Table 3.15.

#### Table 3.15 Expected size of PCR products

Primer pair	Expected size based on Lb. casei draft genome
3SP1S/3SP1AS	792 bp
3SP2S/3SP1AS	794 bp
3SP3S/3SP1AS	630 bp
3SP4S/3SP1AS	1090 bp
3SP1S/3SP2AS	360 bp
3SP2S/3SP2AS	362 bp
3SP3S/3SP2AS	198 bp
3SP4S/3SP2AS	658 bp

Results for *Lb. casei* ATCC 334 matched the expected product sizes, however the results for strain VUP 12006 showed that, of the eight primer pairs used, only four PCR amplification products were obtained from three primer pairs and these varied in size from the fragments obtained from *Lb. casei*. The results are shown in Figure 3.31. This data suggests that the area outside the gene varies between the species.



### Figure 3.31 PCR amplification of DNA from *Lactobacillus* strain VUP 12006 using primers designed from Lb. casei ATCC 334 draft genome.

Lanes 1, 10 and 19 are marker (generuler, MBI Fermentas). Lanes 2-9 are PCR amplification products of *Lb. casei* strain GCRL-163 DNA, using primers 3SP1S/3SP1AS {792} (Lane 2), 3SP2S/3SP1AS {794} (Lane 3), 3SP3S/3SP1AS {630} (Lane 4), 3SP4S/3SP1AS {1090} (Lane 5), 3SP1S/3SP2AS {360} (Lane 6), 3SP2S/3SP2AS {362} (Lane 7), 3SP3S/3SP2AS {198} (Lane 8), 3SP4S/3SP2AS {658} (Lane 9). Expected product sizes, based on *Lb. casei* ATCC 334 draft genome, are shown in brackets {..} after the primer pairs. Lanes 11-18 are PCR amplification products of *Lb. paracasei* strain VUP 12006 using the same primer pairs.

Subsequently, the products obtained from strain VUP 12006 from those primers, were sequenced in duplicate. After a BLASTn search of the *Lb. casei* draft genome, sequence data from two PCR products, obtained from primer pairs 3SP4S/3SP1AS and 3SP4S/3SP2AS, were found to align with sequence from scaffold 3. The estimated size of these fragments, based on comparison to a generuler size marker (Fermentas), was approximately 3000 and 2500 bp.

A consensus sequence based on the alignment of the sequences obtained for these fragments and for the previous two rounds of PCR is shown in Figure 3.32. Consensus sequence data was obtained for each primer pair, individually for both foward and reverse primers, using the multiple alignment program ClustalW and CAP contig, a contig assembly program based on sensitive detection of fragment overlaps (BioEDIT). This process was repeated using the consensus sequence for each individual primer pair to build an overall consensus sequence.

#### 3.6.8 Submission of sequence data to ANGIS for analysis

The consensus sequence data obtained from the three rounds of PCR of interest was submitted to ANGIS to determine the presence of an ORF, to establish the amino acid translation sequence and make-up of the gene within the ORF and to detect the presence of signal peptides. Nucleotide and function similarities to known genes and a ClustalW alignment of the ORFs of *Lb. paracasei* VUP 12006 and *Lb. casei* 334 was also performed as well as an alignment to functionally similar proteins. Hydrophobicity data on the protein was also obtained using a Kyte and Doolittle plot of hydrophobicity.

Flip ORF, accessed through ANGIS, was used to find and translate in all six frames an open reading frame in the consensus nucleotide sequence of *Lb. paracasei* strain VUP 12006 shown in Figure 3.33. Nineteen ORFs were located in the submitted sequence but only one had a length greater than 300 bases and this was 1263 nucleotides in length and encoded a protein of 420 amino acid residues.

#### Figure 3.32 Consensus sequence of strain VUP 12006.

Sequence data based on PCR products derived from eleven primer pairs, was aligned against *Lb. casei* ATCC 334. Alignment is shown between positions 8184 and 10292 of scaffold 3 in the *Lb. casei* ATCC 334 draft genome. Two non-homologous regions can be seen between positions 8281 and 8404, and 8837 and 8884. Black boxes indicate areas of homology. Nucleotide bases are coloured either black (G), red (T), green (A) or blue (C).

		8190	8200	8210	8220	8230	8240	
ATCC 334	8184	GTTCCTGATAA	ACCTCCTTAAC	GTGCCAAAAA	AAGAAGCCAC	AAGGAATGCG	ATCGAACAT	8243
12006	8184	CTGATAA	ACCTCCTTAAC	GTGCCAAAAA	AAGAAG <mark>T</mark> CAC	AAGGAATGCG	ATCGAACAT	8239
		8250	8260	8270	8280	8290	8300	
ATCC 334	8244	TCTTTGTGACT	TCTTTTAGAT	AACTCGGTAG	TETTTAAGAA	AGAGCOTCAG	[]	8303
12006	8240	TC <mark>C</mark> TTGTG <mark>G</mark> CT	TCTTTTAGAT	AACTCGGTAG	TTTTTA		001010000	8276
		8310	8320	8330	8340	8350	8360	
ATCC 334	8304	TTAGAAACCGG	 <b>AGT</b> GTAAGTG(	)   GGCCTGAACG	TGATGACCGG	 GCTTTGGTCA	TTGCGTTCA	8363
12006	8276						-	8276
		8370	8380	8390	8400	8410	8420	
ATCC 334	8364	GGGTCCTTACA	CGCAGGGTTC	I GGTCTGGCG	AACGCCTTTA	GGTTGGG	GTCATCTCA	8423
12006	8276				TTTA	ggttggg <mark>g</mark> tc	GTG <mark>GT</mark> TGA	8299
		8430	8440	8450	8460	8470	8480	
ATCC 334	8424	TCCCAAGACGT	GATTAAGGGT	I	TACCGGTGGA	TATAAACGTA	GCTGCTTGC	8483
12006	8300	TCCCAAGACGT	GATTAAGGGTA	AGGTAAAAAT	TACCGGTGGA	TATAAACGTA	GCTGCTTGC	8359
		8490	8500	8510	8520	8530	8540	
ATCC 334	8484	ACCAGAGATGG	TTTGGGTGTT	CGGGCCCGGTT	GGTGAGCTGA	AGCCCATGCC	GCCTTGGCT	8543
12006	8360	ACCAGAGATGG	TTTGGGTGTT	CGGGCCGGTT	GGTGAGCTGA	AGCCCATGCC	GCCTTGGCT	8419
		8550	8560	8570	8580	8590	8600	
ATCC 334	8544	GATCGTGACAC	TGTCACC <mark>G</mark> GA	GACAGATTGG	AC <mark>A</mark> TAAGCAA	CGTGACC <mark>A</mark> TA	TGAACCATC	8603
12006	8420	GATCGTGACAC	TGTCACCAGAGA	GACAGATTGG	AC <mark>G</mark> TAAGCAA	CGTGACC <mark>G</mark> TA	TGAACCATC	8479
		8610	8620	8630	8640	8650	8660	
ATCC 334	8604	GGCTGTCCATT	GACCGCCAAC	GATTGACCA	GCGGCGAAGA	CGATGATTGA	GCCGGCTGC	8663
12006	8480	GGCTGTCCATT	GACCGCCAAC	GATTGACCA	GCGGCGAAGA	CGATGATTGA	ACCGGCTGC	8039
		8670 •   • • • •   • • • •	8680	8690 	8700	8710	8720 []	
ATCC 334	8664 8540	TGGCGTGTGGT	TAACAGTGAA	GCCAGCTGCT	GC <mark>G</mark> GC <mark>T</mark> GAAC	CCCCCATTG	TGCGCCGTT	8723
12000	0040	IGGCGIGIGGI	TAACAGIGAA	CCAGCAGCI	GCHGCGGAAC	CALCULATIG	GCACCATI	0099
		8730	8740 11	8750	8760	8770	8780	
ATCC 334 12006	8724 8600	GCCCCAGCCAT	TACCAGCCCAT	IG <mark>GT</mark> GCAACT	GA <mark>T</mark> TTGACGT	ACCAGGTGCA	CTGACCCCA	8783 8659
		8700	9900	9910	8820	8830	8840	
_		<u>.    </u>	1				1	
ATCC 334 12006	8784 8660	AGGATA <mark>C</mark> GTGT AGGATA <mark>C</mark> GTGT	TACCTGAACC# TACCTGAACC#	ACTGAAGTCA ACTGAAGTCA	GCGTG <mark>G</mark> CTGC GCGTG <mark>A</mark> CTGC	CACTGCCTGA. CGCTACCTGA.	AGC AG <mark>TCGTGTT</mark>	8837 8719
		8850	8860	8870	8880	8890	8900	
	0007				· · · · · · · · · · · · · · · · · · ·			0050
12006	8837 8720	GTTGTTGATAA	TGGAGGAACTO	GTTGTTTGAA	GAACTTGATG	TGTTGTTGGC	GATGGTGCT GATGGTGCT	8779
		8910	8920	8930	8940	8950	8960	
ATCC 224	9950							8918
12006	8780	GGTTGAAGTTG	CGCTGCTTGAA	ACTGCTGGTA	GTAGAAGAAC	TTGCGGTTGA	AGCGGCAAC	8839
		8970	8980	8990	9000	9010	9020	
ATCC 334	8010	·   · · · ·   · · · ·						8978
12006	8840	GGTCTTCAAAG	CGGCTTGG <u>GT</u>	IGCAGCAGCT IGCAGCAG <u>CT</u>	GCTTCAGACT	GTTCTTGGGC.	AAATTGGCT	8899

		9030	9040	9050	9060	9070	9080	
ATCC 334	8979	TTGCAAAGCAAG	CAACTCATC	TTATGATCA	CAATTTG <mark>T</mark> TI	ATTCAAAGCA	ATCTTGTTT	9038
12006	8900	TTGCAAAGCAAC	CAACTCATC	TTATGATCAG	SCAATTTG <mark>C</mark> TI	ATTCAAAGCA	ATCTTGTTT	8959
		9090	9100	9110	9120	9130	9140	
ATCC 334	9039	ATCGTTAGCAT	TTTTTGTAG	GGAAACCAACI	GTGATTTCTC	GGTTTCCAA	IGAAGCCTT	9098
12006	8960	ATCGTTAGCATO	CTTTTTGTAG	GGAAACCAAC	GTGATTTCTO	GGTTTCCAA	IGAAGCCTT	9019
		9150 •   • • • •   • • • •	9160	9170	9180	9190 • • • •   • • • •	9200	
ATCC 334	9099	AGTTGAAACCA	GCGTCTGACG	GCATTTTCCT	GTTCCGTCTI	CAAGCCAGC	TAATTTATC	9158
12006	9020	AGIIGAAACCA	JUGICIGAUG	GCAT IMICCI	GINCEGICIN	CAAGCCAGC	TAATTTATC	9079
		9210 •   • • • •   • • • •	9220	9230   • • • •   • • • •	9240 • • • • • • • • • •	9250	9260	
ATCC 334 12006	9159 9080	CTTGGCAACTG CTTGGCAACTG	CACGGCATCA CACGGCATCA	AAGCGCATCTT AAGCGCATCTT	TTGCTGGCTTC TTGCTGGCTTC	GCTAAGTTT( GCTAAGTTT(	GCCAACCGT GCCAACCGT	9218 9139
		9270	9280	9290	9300	9310	9320	
-	0.21.0							0270
12006	9219 9140	CATGGTGCGGGG	CAATTAAGIC	AGACAAGTTC	IGGGAATTCAC	GAACAAAGTCO GAACAAAGTCO	GATATAGAC GATATAGAC	9278 9199
		9330	9340	9350	9360	9370	9380	
ATCC 334	9279		L	AGCCTTTTTC	IGCAAGGAGAT	CAACTGATC	TTTCAAGTT	9338
12006	9200	ATTGCCGCTGA	CAGAGTTGCC	AGCCTTTTTT	IGCAAGGAGAT	CAACTGATC	ITTCAAGTT	9259
		9390	9400	9410	9420	9430	9440	
ATCC 334	9339	GTTCTTGCGTG	CTGTGACATT	CTTTTGAGCG	GCAACAATTTC	CACCGCTGAG	IGATGAAAT	9398
12006	9260	GTTCTTGCGTG	CTGTGACATT	CTTTTGAGCG	GCAACAATTTTC	CACCGCTGAG	rgatgaaat	9319
		9450 	9460 	9470	9480 	9490	9500	
ATCC 334	9399 9320	CTTGGCATCAG CTTGGCATCAG	TTGCGCTGAT	CTTAGCCGTT(	GCATCGCTGAT GCATCGCTGAT	TTTGACCATT(	CTTGGTATC CTTGGTATC	9458 9379
		9510	9520	9530	9540	9550	95.60	
	0450							0.510
ATCC 334 12006	.9380	GATCTGCTTGT GATCTGCTTGT	ICAAATIGATI ICAAATTGATI	AACTTCTGTG: AACTTCTGTG:	TTCGCATCTTC	CAATTTGCTT	CAACAGTGC	9439
		9570	9580	9590	9600	9610	9620	
ATCC 334	9519	ATTGGTTTCGC	II ITTTGTTGGA	AGCGATACTG	I		IGGTGTTAC	9578
12006	9440	ATTGGTTTCGC	TTTTGTTGGA	AGCGATACTG	ICGCTTGTAT	CAGCGAAAAC	IGGTGTTAC	9499
		9630	9640	9650	9660	9670	9680	
ATCC 334	9579	GGCGCTAGCAG	AACCTGCTAA	GGTAACTGCA	GCAACCAACG	GATCATGAC	ITTATTGAA	9638
12006	9500	GGCGCTAGCAG	AACCTGCTAA	GGTAACTGCA	GCAACCAACG'	'GATCATGAC'	I'I'I'A'I'I'GAA	9009
		9690	9700	9710 	9720 	9730 	9740 	
ATCC 334 12006	9639 9560	TTTCATTTTGA TTTCATTTTGA	ACCTCACCCT	TAATTTGATT TAATTTGATT	ACTAAAGGC	ACTTTACTG	IGTCAATGT IGTCAATGT	9698 9619
	2000	0750	0760	0770	0790	9790	9800	
	0.000	.						0757
12006	9699 9620	TACGTCAGCGT TACGTCAGCGT	TACAGGACTG	TTAGGAGTTG TTAGGAGTTG	GTACTCAATC	TTCATTTGCG	CTTCAAAAT CTTCAAAAAT	9679
		9810	9820	9830	9840	9850	9860	
ATCC 334	9758	TCCCATCATCA				GTATGAAAA		9817
12006	9680	TGCGATCATGA	ACGTTTTTAA	ICCCAATTTT	IGATGAATTT	GTATGAAAA	AAGCAGCGC	9739

		9870	9880	9890	9900	9910	9920	
				••• ••• •		••••		
ATCC 334	9818 9740	CGTGACGACGCTG	CCTGACAAAA	TTGTAGGTTA TTGTAGGTTA	ATTCGTTTGAT	TTCAGGGCT	TCGACCAT	9877 9799
12006	9/10	CGIGACGACGCIG	CCIGACAAAA	TIGINGTI	ALICOLLIGAT	IICAGGGCI	ICGACCAT	2122
		9930	9940	9950	9960	9970	9980	
	0070				$\ldots   \ldots   .$			0037
ATCC 334	9800	GICCACCCGIIGG	AGICGACGAI	GCGTGAGCC	AGGIGACGACG	CCGGTGAAG	GCCAACAT	9859
12000								
		9990	10000	10010	10020	10030	10040	
ATCC 334	9938	GAGTAACGTGGCC	GTGAGGTAGC	CGACAATGC'	IGATCGTGAGI	GGAAAAACC	ACGGCCTC	9997
12006	9860	GAGTAACGTGGCC	GTGAGGTAGC	CGACAATGC	IGATCGTGAGI	GGAAAAACC	ACGGCCTC	9919
		10050	10060	10070	10090	10000	10100	
						•••		
ATCC 334	9998	AGTTTCGGCTTGA	TAC AAAATC	TAGGCTGTC	AGCAAATTGCC	GAGCAGATA	GCCGAACA	10056
12006	9920	AGTTTCGGCTTGA	ATAC <mark>C</mark> AAAATC	TAGGCTGTC	AGCAAATTGCC	GAGCAGATA	GCCGAACA	9979
		10110	10120	10130	10140	10150	10160	
				1	• • • •   • • • •   •	<u></u>		
ATCC 334	10057	CGATGCCGATCAC		ATGCTCTCA	CGACTGATGTA		CTCGCGAT	10116
12006	9900	CGATGCCGATCA	TGTCAACACA	ATGUTUTUA		ACAIGGIGAC	CICGCGAI	10039
		10170	10180	10190	10200	10210	10220	
NT00 224	10117							10176
ATCC 334 12006	10010	CGAAAAAGCCAAC	GCACCTIGATI	GIGGATAGT GTGGATAGT	TCCCGAATGC	GTTCCGAAAC	GTTGATGT	10099
		10230	10240	10250	10260	10270	10280	
ATCC 334	10177	TATTCAGATTGT	ACAAGACCAC	AATGACAAA	ACCGCACTGA	CAGAATAAA	ААТСАААА	10236
12006	10100	TATTCAGATTGT	ACAAGACCAC	AATGACAAA	ACCGCACTGAG	GCAGAATAAA	AATCAAAA	10159
		10290	10300	10310	10320	10330		
		.     ! .			•••••		· · · ·	
ATCC 334	10237	CAATAGGATCTAA	ACATACCGCTC	CATATTGCTT.	AACGTTTTCTT	TGCTGTCGT	сс <mark>тс</mark> 1029	92
12006	10160	CAATAGGATCTA	ACATACCGCT(	CATATTGCTT.	AACGTTTT <u>CT</u> I	TGCTGTCGT	CG 1021	13

TTAATCAAATTAAGGGTGAGGTCAAAAATGAAATTCAATAAAGTCATGATCACGTTGGTT 1 GCTGCAGTTACCTTAGCAGGTTCTGCTAGCGCCGTAACACCAGTTTTCGCTGATACAAGC 61 GACAGTATCGCTTCCAACAAAAGCGAAACCAATGCACTGTTGAAGCAAATTGAAGATGCG 121 AACACAGAAGTTATCAATTTGAACAAGCAGATCGATACCAAGAATGGTCAAATCAGCGAT 181 GCAACGGCTAAGATCAGCGCAACTGATGCCAAGATTTCATCACTCAGCGGTGAAATTGTT 241 GCCGCTCAAAAGAATGTCACAGCACGCAAGAACAACTTGAAAGATCAGTTGATCTCCTTG 301 CAAAAAAAGGCTGGCAACTCTGTCAGCGGCAATGTCTATATCGACTTTGTTCTGAATTCC 361 CAGAACTTGTCTGACTTAATTGCCCGCACCATGACGGTTGGCAAACTTAGCCAAGCCAGC 421 481 AAAGATGCGCTTGATGCCGTGACAGTTGCCAAGGATAAATTAGCTGGCTTGAAGACGGAA 541 CAGGATAATGCCCGTCAGACGCTGGTTTCAACTAAGGCTTCATTGGAAACCCAGAAATCA CAGTTGGTTTCCCTACAAAAAGATGCTAACGATAAACAAGATGCTTTGAATAAGCAAATT 601 661 GCTGATCATAAAGATGAGTTGGTTGCTTTGCAAAGCCAATTTGCCCAAGAACAGTCTGAA 721 GCAGCTGCTGCAACCCAAGCCGCTTTGAAGACCGTTGCCGCTTCAACCGCAAGTTCTTCT 781 ACTACCAGCAGTTCAAGCAGCGCAACTTCAACCAGCACCATCGCCAACAACACATCAAGT 841 TCTTCAAACAACAGTTCCTCCATTATCAACAACAACAACACGACTTCAGGTAGCGGCAGTCAC 901 GCTGATTACAGCAGTTCAGGCAACACGTATCCTTGGGGTCAGTGCACCTGGTACGTCAAG 961 TCAGTTGCTTCATGGGCTGGTAATGGCTGGGGCAATGGTGCCCAATGGGGTAGTTCCGCT 1021 GCAGCTGCTGGTTTCACAGTCAACCACACGCCAGCAGCCGGTTCAATCATCGTCTTCGCC GCTGGTCAATCTGTTGGCGGTCAATGGACAGCCGATGGTTCATACGGTCACGTTGCTTAC 1081 GTCCAATCTGTCTCTGGTGACAGTGTCACGATCAGCCAAGGCGGCATGGGCTTCAGCTCA 1141 1201 CCAACCGGCCCGAACACCCAAACCATCTCTGGTGCAAGCAGCTACGTTTATATCCACCGG 1202 TAA

LIKLRVRSKMKFNKVMITLVAAVTLAGSASAVTPVFADTSDSIASNKSETNALLKQIEDA
 NTEVINLNKQIDTKNGQISDATAKISATDAKISSLSGEIVAAQKNVTARKNNLKDQLISL
 QKKAGNSVSGNVYIDFVLNSQNLSDLIARTMTVGKLSQASKDALDAVTVAKDKLAGLKTE
 QDNARQTLVSTKASLETQKSQLVSLQKDANDKQDALNKQIADHKDELVALQSQFAQEQSE
 AAAATQAALKTVAASTASSSTTSSSSSATSTSTIANNTSSSSNNSSSIINNNTTSGSGSH
 ADYSSSGNTYPWGQCTWYVKSVASWAGNGWGNGAQWGSSAAAAGFTVNHTPAAGSIIVFA
 AGQSVGGQWTADGSYGHVAYVQSVSGDSVTISQGGMGFSSPTGPNTQTISGASSYVYIHR

421 \*

# Figure 3.33 Nucleotide sequence (above) and amino acid sequence (below) from the ORF determined from the consensus nucleotide sequence of *Lb. paracasei* strain VUP 12006 coding for AphA.

Flip ORF (ANGIS) was used to find and translate an open reading frame in the consensus nucleotide sequence of *Lb. paracasei* strain VUP 12006. AphA is located between positions 9643 and 8437 relative to scaffold 3 of the *Lb. casei* ATCC 334 draft genome. A ribosome binding site, marked in red, was found 18 bases from the start of the ORF. The nucleotide sequence was translated into an amino acid sequence (420 aa's). The predicted secretory signal peptide is marked in blue (bold Methionine marks the start of gene) and the N-terminal amino acid sequence previously determined for this protein is marked in green. The start and stop codons are highlighted in yellow.

A ribosome binding site was found 18 bases from the start of the ORF. The first methionine residue following the ribosome binding site was determined to be the start of the gene, which will now be referred to as AphA. The amino acid sequence was translated from the nucleotide sequence of the ORF and found to be 420 amino acids in length with a mass of 43.49 kDa and pl of 9.61. This information is shown in Figure 3.33. When submitted to ANGIS for SPScan it was found that the gene had the same predicted secretory signal peptide as *Lb. casei* ATCC 334. This data was shown in Figure 3.28. The predicted cleavage site was confirmed by the N-terminal amino acid sequence obtained for VUP 12006. The mature form of the protein is about 39.34 kDa: this is slightly smaller than the size observed by 1-D SDS-PAGE. AphA was also examined for the presence of the C-terminal cell wall anchor motif LPTXTG but no such sequence could be found. The count and percentage of amino acid residues of the ORF is shown in Table 3.16. It can be seen from this data that the gene is relatively rich in serine (15.4%) and alanine (13.3%).

residue	A	C	D	E	F	G	Н	I	К	L	М	N	Р	Q	R	S	Т	V	W	Y
count	56	1	21	9	7	29	5	22	28	26	4	29	5	27	6	65	37	29	6	8
%	13.3	0.23	4.98	2.13	1.66	6.88	1.18	5.22	6.65	6.17	0.95	6.88	1.18	6.41	1.42	15.43	8.78	6.88	1.42	1.9

Table 3.16 Count and percentage of amino acids in ORF sequence.

A ClustalW alignment of the nucleotide and protein sequences comparing strains VUP 12006 and *Lb. casei* ATCC 334 is shown in Figure 3.34. A sequence difference, due to the presence of an additional 45 nucleotides (15 aa's) can be seen between positions 835-880 (positions 8884 and 8839 relative to Scaffold 3 of the *Lb. casei* ATCC 334 draft genome). This region contains two short serine repeats. There are also 29 other base variations between the two genes, of which four result in a change of amino acids and of these four, two changes result in coding for serine. To determine nucleotide and function similarities AphA was submitted to Genbank, Swissprot and spTrEMBL through ANGIS for BLASTn (nucleotide) and BLASTX (function) searches.

# Figure 3.34 ClustalW alignment of the nucleotide and protein sequences comparing the AphA gene of strain VUP 12006 and sequence from scaffold 3 of *Lb. casei* ATCC 334.

Alignment of nucleotide sequence was performed using ClustalW. Shading denotes nucleotides that are identical in that position. A single letter amino acid translation is shown beneath the nucleotide sequence. A sequence difference, due to the presence of an additional 45 nucleotides (15 aa's), can be seen between positions 835-880 (positions 8884 and 8839 relative to Scaffold 3 of the *Lb. casei* ATCC 334 draft genome). This region contains two short serine repeats. An additional area of serine repeats can be seen between positions 773 and 800. The AphA gene is located between positions 9643 and 8437 relative to scaffold 3 of the *Lb. casei* ATCC 334.

		,	10	1	20	1	30	I	40	1	50	6	Û.
ATCC 334	1	ТТААТСА	AATTAA	AGGGTG	AGGTC	AAAA	ATGAAA	TTCA	ATAAA	GTCA	TGATCA	GTTGGTT	60
12006	1	TTAATCA	AATTAA	AGGGTG	AGGTO	AAAA	M K ATGAAA	TTCA	ATAAA	GTCA	TGATCA	CGTTGGTT	60
							MK	F	1; K	Зſ	14 I 7	LV	
		· · · · · · ·	70		80 .		90 • •   • • •	.	100 •••		110 • • • • • • • •	12 .	20
ATCC 334	61	GCTGCAC	V T	ITAGCA	GGTTC	CTGCT. A	AGCGCC	CGTAA V		.GTTT ∑	TCGCTGA	ATACAAGO ) T S	120
12006	61	GCTGCA	TTACC	ITAGCA L A	GGTTC G S	CTGCT.	AGCGCC S A	CGTAA V	CACCA	\GTTT ∵	TCGCTGI	ATACAAGO	120
			130		140		150		160		170	18	80
ATCC 334	121	GACAGT	ATCGCT	ICCAAC	AAAAG	CGAA	ACCAAT	GCAC	TGTTO	AAGC	AAATTG	AGATGCO	180
12006	121	D S GACAGT/	I A ATCGCT	S N	K	E	T N	A	L L TGTTG	K	<u>Q</u> II AAATTGA		180
12000		DS	I z	3 N	K.	E	ΤN	A	L L	K	QIH	DA	
			190		200		210	.	220		230	24	10
ATCC 334	181		GAAGTT	ATCAAT	TTGA		CAGATO	CGATA	CCAAG	SAATG	GTCAAA		240
12006	181	AACACA	GAAGTT.	ATCAAT	TTGA	ACAAG	CAGAT	CGATA		SAATG	GTCAAA	CAGCGAT	240
		11	250	1 0	260	I IX	270	D	290	14	200	30	0
	0.4.1	.	250					.					
ATCC 334	241	GCAACG A T	A K		A 1	CTGAT F D	A K	I	S S	L	S G I		300
12006	241	GCAACG A T	GCTAAG. A K	I S	A T	CTGAT F D	GCCAAC A K	I I	S S	L L	S G I	AATTGTT I ⊽	300
			310		320		330		340		350	36	50
ATCC 334	301	GCCGCT	CAAAAG	AATGTO		∣ CACGC			TGAA	GATC	AGTTGA	ICTCCTTC	360
12006	301	A A GCCGCT	Q K CAAAAG	N V AATGT(		A R CACGC	K N AAGAAG	N CAAC'I	L K TGAAA	D GATC	L AGTTGA	I S L ICTCCTTO	360
		A A	Q K	11 V	ΤA	AR	KN	Ν	ΓK	D	. L :	ISL	
			370 • • • 1 • •	)	380 • • • • •		390 •••••••		400 ••• •••		410	42 	20
ATCC 334	361	CAGAAA Q K	AAGGCT K A	GGCAAC G N	CTCTG	rcagc / S	GGCAA G N	rgtci V	Y I	GACT D	TTGTTC: F V I	<b>IGAATTCO</b> L N S	420
12006	361	CAAAAA O K	AAGGCT K A	GGCAAG G N	CTCTG	<b>ICAGC</b> V S	GGCAA' N	IGTCI V	Y I	CGACT D	TTGTTC: F V 1	IGAATTCO L N S	420
		-	430		440		450		460		470	48	30
ATCC 334	421	CAGAAC	· · ·   · ·		ATTG		ACCAT	GACGO	 TTGGC		TTAGCC	AAGCCAG	480
12006	421		L			A R		T	V G	K	L S (	A S	480
0	121	Q N	L S	D L	I /	A R	ТМ	T	VG	К	LS	2 A S	
		1	490		500	I	510	Ι	520	] .	530	54 	10
ATCC 334	481	AAAGAT	GCGCTT	GATGC	GTGA	CAGTT	GCCAA	GATA	AATT	GCTG	GCTTGA		540
12006	481	AAAGAT	A L GCGCTT	GATGCO	CGTGA	CAGTT	GCCAA	GAT		AGCTG	GCTTGA		540
		ΚD	A L	υA	V .	I V	A K	D	K L	A	, Ц I		20
	_		550 • • • I • •		560 •   • • •		570 •••	.	580 		590 ••• ••	6( 	
ATCC 334	541	CAGGAA. Q E	N A	CGTCAC R Q	T I	IGGTT L V	TCAACT S T	raago K	SCTTCA S	L L	AAACCCA E T (	AGAAATCA D K	600
12006	541	CAGGAT.	AATGCC		GACGC	IGGTT	TCAAC:	raago K	GCTTCA	ATTGG		AGAAATCI D K	<b>A</b> 600

			610		620		630		640		650	66	0
ATCC 334	601	CAGTTO	GTTTCCC	IACAP	AAAAGA	.∣ TGCT.	AACGAI		CAAGAT	GCTT	 ГGААТАА	.     ACAAATT	660
12006	601	L CAGTTO	7 S I GTTTCCC	L Q FACAA	K D AAAAGA	A TGCT.	N D AACGAT	K IAAA	Q D CAAGAT	A İ GCTT	L II K Igaataa	Q I GCAAATT	660
		L	7 S 1	LQ	K D	A	N D	K	QĎ	A I	LIK	QI	
		.	670 • • •   • • •	.	680 ••••••		690 • •   • • •		700 • • •   • •		710 • • •   • • •	72 •   • • • •	0
ATCC 334	661	GCTGAI A D	TCATAAAGA H K	ATGAC D E	GTTGGT L V	TGCT	TTGCA	AGC				GTCTGAA	720
12006	661	GCTGAT	H K	ATGAC	GTTGGT L V	TGCT	TTGCA	AAGC	CAATTI	GCCCZ	AAGAACA	GTCTGAA	720
			730		740		750	Ŭ	760	11	770	78	n
<b>ATTCC 334</b>	721	GCAGCI				.  TTTC		.   .					700
12006	721	A A	A A		A A	L	K T	V	A A				700
12000	, 2 4	A A	AA	r Q	A A	L	КТ	V	A A	ICAA	I A S	S S	780
		1	790	I	800	,	810	ſ	820	1	830	84	0
ATCC 334	781	ACTAC	CAGCAGTT	CAAG	CAGCGC	AACT		CAGC	ACCATC	GCCA	ACAACAC	A	834
12006	781		CAGCAGTT	CAAG	CAGCGC	AACT		CAGC		GCCA	ACAACAC	ATCAAGT	840
		1 1	5 5 3	5	5 A	Т	S T	4*	T 1	A		5	
	004	.			860	.	870 • • I • • •	.   .	••••••••••••••••••••••••••••••••••••••		890	900	)
ATCC 334	834			_					G CI	TCAG	GCAGTGG	CAGCCAC	855
12006	841	TCTTC		GTTCC S S	S I	TATC. I	AACAAC II II		ACGACI T 1	'TCAG(	GTAGCGGG	CAGTCAC S H	900
			910		920		930		940		950	960	)
ATCC 334	856		CTTCAGTG	.   GTTCA	AGGTAA	.   CACC'	IATCCI	.   . TGG	GGTCAG	··∣·· TGCA(	CCTGGTA	CGTCAAA	915
12006	901	A D GCTGAC	F S ( CTTCAGTG	G S GTTCA	G N AGGTAA	T CACG'	Y P TATCCI	W TGG	G Q GGTCAG	C 1 TGCAC	<u>r w</u> y CCTGGTA(	V K CGTCAAG	960
		A D	FS (	G S	G N	Т	Y P	M	i Q	C	C W Y	VK	
		.	970 • • •   • • • •	.	980 •   • • •	.	990 • • • • • • •	.   .	1000 • • • • • • • •		1010 •••   ••• •	102 	0
ATCC 334	916	TCAGTE S V	GCACCATO	GGGC1 N A	IGGTAA' G N	TGGC' G	rggggc W G	CAAC N	GGCGCA G A	CAATO Q V	GGGGCGG7 V G G	TTCAGCC S A	975
12006	961	TCAGTI S V	GCTTCAT	GGGCI N A	IGGTAA' G N	TGGC' G	rggggc W G	CAAT N	GGTGCC G A	CAATO Q V	GGGGTGG:	ITCCGCT	1020
			1030		1040		1050		1060		1070	108	0
ATCC 334	976	GCAGCA		ICACT	.   GTTAA	CCAC		.   . GCA	 GCCGGC	TCAA1		TTCGCC	1035
12006	1021	A A GCAGCI	A G I	T T ICACI	V N	H CCAC/	T P ACGCCA	A	A G GCCGGT	S 1 TCAA1	I I V CATCGTO	F A	1080
		A A	A 🖉 I	F T	V	Н	ΤP	A	A <u>(</u>	· ]	I I V		
		.	1090		1100	.	1110	.   .	1120		1130	114	0
ATCC 334	1036	GCTGGT	CAATCCG	TTGGC	GGTCA	ATGG		GAT	GGTTCA	TATGO	TCACGTT		1095
12006	1081	GCTGGT	CAATCTG	TGGC	GGTCA	ATGG	ACAGCO	GAT	GGTTCA	TACGO	TCACGT	GCTTAC	1140
		17	1150	G	1160	VV	1 A	D	1100	1 10	1100	120	0
ATCC 224	1000							.   .					~ 1155
12004	1141	V O		G G	D S	V V	T I	AGC( S	Q G	GGCA'I G M		S S	1200
	1141	GTCCAA	TCTGTCTC	TGGT	'GACAG'	rgtc/	ACGATO	AGC	CAAGGC	<u>GGCAI</u>	:GGG <u>C'I''I'(</u>	AGCTCA	1∠∪0

		1210				1220				1230 1240				1250			1260					
			• + •	• • •	۱	1		.   .	•••		•••		.   .		۱	• • •		.   .	•••	1	•••	
		P	Т	G	Ρ	N	Т	Q	Т	I	S	G	А	S	S	Y	V	Ŷ	I	Н	R	
12006	1201	CCA	ACC	GGC	CCC	AAC	ACC	CAA	ACC	ATC	TCT	GGT	GCA	AGC.	AGC	TAC	GTT	TAT	ATC	CAC	CGG	1260
		P	Т	G	P	N	Т	Q	Т	I	S	G	A	S	S	Y	V	ľ	I	Н	R	
NECC 334	1216	אמיד	l 12	18																		
ALCC 334	1210	*	i 12																			
12006	1261	TAA	12	263																		
		*																				

The results can be seen in Appendix 5. The direct nucleotide search of completed databases was unsuccessful with only short insignificant matches of less than 30 bases. The protein search was more successful with several matching entries.

Proteins with similarity included: general stress protein GSP-781 (glucan-binding protein) from *Streptococcus mutans*, a secreted 45 kDa protein from *Streptococcus pneumoniae*, secreted antigen SagA from *Enterococcus faecium* and several other secreted proteins including an immunogenic secreted protein from *Streptococcus pyogenes*. It is interesting to note that the closest matching entries had the following lineage in common: Bacteria; Firmicutes; Lactobacillales of which the family Lactobacillales contains all members of the lactic acid bacteria.

A multiple alignment of the amino acid sequence of some of the functionally similar proteins is shown in Figures 3.35. Areas of close to 100 % homology can be seen in the multiple alignment particularly between positions 332 and 372. Some shared homology is also present in the Nterminal and C-terminal regions of the genes. Other similarities shared between these bacteria include the presence of secretory signal peptides and poly-serine domains. The signal sequence of each protein, its length and the size of the encoded proteins is shown in Table 3.17.

Name	Size of encoded	Sequence	Length
	protein (kDa)		(in aa)
Lb. paracasei	43.5	MKFNKVMITLVAAVTLAGSASAVTPVFADT	30
12006			
Lb. casei ATCC 334	40.8	MKFNKVMITLVAAVTLAGSASAVTPVFADT	30
Streptococcus	44.5	MKKRILSAVLVSGVTLSSATT	21
mutans GSP-781			
Strep. pyogenes	42.0	MKRILSAVLVSGVTLGAA	19
Strep. pneumonia	41.7	MKKKILASLLLSTVMVSQVAVLTTAHAET	29

Table 3.17 Signal sequence for proteins determined to be functionally similar to AphA.

Figure 3.35 Multiple alignment of the amino acid sequence of some of the functionally similar proteins to AphA gene in strain VUP 12006.

Black background denoted amino acids with 100 % homology. Similar residues are indicated by grey boxes. The hydrophobic terminus is indicated by (1). The poly-serine domain is indicated by (2). The peptide that was sequenced is indicated by (3).

Key:

Lb. paracasei VUP12006 (VUP 12006)

Lb. casei ATCC 334 (ATCC334)

Strep. mutans (GSP-781) Swissprot Entry Q9AG98

Strep. pyogenes (Strep pyo) Swissprot Entry AE009955

Strep. pneumoniae (sec 45 kDa) Swissprot Entry AE007509

		4	1				3		
		10		20 -   1	<b>3</b> 0	40	50	60	
VUP 12006	1	MKFNKVMIT	AVTLA	G <b>SASAV</b> TE	VFADTSDS IA	SNKSETNA	LKQIEDAN	TEVINL IK	60
ATCC 334	1	MKF KVMITIVA	AVTLA	GSASAVTE	VFADTSDSIA KADDEDAOTA	SNKSETNA SODSKINA		TEVINLIK	60 60
Strep DVO	1	MKKRILSAVLVS	SVTLG	ATTVG	-AEDLSTKIA	KODSIIS	ILTTEOKAAO	NCVSALUA	60 57
sec 45 kDa	1	MKKKIL-JLLS	IV <b>MV</b> SÇ	2VAVLTTA	HAETTDDK	aqd <b>nkis</b> n	ILTAQQQEAQ	KÇVDQIÇE	60
						_		-	
		70		80	90	100	110	120	)
VUP 12006	61	<b>IDTKNGQISDA</b>	T <mark>AKI</mark> SZ	ATDAKISS	LS ETVAAQE	NVTARKNI	<b>ikdolis</b> lq	KKAGNSVS	120
ATCC 334	61	QIDTKNGQI SDA	TAKISA	ATDAKISS	LSCEIVAAQE	N <b>V</b> TARKNI	ILKDQLISLQ	KKA GNSVS	120
GSP-781	61 58	OVSALQTQQAEL	QAENQI TADNITI	RLEAQSAT	LCQ <b>D</b> DTLSS	KIVARNES	SLKQQARSAQ	KSNA	116
sec 45 kDa	61	QVSAIQAEQSIL	OAENDI	RLOAESKI	LE(EITELS)	N <b>IVS</b> RNQS	LEKCARSAQ	TNGA	116
		-	-	-	-	-			
		130		140	150	160	170	180	)
VUP 12006	121	GNVYIDFVLNSØ	NLSDL.	IARTMT <b>V</b>	KLSQASKDAT	DAVTVAKI	) <b>kl</b> a j <b>lk</b> teq	DNARQ <b>TLV</b>	180
ATCC 334	121	GNVYIDF <b>VL</b> NSQ	NLSDL:	LARTMTV	KLSQA <b>SKDA</b> I	DAVTVAKI	KLA LKTEQ	ENARQTLV	180
GSP-781	11/	ATSY INALINSK	SVSDA:	INRVSAIF VNRLVATN	EVVSANEKMI BAVSANAKTI	QQQEQDKA EOOKADKI	AVEQKQQEN KSLEEKODDD	QAAINIVA OTATUTTA	176
sec 45 kDa	117	VTSYINTIVNSK	SITEA	ISRVAAMS	EIVSANNKMI	EQQKADK	KAISEKQVA -	DAI: TVI	176
				-		-			
		190		200	210	220	230	240	)
VUP 12006	181	<b>STKASLETQKSQ</b>	LVSLOI	KDANDKQI	ALNKQIA	DHKDELVA	LQSQFAQEQ	SEAAAATQ	237
ATCC 334	181	STKASLETQKSQ	LVSLQI	KDANDKQL	ALNKQIA	DHKDELVA	LQSQFAQEQ	SEAAAATQ	237
GSP-781	174	ANQETIAQNTNA Anmamafenont	LNTQQ/	AQLEAAQI	NLQAELTTAC	DOKATLVA	Q <b>K</b> AAAEEAA		236
soc 45 kDa	177								235
Bec al KDa	111	MNOOLTHODAOA	TING A	ARTIVAVEL	SLAAEKATAI	GEKASLLE	:Q <b>K</b> AAAEAEA	RAAAVAEA	230
BEC 45 KDA	1,1	MNQQATADDAQA					QKAAAEAEA		230
BEC 45 KDA	1,,						290		230
VUP 12006	238	250 		260 ••••••••••••••••••••••••••••••••••••	270 270 SSSSSATS	2 2 280 	290 .     STIANNTSSS		230
VUP 12006 ATCC 334	238 238	250    AALKTV		260 	270 I	2 2 280 280 280 280 280 280 280 280 280	290 		230 280 270
VUP 12006 ATCC 334 GSP-781 Strep pvo	238 238 237 234	250    AALKTV AALKTV AAEAKAAAEAKA AAKAKA		260 STASSSTI STASSSTI QAQAAAN	270 270 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS	2 2 280 280 280 280 280 280 280 280 280	290 	300 	230 280 270 296 280
VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa	238 238 237 234 237	250   AALKTV AALKTV AAEAKAAAEAKA AAKVKA AYKEKR		260 	270 270 25SSSSATS 2SSSSSSATS 2SSSSATS 2SSSSSATS 2SSSSSATS 2SSSSSATS 2SSSSSATS 2SSSSSATS 2SSSSATS 2SSSSSATS 2SSSSATS 2SSSSATS 2SSSSSATS 2SSSSATS 2SSSSATS 2SSSSATS 2SSSSATS 2SSSSATS 2SSSSATS 2SSSSSATS 2SSSSSATS 2SSSSSATS 2SSSSSATS 2SSSSATS 2SSSSATS 2SSSSATS 2SSSSATS 2SSSSSATS 2SSSSATS 2SSSSATS 2SSSSATS 2S	2 2 280 280 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	290  STIANNTSSS STIANNX-X- HTQAAQT©DS ATPAAQSSNA QVQAV	300 	230 280 270 296 280 272
VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa	238 238 237 234 237	250    AALKTV AALKTV AAEAKAAAEAKA AAKVKA AYKEKR		260 STASSSTI STASSSTI AQAQAAAN AQQAASVE SQQQSVLA	270 270 25SSSSATS 2SSSSSATS NNTQATDASI AAKSAITP SANTNLTA	2 2 280 280 280 280 280 280 280 280 280	290  STIANNTSSS STIANNX-X- NTQAAQTODS ATPAAQSSNA QVQAV	300 SNNSSSII TEQSAAQA IEPAALTA SESAAAPV	236 280 270 296 280 272
VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa	238 238 237 234 237	250     AALKTV AALKTV AAEAKAAAEAKA AXKEKR 310		260 STASSSTI STASSSTI STASSSTI AQAQAAAN AQQAASVE SQQQSVLA 320	270 270 25SSSSATS 2SSSSSATS 2SSSSSATS 2SSSSATS 2SSSSATS 2SATDASI 2AAKSAITP 2SANTNLTA 330	2 2 280 	290 	300 	230 280 270 296 280 272
VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa	238 238 237 234 237 281	250    AALKTV AALKTV AAEAKAAAEAKA AAKVKA AYKEKR 310 		260 	270 270 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 270 270 270 270 270 270 270 270	2 2 2 2 2 2 2 80 7 7 80 7 7 80 80 80 80 80 80 80 80 80 80	290  STIANNTSSS STIANNX-X- ITQAAQT@DS ATPAAQSSNA QVQAV 350 	300 SNNSSSII TEQSAAQA IEPAALTA SESAAAPV 360 	230 280 270 296 280 272 325
VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334	238 238 237 234 237 234 237 281 270	250 ALKTV ALKTV AALKTV		260 260 STASSST1 STASSST1 STASSST1 AQAQAAAN AQQAASVE SQQQSVLA 320 	270 270 255555ATS 25555575ATS 2555575757575757575757577575777757777777	2 280 280 	290 	300 SNNSSSII TEQSAAQA IEPAALTA SESAAAPV 360 CI WGNCA GN WGNCA	230 280 270 296 280 272 325 311
VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334 GSP-781 Strep pyo	238 238 237 234 237 281 270 297 281	250 AALKTV	AA AA LQEQA - QEQA A GSGSH/ GSGSH/ GSGSH/ SARPO SARPO	260 	270 270 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 25555ATS 25555ATS 25555ATS 25555ATS 25555ATS 270 270 270 270 270 270 270 270	2 2 2 2 2 2 2 2 8 2 2 8 2 2 8 2 2 8 0 2 8 0 2 8 0 2 8 0 1 1 1 1 1 1 1 1 1 1 1 1 1	290 . J J STIANNTSSS STIANNX-X- ITQAAQT©DS ATPAAQSSNA 	300 SNNSSSII SNNSSSII TEQSAAQA IEPAALTA SESAAAPV 360 	280 270 296 280 272 312 311 356 322
VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa	238 238 237 234 237 234 237 281 270 297 281 273	250 AALKTV		260 260 3TASSSTI STASSSTI STASSSTI AQAQAASVE 3QQQSVLA 320 	270 270 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 270 270 270 270 270 270 270 270	2 2 2 2 2 2 2 2 2 2 2 2 2 2	290 	300 SNNSSSII SNNSSSII TEQSAAQA IEPAALTA SESAAAPV 360 CN WGNCA GN WGNCA GN WGNCA GN WGNCA GN WGNCA GN WGNCA	280 270 296 280 272 325 311 356 322 311
VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa	238 238 237 234 237 281 270 297 281 273	250 AALKTV AALKTV AAEAKAAAEAKA AAKVKA AYKEKR AYKEKR NNNTTS 	AA AA LQEQA - QEQA AA CSGSHI GSGSHI GSGSHI EAQPSI SARPQ KVRPTY	260 260 STASSSTI STASSSTI AQAQAAAN AQQAASVE SQQQSVLA 320 100000000000000000000000000000000000	270 270 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 25555ATS 25555ATS 270 270 270 270 270 270 270 270	2 280 280 280 280 280 280 280 28	290 	300 SNNSSSII SNNSSSII TEQSAAQA IEPAALTA SESAAAPV 360 	280 270 296 280 272 325 311 356 322 311
VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa	238 238 237 234 237 281 270 297 281 273	250 ALKTV		260 260 3TASSSTI STASSSTI STASSSTI AQAQAAAN AQQAASVE 3QQQSVLA 320 	270 270 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 270 270 270 270 270 270 270 270	2 2 2 2 2 2 2 2 2 2 2 2 2 2	290 	300 SNNSSSII SNNSSSII TEQSAAQA IEPAALTA SESAAAPV 360 CN WGNCA GN WGNCA GN WGNCA GN WGNCA GN WGNCA GN WGNCA GN WGNCA GN WGNCA CN WGN	280 270 296 280 272 311 356 322 311
VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa	238 238 237 234 237 281 270 297 281 273 326	250 AALKTV	AA AA AA LQEQA - QEQA AA CSSGSHI GSGSHI GSGSHI SARPQT KVRPTY 	260 260 STASSSTI STASSSTI AQAQAAAN AQQAASVE 320 320 320 ADYSS ADYSS ADFSG ASSASTAF ISYD 380 	270 270 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 25555ATS 25555ATS 25555ATS 270 270 270 270 270 270 270 270	2 2 2 2 2 2 2 2 8 2 2 8 2 2 8 8 2 2 8 8 8 8 8 8 8 8 8 8 8 8 8	290 	300 SNNSSSII SNNSSSII TEQSAAQA IEPAALTA SESAAAPV 360 	280 270 296 280 272 325 311 356 322 311
VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334	238 238 237 234 237 281 270 297 281 273 326 312	250 AALKTV		260 	270 270 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 25555ATS 25555ATS 25555ATS 25555ATS 25555ATS 25555ATS 25555ATS 25555ATS 270 270 270 270 270 270 270 270	2 2 2 2 2 2 2 2 2 2 2 2 2 2	290 	300 SNNSSSII TEQSAAQA IEPAALTA SESAAAPV 360 	280 270 296 280 272 311 356 322 311 384 370
VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334 GSP-781 Strep pyo	238 238 237 234 237 234 237 281 270 297 281 273 326 312 357 323	250 AALKTV	AA AA LQEQA - QEQA -	260 260 3TASSSTI STASSSTI STASSSTI AQQAASVE SQQQSVLA 320 	270 270 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 25555ATS 25555ATS 270 270 270 270 270 270 270 270	2 280 280 280 755 280 755 280 755 280 755 280 755 755 755 755 755 755 755 75	290 	300 SNNSSSII TEQSAAQA IEPAALTA SESAAAPV 360 CN WGNCA GN WGNCA GN WGNCA GN WGNCA GN WGNCA GN WGNCA CN WGN	280 270 296 280 272 325 311 356 322 311 384 370 404
VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa	238 238 237 234 237 281 270 297 281 273 326 312 357 323 312	250 	AA AA LQEQA - QEQA - QEQA AA GSGSHI GSGSHI GSGSHI EAQPSI SARPQ KVRPTY A 	260 STASSSTI STASSSTI STASSSTI AQAQAASVE SQQQSVLA 320 AQAASVE SQQQSVLA AQAASVE AQAASVE AQAASVE AQAASVE AQAASVE AQAAANA AQAAASVE AQAAANA AQAAANA AQAAAANA AQAAAANA AQAAAANA AQAAAANA AQAAAANA AQAAAANA AQAAAANA AQAAAANA AQAAAANA AQAAAANA AQAAAANA AQAAAAANA AQAAAAANA AQAAAAANA AQAAAAANA AQAAAAANA AQAAAAANA AQAAAAANA AQAAAAANA AQAAAAAANA AQAAAAANA AQAAAAAANA AQAAAAAANA AQAAAAAAANA AQAAAAAAAAAA	270 270 270 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 255555ATS 270 270 270 270 270 270 270 270	2 2 2 2 2 2 2 2 2 2 2 2 2 2	290 .     STIANNTSSS STIANNX-X- ITQAAQTODS ATPAAQSSNA QVQAV 	300 SNNSSSII TEQSAAQA IEPAALTA SESAAAPV 360 CN WGNCA GN WGN	280 270 296 280 272 311 356 322 311 384 370 404 371 360
VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa	238 238 237 234 237 234 237 281 270 297 281 273 326 312 357 323 312	250 AALKTV	AA AA LQEQA - QEQA -	260 260 3TASSSTI STASSSTI STASSSTI AQQAASVE SQQQSVLA 320 	270 270 270 255555ATS	2 2 2 2 2 2 2 2 2 2 2 2 2 2	290 . J J 290 . J 37 IANNTSSS 37 IANNX-X- 17QAAQTODS 17PAAQSSNA  17QAQTODS 17PAQSSNA  100  100  100  100  100  100  100  100 	300 SNNSSSII TEQSAAQA IEPAALTA SESAAAPV 360 CN WGNCA GN WGNCA GN WGNCA CN WGN	280 270 296 280 272 325 311 356 322 311 384 370 404 371 360
VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa	238 238 237 234 237 281 270 297 281 273 326 312 357 323 312	250    AALKTV	AA AA LQEQA - QEQA AA CSSGSHJ GSGSHJ GSGSHJ EAQPSJ SARPQ KVRPTY VNHTPJ VNHTPJ VNHTPJ VGSTPS TGSTPC	260 260 STASSSTI STASSSTI STASSSTI AQAQAASVE SQQQSVLA 320 AQAASVE SQQQSVLA AQAASVE AQAASVE AQAAASVE AQAASVE AQAASVE AQAASVE AQAAASVE AQAASVE AQAAASVE AQAAASVE AQAAASVE AQAAASVE AQAAASVE AQAAASVE AQAAASVE AQAAASVE AQAAASVE AQAAASVE AQAAASVE AQAAASVE AQAAASVE AQAAASVE AQAAAASVE AQAAASVE AQAAAA AQAAAA AQAAAA AQAAAA AQAAAA AQAAAA AQAAAA AQAAAA AQAAAAA AQAAAAA AQAAAAA AQAAAAAA AQAAAAAA AQAAAAA AQAAAAAA AQAAAAAA AQAAAAAA AQAAAAAAA AQAAAAAAAA	270 270 270 255555ATS 25555ATS 25555ATS 25555ATS 25555ATS 25555ATS 25555ATS 25555ATS 25557 25555ATS 25557 25555ATS 25557 255777 25577 255777 255777 255777 255777 255777 255777 255777 255777 2557777 2557777 2557777 2557777 25577777 25577777777 2557777777777	2 2 2 2 2 2 2 2 2 2 2 2 2 2	290 .     STIANNTSSS STIANNX-X- ITQAAQTODS ATPAAQSSNA QVQAV 	300 SNNSSSII TEQSAAQA IEPAALTA SESAAAPV 360 CN WGNCA GN WGNCA SESAAAPV 100 100 100 100 100 100 100 10	280 270 296 280 272 311 356 322 311 384 370 404 371 360
VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa VUP 12006 ATCC 334 GSP-781 Strep pyo sec 45 kDa	238 238 237 234 237 281 270 297 281 273 326 312 357 323 312 385	250 AALKTV	AA AA LQEQA - QEQA -	260 260 3TASSSTI STASSSTI STASSSTI AQAQAAAN AQQAASVE SQQQSVLA 320 	270 270 270 2555555555555555555555555555555555555	2 2 2 2 2 2 2 2 2 2 2 2 2 2	290 . J J 290 . J 30 31 31 31 350 1 350 . J 350 . J 350 . J 350 . J 350 . J 350 . J 350 . J	300 SNNSSSII TEQSAAQA IEPAALTA SESAAAPV 360 CN WGNC CN WGNC	280 270 296 280 272 311 356 322 311 384 370 404 371 360
VUP 12006           ATCC 334           GSP-781           Strep pyo           sec 45 kDa           VUP 12006           ATCC 334           GSP-781           Strep pyo           sec 45 kDa           VUP 12006           ATCC 334           GSP-781           Strep pyo           sec 45 kDa           VUP 12006           ATCC 334           GSP-781           Strep pyo           sec 45 kDa           VUP 12006           ATCC 334           CSP 731	238 238 237 234 237 281 270 297 281 273 326 312 357 323 312 385 371	250 	AA AA LQEQA - QEQA - QEQA - QEQA AA GSGSHJ GSGSHJ GSGSHJ GSGSHJ SARPQ KVRPTY VNHTPJ VNHTPJ VNHTPJ VNHTPJ VNHTPJ TGSTPQ TGSTPC TQTIS- TQTIS-	260 	270 270 270 255555ATS 255557 25557 255557 25557	2 2 2 2 2 2 2 2 2 2 2 2 2 2	290 .     STIANNTSSS STIANNX-X- ITQAAQTODS ATPAAQSSNA   	300 SNNSSSII TEQSAAQA IEPAALTA SESAAAPV 360 SN WGNG GN WGNG GN WGNG GN WGNG GN WGNG GN WGNG GN WGNG GD WGNG	280 270 296 280 272 311 356 322 311 356 322 311 384 370 404 371 360
VUP 12006         ATCC 334         GSP-781         Strep pyo         sec 45 kDa         VUP 12006         ATCC 334         GSP-781         Strep pyo         sec 45 kDa         VUP 12006         ATCC 334         GSP-781         Strep pyo         sec 45 kDa         VUP 12006         ATCC 334         GSP-781         Strep pyo         sec 45 kDa         VUP 12006         ATCC 334         GSP-781         Strep pyo         sec 45 kDa	238 238 237 234 237 281 270 297 281 273 326 312 357 323 312 385 371 405 372	250 AALKTV	AA AA LQEQA - QEQA - QEQA - QEQA - QEQA - QEQA - QEQA AA GSGSH/ GSGSH/ GSGSH/ EAQPS/ SARPQT KVRPTY A AA 	260 260 3TASSSTI STASSSTI STASSSTI AQAQAAAA AQQAASVE 3QQQSVLA 320 	270 270 270 2555555555555555555555555555555555555	2 2 2 2 2 2 2 2 2 2 2 2 2 2	290 .     STIANNTSSS STIANNX-X- ITQAAQTODS ATPAAQSSNA    	300 SNNSSSII TEQSAAQA IEPAALTA SESAAAPV 360 CN WGNCA GN WGNCA GN WGNCA GN WGNCA GN WGNCA GD WGNCA CN WGNCA GD WGNCA CN WGNCA GD WGNCA ST J CN WGNCA CN WGNCA CN WGNCA ST J CN WGNCA CN WGNCA CN WGNCA ST J CN WGNCA CN WGNCA ST J CN WGNCA CN WGNCA CN WGNCA ST J ST J	280 270 296 280 272 311 356 322 311 384 370 404 371 360
VUP 12006         ATCC 334         GSP-781         Strep pyo         sec 45 kDa         VUP 12006         ATCC 334         GSP-781         Strep pyo         sec 45 kDa         VUP 12006         ATCC 334         GSP-781         Strep pyo         sec 45 kDa         VUP 12006         ATCC 334         GSP-781         Strep pyo         sec 45 kDa         VUP 12006         ATCC 334         GSP-781         Strep pyo         sec 45 kDa	238 238 237 234 237 281 270 297 281 273 326 312 357 323 312 385 371 405 372 361	250 	AA AA LQEQA - QEQA - QEQA - QEQA - QEQA AA GSGSH GSGSH GSGSH GSGSH SARPQ KVRPTY VNHTP VNHTP VNHTP VNHTP VNHTP TGSTP TGSTP TGSTP TGSTP TGSTP CUTIS- RGWFNE RGWFNE RGWFNE	260 	270 270 270 2555555555555555555555555555555555555	2 2 2 2 2 2 2 2 2 2 2 2 2 2	290 .     STIANNTSSS STIANNX-X- ITQAAQTODS ATPAAQSSNA   	300 SNNSSSII TEQSAAQA IEPAALTA SESAAAPV 360 SN WGNG GN WGNG GN WGNG GN WGNG GN WGNG GN WGNG GN WGNG GN WGNG GD WGN GD WGN	280 270 296 280 272 311 356 322 311 356 322 311 356 322 311 356 322 311

A Kyte and Doolittle plot of hydrophobicity is shown in Figure 3.36. A hydrophobic region, which can be seen at the start of AphA between positions 8 and 23, is consistent with the presence of a signal sequence. There is also another hydrophobic region around position 350, which may correspond to another membrane-spanning region.

#### 3.7 Discussion

In this study, the response of *Lb. paracasei* strain VUP 12006 to environmental stress, principally heat and acid stress, has been examined. The response to acid stress has been particularly intriguing, with novel changes in protein profiles and fatty acid methyl ester profiles being observed. The results presented and their significance are discussed below.

#### 3.7.1 Identification of test strains

This project initially planned to test the responses to an acid environment of strains of *Lb. acidophilus* principally because of the interest in this species as a probiotic supplement. Subsequently, twenty-five strains reported to be *Lb. acidophilus* were obtained from a variety of sources. However, only nineteen showed the basic colony and Gram stain morphology, which progressed these candidate strains through to the next stage of identification. Although this is a very small sample, it is interesting to note that 24% of strains from culture collections failed to match the expected colony and Gram stain morphology.

It is not known whether any of the strains had previously been tested using molecular biology techniques. However, it is known that the strains had been propagated from frozen stocks prior to receipt at the VUT laboratories and the possibility of strain contamination cannot be excluded. Such discrepancies have also been noted by Yeung (2002) who found a strain of *Streptococcus sanguis* in a commercial product labelled as containing *Lb. acidophilus*.



#### Figure 3.36 A Kyte and Doolittle plot of hydrophobicity.

A mean hydrophobicity profile was performed for AphA using the Kyte and Doolittle method using default parameters (BioEdit). Positive scores indicate a hydrophobic region. A hydrophobic region between positions 8 and 23 is consistent with the presence of a signal sequence. The poly serine area is located in a hydrophilic region between positions 249 and 279. The peak at position 350 may also correspond to possible membrane-spanning domain.

#### 3.7.1.1 Traditional carbohydrate tests

Carbohydrate utilisation test kits are commercially available for the identification of LAB, however the cost per kit made the screening of multiple isolates prohibitive in this project. Identification of a small number of control strains was performed using lactobacilli fermentation broth, but difficulties were noted in interpretation of positive results and reproducibility was poor so it was then decided to develop a cheap and rapid method that could be the first step in identification of a variety of LAB to the species level. Three separate trials focussed on the use of minimal media using agar and liquid bases with 2% selected carbohydrate sources. In the first two trials different agar bases were tested but growth was generally very restricted in such minimal media and even when spectrophotometric assays were used to quantify changes in growth, interpretation of the results was very difficult. These trials were abandoned in favour of a method using a liquid based minimal media and an ELISA plate reader to measure changes in optical density. The advantages of this method are:

- results are reproducible and can easily be performed in duplicate on the same microtitre plate;
- the method works well on any air-tolerant bacteria;
- the volumes of broth used are very small (200µL per well), which keeps costs low;
- the plate set-up can be changed to suit the test protocol, for example the plate may be set up to test fermentation of a single carbohydrate against 80 different tests or a single test against multiple carbohydrates, although control strains using this method take up a number of the test wells; and
- the results are obtained in one working day.

Traditional carbohydrate tests are no longer used as the full means of identification of LAB, but when compared with molecular biology based methods, they rnust be considered as the preferred starting point in a step-wise identification scheme because of the low cost. It has been stated by Du Plessis (1996) that differentiation and identification to species level based on biochemical and physiological tests is unreliable in *Lactobacillus* species. However, the experience of this study was that rapid carbohydrate tests in conjunction with Gram stain and growth characteristics, was the best option for first step identification when screening and comparing large numbers of isolates.

#### 3.7.1.2 FAME fingerprints

It has been confirmed in this work that both SDS-PAGE and FAME fingerprinting require highly standardised methodology. Fingerprinting using FAME profiles has been successfully used as a way of differentiating both yeast and bacterial strains (Augustyn *et al.*, 1990, Johnsson *et al.*, 1995, Rementzis *et al.*, 1996). However, because the fatty acid composition of bacterial and yeast cells is dependent upon growth temperature, pH, growth medium and growth phase, comparison must be made using cultures grown under a rigid standard set of conditions. The standard deviation achieved in these results, <7% over 2 runs, is comparable to other studies, <5% by Blignaut, (1996), but the time and cost involved in this procedure suggest that, at least in this species, molecular based methods would be preferred as a means of strain identification, before using FAME profiles. The presence of lactobacillic acid in the profiles did, however, confirm species identification.

#### 3.7.1.3 SDS-PAGE profiles

The same rigid culture conditions, as used in obtaining FAME profiles, are required when performing SDS-PAGE. Other studies using SDS-PAGE as a means of identification, have shown that the method is only reproducible when a standard set of whole cell proteins is used for normalisation between runs (Pot *et al.*, 1993) and these runs are required at least in triplicate.

The study by Dykes (1993) showed that triplicate readings of SDS-PAGE profiles performed on the same organism in independently run gels gave only 85-92% sirnilarity; other authors have given figures of 85 to 93%. Furthermore, in order to handle complex identifications, such as comparisons of multiple species of LAB, computer-assisted comparison must be done using a databank of known LAB accessed using a software package such as GELCOMPAR (Pot *et al.*, 1993, Pot *et al.*, 1994). However, once a fully-standardised technique is used there appears to be good correlation with the results obtained by 16S rRNA gene analysis (Pot *et al.*, 1993).

While this work may have compared only a small number of cultures, it was found that over duplicate runs, visual comparison was sufficient to differentiate all but two of the test strains and of those two strains, one was a mutant of the other. With the exception of strain VUP 12006, that differentiation was based on changes in minor protein bands. In some regards this may suggest that, in this small sample, the major bands were good species markers. This appears to be supported with the observation of the marked differences in the major banding pattern of the strain VUP 12006. The principal difference of strain VUP 12006 to the other test strains was the presence of four highly-expressed protein bands, which although present in the other strains, were not expressed at such high levels. As over-expression of proteins may result from years of selection for flavour or functional properties in cultures used in manufacturing or industrial applications as well as species differences, this technique may not be as useful in identification of bacterial strains used by the manufacturing industry, at least when comparing profiles to known databanks. Apart from that observation, the method appears to be a reasonable second step for strain differentiation when screening large numbers of isolates in a situation similar to the type of identification process used in this project. The cost of performing these tests is not prohibitive and the time spent in extracting proteins from cultures is neither difficult nor arduous.

#### 3.7.1.4 PCR (What's in a name?)

The identification of LAB has traditionally been performed using biochemical tests that are used to classify the organisms into one of three major fermentation groups: homofermentative lactobacilli, facultative heterofermentative lactobacilli and the obligate heterofermentative lactobacilli. These groups, however, do not reflect the phylogenetic relationship between species, either the three clusters proposed by Collins (1991) or more recently the five phylogenetic groups proposed by Schliefer and Ludwig (1995). This section will further discuss the difficulties of classification of unknown LAB and the use of molecular techniques to arrive at a valid identification.

Clearly, identification based on phylogenetic rather than phenotypic characteristics has now become a definitive means of identification, and the molecular biology based method used in this study was definitive in identifying the test strain. Nevertheless, the cost and time involved in obtaining those results make this method the third choice in the step-wise identification scheme. This method is probably best used in situations where tracking changes in the genetic make-up is important, for example strain substitutions in industrial applications. More recently, phylogenetic analysis based on partial HSP60 gene sequences has been developed (Jian *et al.*, 2001). The technique in which partial HSP60 genes are sequenced using degenerate PCR primers appears to have a higher resolving power than the 16S rRNA technique when determining differences between closely related strains. This technique may have a place in this type of study, where separation of closely related bacteria has proven difficult.

Molecular biology techniques were seen as the only way to confirm the identification of strain VUP 12006. The initial studies focussed on placing strain VUP 12006 in the correct genus. Dubernet *et al.* (2002) had recently published details of a genus-specific universal PCR primer pair, R16-1 and LbLMA1-rev. These authors had shown that PCR using the universal primers generated a single product of approximately 250 bp (exact size appeared to be related to species) and was specific for 21 *Lactobacillus* species including *Lb. acidophilus* and *Lb. casei.* 

They also demonstrated that using DNA from *E. coli, Listeria, Staphylococcus* and LAB not on the list of 21 species, PCR using the same primers failed to generate any fragments. When strain VUP 12006 was tested using the universal primers a fragment of  $\cong$ 250 bp was obtained, which was consistent with the strain being placed within the genus *Lactobacillus*. Interestingly, two control strains, *Lb. casei* GCRL-163 and *Lb. acidophilus* ASC 388 were run at the same time. *Lb. acidophilus* ASC 388 yielded a product of slightly larger than both strain VUP 12006 and *Lb. casei* GCRL-163. This variation in fragment size and slight differences in FAME data were the first indications that strain VUP 12006, was not an *acidophilus* strain.

In order to speciate strain VUP 12006 it was decided to perform a partial 16S rDNA sequence and compare the results to sequence data in Genbank. Although several potential PCR primer pairs have been published, it was decided to use PAF and 536R. PAF and 536R, which had been shown to identify a range of LAB, yielded products of 500 bp. These products were sequenced and the sequence data submitted for a BLASTn search.

It was found that the initial BLASTn search of sequence based on data from the forward primer identified strain VUP 12006 as *Lb. paracasei* while the data from the reverse primer most closely matched *Lb. casei* strain KH-1. Similar results were also obtained for *Lb. casei* ATCC 334 from the draft genome, in that the 16S sequence appeared to most closely match *Lb. paracasei*. The control strain of *Lb. casei* GCRL-163 sequenced at the same time yielded similarly unusual results while the *Lb. acidophilus* sequenced as expected and was most closely related to *Lb. acidophilus*.

This apparent contradiction in identification of the casei strains was made slightly clearer after examination of the alignment of strain VUP 12006 in the V1 region of the 16S rRNA gene with sequence data from type strains of both *Lb. casei* and *Lb. paracasei*. The sequence obtained from these variable regions had been shown by Ward and Timmins (1999) to differentiate

members of the *casei* complex. Examination of this region demonstrated that strain VUP 12006 was an exact match to *Lb. paracasei*. It should be noted that *Lb. casei* ATCC 334 and GCRL-163 showed similar sequence data in this region and it cannot easily be explained as to why they are not identified as *Lb. paracasei* as they differ in sequence to the type strain *Lb. casei* ATCC 393. Given these differences there is certainly further scope for study into the identification protocol for the *casei* group. Future work may well concentrate on strain classification based on other differences between the strains, particularly on genes other than the 16S rRNA gene and may well include the gene for HSP60 or similarly conserved genes.

#### 3.7.1.5 Summary of identification techniques

In summary, as expected, carbohydrate utilisation studies were unable to speciate strain VUP 12006 and only minor differences were seen between the strains tested. Although sorbitol fermentation was negative, strain VUP 12006 had the typical fermentation pattern of *Lb. acidophilus*. It should also be pointed out that the range of carbohydrate tests was probably not extensive enough to differentiate the species being tested. FAME and SDS-PAGE profiles were equally unhelpful as an aid to identification, although strain differences were seen. A low level of discrimination between the strains is apparent and the fingerprints produced by both methods were easily influenced by the growth conditions and the growth phase prior to harvest. Molecular based identification, despite its cost and time-consuming methods, clearly produced the best results but these were not unambiguous either. Table 3.18 summarises the results relating to the usefulness of each technique used in this work.

#### 3.7.2 Characterisation of *Lb. acidophilus* strain VUP 12006

#### 3.7.2.1 Growth kinetics

Although it is hard to compare growth kinetics because of variation in culture conditions and species used, only one report has been published that would allow comparison of growth kinetics found in this study.

# Table 3.18 Summary of the identification procedures used in this study.

Comparison was made on the basis of the apparent reliability of identification, cost, and time factors relating to the identification procedures used in this study. Cost per test decreases when performing more than 10 FAME profiles in a run, due to reduced set-up costs; approximately 56 samples were analysed to obtain FAME profile results. Costs for FAME profiles are marked Medium toHigh as each batch was done in runs of 10 or less: had the analysis been done in one or two batches, the cost would be Medium.

Test	Level of identification	Apparent reliability of result	Suitability for screening large numbers of isolates	aCost	Minimum time taken to obtain results (does not include culture time)
Carbohydrate tests	Species	Good	No Yes	Single vial testing-medium Microtitre plate testing-low	24 hours 1 working day
FAME profiles	Species /strain	Good	Yes	Medium to High	24 hours
SDS-PAGE profiles	Strain	Very good	Yes	Medium	8 hours
PCR	Strain	Very good	Yes	High	>1 week

a cost of testing based on commercial rates. Low-< \$10 per result. Inequim-\$10-\$100 per result. Fight ->\$100 per result

The growth kinetics for *Lb. paracasei* strain VUP 12006, determined during growth in fermenters, are similar to the values published in the literature for *Lb. acidophilus* (Kim *et al.*, 2001). In that study mid-log phase was reached in approximately 4 hours and stationary phase was established after around 14 hours. The work presented in section 3.4.2 demonstrated a slightly longer time until mid-log phase, around 6 hours, and stationary phase was reached after around 13 hours; but only an indirect comparison can be made when all parameters of the study were not included in the publication methods, particularly starter inocula and broth. A study by Gadgil (1997), working on growth modelling in batch cultures of *Lb. bulgaricus*, determined that the  $\mu_{max}$  at pH 5.6 was around 0.45 h(-1). Batch fermenter conditions were similar to those described in this work, including the use of MRS broth, however the flasks were agitated without aeration and no nitrogen sparging occurred. In this study, at pH 5.0, the  $\mu_{max}$  was 0.50 h(-1) and at pH 6.0 the  $\mu_{max}$  was 0.59 h(-1), which is slightly higher than that described for *Lb. bulgaricus*, but the difference may possibly be explained in strain variation and exposure of the flasks to room air.

#### 3.7.2.2 Optimum growth conditions

The optimum growth conditions determined in this study, compared to published values of other *Lactobacillus* species, are shown below. While the value for temperature falls in the range of expected optimum values, the value for pH is slightly higher than that observed in some other species of *Lactobacillus*. As stated previously, it is critically difficult to compare these values, as the growth conditions were not fully elucidated in the publication.

	Values determined in this study for strain VUP 12006	Published values for the genus Lactobacillus (Gomez et al., 1999)
Optimum temperature range	37 °C	35-40°C
Optimum pH	6.3 ± 0.2	5.5-6.0

#### 3.7.2.3 Protein profiles and Western blot reactions

Stress, resulting from exposure to acid and heat, is commonly experienced by bacteria used in the food industry, particularly during the manufacture of fermented dairy products. One of the goals of this study was to detect proteins induced during acid and heat stress to identify the target areas within the cell responding to the stress. The results shown in this thesis, sections 3.4.9 and 3.4.10, indicate that growth at increased temperature (due to a 7°C rise in temperature above the optimum) resulted in a four-fold up-regulation of analogues of HSP60 and HSP70.

The presence of the *gro*ESL operon, encoding the GroES and GroEL chaperone proteins, has previously been localised on the chromosome of both *Lb. johnsonii* and *Lb. helveticus* (Broadbent *et al.*, 1998, Carey Walker *et al.*, 1999). The finding of the analogues is consistent with the presence of a similar operon in the tested strain of *Lb. paracasei*. Studies of the protein profiles showed that growth at low pH and high temperature, compared to growth at optimum pH and temperature, resulted in dissimilar changes to the banding pattern, suggesting that a universal stress response does not occur. The changes caused by decreasing pH during growth were rapid and appeared within one hour of the pH being lowered to pH 4.0, although a marked decrease in the generation time was observed at the same time.

#### 3.7.2.4 FAMEs (Effect of acid and temperature shift on total cellular FAMEs )

The study of Decallone and others (1991) has clearly demonstrated that the age of a culture when sampled for comparison of FAME fingerprints is of fundamental importance. The profile clearly changes as a culture ages through exponential and growth phase. It was considered that this analysis of FAME profiles may also be a useful technique in understanding the mechanisms of adaptive tolerance to environmental change, so a study was made of the changes in FAME profiles taken at the same stage of growth but from cultures grown in different environments. The results were presented in section 3.4.6.

Bacterial cell membranes perform a vital function in protecting the cell from changes in the external environment such as increasing temperature or decreasing pH. Temperature dependent changes to fatty acid composition have been demonstrated in several bacteria including *Lb. plantarum, L. monocytogenes* and *C. botulinum* (Russell *et al.,* 1995, Evans *et al.,* 1998). In these organisms membrane fluidity at low temperatures was maintained by a decrease in chain length and an increase in the proportion of unsaturated fatty acid (UFA) chains as growth temperature decreased. In this study, it was determined that there was an increase in the presence of saturated C<sub>18:0</sub> (stearic acid) during heat stress from 6 to 14% and a decrease in C<sub>18:1</sub> (oleic acid) from 26 to 21%. The probable conversion, in part, of C<sub>18:1</sub> to C<sub>18:0</sub> by removal of one double bond would significantly increase the melting point and make the membranes less fluid at the higher temperature. Stearic acid has a high melting point of 69.6°C and oleic acid has a melting point of 13.4°C.

Following acid stress, marked changes were also noted in the pattern of fatty acids. Most changes were similar to those noted after the culture had undergone heat stress, particularly the decrease in peaks X1, C<sub>14.0</sub>, C<sub>18.0</sub> and C<sub>19.0</sub> and increase in peaks X3, and C<sub>18.1</sub>; however, C<sub>16.1</sub> increased from 2 to 10 % while no similar change occurred in a heat stressed culture. The overall similarity of changes suggests two possibilities. One possibility is of a general response to environmental stress rather than to a specific heat or acid stress response. The second is that heat and acid stress provoke similar responses to quite different stresses but not the responses that may be provoked by, for example, an oxidative stress. It would be expected that a different fatty acid profile would be seen after a decrease in temperature, so the idea of a single stress response seems unlikely and it must be questioned as to why the responses appear so similar. The answer may be found in the presence of cyclopropane fatty acids (CFAs) in the profile. The roles of cyclopropane fatty acids in the membranes of LAB are poorly understood, although it is known that in *E. coli* CFA synthesis occurs largely as the cells enter the stationary phase and

when starved for amino-acids (Eichel et al., 1999). The function of CFAs is believed to be related to changes in the fluidity of the membrane in a similar manner to the way the presence of polyunsaturated fatty acids (PFAs) affect membrane fluidity, that is, they have a stabilizing effect on the membrane (Johnsson et al., 1995, Grogan et al., 1997). The presence of a CFA, C19:0 cyc (lactobacillic acid) was observed in both control and stressed cultures. After heat and acid stress, the percentage composition of C19:0cyc was found to decrease from 26% to 12% and 16% respectively, indicating lactobacillic acid plays only a limited role during heat and acid stress. Previously there have been only a few attempts at trying to explain how CFAs aid the cell in tolerating stress covering: growth over a limited temperature range, acid habituation and ethanolinduced changes to fatty acid composition (Khuller et al., 1974, Couto et al., 1996, Brown et al., 1997, Evans et al., 1998). When observing the changes in CFA composition of growth over a wide temperature range of two different strains of Clostridium sp., Khuller (1974) and Evans (1998) observed quite dissimilar effects. Khuller demonstrated an increase in CFA composition towards the cooler temperatures while in contrast Evans found the increase in warmer temperatures. Current information suggests that CFAs should be found in increased amounts in heat-affected cells, as they are expected to reduce the effect of temperature on membrane fluidity. However, this study and Khuller found otherwise and it may be considered that changes in other components which affect membrane fluidity that are perhaps less energetically expensive, may in part explain this difference. Following acid habituation, Brown et al, (1997) found increased levels of cyclopropane fatty acids (C<sub>17:0</sub>cyc and C<sub>19:0</sub>cyc) in *E. coli*. No similar changes were seen in this study. Acid habituation, compared to a sudden acid exposure may produce different FAME profiles and perhaps the two methods of acid exposure cannot be compared directly and more work needs to be performed in this area. In conclusion, in strain VUP 12006 it appears that CFAs do not play a significant role in the protection of the cell to heat and acid stress.

#### 3.7.2.5 Lactic acid challenge

It has been shown previously that acidification of the cell is a principal factor in the cessation of cell division (Rallu *et al.*, 1996) and results shown in section 3.3.4 support these observations. In this experiment lactic acid was not used to provide a low pH environment. It can be seen that there is a logarithmic relationship between the  $\mu_{max}$  value and lactate concentration indicating an inhibitory effect from the organic acid rather than the presence of [H+]. Further studies of the affect of lactate were not pursued but may be followed up in future work.

#### 3.7.3 Response of Lb. acidophilus strain VUP 12006 to challenge at lethal pH

#### 3.7.3.1 Acid resistance is dependent on growth phase

The ability of Lb. acidophilus to survive at pH 2.5 for extended periods reflects an ability to control the internal pH of the cell by exclusion of hydrogen ions or by having the ability to pump protons from the cell. This study has shown that acid resistance is very dependent upon growth phase. In particular the ability to survive a lethal pH challenge was most ably demonstrated during stationary phase and lag phase where 50% and 45% of challenged bacteria, respectively, were able to survive for two hours. During exponential phase this value dropped to around 28%. This can be compared to the enteric pathogen Shigella flexneri, which was unable to demonstrate acid resistance until the late exponential phase (Gorden et al., 1993). The results from the current study indicate that Lb. paracasei strain VUP 12006 is acid resistant according to the criteria of Gorden and Small (1993) and this was equivalent to many of the Shigella strains used in their study of acid resistance. Interestingly, it is noted that even after twenty-four hours the inoculum of Lb. paracasei continues to survive even if at very low numbers (<1%). An extensive study of acid resistance of the eight candidate LAB strains was not undertaken. However, when VUP 12000 (Lb. acidophilus type strain) was tested similarly, it was more rapidly killed by exposure to pH 2.5, demonstrating the relatively greater acid tolerance of VUP12006.

#### 3.7.4 Comparison of test strains; the response to acid stress

The response to growth at low pH has been shown to vary between *Lactobacillus* strains. In the data shown in Figure 3.26 it can be seen that strain 12006, when compared with strains 12001 and 12007, was the only strain to up-regulate a 42 kDa protein during growth at pH 4.0. Although the optimum pH for growth is not known, strains 12001 and 12007 demonstrated an up-regulated protein of similar size (42 kDa) during growth at pH 6.3, but no analysis of that protein was undertaken to show whether the protein was the same as the protein produced by strain VUP 12006. These differences may be due to either species or strains variations, but clearly further work would need to be performed to determine if there was any link between the proteins produced by the strains. This could include N-terminal sequencing or antibody studies, using antibody prepared against the AphA protein of *Lb. paracasei*.

#### 3.7.5 Lb. paracasei strain VUP 1200; structure and function of AphA

Several difficulties were experienced in attempts to identify as well as find the structure and function of the 42 kDa protein up-regulated in *Lb. paracasei* strain VUP12006 when exposed to low pH and these will be reported in this section.

#### 3.7.5.1 identification of AphA

Initial investigations to establish the identification of AphA were only partially successful. The Nterminal sequence of 20 amino acids was obtained; however, the attempts at matching that sequence to a sequence in known protein databases were unproductive until the release of the *Lb. casei* ATCC 334 draft genome. Prior to the release of the draft genome several attempts were made to clone AphA into a representative genomic library, but the resulting transformants yielded only small inserts that showed no sequence homology to the N-terminal sequence. With the release of the draft genome this approach was changed to using PCR based techniques to obtain gene sequence.

The first round of PCR using 2 primer pairs designed around the upstream region and a third primer pair around the N-terminal region were successful in obtaining amplification products. These fragments showed 100% homology to sequence in *Lb. casei* 334, indicating that the N-terminal amino acid sequence obtained from *Lb. paracasei* VUP 12006 was coded in the region of scaffold three under investigation.

A second round of PCR was designed to find the complete sequence of the gene coding AphA. Using the known *Lb. casei* ATCC 334 sequence, primers were designed that would have covered the entire sequence if that sequence shared significant homology with the gene in *Lb. paracasei*. Although two primer pairs that used the same forward primer and had been designed to amplify DNA in the downstream region of the gene failed to produce products, six other primer pairs did produce products, which were sequenced and submitted to a BLASTn search of the draft *Lb. casei* genome. It was also found that the size of the expected PCR products was slightly lower for some primer pairs compared to the actual sizes obtained, suggesting differences between the genes in *Lb. casei* and *Lb. paracasei*. Analysis of the BLASTn results and subsequent alignment of the results with the draft genome of *Lb. casei* ATCC 334 revealed a region with 45 additional nucleotides compared to the draft genome (starting at position 8838) and this area was high in serine residues. This would appear to explain the size differences between some of the PCR products.

The data to this point provided over 90% of the AphA gene in *Lb. paracasei*. However, there was a gap between positions 1229 and 1263, so that the ORF was not defined. Consequently, anchored PCR was used in an attempt to walk down the gene from the region where the sequence was known. Although amplification products were obtained, the amplification products were all less than 800bp so that sequences beyond what was already known were not obtained. The reason small products were obtained from ligation of vector to *Lb. casei* VUP 12006 is not
immediately evident, but may be related to the selection of restriction endonuclease used or the vector.

PCR products obtained using these primers yielded expected results, in terms of fragment size, for *Lb. casei* ATCC 334, but only four amplification products from three primer pairs were observed when DNA from strain VUP 12006 was used. The size of the fragments was considerably larger than expected and they had to be sequenced to determine any relationship to AphA. Two of the amplification products were not from scaffold 3 and were not considered further, but the remaining two were found to share sequence homology in scaffold 3 of the draft genome of *Lb. casei* ATCC 334; the forward primer was the same for both these fragments and duplicate sequencing from the forward primer was identical for over 400 bases of sequence data obtained from *Lb. casei* ATCC 334.

It was found that the first  $\approx$ 120 bases of the sequence for the forward primers were found to share 100% homology with sequence from *Lb. casei* ATCC 334 starting at position 7777 of scaffold 3. Beyond position 7897, little homology was observed between the remaining 290+ bases to sequence from *Lb. casei* ATCC 334. Given the size of the PCR fragments obtained from *Lb. paracasei*, an additional 2,000+ bases remain to be sequenced.

The sequence from the reverse primer 3SP1AS was used to obtain data that included a stop codon for AphA. The data obtained for this primer showed discontinuous homology. The sequence data obtained for *Lb. paracasei* matched *Lb. casei* between positions 8188 and 8281 and positions 8405 and 8851 with a non-matching region of 124 nucleotides between positions 8281 and 8405. Although this sequence data was continuous for strain VUP 12006 the gap was seen when a ClustalW alignment was performed; refer to Figure 3.33. The apparent difference between the two strains may account for the lack of amplification products when primer 2SP1S was used to amplify DNA from strain VUP 12006.

#### 3.7.5.2 Structure and potential function of AphA of Lb. paracasei

The ORF encoded a protein of 420 amino acids with a predicted molecular mass of 43.49 kDa. A 28 amino acid signal peptide was predicted by SPScan and the cleavage site was confirmed by the N-terminal sequence. Thus the mature form of the protein is about 39.34 kDa, which is slightly smaller than the size observed by 1-D SDS-PAGE. From these results it can be concluded that AphA is translocated across the cytoplasmic membrane (due to the presence of the hydrophobic signal peptide) and is a secreted protein (the signal peptide has been removed as determined by the N-terminal sequence).

AphA did not posses the C-terminal cell wall anchor motif LPTXTG, which may be consistent with the protein being transitory in the cell wall before being excreted into the extracellular environment. No examination was made of the growth media, to look for the presence of AphA, in the current study so the presence or absence of the protein in the extracellular medium would have to be the subject of further study.

BLASTn (gene sequence) and BLASTx (protein sequence) were used to determine the relationship of AphA to known genes and proteins. The gene sequence was found not to have any significant matches in the known databases, although it was 94% homologous to an ORF found in the draft genome of *Lb. casei* ATCC 334. With the exception of that match, this gene was not conserved across unrelated or related genera. Protein relatedness was more apparent and there was significant homology to several members of the family *Lactobacillalles*. Many of the significant matches also showed the presence of secretory signal peptides and poly-serine domains and the proteins were of similar size, around 40-45 kDa.

The closest match was GSP-781, a glucan-binding protein from *Strep. mutans*. GSP-781 (general-stress protein) has been shown to be up-regulated significantly when *Strep. mutans* was exposed to high osmolarity and temperature, as well as low pH (Chia *et al.*, 2001). According to

Smith (1996) at least 6 proteins with the ability to bind  $\alpha$ -1,6-glucan, glucan-binding proteins or Gbp have been identified in *Streptococcus* species. These include GpbA, GbpB and GbpC. One of these, GbpB, is thought to play a role in the molecular pathogenesis (dental caries) although the role has not been resolved. However, despite the presence of homologous areas there were no amino acid domains that suggested the biological function of the protein, although it has been suggested that GSP-781 may be associated with the maintenance of the stability of the cell wall or cell membrane (Chia *et al.*, 2001).

No study has been made of the putative proteins from *Strep. pneumoniae* or *Strep. pyogenes*. The Genbank references are related to whole genome sequences and not a study of a particular gene.

### 3.7.5.3 Further studies

The role of AphA in acid stress response or acid tolerance in *Lb. paracasei* is not known, nor is the physiological role defined. Further analysis may include genetic analysis through gene manipulation, which may correlate the role this protein plays in generalised stress response, acid tolerance or cell-surface maintenance. However, such studies must await the development of specific genetic tools to allow inactivation of the AphA gene (presuming that this would not be fatal). Further studies on the sub-cellular location of the protein need to be done, given the observation of the signal sequence and attempts at purification showed the loss of the AphA protein in the homogenates after centrifugation, suggesting the protein was located in the membrane-associated fraction.

Other areas of further study could also include the following:

• Obtain further up-stream and downstream sequence, which would help to determine the presence of an operon. As was noted in chapter 3, the downstream regions of the AphA

gene in *Lb. paracasei* strain VUP 12006 and the homologous region in *Lb. casei* ATCC 334 shared little homology, and over 2,000 bases of a PCR fragment remain to be sequenced.

Preparation of antibodies against the 42 kDa protein identified during growth at low pH in *Lb.* paracasei strain VUP 12006. It would be useful to monitor the regulation of the gene through Western blot analysis. These antibodies could also be used to determine the presence of analogues of AphA in related strains, related species and non-related species as well as the quantification of AphA protein secreted into the growth medium. This protein has already been shown to have altered regulation in two related species compared to *Lb. paracasei* strain VUP 12006, see section 3.5.1.

### 3.8 Conclusion

- FAME, SDS-PAGE and carbohydrate utilisation profiles could not sufficiently differentiate the strains of LAB to enable them to be adequately identified to species level. Fingerprints produced by both methods were easily influenced by the growth conditions and the growth phase prior to harvest.
- 16S rRNA was able to differentiate to species level and strain VUP12006 was identified as *Lb. paracasei*. Molecular based identification, despite its cost and time-consuming methods, clearly produced the best results but these were not unambiguous either.
- Lb. paracasei VUP 12006 grew optimally at  $37^{\circ}C \pm 0.5$  and at pH 6.3  $\pm 0.2$ .
- When acid stressed, *Lb. paracasei* VUP 12006 produced a protein of approximately 42 kDa which was found to be cell membrane- or cell-surface-associated
- PCR based techniques were used to determine the nucleotide sequence of the gene known to encode the 42kDa AphA protein
- The genomic arrangement of the AphA gene was different in *Lb. paracasei* VUP 12006 compared to *Lb. casei* ATCC 334.

## Chapter Four

## STRESS RESPONSES IN *BIFIDOBACTERIUM* ANIMALIS (SYN. *LACTIS*) STRAIN VUP Bb 12

### 4.1 Introduction

At the time research for this thesis commenced, both *Lactobacillus* and *Bifidobacterium* strains were being investigated in the CBFT laboratories in terms of viability of probiotics in fermented dairy products. While investigations initially involving *Lb. paracasei* strain VUP 12006 were underway, as described in chapter 3, a strain of *Bifidobacterium* was also selected on which to undertake some preliminary studies on stress responses. Strain selection was based on known characteristics and use as a probiotic adjunct as described below:

- At the time this project commenced, the bacterium was commonly supplied for use as a probiotic adjunct in yoghurt manufacture;
- The strain had been previously shown to display functional characteristics (health benefits); and
- Other features of the strain such as identification and survival characteristics of the strain had been described previously in the literature (Clark and Martin, 1994; Sgorbati *et al.*, 1995; Meile *et al.*, 1997; Kailaspathy and Chin, 2000; Shin *et al.*, 2000; Alander *et al.*, 2001; Crittenden *et al.*, 2001; O'Riordan *et al.*, 2001).

On this basis, *B. animalis* (syn. *lactis*), (stored as strain VUP Bb 12), was selected for characterisation of stress responses. *B. animalis* (syn. *lactis*) strain VUP Bb 12 has been used by the starter culture manufacturer, Christian Hansen, since 1986, and is included in several product

lines including ABT cultures, single cultures and anti-diarrhoea tablets. ABT cultures contain *Lb. acidophilus, S. thermophilus* and *B. animalis* syn. *lactis* strain Bb 12 (Wind, personal communication).

### 4.2 Identification of test strain

Genus and species designation of strain VUP Bb 12 was confirmed using the oligonucleotide primers described in Table 4.1. Lm3/Lm26 are genus specific primers, Ban2/23Si was used to confirm species and PAF/536R was used for sequencing the 16S rRNA gene.

Table 4.1 Primers used fo	r identification of strain VUP	12006.
---------------------------	--------------------------------	--------

Name	Description	Tm
Lm3	CGGGTGCTNCCCACTTTCATG	63.4
Lm26	GATTCTGGCTCAGGATGAACG	54.6
Ban2	CATATTGGATCACGGTCC	50.5
23Si	CATTCGGACACCCTGGGATC	62.0
PAF	AGAGTTTGATCCTGGCTCAG	54.0
536R	GTATTACCGCGGCTGCTG	57.5

The PCR reaction mixture (50µL) for the first primer pair, Lm3/Lm26 contained 30 pmol of each primer, 200µM each of dATP, dTTP, dCTP and dGTP, approximately 100ng of bacterial DNA, 2.5 U of *Taq* DNA polymerase (Sigma) and PCR buffer with 50mM KCl pH 8.3, 10 mM Tris HCl and 1.5 mM MgCl<sub>2</sub>.

DNA was amplified as follows: initial denaturation at 94°C for 4 minutes, followed by 30 cycles consisting of denaturation at 94°C for 30 seconds, annealing at 57°C for 30 seconds, extension at 72°C for 60 seconds and a 7 minute final extension step at 72°C. The products were stored at 4°C until required. Aliquots of the reaction mixture (10µL) were separated on a 0.8% agarose gel, stained with ethidium bromide and visualised under UV light. The results are shown in Figure 4.1.



Figure 4.1 PCR amplification of DNA from *Bifidobacterium* strain VUP Bb 12 using primers Lm3 and Lm26.

Lane 1 is marker (EcoRI/Hind III digest, Progen). Lanes 2 and 3 are PCR amplification products of DNA using primers Lm3 and Lm26 (in duplicate). Products were obtained from the genomic DNA of strain VUP Bb 12. Products size was approximately 1770 bp.

Genomic DNA from strain VUP Bb 12 amplified using Lm3/Lm26 yielded a PCR product of approximately 1770 bp. Lm3/Lm26 have been shown to amplify all *Bifidobacterium* type strains and production strains using a 21 nucleotide sequence of the 16S gene that is highly conserved in *Bifidobacterium* (Kaufmann *et al.*, 1997). These results are therefore consistent with strain Bb 12 having a genus specific identification of *Bifidobacterium*.

Species level identification was confirmed using the primer pair PAF and 536R. The reaction mixture (50µL) for PAF/536R contained 30 pmol of each primer, 200µM each of dATP, dTTP, dCTP and dGTP, approximately 100ng of bacterial DNA, 2.5 U of Tag DNA polymerase (Sigma) and PCR buffer without MgCl<sub>2</sub>. DNA was amplified as follows: initial denaturation at 94°C for 2 minutes, followed by 30 cycles consisting of denaturation at 94°C for 45 seconds, annealing at 55°C for 45 seconds, extension at 72°C for 60 seconds and a 7 minute final extension step at 72°C. The products were stored and electrophoresed as described above. The results are shown in Figure 4.2. Genomic DNA from strain Bb 12 amplified using PAF/536R yielded a PCR product of approximately 550 bp. Sequence of the PCR products was determined using a CEQ 2000 (Beckman Coulter) according to the manufacturer's instructions. The sequences in both the forward and reverse direction were edited in the software package 'bio-edit' to about the first 450 bases and submitted to ANGIS for a BLASTn search (2.6.14). The data for sequences showing significant alignment is shown in Figure 4.3. These results are consistent with strain Bb 12 having a species identification of *B. animalis*. As strain BB 12 was stated to be a junior synonym sub-species (lactis) of B. animalis, a third round of PCR, using the primer pair Ban 2/23Si, was used to confirm the final identification. The primer Ban 2 was designed by Ventura and Zink (2002) to anneal to a conserved region of *B. animalis* but not to *B. lactis*. Ventura and Zink feel B. lactis is not a junior synonym of B. animalis but rather a subspecies of B. animalis and they used these primers to investigate the taxonomic differentiation of the *B. lactis*.



## Figure 4.2 PCR amplification of DNA from *Bifidobacterium* strain VUP Bb 12 using primers PAF and 536R.

Lane 1 is marker (EcoRI/Hind III digest, Progen). Lanes 2 and 3 are PCR amplification products of DNA using primers PAF and 536R (in duplicate). Products were obtained from the genomic DNA of strain VUP Bb 12. Product size was approximately 550 bp.

Mato	hing Entry (in Bank Main)	Begin-End	Description	Score	E Value
F	AB050138	[1-460]	Bifidobacterium animalis gene for 16S rRNA, partial sequence, strain:YIT 4090.	842.0	0.0
F	AB050137	<u>[1-460]</u>	Bifidobacterium animalis gene for 16S rRNA, partial sequence, strain: YIT 4119.	<u>842.0</u>	0.0
Г	AB050136	<u>[1-460]</u>	Bifidobacterium lactis gene for 16S rRNA, partial sequence, strain: YIT 4121.	<u>842.0</u>	0.0
Г	AB050134	<u>[1-460]</u>	Bifidobacterium animalis gene for 16S rRNA, partial sequence, strain:Y 93050.	<u>842.0</u>	0.0
L	AB050133	[1-460]	Bifidobacterium animalis gene for 16S rRNA, partial sequence, strain:Y 96007.	<u>842.0</u>	0.0

AB050138 Bifidobacterium animalis gene for 16S rRNA, partial sequence, strain:YIT 4090. Identities = 456/461 (99%) Strand = Plus / Minus

Query:	1	ggtgcttattcgaacaatccactcaacacggccgaaaccgtgccttgcccttgaacaaaa	60
Sbjct:	477	ggtgcttattcgaacaatccactcaacacggccgaaaccgtgccttgcccttgaacaaaa	418
Query:	61	gcgggttacaacccgaaggcctccatcccgcacgcggcgtcgctgcatcaggcttgcgcc	120
Sbjct:	417	gcggtttacaacccgaaggcctccatcccgcacgcggcgtcgctgcatcaggcttgcgcc	358
Query:	121	cattgtgcaatattccccactgctgcctcccgtaggagttctgggccgtatctcagtccc	180
Sbjct:	357	cattgtgcaatattccccactgctgcctcccgtaggagt-ctgggccgtatctcagtccc	299
Query:	181	aatgtggccggtcaccctctcaggccggctacccgtcaacgccttggtgggccatcaccc	240
Sbjct:	298	aatgtggccggtcaccctctcaggccggctacccgtcaacgccttggtgggccatcaccc	239
Query:	241	cgccaacaagctgataggacgcgaccccatcccatgccgcaaaagcatttcccaccccac	300
Sbjct:	238	cgccaacaagetgataggacgcgaccccatcccatgccgcaaaagcatttcccaccccac	179
Query:	301	catgcgatggagcggagcatccggtattaccacccgtttccaggagctattccggtgcac	360
Sbjct:	178	catgcgatggagcggagcatccggtattaccacccgtttccaggagctattccggtgcac	119
Query:	361	agggcaggttggtccacgcattactcacccgttcgccactctcaccccgacagcaagctg	420
Sbjct:	118	agggcaggttggt-cacgcattactcacccgttcgccactctcaccccgacagcaagctg	60
Query:	421	ccagggatcccgttcgacttgcaatgtg-taagcacgccgc 460	
Sbjct:	59	ccagggatcccgttcgacttgc-atgtgttaagcacgccgc 20	

Figure 4.3 Result of BLASTn search of GenBank using sequence data from primer pair PAF/536R and DNA from strain VUP Bb 12.

16S rRNA gene sequence data from strain VUP Bb 12 and alignment of the closest matching GenBank entry is shown below the top 5 BLASTn results.

The conserved region was two insertions of eight nucleotides each. The PCR reaction mixture (50μL) for Ban 2/23Si contained 50 pmol of each primer, 200μM each of dATP, dTTP, dCTP

and dGTP, approximately 100ng of bacterial DNA, 2.5 U of Taq DNA polymerase (Sigma) and PCR buffer with 50mM KCl pH 8.3, 10 mM Tris HCl and 1.5 mM MgCl<sub>2</sub>. DNA was amplified as follows: initial denaturation at 95°C for 5 minutes, followed by 30 cycles consisting of denaturation at 95°C for 30 seconds, annealing at 54°C for 60 seconds, extension at 72°C for 90 seconds and a 7 minute final extension step at 72°C. After running three trials no PCR products were obtained using the reaction mixture and conditions described above, although no positive control was available to confirm the result. According to Ventura and Zink, this negative result indicates a closer relationship of strain VUP Bb 12 to *B. lactis* rather than *B. animalis*. Based on currently accepted nomenclature and classification, these results are consistent with the identification remaining B. animalis syn. lactis. Evidence of the presence of the 'bifid shunt' has also been confirmed in this strain by use of the assay for the diagnostic enzyme fructose-6-phophate (Crittenden et al., 2001, Sgorbati et al., 1995). The phenotypic characterisation of strain VUP Bb 12 to genus level was checked at 3-monthly intervals using carbohydrate utilisation tests and the strain was monitored weekly for typical Gram stain and growth characteristics on MRS agar (section 2.2). Table 4.2 shows the parameters used for this characterisation.

## 4.3 Growth kinetics, protein profiling and effect of elevated temperature, low pH and aeration on growth curves of strain VUP Bb 12

The test strain was initially characterised by growth rate and protein profiles across a range of temperatures, pH levels and dissolved gases. Studies were conducted in one-litre fermenters according to the methods described in section 2.3.7. Cultures were grown in MRS broth with the addition of 0.05% cysteine-HCI and the fermenter was stirred at 150 rpm and sparged with nitrogen. Samples of fermenter contents were collected hourly for at least eight hours and again at twenty-fours hours after growth commenced.

## Table 4.2 Characteristics of *B. animalis* (syn. lactis) strain VUP Bb 12.

Summary of phenotypic characteristics of *B. animalis* (syn. lactis) strain VUP Bb 12.

Characteristic	Description
Carbohydrate fermentation	Glucose (+), Galactose (+), Lactose (+)
Gram stain	Gram positive bacilli with many bifurcated forms and club shapes
Growth characteristics	Grew on MRS agar as small 1-2 mm, white-grey colonies after 48 hours anaerobic incubation at 42°C.
	Cultures failed to grow after 48 hours aerobic incubation on MRS agar.

#### 4.3.1 Optimum temperature and pH for growth

#### 4.3.1.1 Determination of optimum growth conditions

Specific growth rates for *B. animalis* strain VUP Bb 12 were calculated from absorbance values measured during growth over a range of temperatures (22 to 47°C) and pH values (4 to 7). The highest specific growth rate during the growth cycle ( $\mu_{max}$ ) was used to determine optimum growth conditions, as described in section 2.3.8. Figure 4.4 (A) presents the maximum specific growth rate plotted against temperature. The temperature values at the top of the peak formed by this plot were determined to be the optimum temperature range for growth. Optimum growth temperature was determined to be 42°C ± 0.5°C. Figure 4.4 (B) presents the maximum specific growth rate plotted against pH. The pH values at the top of the peak formed by this plot were determined to be 42°C ± 0.5°C. Figure 4.4 (B) presents the maximum specific growth rate plotted against pH. The pH values at the top of the peak formed by this plot were determined to be the optimum pH range for growth. Optimum pH was determined to be 6.3 ± 0.3.

#### 4.3.1.2 Growth kinetics

Strain VUP Bb 12 was cultivated in batch culture in a one-litre fermenter. The inoculum was prepared by harvesting stationary phase cells of an overnight culture that was not pH controlled during growth (section 2.3.7.1). The cells were centrifuged and made up in pre-warmed MRS broth. When grown in optimum conditions (42°C, pH 6.3 in MRS broth with nitrogen sparging and agitation at 150 rpm with a starting OD<sub>600</sub> of 0.2.), mid-exponential phase was reached after approximately two hours and stationary phase was reached after approximately four hours. The maximum specific growth rate reached during growth in optimum conditions was 0.74 h<sup>-1</sup>.

## 4.3.2 Observations of protein profiles of Bb 12 cultures grown between 22°C and 47°C

SDS-PAGE was used to study changes in protein profiles of strain VUP Bb 12 grown over a range of temperatures (22°C and 47°C).



Figure 4.4 Optimum temperature and pH for growth of B. animalis (syn. lactis) Bb 12.

Maximum specific growth rate ( $\mu_{max}$ ) was calculated at different temperatures and pH levels. *B. animalis* (syn. *lactis*) strain VUP Bb 12 was grown in MRS broth with 0.05% cysteine-HCl in onelitre fermenters. Temperatures were set between 22 and 47°C with the pH fixed at 6.3 (A), or at 42°C at fixed pH levels between 4 and 7 (B). Nitrogen gas was sparged through the fermenter at 0.1 L/min. The peak of the resultant curve is the optimum growth range for temperature (A) and pH (B). Cultures were grown in MRS broth with 0.05% cysteine-HCl in one-litre fermenters at constant pH (6.3) and the chosen temperature (Figure 4.5). The inoculum was prepared by harvesting stationary phase cells of an overnight culture that was not pH controlled during growth (section 2.3.7.1). The cells were centrifuged and made up in pre-warmed MRS broth. The protein profiles showed several areas of altered expression during the growth cycle across the range of temperatures shown and these areas are circled in Figure 4.5. Protein bands that were up-regulated during growth at 47°C, which could be detected by laser densitometry, included bands of approximately 19, 25, 40 and 65 kDa.

## 4.3.3 Observations of protein profiles of Bb 12 cultures grown between pH 4 and pH 7

SDS-PAGE was also used to study changes in protein profiles of strain VUP Bb 12 grown over a range of pH values (4.0 to 7.0). Cultures were grown in MRS broth with 0.05% cysteine-HCl in one-litre fermenters at constant temperature (42°C) and the chosen pH. The results are shown in Figure 4.6. The protein profiles for growth at neutral (pH 7.0) and mildly acid pH (pH 6.0) were essentially identical but at pH 5.0 an increase was noted in the density of a band at around 33 kDa. The same band was markedly increased during growth at pH 4.0.

Laser densitometry was performed on Coomassie stained SDS-PAGE gels prepared from samples collected from cultures grown at pH 4.0. It was determined that the band at 33 kDa made up around 4% of the amount of all bands present at the time growth commenced. This had increased to 21% after one hour, 33% after 3 hours, and 50% after 5 hours (Figure 4.7). The proportion of the up-regulated band peaked at 58% after overnight growth at pH 4.0. Laser densitometry also determined that the band is a doublet, but this can only be seen early in the growth cycle and after the second hour of growth at pH 4.0 the doublet merges and appears as a single band. A second band, of approximately 36 kDa, was also up-regulated during growth at pH 4.0, but not as strongly as the 33 kDa band.

## Figure 4.5 SDS-PAGE profiles of *B. animalis* (syn. *lactis*) strain VUP Bb 12 grown over a range of temperatures.

Cultures of *B. animalis* (syn. *lactis*) strain VUP Bb 12 were grown at constant temperature and pH in one-litre fermenters containing MRS broth with 0.05% cysteine-HCl. The inoculum was prepared by harvesting stationary phase cells of an overnight culture that was not pH controlled during growth at 42°C (section 2.3.7.1). The cells were centrifuged and made up in pre-warmed MRS broth before inoculating to an initial OD<sub>600</sub> of approximately 0.2. Temperatures were set at 32° (A), 37° (B), 42° (C) or 47°C (D) and the pH at 6.3. Samples of fermenter contents were taken hourly, concentrated by centrifugation and boiled with double strength loading buffer. The treated samples were centrifuged to remove cell debris and approximately 1µg of protein was applied to each lane. Protein bands were visualised by silver staining. Bands with altered expression during the growth cycle are circled in yellow.

#### <u>Key</u>

Gel (A) Lane 1 MW Standard, Lanes 2-9 samples collected hourly Gel (B) Lane 1 MW Standard, Lanes 2-8 samples collected hourly Gel (C) Lane 9 MW Standard, Lanes 1-8 samples collected hourly Gel (D) Lane 3 MW Standard, Lanes 1-2, 4-9 samples collected hourly

Molecular Weight Standards (MWS):

- 1. Phosphorylase B-97.4 kDa.2. Bovine serum albumin-66.2 kDa3. Ovalbumin-45 kDa4. Carbonic anhydrase-31 kDa
- 5. Trypsin inhibitor-21.5 kDa 6. Lysozyme-14.4 kDa



(C)

## **MWS**<sup>1</sup> 2 3 4 5 6 7 8





(D)

### Figure 4.6 SDS-PAGE profiles of strain VUP Bb 12 grown over a range of pH levels.

Cultures of *B. animalis* (syn. *lactis*) strain VUP Bb 12 were grown at constant temperature and pH in one-litre fermenters containing MRS broth with 0.05% cysteine-HCI. The inoculum was prepared by harvesting stationary phase cells of an overnight culture that was not pH controlled during growth at 42°C (section 2.3.7.1). The cells were centrifuged and made up in pre-warmed MRS broth before inoculating to an initial OD<sub>600</sub> of approximately 0.2. Temperature was set at 42°C and the pH at 4.0 (A), pH 5.0 (B), pH 6.0 (C) or 7.0 (D). Samples of fermenter contents were taken hourly, concentrated by centrifugation and boiled with double strength loading buffer. The treated samples were centrifuged to remove cell debris and approximately 1µg of protein rich supernatant was applied to each lane. Protein bands were visualised by silver staining. Black arrow at left of gel (A) indicates the area of an up-regulated protein produced during growth at pH 4.0.

#### <u>Key</u>

Gel (A) Lane 10 MW Standard, Lanes 1-9 samples collected hourly during growth at pH 4.0

Gel (B) Lane 1 MW Standard, Lanes 2-10 samples collected hourly during growth at pH 5.0 Gel (C) Lane 10 MW Standard, Lanes 1-9 samples collected hourly during growth at pH 6.0 Gel (D) Lane 5 MW Standard, Lanes 1-4, 5-9 samples collected hourly during growth at pH 7.0

Molecular Weight Standards (MWS): 1. Phosphorylase B-97.4 kDa. 3. Ovalbumin-45 kDa

- 2. Bovine serum albumin-66.2 kDa
- 4. Carbonic anhydrase-31 kDa







A culture of *B. animalis* (syn. *lactis*) strain VUP Bb 12 was grown at constant temperature (42°C) and pH (4.0) in a one-litre fermenter containing MRS broth with 0.05% cysteine HCl. The inoculum was prepared by harvesting stationary phase cells of an overnight culture that was not pH controlled during growth at 42°C (section 2.3.7.1). The cells were centrifuged and made up in pre-warmed MRS broth before inoculating to an initial OD<sub>600</sub> of approximately 0.2. Samples were collected hourly for SDS-PAGE protein profiles and the resulting profiles studied by laser densitometry. Selected Coomassie stained SDS-PAGE laser densitometry profiles are shown above. Samples for the profiles shown were collected at the start of growth (Tzero ----) and then after 2 (T2---) and 24 hours (T24---).

The molecular weight standards are shown in black and correspond to:

Molecular Weight Standards:

- 1. Phosphorylase B-97.4 kDa.
- 3. Ovalbumin-45 kDa
- 5. Trypsin inhibitor 21.5 kDa

- 2. Bovine serum albumin-66.2 kDa
- 4. Carbonic anhydrase-31 kDa

## 4.3.4 Observations of protein profiles of Bb 12 cultures without pH control

A culture of *B. animalis* (syn. *lactis*) strain VUP Bb 12 was grown in MRS broth with 0.05% cysteine-HCl in a one-litre fermenter at 42°C. The fermenter was sparged with nitrogen at 0.1 L/min. The starting pH was 6.3 but pH was not controlled and the pH of the fermenter reached approximately pH 5.2 after three hours and approximately 4.5 after eight hours. Over duplicate runs, samples were collected every two hours and prepared for SDS-PAGE as described in section 2.5.1.1 and run as described in section 2.5.1. The results are shown in Figure 4.8. An increase can be seen in the regulation of a protein of approximately 33 kDa. This protein has the same molecular weight as the protein up-regulated during growth at pH 4.0. A second protein of approximately 36 kDa was also up-regulated during growth without pH control.

## 4.3.5 Observations of protein profiles of VUP strain Bb 12 cultures grown in a fermenter sparged with instrument air

The effect of aeration of strain VUP Bb 12 was studied by exposing bacterial cells grown in a fermenter sparged with instrument air from the time of inoculation. Two fermenters were set up at the same time; one was continuously sparged with nitrogen (99.9% nitrogen with a maximum of 1 ppm oxygen) at 0.1 L/min (control), and the other continuously sparged with instrument air (20.9% oxygen, 78.1% nitrogen and 0.9% argon) at 0.1 L/min (potentially causing oxidative stress).

SDS-PAGE was used to observe protein profiles of strain VUP Bb 12 sampled while growing with aeration. Cultures were grown in MRS broth with 0.05% cysteine-HCl in one-litre fermenters at constant pH (6.3) and temperature (42°C). Samples were collected every two hours after inoculation and the results for SDS-PAGE analysis are shown in Figure 4.9. The protein profiles produced during aerated growth are very similar to the control: no protein bands were obviously up-regulated during growth in instrument air but several minor bands of molecular weight greater than 97.4 kDa and around 25 kDa were down-regulated.



## Figure 4.8 SDS-PAGE profile of strain VUP Bb 12 grown without pH control.

A culture of strain VUP Bb 12 was grown at constant temperature in a one-litre fermenter containing MRS broth with 0.05% cysteine-HCl. The inoculum was prepared by harvesting stationary phase cells of an overnight culture that was not pH controlled during growth at 42°C (section 2.3.7.1). The cells were centrifuged and made up in pre-warmed MRS broth before inoculating to an initial OD<sub>600</sub> of approximately 0.2. Temperature was set at 42°C and the pH initially at pH 6.3. Samples of fermenter contents were taken every 2 hours (commencing two hours after inoculation), concentrated by centrifugation and boiled with double strength loading buffer. The treated samples were centrifuged to remove cell debris and approximately 1 $\mu$ g of protein was applied to each lane. Protein bands were visualised by silver staining. Black arrows at right of gel indicate area of up-regulated protein production during growth. The lower arrow indicates a protein band that has the same molecular weight as the protein band up-regulated during growth at pH 4.0.

### <u>Key</u>

Lane 1 MW Standard, Lanes 2-5 samples collected 2 hourly

Molecular Weight Standards (MWS):

- 1. Phosphorylase B-97.4 kDa.
- 3. Ovalbumin-45 kDa

- 2. Bovine serum albumin-66.2 kDa
- 4. Carbonic anhydrase-31 kDa





### oxygen environment.

Cultures of *B. animalis* (syn. *lactis*) strain VUP Bb 12 were grown at constant temperature (42°C) and pH (6.3) in one-litre fermenters containing MRS broth with 0.05% cysteine-HCl. The inoculum was prepared by harvesting stationary phase cells of an overnight culture that was not pH controlled during growth at 42°C (section 2.3.7.1). The cells were centrifuged and made up in pre-warmed MRS broth before inoculating to an initial OD<sub>600</sub> of approximately 0.2. The culture in lanes 1-4 (A) was sparged with nitrogen during growth while the second culture (B) was grown in the same conditions except that instrument air (20.9% oxygen and 78.1% nitrogen) was being sparged through the system instead of nitrogen (99.9% nitrogen). Molecular weight standards (MWS) are shown on the left.

### Key

- Gel (A) Lanes 1-4, samples collected every two hours
- Gel (B) Lanes 1-4, samples collected every two hours

Molecular Weight Standards (MWS):

- 1. Phosphorylase B-97.4 kDa.
- 3. Ovalbumin-45 kDa
- 5. Trypsin inhibitor-21.5 kDa

- 2. Bovine serum albumin-66.2 kDa
- 4. Carbonic anhydrase-31 kDa

### 4.3.6 N-terminal sequencing

The 33 kDa protein that was seen to be up-regulated during growth at pH 4.0 was submitted once for N-terminal sequencing to the Australian Proteome Analysis Facility. Protein samples were prepared from cultures grown at pH 4.0 and 37°C. Cells from the fermenter were harvested after six hours, concentrated by centrifugation and boiled with double strength loading buffer. The treated samples were centrifuged to remove cell debris and analysed by 1-D SDS-PAGE. After the presence of the 33 kDa protein was confirmed by Coomassie staining, the process was repeated except that the gel was electro-blotted to transfer the band of interest to PVDF membrane.

Amido Black 10B was used to visualise the band, which was excised from the membrane and submitted for sequencing (section 2.5.7). Results indicated that the N-terminus was blocked and sequencing data was not obtained.

**4.3.7 Effect of heat, acid and oxidative stress on growth curves of strain VUP Bb 12** After determining the pH and temperature conditions for fastest growth (section 4.3.1.1) and performing preliminary SDS-PAGE gels on strain VUP Bb 12 cells, a separate set of experiments was performed to determine suitable conditions to study stress responses in detail. The temperature selected for heat stress was 45°. At 45° there was a significant effect on growth rate but an increase in biomass during growth was observed initially. The selected level of acid stress was initially pH 4.0, however this was later changed to pH 4.3 for studies into sudden decreases in pH or continuous exposure to low pH during growth. This level (pH 4.0) was determined from both preliminary SDS-PAGE gels and growth kinetics data. Cultures grown at pH 4.0 had shown the development of a low molecular weight protein that was not seen at high pH values; however the maximum specific growth rate at pH 4.0 was very low (mean value of 0.111 over three experiments). At pH 4.0 cultures produced insufficient levels of biomass for protein studies, but this was achieved at pH 4.3 while retaining SDS-PAGE gel characteristics.

The effect of elevated temperature, low pH and exposure to oxygen on the extent of growth was measured by changes in OD<sub>600</sub> and compared to a control culture: typical results are shown in Figure 4.10. A starter culture of strain VUP Bb 12 was grown overnight as described in section 2.3.7.1 and used to inoculate the fermenters that contained MRS broth pre-heated to the selected temperature.

The relative changes in absorption for each of the fermenters varied with the different growth conditions. The biomass in the control cultures (42°C and pH 6.3) increased was approximately 11-fold after seven hours of incubation. Cultures grown at low pH increased biomass approximately 2.5-fold, reaching approximately one quarter the OD<sub>600</sub> of the control. The culture grown at low pH grew slowly compared to the control strain, and failed to reach stationary phase during the first seven hours of growth at pH 4.0.

The four fermenters were set up as follows:

	Condition	Temperature and pH	Gaseous environment
(1)	Optimum growth conditions	42°C and pH 6.3	Nitrogen sparged, 0.1 L/min
(2)	Elevated temperature	45°C and pH 6.3	Nitrogen sparged, 0.1 L/min
(3)	Low pH	42°C and pH 4.0	Nitrogen sparged, 0.1 L/min
(4)	Aeration	42°C and pH 6.3	Instrument air sparged, 0.1 L/min

The biomass in culture grown at 45°C was essentially unchanged after seven hours, although the OD<sub>600</sub> initially increased slightly to almost twice the the starting value after the first hour.



## Figure 4.10 Effect of low pH and elevated temperature on the growth of strain VUP Bb 12 as measured by changes in optical density.

Cultures of strain VUP Bb 12 were grown in MRS broth in one-litre fermenters under controlled pH and temperature conditions. The inoculum was prepared by harvesting stationary phase cells of an overnight culture that was not pH controlled during growth at 42°C (section 2.3.7.1). The cells were centrifuged and made up in pre-warmed MRS broth before inoculating to an initial  $OD_{600}$  of approximately 0.2. The starter was used to inoculate four fermenters containing MRS broth preheated to the selected temperature. The set conditions of the fermenters were (1) 42°C and pH 6.3 (  $\blacktriangle$  ), (2) 45°C and pH 6.3 (  $\bullet$  ), (3) 42°C and pH 4.0 ( ) and (4) 42°C and pH 6.3 (  $\checkmark$  ) with aeration. Nitrogen was sparged through fermenters 1-3 at 0.1 L/min and instrument air was sparged through fermenter 4 at 0.1 L/min.

This transient increase in the OD of the culture grown at 45°C was slightly higher than that for the optimally-grown culture, but in the second hour the optical density remained stationary and in the following four hours the optical density decreased slowly. This data suggests that the temperature selected for studying heat stress may have been too high and the reduction in OD may be due to cell lysis. Cultures grown in MRS broth sparged with instrument air increased biomass around 6.5-fold. These cultures had a lower  $\mu_{max}$  than the control strain and stationary phase was delayed by around two hours.

## 4.4 Analysis of physiological and biochemical changes in *B. animalis* (syn. *lactis*) strain VUP Bb 12 following growth in different pH conditions

Under the experimental conditions used, growth at elevated temperature and exposure to air during growth did not cause large changes in the cellular protein profile of strain VUP Bb 12 as detected by 1-D SDS-PAGE. In the case of growth at 47°C, this may have been due to cell death at the elevated temperature, shortening the period over which stress proteins could be expressed (see Figure 4.10). It was therefore decided that studies beyond the initial characterisation should focus on the effects of acid stress on strain VUP Bb 12, given that growth at low pH resulted in significant changes in cellular protein profile. The principle area of on-going examination then focussed on the responses of growing cultures to sudden decreases in pH and continuous exposure to low pH. The results of these experiments are described in the following sections.

## 4.4.1 Growth characteristics of VUP strain Bb 12 following sudden decrease in pH or continuous exposure to low pH

The effect of sudden decrease in pH or continuous exposure to low pH on strain VUP Bb 12 was determined by comparing changes in optical density during growth, viable cell counts, carbohydrate utilisation, 1 and 2-D SDS-PAGE profiles and FAME profiles.

This experiment was performed in duplicate, using seven 1L flasks, fitted with screw-capped lids and sampling ports, instead of fermenters; there were insufficient fermenters available to perform this experiment and at the same include all the selected growth conditions. This experimental setup ensured that identical starter cultures and fixed growth conditions were applied to each flask. The flasks were rotated at 50 rpm on a rotating platform and were sparged through the sampling port with nitrogen immediately before inoculation and after each sampling. All flasks were incubated at 42°C. The pH was maintained using a citrate phosphate buffer system fixed at the desired pH (section 2.3.1.1). The ability of the citrate phosphate buffer to control pH at specific levels was confirmed during each experiment by the removal of an aliquot of the growth media and checking the pH was within 0.1 pH units of the set value. With the exception of Flask 7, the starter cultures were grown overnight at pH 6.3. The starter culture for Flask 7 was grown at pH 4.3. Seven flask fermentations were set up as follows:

Flask	Growth conditions.
Flask 1	Optimum pH 6.3
Flask 2	pH changed from 6.3 to 5.3 after 2 hours
Flask 3	pH changed from 6.3 to 4.3 after 2 hours
Flask 4	Constant pH 5.3
Flask 5	Constant pH 4.3
Flask 6	Changed from 6.3 to 5.3 after one hour then to 4.3 at the second hour
Flask 7	Inoculum grown at pH 4.3, constant pH of 4.3

### 4.4.1.1 Effect of acid stress on growth kinetics (as determined by optical density)

The effect of sudden decrease in pH, or continuous exposure to low pH on growth kinetics was determined by optical density of strain Bb 12 (Figure 4.11). In Figure 4.11 (A) the effect of a

sudden decrease in pH can be seen on cells grown at optimum pH (6.3) that were pH shifted to either pH 5.3 or 4.3. Clearly the most severe effect is on cells shifted to pH 4.3, where the optical density decreased significantly, almost back to the value at the start of the growth cycle. The cells also failed to recover from this sudden change in pH and the growth curve remained relatively flat after the exposure. A decrease in optical density was also noted in the cells shifted to pH 5.3, although the change was not as severe. A continuous increase in optical density was noted in the control cells, which was consistent with previous experiments.

In Figure 4.11 (B) the effect of continuous growth at pH 4.3 and 5.3 is shown compared to a control grown at pH 6.3. It can be seen that the growth rate from cells grown at pH 5.3 was very similar to that of cells grown at pH 6.3, but that rate decreased significantly late in the growth cycle. By comparison with results shown in Figure 4.11(A), it can be seen that the extent of growth of cells grown continuously at either 4.3 or 5.3 is higher than for cells shifted to that specific pH during exponential phase. This suggests that the ability of the strain to cope with low pH may be related to the growth phase.

To determine if a brief exposure to an intermediate pH was protective of exposure to lower pH, the pH of test flasks was shifted from pH 6.3 to 4.3 (flask 3) or from 6.3 to 5.3 then to 4.3 (flask 6). Figure 4.11 (C) shows that change in  $OD_{600}$  for flasks 3 and 6 were similar, which suggests that prior culture at pH 5.3 for 2 hours did not improve subsequent growth at pH 4.3. However, when the results are compared to viable cell counts (see Figure 4.12), this data suggested that a step down in pH to 5.3 prior to final transfer to pH 4.3 might have improved final cell viability.

Figure 4.11 (D) shows the comparison of two flasks grown continuously at pH 4.3. One flask was inoculated with a starter culture grown overnight at pH 4.3 (flask 7) and the other starter was grown at pH 6.3 then inoculated into broth at pH 4.3 (flask 5).

# Figure 4.11 Optical density changes seen during culture of *B. animalis* (syn. *lactis*) strain VUP Bb 12 at different pH values and pH shift during growth.

Growth characteristics of *B. animalis* (syn. *lactis*) strain VUP Bb 12 during growth at low pH were determined by measuring changes in optical density. All flasks were held at constant temperature (42°C) and grown in MRS broth with 0.05% cysteine-HCI. The flasks were sparged with nitrogen, through a sampling port, every hour after sampling. Flasks were rotated at 50 rpm. Except for Flask 7, the inoculum was prepared by harvesting stationary phase cells of an overnight culture that was pH controlled at 6.3 during growth at 42°C (section 2.3.7.1). The cells were centrifuged and made up in pre-warmed MRS broth before inoculating to an initial OD<sub>600</sub> of approximately 0.2, as described in section 2.3.7. The inoculum for Flask 7 was grown at pH 4.3 but otherwise was treated the same way.

Flask	Key	Growth conditions
Flask 1	•	Optimum pH (6.3)
Flask 2		pH changed from 6.3 to 5.3 after 2 hours
Flask 3		pH changed from 6.3 to 4.3 after 2 hours
Flask 4		Constant pH 5.3
Flask 5		Constant pH 4.3
Flask 6	•	Changed from pH 6.3 to 5.3 after 2 hours then to 4.3 after the third hour
Flask 7	$\odot$	Constant pH 4.3 (Inoculum grown at pH 4.3)

The seven flasks were set up as follows:

Graph (A) compares growth between the control (flask 1) and two flasks that started at pH 6.3 but the pH was changed after two hours to pH 5.3 (flask 2) and pH 4.3 (flask 3) respectively. Graph (B) compares growth between the control (flask 1) and two flasks grown continuously at pH 5.3 (flask 4) and pH 4.3 (flask 5). Graph (C) compares growth between three flasks. All flasks started at pH 6.3. The pH of two flasks was changed after two hours to pH 5.3 (flask 2) and pH 4.3 (flask 3). The third flask was changed to pH 5.3 after two hours and changed again after a further hour to pH 4.3. Graph (D) compares growth between two cultures grown at pH 4.3. One flask was pre-acid-stressed by growing the starter culture at pH 4.3 (flask 7) and the other starter was grown at pH 6.3 (flask 5).



## Time (hours)

### 4.4.1.2 Effect of acid stress on viable cell counts

The effect of sudden decrease in pH, or continuous exposure to low pH, on viable cell counts was determined for strain VUP Bb 12 (Figure 4.12). Viable cell counts were performed by plating serial dilutions onto MRS agar plates and incubating the plates for 48 hours under anaerobic conditions (section 2.3.9.4). In Figure 4.12 (A) the effect of a sudden decrease in pH can be seen on cells growing at optimum pH then shifted to either pH 5.3 or 4.3. The viable cell count of the control culture grown at pH 6.3 increased over two orders of magnitude in the first eight hours and viability dropped only slightly in samples taken at 8 and 24 hours. Viable cell numbers in the culture grown at pH 5.3 increased almost two orders of magnitude in the first eight hours but viability decreased significantly, by over one order of magnitude, between the samples taken at 8 and 24 hours. After 24 hours growth at pH 4.3, no viable cells could be detected. Viable cell numbers increased only one order of magnitude in the first eight hours of growth at that pH. This data indicate that prolonged exposure to low pH had a significant impact on cell viability.

In Figure 4.12 (B) the effect of continuous growth at pH 4.3 and 5.3 is shown compared to a control grown at pH 6.3. At both pH 5.3 and 4.3, it can be seen that the viability of cells decreased significantly after eight hours growth. The decreases are similar to cultures that experience sudden decrease in pH during growth (flask 3). However, it must be noted that viable cells could still be detected in cultures grown continuously at pH 4.3 after 24 hours, whereas no viable cells could be detected in the culture experiencing the sudden pH decrease to pH 4.3.

To determine if a brief exposure to an intermediate pH decrease was protective of subsequent exposure to low pH, the pH of test flasks was shifted from pH 6.3 to 4.3 (flask 3) or from 6.3 to 5.3 then to 4.3 (flask 6). Figure 4.12 (C) shows the results of those changes. The findings showed that after twenty-four hours viable cells are still present in flask 6, while no viable cells could be detected in flask 3.

## Figure 4.12 Viable cell count changes seen during culture of B. animalis (syn. lactis) strain

VUP Bb 12 at different pH values and pH shift during growth.

Growth characteristics of *B. animalis* (syn. *lactis*) strain VUP Bb 12 during growth at low pH were determined by measuring changes in viable cell count. All flasks were held at constant temperature (42°C) and cultures grown in MRS broth with 0.05% cysteine-HCl. The flasks were sparged with nitrogen, through a sampling port, every hour after sampling. Flasks were rotated at 50 rpm. Except for Flask 7, the inoculum was prepared by harvesting stationary phase cells of an overnight culture that was pH controlled at 6.3 during growth at 42°C (section 2.3.7.1). The cells were centrifuged and made up in pre-warmed MRS broth before inoculating to an initial OD<sub>600</sub> of approximately 0.2, as described in section 2.3.7. The inoculum for Flask 7 was grown at pH 4.3 but otherwise was treated the same way.

Flask	Key	Growth conditions
Flask 1		Optimum pH 6.3
Flask 2		pH changed from 6.3 to 5.3 after 2 hours
Flask 3		pH changed from 6.3 to 4.3 after 2 hours
Flask 4	-	Constant pH 5.3
Flask 5		Constant pH 4.3
Flask 6	٠	Changed from pH 6.3 to 5.3 after 2 hours then to 4.3 after the third hour
Flask 7	•	Constant pH 4.3 (Inoculum grown at pH 4.3)

The seven flasks were set up as follows:

Graph (A) compares growth between the control (flask 1) and two flasks that started at pH 6.3 but the pH was changed after two hours to pH 5.3 (flask 2) and pH 4.3 (flask 3) respectively. Graph (B) compares growth between the control (flask 1) and two flasks grown continuously at pH 5.3 (flask 4) and pH 4.3 (flask 5). Graph (C) compares growth between three flasks. All flasks started at pH 6.3. The pH of two flasks was changed after two hours to pH 5.3 (flask 2) and pH 4.3 (flask 3). The third flask was changed to pH 5.3 after two hours and changed again after a further hour to pH 4.3. Graph (D) compares growth between two cultures grown at pH 4.3. One flask was pre-acid-stressed by growing the starter culture at pH 4.3 (flask 7), the other starter was grown at pH 6.3 (flask 5). When no viable cells were detected (<10/mL), this is shown as 1e+0 in all panels.



Time (hours)

These results suggest some protection of cell viability may have occurred through prior exposure to pH 5.3 before shifting to pH 4.3, an observation that could not be seen from OD<sub>600</sub> measurements. Figure 4.12(D) shows the comparison of two flasks grown continuously at pH 4.3. One flask was inoculated with a starter culture grown overnight at pH 4.3 and the other starter was grown at pH 6.3 then sub-cultured into broth at pH 4.3. Compared to the results observed for optical density, where pre-stressed cells demonstrated slightly poorer growth than the control, two minor differences were observed between the two cultures. Firstly, the increase in viable cells from the culture grown using an inoculum grown at pH 4.3 was over two orders of magnitude compared to an increase of less than half that in the non-stressed starter. Secondly, viable cells could be detected after twenty-four hours growth of the pre-stressed cultures. This data suggests that prior growth at low pH may assist subsequent growth and survival at low pH.

#### 4.4.1.3 Effect of acid stress on carbohydrate utilisation

HPLC was used to measure changes in carbohydrate concentration and to determine what metabolic by-products were produced during growth. Examples of HPLC traces of standards and samples are shown in Figure 4.13. Comparison to the retention time of known standards was used to show utilisation of glucose and production of lactate and acetate. The limit of detection for glucose, acetate, citrate and formate was 0.2 g/l, and 0.1 g/L for lactate. Changes in citrate concentration reflected the concentration of the citrate-phosphate buffer used to control pH. Four other peaks were detected by HPLC, including formic acid, but the concentration of these analytes did not change either during growth under optimum conditions or at low pH (Figure 4.13).

The results for glucose utilisation and lactate production are shown in Figure 4.14. The glucose concentration was 20 g/L in all flasks at the start of growth. After 24 hours growth, the glucose level of both the control, growing at pH 6.3, and the flask growing constantly at pH 5.3 fell to around 14 g/L.

1
and the second s
And the second se
1.00 2.00 3.00 4.00 5.00 6.00 7.00 8.00 9.00 10.00 11.00 12.00 13.00 14.00 15.00 16.00 17.00 18.00 19.00 20 Minutes
SampleName G2 Vial 2 Injection 1 Channel 410 Date Acquired 21/07/01 SampleName G2 Vial 2 Injection 1 Channel 410 Date Acquired 21/07/01
SampleName G2 Vial 2 Injection 1 Channel 410 Date Acquired 21/07/01 SampleName G2 Vial 2 Injection 1 Channel 410 Date Acquired 21/07/01



### Figure 4.13 HPLC standards and samples.

Growth characteristics of *B. animalis* (syn. *lactis*) strain VUP Bb 12 were determined by measuring changes in glucose and lactate concentration by HPLC. Glucose standards (A) (0.2 to 1.0 g/L) were prepared from 1.0 g/L stock. Samples (examples were collected from a single fermenter at the time intervals indicated) are shown in (B).
Figure 4.14 Effect of a sudden decrease in pH or constant exposure to acidic pH on glucose (Figure 4.14 A) and lactate concentration (Figure 4.14 B) on B. animalis (syn. lactis) strain VUP Bb 12.

Growth characteristics of B. animalis (syn. lactis) strain VUP Bb 12 during acid stress were determined by measuring changes in glucose and lactate concentrations by HPLC. All flasks were held at constant temperature (42°C) and cultures grown in MRS broth with 0.05% cysteine-HCI. The flasks were sparged with nitrogen every hour after sampling. Flasks were rotated at 50 rpm. Starter cultures were prepared as described in section 2.3.7, one was grown at pH 6.3 and used to inoculate flasks1-6, the other was grown at pH 4.3 and was used to inoculate flask 7. Samples were taken every hour over the first nine hours of growth and again after 24 hours.

Flask	Кеу	Growth conditions
Flask 1	•	Optimum pH 6.3
Flask 2		pH changed from 6.3 to 5.3 after 2 hours
Flask 3		pH changed from 6.3 to 4.3 after 2 hours
Flask 4	V	Constant pH 5.3
Flask 5	•	Constant pH 4.3
Flask 6		Changed from pH 6.3 to 5.3 after 2 hours then to 4.3 after the third hour
Flask 7		Constant pH 4.3 (Inoculum grown at pH 4.3)

covon flacks word sot up as follows:

(A) Glucose utilisation

(B) Lactate production





The cultures in which the pH was changed to 5.3 after 2 hours reached around 15 g/l after the same time.

The cultures that were grown at pH 4.3 or shifted to pH 4.3 during the growth cycle, failed to utilise significant amounts of glucose once at pH 4.3, which is consistent with the relatively poor growth and subsequent viability of cells seen in these flasks (see Figure 4.12). In general, increased lactate levels were not detected in any flask during the first five hours of growth and levels mainly increased during the period between the ninth and twenty-fourth hours.

However, cultures grown at pH 4.3, including those that experienced sudden changes to pH 4.3 produced only very minor levels of lactate during the twenty-four hours over which samples were collected. The control, grown at pH 6.3, and the culture grown constantly at pH 5.3 produced the highest levels of lactate, around 0.42 g/L and 0.45 g/L respectively. The culture that was shifted to pH 5.3 after two hours produced around 0.37 g/L of lactate. This suggests that the ability to grow at pH 4.3 was very limited.

#### 4.4.2 Effect of growth at low pH on protein profiles: 2-D SDS-PAGE Profiles

1-D SDS-PAGE profiles revealed little in terms of regulation of minor protein during sudden decrease in pH or continuous exposure to low pH, so it was decided to perform 2-D SDS-PAGE to better assess protein regulation during acid stress.

A method that was based on extraction of proteins in the presence of solvents was developed in an attempt to improve the extraction of hydrophobic proteins. This method development is shown in the following section. A more traditional method of separation of cytosolic proteins based on iso-electric focussing was also used to assess protein regulation during growth at low pH. This data is shown in section 4.4.2.3.

#### 4.4.2.1 Methods development

As part of the preparation for 2-D SDS-PAGE, cell homogenates were prepared using a Bead beater-8 <sup>®</sup> and 40mM Tris pH 7.0, as described in section 2.4.1.3. The protein profiles prepared using this technique had a major disadvantage, however, as the 33 kDa protein seen during acid stress was not detected in these cell-free extracts. This suggested that the protein may be a cell surface or membrane protein that would be retained in the cell debris after centrifugation of cell lysates. The cell debris, from broken cells, was therefore collected and prepared for SDS-PAGE in the same manner as whole cells. Analysis of protein profiles from this procedure revealed that the protein of interest was indeed retained with the cell debris and a method of extracting the protein would have to be found if the 33 kDa protein were to be tracked using 2-D SDS-PAGE. These results are shown in Figure 4.15. It should be noted that fine cellular material, which was not centrifuged from the samples, has stained as a strong dark background in the lane loaded with proteins prepared from cellular debris.

Cell stressors, by their nature, are often in close contact with cell membranes, and it was considered important to develop a method that preferentially extracted hydrophobic proteins so that changes in membrane proteins during cell stress could be observed. Given the difficulties associated with the use of 2-D SDS-PAGE as an analytical tool to study membrane proteins, in particular poor solubilization of membrane proteins, a literature search was performed to indicate suitable methodology. The literature suggested detergent partition and solvent extraction as two possible methods to extract hydrophobic proteins (Lopez *et al.*, 1991; Kobayashi *et al.*, 1998; Ferro *et al.*, 2000). In the end, two methods were examined: the first, using detergent extraction, was adapted from commercially available methods (Sigma, ProteoPrep Protein Precipitation Kit), and the second, extraction of protein using various solvents, was tested. In the first method, proteins were extracted in a bead beater cell in the presence of non-denaturing zwitterionic detergents made up in 40 mM Tris pH 7.



# Figure 4.15 Comparison of protein banding between samples prepared three ways.

Proteins were prepared from a single culture of *B. animalis* (syn. *lactis*) in a bead beater. Cells were resuspended in 40mM Tris pH 7. After bead beating the sample was centrifuged and both the clear supernatant and pellet were collected. A control sample of centrifuged whole cells was also collected. Approximately 200  $\mu$ L of each sample was added to an equal volume of double strength loading buffer and heated to 100°C for 5 minutes. The samples were centrifuged and 20 $\mu$ L was loaded onto a 1-D SDS-PAGE and visualised by silver staining.

Lane	Sample	Solvent solution
1	Supernatant removed from cellular debris	40mM Tris pH 7
2	MW Standard	
3	Whole cells	40mM Tris pH 7
4	Pellet of cellular debris left after bead beating	Double strength loading buffer

The detergents used were CHAPS (3-[(3-Cholamidopropyl)dimethylammonio]-1-propane sulfonate) and Triton X-100 made up as described in Table 4.3. Cells were obtained from a culture grown at pH 4.3 and were harvested after 6 hours. These zwitterionic detergents do not denature proteins and they were also selected for compatibility with the reagents used in the first dimension separation of 2-D SDS-PAGE.

The extractions were performed as described in section 2.4.1.3. Protein profiles were compared using 1-D SDS-PAGE. The results are shown in Figure 4.16. Figure 4.16 shows that the single detergent extractions have an almost identical banding pattern to the samples prepared using TRIS alone. The banding pattern produced from the mixture of detergents (CHAPS and Triton X-100) showed only a slightly different banding pattern to that produced from the single detergent extractions. While this may appear to indicate that hydrophobic proteins might have a slightly higher representation in samples prepared using the detergent mixture, the banding patterns were not dissimilar enough to pursue this method as a means of extracting hydrophobic proteins. The second method used to preferentially extract hydrophobic proteins examined a number of common solvents, including octanol, acetone, chloroform, ethanol, methylated spirit, diethyl ether and n-hexane. The solvents used in this method varied in boiling point (34-195°C), density (0.60-1.492 g/mL) and miscibility with water, see Table 4.4.

Extractions were performed in a bead beater-8, as described in section 2.4.1.3. Extracts were prepared in duplicate from cells grown overnight in MRS broth at 42°C. The cells were centrifuged and divided equally (0.5 g wet weight) into bead beater tubes prior to solvent extraction. After treatment both the solvent and aqueous layers were removed separately and dried under nitrogen. After drying the samples were made up in 200  $\mu$ L of 40 mM Tris pH 7.

# Table 4.3. Solutions used for protein extraction from cell homogenates of strain VUP Bb

## 12.

· · · · · · · · · · · · · · · · · · ·	Extraction solution
Control 1	40mM Tris pH 7
Treatment 1	CHAPS 4% (w/v) in 40mM Tris pH 7
Treatment 2	Triton X-100 2% (w/v) in 40mM Tris pH 7
Treatment 3	CHAPS 4% (w/v) and Triton X-100 2% (w/v) in 40mM Tris pH 7
Control 2	Protein extract prepared by boiling the cells in double strength loading buffer

Table 4.4. Physical properties of solvents used to extract proteins from homogenates of strain VUP Bb 12.

	Boiling Point (Range)	Density d <sup>20</sup> <sub>4</sub> (g/ml)	Miscibility with water
Octanol	193-195 °C	0.82	no
Acetone	55.5- 56.5 °C	0.790-0.792	yes
Chloroform	60.5-61.5 °C	1.492	no
Ethanol (95%)	80 °C	0.79	yes
Di-ethyl ether	34-35 °C	0.713-0.714	no
n-Hexane	69 °C	0.66	no
Methylated spirit	76-79 °C	0.806	yes



## Figure 4.16 Comparison of detergent extraction methods.

Proteins were extracted from a single culture of *B. animalis* (syn. *lactis*) in a bead beater cell using non-denaturing zwitterionic detergents (CHAPS and Triton X-100). The extractions were performed as described in section 2.2.6.4. Protein profiles were compared using 1-D SDS-PAGE.

Lane	Sample	Extraction solution
1	Standards	1. 97.4 kDa 2. 66.2 kDa 3. 45 kDa 4. 31 kDa 5. 21.5 kDa 6. 14 kDa
2	Control 1	40mM Tris pH 7
3	Treatment 1	CHAPS 4% (w/v) in 40mM Tris pH 7
4	Treatment 2	Triton X-100 2% (w/v) in 40mM Tris pH 7
5	Treatment 3	CHAPS 4% (w/v) and Triton X-100 2% (w/v) in 40mM Tris pH 7
6	Control 2	Protein extract prepared by boiling the cells in double strength loading
		buffer

The aqueous layer in the octanol, chloroform and n-hexane treatments produced a gel rather than a liquid, which was difficult to handle and dry. 1-D SDS-PAGE gels were loaded with  $20\mu$ L of protein-rich extract prepared by boiling 200µL of protein-rich extract with 200µL of double strength loading buffer. The results are shown in Figure 4.17. Several quite different banding patterns were obtained from a single culture. A total of 27 major bands could be identified from all of the extraction techniques combined; a major band constituted more than 5% of the sum of protein contained in all bands in a particular lane. Fifteen major bands were observed in protein samples for cells lysed in the presence of Tris buffer and glass beads alone, the highest of any treatment. Only one major band was observed in the octanol-treated samples. The banding patterns that most closely resembled the samples prepared in the presence of Tris buffer and glass beads alone, came from the water-miscible solvents, acetone, ethanol and methylated spirit. The size distribution of the major bands observed in the water-miscible solvents was almost identical to the samples extracted in the presence of Tris buffer and glass beads alone; however, some bands were present in higher amount compared to the Tris samples. In both the acetone and ethanol treated samples, bands at approximately 92, 68, 64, 61 and 51 kDa appeared to be preferentially extracted. Between 20 and 50% more of these bands was extracted by the addition of the solvents.

The water immiscible solvents produced markedly different banding patterns to the Tris-treated samples. Both octanol and chloroform produced a single major band of approximately 102 kDa. This band was seen in the aqueous and solvent layer of the octanol treated cells, although not as strongly in the aqueous layer. The aqueous layer of the chloroform treated cells appeared more like the Tris suspended cells but the clarity of the banding pattern was affected by the presence of strong background staining probably caused by cell debris caught up in the gel matrix from which the sample was prepared.

# Key

Lane	Solvent used to assist extraction of proteins	Sample
1	-	Molecular Weight Standards
		97.4 kDa
ļ		66.2 kDa
		45 kDa
		31 kDa
2	Octanol	aqueous layer
3	Acetone	solvent layer
4	Chloroform	solvent layer
5	Ethanol	aqueous layer
6	Ether	solvent layer
7	n-Hexane	solvent layer
8	Methylated spirit	aqueous layer
9	-	Tris-extracted cells
10	-	Tris-extracted cells
11	-	Biomass (whole cells boiled with equal volume of double strength loading buffer)
12	Octanol	solvent layer
13	Chloroform	aqueous layer
14	Ether	aqueous layer
15	n-Hexane	aqueous layer

LANE





*B. animalis* (syn. *lactis*) strain Bb 12 was grown overnight in MRS broth at 42°C. The culture was centrifuged and the cells divided into equal wet weights of 0.5 g. The cells were disrupted using the bead beater system, as described in section 2.2.6.4. A range of solvents was added to each sample to assist extraction of hydrophobic proteins. Where the solvent was immiscible in water, both the solvent and aqueous layers were collected then dried under nitrogen. A single sample was collected for water miscible solvents, which was also dried under nitrogen. These samples and whole cells were further prepared by boiling in double strength loading buffer. Equal loadings of 20  $\mu$ L of the centrifuged supernatant were then loaded onto 1-D SDS-PAGE to compare the effectiveness of the extraction procedure.

The other water immiscible solvents, diethyl ether and n-hexane, produced very similar banding patterns to each other. Only six major bands were produced, which were all present in the Trissuspended samples.

Using the results from the solvent extraction experiment, trials were made to determine if mixtures of solvents could be used to extract proteins from bacterial cells. These trials indicated that extracting proteins in the presence of a mixture of solvents of either 2 or 3 solvents drastically reduced the yield of protein for each sample. For example, a mixture of octanol, ether and ethanol or a mixture of hexane and chloroform produced insufficient protein to be visualised on a 1-D SDS PAGE gel. The cause of the reduction in yield is unclear.

Sufficient protein for testing was, however, obtained by mixing equal concentrations of the proteins extracted in the presence of single solvents. In a single trial, proteins separately extracted in chloroform and diethyl ether were dried under nitrogen, then mixed in equal concentrations after they had each been re-dissolved in Tris buffer. Comparison of proteins extracted in the presence of chloroform and diethyl ether and the mixture described above is shown in Figure 4.18.

The results indicated that a benefit (a reduction in the number of 2-D SDS PAGE gels that may be needed for comparison of profiles) could be obtained from preparing 2-D SDS PAGE gels from a 50:50 mixture of two water immiscible solvents with different banding patterns. Water miscible solvents had been shown to have banding patterns that were very similar to that of proteins extracted only in the presence of Tris buffer, and it would not have been advantageous to use one of those solvents in a mixture. In the following sections 'proteins extracted using solvents' refers to the combined chloroform and diethyl ether extracts mixed as described above.



# Figure 4.18 Comparison of proteins extracted in the presence of chloroform, diethyl ether and a mixture of chloroform and diethyl ether.

*B. animalis* (syn. *lactis*) strain VUP Bb 12 was grown overnight in MRS broth at 42°C and pH 4.3. The culture was centrifuged and the cells divided into equal wet weights of 0.5 g. The cells were disrupted using the bead beater system, as described in section 2.2.6.4. Either chloroform or diethyl ether was added to each sample to assist extraction of hydrophobic proteins. The solvent layer from each sample was collected, dried under nitrogen then re-dissolved in 200  $\mu$ L of Tris buffer. 100  $\mu$ L of each of these extracts were mixed to make a 50:50 mixture of each of the chloroform and diethyl ether extracts. All three samples were further prepared by boiling in an equal volume of double strength loading buffer. Equal loadings of 20  $\mu$ L of the centrifuged supernatant were then loaded onto 1-D SDS-PAGE and compared after visualisation using a silver stain.

Lane	Sample	Lane	Sample
1	MW Standards	3	Diethyl ether treatment
2	Chloroform treatment	4	50:50 mixture of chloroform and diethyl ether extracts

#### 4.4.2.2. 2-D SDS-PAGE for analysis of solvent-extracted proteins

2-D SDS-PAGE was performed on cultures grown at pH 6.3 and pH 4.3 (optimum pH and acid stressed conditions) in the conditions described for flask 2 and flask 5 in section 4.4.1. Up to 5 gels were run on the same samples using varying protein concentrations. Varying the loading allowed examination of all protein spots without problems of overloading masking proteins present in low abundance. The gels shown here are the ones that allowed proteins with altered regulation to be most easily observed. Proteins were extracted in the presence of either Tris or solvent, as described in sections 2.4.1.3 and 2.5.2. Samples were collected six hours after inoculation of the flasks. The results are shown in Figures 4.19, 4.20, 4.21 and 4.22. Figures 4.19 and 4.20 show 2-D SDS-PAGE profiles of cultures that were grown at pH 6.3. Figure 4.20 shows proteins that were present when cells were lysed in 40 mM Tris pH 7.0 and Figure 4.20 shows proteins extracted using solvent. Figures 4.21 and 4.22 show 2-D SDS-PAGE profiles of cultures that were extracted using 40 mM Tris pH 7.0 and Figure 4.22 shows proteins extracted using 4.21 shows proteins that were extracted using 40 mM Tris pH 7.0 and Figure 4.22 shows proteins that were grown at pH 4.3. Figure 4.21 shows proteins that were extracted using 40 mM

Over 150 protein spots could be detected in samples that were present when cells were lysed in Tris buffer in cultures grown at pH 6.3 and 4.3, and the spot profiles vary for each of the samples. In particular at least six proteins are up-regulated during growth at pH 4.3. These proteins are circled in red in Figure 4.21. Samples that were extracted in the presence of solvents demonstrated around half the number of protein spots that could be seen in Tris-extracted samples, but a number of up-regulated proteins could be seen and these are circled in Figure 4.22. Significantly, none of the proteins that were seen to be up-regulated in the Tris-extracted samples equated to the spots in the solvent-extracted samples in terms of pI and molecular weight. The 33 kDa protein could also be detected, but always appeared as a smear rather than a distinct spot. The best estimation at a pI was about 4.9. The 33 kDa protein is circled in blue in Figure 4.22.

Figu Trie)	e 4.19 Computer processed image of two dimensional SDS-PAGE of I	3. ani	malis (syn lactis) strain VUP Bb 12 grown at pH 6.3 (cleared lysate in
2-D 2-D hour prote Amr stack	electrophoresis was performed on samples of proteins extracted from st chloride. The temperature was controlled at $42^{\circ}$ C and the pH at 6.3. Nitroc after inoculation of the fermenter and protein extracted according to the r n concentration measured. First dimension isoelectric focussing of a 20 $\mu$ c id Pharmacia Biotech) according to section 2.5.2. The second dimension no del. Protein detection was performed by silver staining according to section	en w ethoc SDS- SDS- SDS-	'UP Bb 12 grown in 1L fermenters in MRS broth with 0.05% cysteine is sparged through the fermenter at 0.1 L/min. Samples were collected six described in section 2.4.1.3. The protein samples were desalted and the in sample was performed on a linear pH 4-7 immobilised pH gradient strip PAGE was carried out on a homogeneous resolving gel (12%) and a 4% .3.2.
<u>Key</u> Posit	on (lane or strip)		
-	Molecular weight standards	e	Molecular weight standards
	1. 97.4 KDa		
	2. 66.2 kDa		
	3. 45 kDa		
	4. 31 kDa		
	5. 21.5 kDa		
	6. 14 kDa		
2	IPG strip of strain VUP Bb 12 - culture grown at pH 6.3		



က

e 4.20 Computer processed image of two dimensional SDS-PAGE of <i>B. animalis</i> ( n).	lactis) strain VUP Bb 12 grown at pH 6.3 (solvent extracted
ectrophoresis was performed on samples of proteins extracted using chloroform and die cysteine hydrochloride. The temperature was controlled at 42°C and the pH at 6.3. Nitro ed six hours after inoculation of the fermenter and protein extracted according to the meth e protein concentration measured. First dimension isoelectric focussing of a 20 μg prot nt strip (Amrad Pharmacia Biotech) according to section 2.5.2. The second dimension \$4% stacking gel. Protein detection was performed by silver staining according to section \$20 μg.	ether. Cultures were grown in 1L fermenters in MRS broth with n was sparged through the fermenter at 0.1 L/min. Samples were described in section 2.4.1.3. The protein samples were desalted sample was performed on a 7cm, linear pH 4-7 immobilised pH S-PAGE was carried out on a homogeneous resolving gel (12%) .3.2.
in (lane or strip)	
Molecular weight standards 2 IPG stri   1. 97.4 kDa 2 2   2. 66.2 kDa 3 45 kDa   3. 45 kDa 4 31 kDa   5. 21.5 kDa 6 14 kDa	f strain VUP Bb 12 – grown at pH 6.3





Fig Tris	ure 4.21 Computer processed image of two dimensional SDS-PAGE of )	B. an	malis (syn lactis) strain VUP Bb 12 grown at pH 4.3 (cleared lysate in
2-D of them Prot	electrophoresis was performed on samples of proteins extracted from strair perature was controlled at 42°C and the pH at 4.3. Nitrogen was sparged thruch fermenter and protein extracted according to the method described in suscured. First dimension isoelectric focussing of a 50 μg protein sample wirmacia Biotech) according to section 2.5.2. The second dimension SDS-PA(iein detection was performed by silver staining according to section 2.5.3.2. F	vUP ough th ection as per 3E wa	Bb 12 grown in 1L fermenters in MRS broth with 0.05% cysteine HCl. The e fermenter at 0.1 L/min. Samples were collected six hours after inoculation 2.4.1.3. The protein samples were desalted and the protein concentration formed on an 18 cm, linear pH 4-7 immobilised pH gradient strip (Amrad s carried out on a homogeneous resolving gel (12%) and a 4% stacking gel. spots that are circled in red are up-regulated compared to pH 6.3.
Pos	ition (lane or strip)		
<b>~</b>	Molecular weight standards 1. 97.4 kDa 2. 66.2 kDa 3. 45 kDa 4. 31 kDa 5. 21.5 kDa	m	IPG strip of strain VUP Bb 12 – grown at pH 4.3
2	Molecular weight standards	4	Molecular weight standards



Figur	ure 4.22 Computer processed image of two dimensional SDS-PAGE of <i>B. animalis</i> (syn <i>lactis) strain VUP Bb 1</i> 2 grown at pH tein).	.3 (solvent extracted
2-D el MRS Samp were ( immot resolv resolv compa	<sup>1</sup> electrophoresis was performed on samples of proteins extracted using chloroform and diethyl ether (see 4.4.2.1). Cultures were grc S broth with 0.05% cysteine HCl. The temperature was controlled at 42°C and the pH at 4.3. Nitrogen was sparged through the mples were collected six hours after inoculation of the fermenter and protein extracted according to the method described in section 2.4.1 to desaited and the protein concentration measured. First dimension isoelectric focussing of a 20 µg protein sample was performed obliked pH gradient strip (Amrad Pharmacia Biotech) according to section 2.5.2. The second dimension SDS-PAGE was carried alving gel (12%) and a 4% stacking gel. Protein detection was performed by silver staining according to section 2.5.3.2. Proteins circlec uparted to pH 6.3. Protein circled in blue is 33 kDa protein seen in earlier SDS-PAGE gels.	in in 1 L fermenters in ermenter at 0.1 L/min. The protein samples in a 7cm, linear pH 4-7 ut on a homogeneous in red are up-regulated
<u>Key</u> Positic	ition (lane or strip)	
	Molecular weight standards 2 IPG strip of strain VUP Bb 12 – grown at pH 4.3 1. 97.4 kDa	
	2. 66.2 kDa	
	3. 45 kDa	
	4. 31 kDa	
	5. 21.5 kDa	
	6. 14 kDa	



The same procedure was used to compare proteins extracted from *B. animalis* (syn. *lactis*) strain VUP Bb 12 obtained as freeze-dried powder (intended for use as a starter culture) to determine if the protein profiles were similar to either a culture grown at pH 6.3 or 4.3 and from a fermenter where the culture was grown without pH control. Figures 4.23 and 4.24 show 2-D SDS-PAGE profiles of protein extracted from the freeze-dried starter culture following suspension in 40 mM Tris pH 7.0 (Figure 4.23) and solvent (Figure 4.24). Figures 4.25 and 4.26 shows 2-D SDS-PAGE profiles of proteins extracted from a culture grown without pH control using 40 mM Tris pH 7.0 (Figure 4.25) and solvent (Figure 4.26).

Nearly 200 spots could be detected in 2-D SDS-PAGE profiles of proteins prepared from freezedried cultures strain VUP Bb 12. The protein profile most closely resembles Figure 4.22, which was prepared from the culture grown at pH 4.3. Solvent-extracted proteins from the freeze-dried cells also contained the 33 kDa protein as did the solvent preparation from a culture grown without pH control. Of particular interest was the very close similarity between samples prepared from cultures grown without pH control and a culture grown at pH 4.3.

Table 4.5 summarises the molecular weight and pl of proteins up-regulated during growth at pH 4.3. Known standards (Serva IEF markers, supplied by Invitrogen) were run on 2-D SDS-PAGE to confirm the linearity of the immobilised pH gradient on the IEF strips.

#### 4.4.2.3 2-D SDS-PAGE profiles of cytosolic proteins

2-D SDS-PAGE profiles were obtained for cytosolic protein samples prepared by isoelectric focussing (IEF) using a Rotofor cell (BioRad), see section 2.5.5. Cell homogenates were prepared from cultures grown at pH 4.3 and 6.3 as described in section 4.4.1. Cell-free protein extracts were prepared using a Bead beater-8 <sup>®</sup> and 40mM Tris pH 7.0, as described in section 2.4.1.3.

Fig	ure 4.23 Computer processed image of two dimensional SDS-PAGE of <i>B.</i> . s).	anim	nlis (syn. lactis) strain VUP Bb 12 (Protein extracted from freeze dried
2-C iso sec by	electrophoresis was performed on samples of proteins extracted from freeze c electric focussing of a 50 $\mu$ g protein sample was performed on an 18 cm, lines tion 2.5.2. The second dimension SDS-PAGE was carried out on a homogene silver staining according to section 2.5.3.2.	dried ( ar pH ous re	ells into pH 7.0 Tris buffer as described in section 2.4.1.3. First dimension 4-7 immobilised pH gradient strip (Amrad Pharmacia Biotech) according to solving gel (12%) and a 4% stacking gel. Protein detection was performed
Ke			
БÖ	ition (lane or strip)		
-	Molecular weight standards 1. 97.4 kDa 2. 66.2 kDa 3. 45 kDa 4. 31 kDa 5. 21.5 kDa 6. 14 kDa 6. 14 kDa	5	EF sample of protein extracted into Tris buffer from strain VUP Bb 12 - freeze dried cells



Fig free	ure 4.24 Computer processed image of two dimensional SDS-PAGE of ize dried cells).	B. al	nimalis (syn lactis) strain VUP Bb 12 (solvent extracted protein from
2-D dim acc	electrophoresis was performed on samples of proteins extracted from freeze ( ension isoelectric focussing of a 50 μg protein sample was performed on ar ording to section 2.5.2. The second dimension SDS-PAGE was carried out c performed by silver staining according to section 2.5.3.2. Protein circled in blu	dried ( 18c) n a h ue is 3	cells using chloroform and diethyl ether as described in section 2.4.1.3. First m, linear pH 4-7 immobilised pH gradient strip (Amrad Pharmacia Biotech) omogeneous resolving gel (12%) and a 4% stacking gel. Protein detection 33 kDa protein of interest.
Pos	ition (lane or strip)		
<del>.</del>	Molecular weight standards 1. 97.4 kDa 2. 66.2 kDa	4	Protein extracted by boiling cells with double strength loading buffer- freeze dried cells
	3. 45 kDa 4. 31 kDa		
2	1 µg of protein sample used to run IEF	2 2	Molecular weight standards
ო	IEF sample of solvent extracted protein from strain VUP Bb 12 - freeze dried cells		
		ł	



Position (lane or strip)

electrophoresis was performed on samples of proteins extracted into pH 7.0 Tris. Cultures were grown in 1L fermenters in MRS broth with 0.05% cysteine-HCl. temperature was controlled at 42°C but pH was not controlled. Nitrogen was sparged through the fermenter at 0.1 L/min. Samples were collected six hours inoculation of the fermenter and protein extracted according to the method described in section 2.4.1.3. The protein samples were desafted and the protein entration measured. First dimension isoelectric focussing of a 50 μg protein sample was performed on an 18cm, linear pH 4-7 immobilised pH gradient strip ad Pharmacia Biotech) according to section 2.5.2. The second dimension SDS-PAGE was carried out on a homogeneous resolving gel (12%) and a 4% sing gel. Protein detection was performed by silver staining according to section 2.5.3.2.	Molecular weight standards 1. 97.4 kDa 2. 66.2 kDa 3. 45 kDa 4. 31 kDa 5. 21.5 kDa
2-D electro The temport after inocu concentrat (Amrad Pt stacking go Key	

Figure 4.25 Computer processed image of two dimensional SDS-PAGE of *B. animalis* (syn. *lactis) strain VUP Bb 1*2 grown without pH control (cleared lysate in Tris)





Fig ext	<sup>-</sup> igure 4.26 Computer processed image of two dimensìonal SDS-PAGE of <i>B. animalis</i> (syn. <i>lactis) strain VUP Bb 1</i> 2 grown without pH contro extracted protein).	ol (solvent
2-D sec Pha Pro	4-D electrophoresis was performed on samples of proteins extracted from a culture grown without pH control, using chloroform and diethyl ether as de ection 2.4.1.3. First dimension isoelectric focussing of a 50 μg protein sample was performed on a linear, 18 cm, pH 4-7 immobilised pH gradient st <sup>3</sup> harmacia Biotech) according to section 2.5.2. The second dimension SDS-PAGE was carried out on a homogeneous resolving gel (12%) and a 4% str <sup>3</sup> rotein detection was performed by silver staining according to section 2.5.3.2.	escribed in rip (Amrad acking gel.
<u>Pos</u>	<u>(ey</u> osition (lane or strip)	
<b>~</b>	Molecular weight standards 1. 97.4 kDa	
	2. 66.2 kDa 3. 45 kDa	
	4. 31 kDa	
	5. 21.5 kDa	
2	IEF sample of solvent extracted proteins from strain VUP Bb 12 - pH uncontrolled	



# Table 4.5 Molecular weight and pl of protein up-regulated during acid stress.

2-D SDS-PAGE was performed on proteins extracted from samples grown at pH 4.3 and pH 6.3. Estimations of size and pI were performed on proteins that were up-regulated during growth at pH 4.3 compared to pH 6.3. The method of treatment refers to either, proteins extracted using the bead beater with glass beads and Tris buffer (Tris) or proteins extracted in the presence of chloroform and di-ethyl ether (Solvent). Proteins extracted in the presence of solvents were prepared from equal concentrations of protein extracted in the presence of chloroform and di-ethyl ether. These preparations were then re-dissolved in Tris and mixed together in equal volumes.

	Method of treatment	Estimated molecular weight (kDa)	Estimated pl
1	Tris	43	6.2
2	Tris	27	4.1
3	Tris	25	5.0
4	Tris	23	5.1
5	Tris	23	5.2
6	Tris	26	5.3
7	Solvent	33	4.9
8	Solvent	35	5.4
9	Solvent	36	6.2
10	Solvent	22	5.2
11	Solvent	13	4.6
12	Solvent	5	4.9

The crude protein mixture, prepared by bead-beating (section 2.4.1.3) was de-salted by dialysis against distilled water. The samples were then mixed with broad range carrier ampholytes pH 3-10 (2% v/v) to make a final volume of 55 mL. The protein-ampholyte solution was injected into the sample chamber and large air bubbles removed. Constant power (12 W) was applied to the Rotofor until the current had stabilised for one hour. Each run took approximately four hours. Twenty fractions were collected from each run and 100µL was mixed with an equal quantity of double strength loading buffer and heated to 100°C for 5 minutes. The samples were then centrifuged and loaded onto a 1-D SDS-PAGE gel.

Comparison of the results for cells grown at pH 6.3 and 4.3 are shown in Figures 4.27 and 4.28. No bands were observed in fractions 16-20 for samples prepared from cultures grown at either pH. Visual comparison of the fractions 1-15 indicated a number of differences. In particular where there were fewer bands, as in fractions 10, 11 and 12, a number of changes in banding patterns could be seen. Proteins between approximately 40-45 kDa and 10-20 kDa appeared to be up-regulated during growth at pH 4.3. Higher molecular weight proteins of approximately 90 and 70 kDa appear to be down-regulated at pH 4.3. Other differences could be seen in low molecular weight proteins in fractions 5-9 of samples prepared from cells grown at low pH, but they were primarily down-regulation changes.

Samples from fractions 10, 11 and 12 (both pH 6.3 and 4.3-derived samples) were selected for further separation by 2-D SDS-PAGE because they contained proteins that appeared to be up-regulated during growth at low pH and relatively low numbers of bands meant that spots would be easier to identify and handle during subsequent collection of protein for N-terminal sequencing. A pre-formed immobilised pH gradient on a 7 cm plastic support strip (Immobiline DryStrip pH 4-7, Amrad Pharmacia Biotech) was rehydrated overnight by placing the strip in a sterile screw-capped test tube containing 20  $\mu$ g of protein in 40  $\mu$ L of rehydration solution.

4.27 Computer processed image of one dimensional SDS-PAGE of B. animalis (syn. lactis) strain Bb12 grown with pH control at pH 6.3 (rotofor	n 1-12) and 4.3 (rotofor fractions 1-13).
ure 4.27	ction 1-1

position 13, 16 and 28 are blank (contain no samples). Lanes 1, 15 and 31 contain molecular weight standards (MWS) 97.4 kDa, 66.2 kDa, 45 kDa, 31 kDa and 21.5 Constant power (12W) was applied for approximately 4.5 hours until the proteins were separated according to their isoelectric point. The fractions, numbered from the -D electrophoresis was performed on samples of proteins extracted from B. animalis (syn. lactis) grown in 1L flasks in MRS broth. The temperature was controlled at 42°C but the pH of each flask was varied. Samples shown in lanes 2-14 were grown at pH 6.3 and samples shown in lanes 17-30 were grown at pH 4.3. Lanes at kDa. Nitrogen was sparged through the flasks prior to flask inoculation and after collection of each sample. Samples were collected six hours after inoculation of the flask and protein extracted according to the method described in section 2.4.1.3. The crude protein mixtures prepared by bead-beating were de-salted and the protein concentration measured. The protein extract was mixed with carrier ampholytes (2%) to a final volume of 51 mL and injected into the Rotofor (BioRad) chamber. acidic end of the rotofor, were applied to the lanes as listed below. Proteins up-regulated at pH 4.3 compared to pH 6.3 are circled in blue. Proteins down-regulated at pH 4.3 compared to pH 6.3 are circled in red

or	on											
Rotof	Fracti	2	00	,	6	6 0	6 11 10	-110	9 11 9   12 11 10 9	13 2 11 10 9	13 2 11 0 0	13 2 11 0 9
ЬН		4.3	4.3		4.3	4.3 4.3	4.3 4.3 4.3	4.3 4.3 4.3 No sample	4.3 4.3 4.3 No sample 4.3	4.3 4.3 4.3 No sample 4.3 4.3	4.3 4.3 4.3 No sample 4.3 MWS	4.3 4.3 4.3 No sample 4.3 MWS
Lane		23	24		25	25 26	25 26 27	25 26 27 28	25 26 27 28 28 29	25 26 27 28 29 30	25 26 27 28 28 30 31 31	25 26 27 28 28 30 31
Rotofor	Fraction	11	6		12	12	12	1 12	12	12 2 3	12 2 1 3 3 4	12 2 1 5 4 4
Hd		6.3	No sample		6.3	6.3 MWS	6.3 MWS No sample	6.3 MWS No sample 4.3	6.3 MWS No sample 4.3	6.3 MWS No sample 4.3 4.3	6.3 MWS No sample 4.3 4.3 4.3 4.3	6.3 MWS No sample 4.3 4.3 4.3 4.3 4.3
Lane		12	13		14	14 15	15 15	14 15 16 17	14 15 17 18	14 15 16 17 18 19	14 16 17 20 20	14 15 13 19 20 20 21
Rotofor	<sup>-</sup> raction		_									
H		1WS	.3		6		6. 6. 6.	<u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u></u>				
ane	-		2		0	00	5 4 0 6 0 0	6 6 6 6	0 0 0 0 0	0 0 0 0 0 0	000000000	0 0 0 0 0 0 0 0 0



LANE

for	
(roto	
1 6.3	
at ph	
itrol	
4 cor	
thp	
'n wi	
grow	
3b12	
ain B	
s) str	
lactis	
syn.	
alis (	
anima	
ф. Э	
GE o	
S-PA	
I SD	
siona	
mens	
ne di	ć
of ol	14-2(
nage	ions
ed in	fract
cess	tofor
r pro	3 (rot
pute	nd 4.
Com	20) al
4.28	13-1
gure	Iction
'≓	2

at 42°C but the pH of each flask was varied. Samples shown in lanes 2-9 were grown at pH 6.3 and samples shown in lanes 11-17 were grown at pH 4.3. Lane at Nitrogen was sparged through the flasks prior to flask inoculation and after collection of each sample. Samples were collected six hours after inoculation of the flask Constant power (12W) was applied for approximately 4.5 hours until the proteins were separated according to their isoelectric point. The fractions, numbered from position 18 is blank (contain no sample). Lanes 1,10 and 19 contain molecular weight standards (MWS) 97.4 kDa, 66.2 kDa, 45 kDa, 31 kDa and 21.5 kDa. 1-D electrophoresis was performed on samples of proteins extracted from B. animalis (syn. lactis) grown in 1L flasks in MRS broth. The temperature was controlled and protein extracted according to the method described in section 2.4.1.3. The crude protein mixtures prepared by bead-beating were de-salted and the protein concentration measured. The protein extract was mixed with carrier ampholytes (2%) to a final volume of 51 mL and injected into the Rotofor (BioRad) chamber the acidic end of the rotofor, were applied to the lanes as listed below.

Lane	Hd	<b>Rotofor Fraction</b>	Lane	Hd	Rotofor Fraction
-	SWM		11	4.3	14
2	6.3	13	12	4.3	15
3	6.3	14	13	4.3	16
4	6.3	15	14	4.3	17
5	6.3	16	15	4.3	18
9	6.3	17	16	4.3	19
7	6.3	18	17	4.3	20
8	6.3	19	18	No sample	
ത	6.3	20	19	MWS	
10	SWM			Ĩ	


LANE

The strip was focussed in three stages using a Multiphor II electrophoresis unit with 2-D application kit according to the manufacturer's instructions (Amrad Pharmacia Biotech). Times were at the high end of the recommended range. After focussing, the strip was removed and equilibrated in solutions A and B (ten minutes in each solution). The second dimension of separation was performed by placing the equilibrated strip onto a 12% vertical gel sandwich (CBS triple wide system). The second dimension was run for 120 volt hours (600 V, 40 mA, 10 W) at constant current before the plastic support strip was removed. The gels were then allowed to run until the bromophenol blue dye front had reached the lower edge of the acrylamide gel. These results are shown in Figures 4.29 and 4.30. Figure 4.29 shows 2-D SDS-PAGE of proteins extracted from a culture grown at pH 6.3. Figure 4.30 shows 2-D SDS-PAGE of proteins extracted from a culture grown at pH 4.3. The gels were compared to confirm the presence of proteins that were up-regulated during growth at pH 4.3 and these compared with earlier results using 1-D SDS PAGE gels. 2-D SDS-PAGE was repeated on the pH 4.3-derived sample using pooled fractions 11, 12 and 13. This was done to achieve the high level of protein required for Nterminal sequencing.<sup>1</sup> A pre-formed immobilised pH gradient on an 18 cm plastic support strip (Immobiline DryStrip pH 4-7, Amrad Pharmacia Biotech) was rehydrated overnight by placing the strip in a sterile screw-capped test tube containing 120 µg of protein in 350 µL of rehydration solution. 2-D SDS PAGE was performed as described above. After electrophoresis, protein bands were electro-blotted using the technique described for Western blotting (section 2.5.4) to PVDF membrane using CAPS buffer and visualised using an Amido Black 10B stain. Seven spots of interest, labelled Bsb-2 to Bsb-8, were excised and sent to LaTrobe University, Department of Biochemistry. The spots of interest were proteins that had been seen to be upregulated during growth at pH 4.3. Of the seven spots sent for sequencing, spots Bsb-2, Bsb-3 and Bsb-4 were found to have insufficient protein for sequencing and spots Bsb-5 and Bsb-6 had identical sequence patterns. Sequence data was obtained for spots Bsb-5, 6, 7 and 8.

Figure 4.29 Computer processed ir	image of two dimensional SDS-PAGE of <i>B. animalis</i> (syn. <i>lactis</i> ) strain Bb12 grown with pH control at pH 6.3.
2-D electrophoresis was performed c broth at pH 6.3 and at 42°C. Samp 2.4.1.3. The crude protein mixtures ampholytes (2%) to a final volume of the proteins were separated accordi immobilised pH gradient strip (Amra resolving gel (15%) and a 4% stackir	on Rotofor fractions 11, 12 and 13. Samples of proteins were extracted from <i>B. animalis</i> (syn. <i>lactis</i> ) grown in 1L flasks in MRS ples were collected six hours after inoculation of the flask and protein extracted according to the method described in section prepared by bead-beating were de-salted and the protein concentration measured. The protein extract was mixed with carrier of 51 mL and injected into the Rotofor (BioRad) chamber. Constant power (12W) was applied for approximately 4.5 hours until ing to their isoelectric point. First dimension isoelectric focussing of a 20µg protein sample was performed on a linear pH 4-7 ing gel. Protein detection was performed by silver staining according to section 2.5.3.
<u>Kev</u> Position (lane or strip)	
	Molecular weight standards (MWS) 97.4 kDa, 66.2 kDa, 45 kDa, 31 kDa, 21.5 kDa and 14 kDa
	Rotofor fraction 11 (sample grown at pH 6.3)
1 0	MWS
04	Rotofor fraction 12 (sample grown at pH 6.3)
	SMM
	Rotofor fraction 13 (sample grown at pH 6.3)
2	SMW

LANE



	Figure 4.30 Computer processed image of two dimensional SDS	-PAGE of <i>B. animalis</i> (syn. <i>lactis</i> ) strain Bb12 grown with pH control at pH 4.3.
	2-D electrophoresis was performed on Rotofor fractions 11, 12 and 1 broth at pH 4.3 and at 42°C. Samples were collected six hours af 2.4.1.3. The crude protein mixtures prepared by bead-beating were ampholytes (2%) to a final volume of 51 mL and injected into the Ro the proteins were separated according to their isoelectric point. Fin immobilised pH gradient strip (Amrad Pharmacia Biotech) accordin resolving gel (15%) and a 4% stacking gel. Protein detection was p terminal sequenced.	3. Samples of proteins were extracted from <i>B. animalis</i> (syn. <i>lactis</i> ) grown in 1L flasks in MRS er inoculation of the flask and protein extracted according to the method described in section de-salted and the protein concentration measured. The protein extract was mixed with carrier tofor (BioRad) chamber. Constant power (12W) was applied for approximately 4.5 hours until st dimension isoelectric focussing of a 20µg protein sample was performed on a linear pH 4-7 ng to section 2.5.2. The second dimension SDS-PAGE was carried out on a homogenous erformed by silver staining according to section 2.5.3.2. Proteins within the red circles were N-
308	<u>Key</u> Position (lane or strip)	
	1 Molecular weight standard	s (MWS) 97.4 kDa, 66.2 kDa, 45 kDa, 31 kDa, 21.5 kDa and 14 kDa
	2 Rotofor fraction 11 (sample	grown at pH 4.3)
	3 MWS	
	4 MWS	
	5 Rotofor fraction 12 (sample	grown at pH 4.3)
	6 MWS	
	Z MWS	
	8 Rotofor fraction 13 (sample	e grown at pH 4.3)
	6 B	

LANE



This data was submitted for a BLASTP search to GenBank through ANGIS and the draft genome of *B. longum* was searched through ORNL. The results from these searches are shown in Table 4.6. The matches to Bsb-6 and 8 were elongation factor-P and a 50S ribosomal protein L7/L12. Otherwise, only hypothetical proteins of no known function could be matched to Bsb-7.

### 4.5 Discussion

This chapter concentrated on the effects of acid stress on the growth kinetics and proteins made by *B. animalis* (syn. *lactis*) strain VUP Bb 12. The results presented in the previous sections are discussed below and the significance of the findings presented.

## 4.5.1 Characterisation of *B. animalis* (syn. *lactis*) strain VUP Bb 12 4.5.1.1 Optimum temperature and pH for growth

Table 4.7 summarises the optimum growth temperature and pH and temperature and pH range determined in this study and compares them to published values. By contrast to the published values for *Bifidobacterium* sp., the strain used in this study has a wider temperature growth range. The fastest growth rate for temperature and upper temperature limit for growth are similar to the values stated for this genus. Although the values obtained in this study (section 4.3.1) show that VUP Bb 12 was able to initiate growth at 45°C, it was not able to sustain growth beyond the first few hours. The lower temperature limit of growth was approximately 22°C, at least 3°C lower than published values. The findings for upper temperature limit are consistent with those found in the work by Crittenden *et al.* (2001), which examined two strains of *B. lactis* (B94 and DS 920) that were shown to have optimum growth temperatures around 40°C.

The pH determined to be the optimum for growth in this study, pH 6.3, falls within the expected range for this species, but the lower growth limit appears equal to (or possibly lower) than the previously published values for this genus.

<u>с</u>
4
I
<u>d</u>
at
2
5
ä
E
ai
St.
č
tis
S
-10
Ľ.
Ś
) s
ili.
nő
iir
al
œ
Ť.
0
f
Š
ŭ
- Dí
Ĕ
Ï
ď
σ
fe
la I
<u>б</u>
ē
<u>م</u>
ns
ei.
đ
Ď
J
i l
Š
우
С С
÷
_
0
ng o
icing a
encing o
quencing o
equencing o
l sequencing o
ial sequencing o
ninal sequencing o
rminal sequencing o
Terminal sequencing o
V-Terminal sequencing o
f N-Terminal sequencing o
of N-Terminal sequencing o
ts of N-Terminal sequencing o
ults of N-Terminal sequencing o
esults of N-Terminal sequencing o
Results of N-Terminal sequencing o
.6 Results of N-Terminal sequencing o
4.6 Results of N-Terminal sequencing o
vie 4.6 Results of N-Terminal sequencing o
able 4.6 Results of N-Terminal sequencing o

Cytosolic proteins that were seen to be up-regulated during growth at pH 4.3 were N-terminally sequenced by The Department of Biochemistry at LaTrobe University. The data obtained from this sequencing was submitted to both Genbank and ORNL for BLASTP searches. Completed genomes were searched in Genbank and the draft genome of B. longum was searched in ORNL. The molecular weight, estimated pl and possible function of the closest protein matches are shown in columns 2, 3 and 6 respectively. The N-terminal sequence of protein matches in the B. longum draft genome is shown in column 5. The sequence of the closest protein match is shown in column 7.

Name         Wit         Pit         Neterninal sequence         B.         Iongum         N-terninal         Possible function         Sequence of related proteins           BSb-5         20654         5.05         same as Bsb-6         same as Bsb-6         mod ThubiKuGSVLLDGGLWTMKFGHVKPGKFBAFVFT           BSb-5         20654         5.05         same as Bsb-6         same as Bsb-6         mod ThubiKu GSVLN         mod ThubiKu GSVLN           BSb-6         20654         5.05         A(0)(T)(T)N D()(K N G - L N         MaQTTND[KN GSVLN         "elongation factor P"         Micruitacipacyclicitic/GAGE/GAP/KFGKFBAFVFT           BSb-7         20654         5.05         A(0)(T)(T)N D()(K N G - L N         MaGTTND[KN GSVLN         "elongation factor P"         Micruitacipacyclicitic/GAGE/GAF/KAFT           BSb-7         20654         5.05         A(0)(T)(T)N D()(K N G - L N         MaGTTND[KN GSVLN         "Elongation factor P"         "Elongation factor P"           BSb-7         20654         5.05         A(0)(T)(T)N D()(K N G - L N         MaGTTND[KN GSVLN         "Elongation factor P"           BSb-7         26973         4.64         SCHARWAXXMAKAADARGHCFAARDARGHCFAA					
Name (k0a)W plN-terminal sequenceB. <i>longun</i> N-terminal sequenceBsb-5206545.05same as Bsb-6same as Bsb-6Bsb-6206545.05adm as Bsb-6same as Bsb-6Bsb-7206545.05A(C)(T)(T)N D()K N GL NMACITNDIKN GSVLNBsb-7269734.64SGHXKWAXXDAMSGHSKWATTKHKKAAIDA??Bsb-8132474.56AK FT GD E LLE AFMAKYTNDELLEAF50S RIBOSOMAL PROTEIN L7/L12	Sequence of related proteins		MAQTTNDIKNGSVLNLDGQLWTVMKFQHVKPGKGPAFVRT IKNVLSGKIVDKTFNAGMKMEFETVDNRTLQYSYEDGDN FVFMDMTTYDQIMVPKTLLGDKAKFLLEGTDCLVSFHDGTPLSVDLPGSV VLTITHTPGLQGNRSNAGTKPATVETGAE IQVPLFINEG DRVKINTEDG SYTGPENN	MSGHSKWATTKHKKAAIDAKRGKLFAKLIKNIEIAARLGGGDPDGNPSLY DAIYKAKASMPADNIARAVKRGGAEDGAANYEDIVYEGYAPAGVGLII ECLTDNRNRAAAEVRSTLTKGNGSLATSGSVSFNFERKGQ IVVPSEGVDFDKLFETAAEAGAEDVTDDGEVFTVVTGPSD LFTVRKALQEAGFDYDSADQVMQPKNEVELSLEDARKVSK LIDNLDDLDDVQNIYSNWTASDEVMAQLDEE	MAKYTNDELLEAFGEMTLVELSEFVKAFEEKFDVEAAAPV AAVAAVAGAAAPAEEEKDEFDVILSAVGDKKIQVIKAVRAITNLGLAEAK ALVDGAPKAVLEKAKKEDAEKAKAQLEEAGASVELK
NameWW (kDa)PlN-terminal sequenceB.IongumN-terminalBSb-5206545.05same as Bsb-6same as Bsb-6same as Bsb-6BSb-6206545.05A(O)(T)(T)N D(I)K N G - L NMAOTTNDIKN GSVLNBSb-7269734.64SGHXKWAXXKHKXAAXDAMSGHSKWATTKHKKAAIDABSb-7269734.64SGHXKWAXXKHKXAAXDAMSGHSKWATTKHKKAAIDABSb-8132474.56AK F T Q D E L L E A FMAKYTNDELLEAF	Possible function		"elongation factor P"	62	50S RIBOSOMAL PROTEIN L7/L12
Name (kDa)MW (kDa)PlN-terminal sequenceBSb-5206545.05same as Bsb-6BSb-6206545.05A(Q)(T)(T)N D(I)K N G - L NBSb-72065734.64SGHXKWAXXKHKXAAXDABSb-7269734.64SGHXKWAXXKHKXAAXDABSb-8132474.56A K F T Q D E L L E A F	B. longum N-terminal	sequence same as Bsb-6	MAQTTNDIKN GSVLN	MSGHSKWATTKHKKAAIDA	MAKYTNDELLEAF
Name         MW (kDa)         PI           Bsb-5         20654         5.05           Bsb-6         20654         5.05           Bsb-7         26973         4.64           Bsb-8         13247         4.56	N-terminal sequence	same as Bsb-6	A(Q)(T)(T)N D(I)K N G L N	SGHXKWAXXKHKXAAXDA	AKFTQDELLEAF
Name         MW (kDa)           Bsb-5         20654           Bsb-6         20654           Bsb-7         26973           Bsb-8         13247	٩	5.05	5.05	4.64	4.56
Name Bsb-5 Bsb-6 Bsb-6 Bsb-7 Bsb-8	MM	20654	20654	26973	13247
· · · ·	Name	Bsb-5	Bsb-6	Bsb-7	Bsb-8

# Table 4.7 Comparison of optimum growth temperature and pH to published values.

	Values determined in this study for strain VUP Bb 12	Published values for <i>Bifidobacterium</i> spp. (Gomez and Malcata, 1999)
Optimum temperature	42°C	37-41°C
range		
Optimum pH	6.3 ± 0.2	6-7
Growth (temperature)		
range		
low limit	≈22°C	No growth below 25-28°C
high limit	Not determined in this study	No growth above 43-45°C
Growth (pH) range		
low limit	≈4.0	No growth below 4.5-5.0
high limit	Not determined in this study	No growth above 8.0-8.5

Data shown in section 4.3.1 is problematical in its interpretation given the relatively few points used to determine growth kinetics. However, later data (Figure 4.10) clearly showed that strain VUP Bb 12 was able to both initiate growth and grow at pH 4.0. This capacity could offer significant advantage over other strains when considering the selection of bifidobacteria for use in fermented dairy products.

The observations of extended growth range, reflect the origin and development of the test strain. As a dairy starter culture and adjunct, the strain has undergone a stringent selection process over the last 20 years. Strain VUP Bb 12 was specifically selected through a series of mutations and propagations to make it suitable for use in dairy products (Mogensen, 2002).

The specific selection criteria were:

- the ability to withstand the high acidity and temperatures associated with dairy fermentation processes
- good survival characteristics (in terms of product shelf life) and
- contribution to good flavour in fermented products.

In terms of being able to survive the temperatures and pH values experienced during the manufacture of dairy products, the findings of this study support the suitability of strain VUP Bb 12 as a dairy adjunct.

### 4.5.1.2 Growth kinetics

This study showed that when strain VUP Bb 12 is grown in a complex synthetic medium, such as MRS broth, in a fermenter where fine control is maintained over temperature, pH, agitation and gaseous environment,  $\mu_{max}$  values can reach 0.7 h<sup>-1</sup>.

Few reports have been made of the growth kinetics of Bifidobacterium species and in particular B. animalis (syn. lactis), as most research is targeted on survival in sub-optimum conditions rather than growth in optimum conditions. However, in a publication by Alander (2001) on the survival and persistence of B. animalis Bb-12 in the gastrointestinal tract, the growth kinetics of the strain was shown as a plot of OD versus time. In those growth curves, growth in TPY medium with 1% fructo-oligosaccharides or 1% gluco-oligosaccharides was compared to growth in TPY medium with 1% glucose (w/v). It was stated that cultures were grown in microtitre plates, at 37°C in anaerobic conditions and no other details were given. For cultures grown in 1% glucose, the growth curves showed a very long lag phase, extending nearly 12 hours. In this study lag phase lasted only one hour but this cannot be compared to the earlier study as the size of the inocula and prior growth conditions of the starter culture are unknown. Commencing at an OD of around 0.1 the final OD of the culture was 0.6 after 24 hours, which was not only much slower than the results achieved in this study but the extent of growth was also much less. These results are equally difficult to compare to the results from the study reported in this thesis, as the measurement of OD in microtitre plates cannot be directly compared to a measurement of OD in a spectrophotometer. The light path in a spectrophotometer is generally fixed at 1 cm while the volume of media in the microtitre plate well will determine the length of the light path and that will affect the OD values. Growth curves for the media containing oligosaccharides showed even longer lag phases and lower extents of growth compared to the glucose control (Alander *et al.*, 2001).

Other studies, using two commercial strains of *B. bifidum* (Bf-1 and Bf-6), have demonstrated mean doubling times of 237  $\pm$ 10 minutes and 242  $\pm$ 13 minutes respectively (Shin *et al.*, 2000).

In the present study, the calculated doubling time for *B. animalis* (syn. *lactis*) strain VUP Bb 12 grown in optimum conditions was  $56 \pm 1$  minute. Apart from variation due to strain differences,

the marked differences between the results may be accounted for in the differences in culture methods. In the study by Shin (2000) several factors would have contributed to a slower growth rate, such as:

- the bacteria were cultured in 12% (w/w) reconstituted non-fat dry milk, not MRS with cysteine-HCI. Milk as a growth medium has been shown to provide essential nutrients but the lack of amino acids and small peptides does not to support 'extended growth' (Rasic and Kurmann, 1983);
- the cultures were grown in tubes (no volumes were stated) and the tubes were not mixed during culture by nitrogen gas sparging or stirred agitation. This alone may have significantly extended the doubling time and
- the tubes were incubated using GasPaks, an anaerobic pouch incubation system. Poor gas diffusion in liquid media may have resulted in high oxygen levels remaining in the growth medium for an extended period, which would also extend the doubling time.

### 4.5.1.3 Observations of protein profiles of strain Bb 12 grown between 32 °C and 47 °C

The only differences observed in the protein profiles prepared from cultures grown between 32°C and 47°C were noted in the cultures grown at the highest temperature (47°C), which was 5°C above optimum.

When bacterial cells are shifted to a temperature significantly higher than the optimum, the synthesis of heat shock proteins is rapidly induced (Lindquist and Craig, 1988; Yura *et al.*, 1993). The typical response expected to be seen in 1-D SDS-PAGE profiles in Gram-negative bacteria undergoing heat stress would be an increase in proteins of molecular weight corresponding to induction of some or all of the heat shock proteins, including HSP 60, HSP 70, HSP 90 and small HSPs. Strain VUP Bb 12 did demonstrate a heat stress response: in particular, up-regulation of

the higher molecular weight proteins corresponding to induction of HSP 70 and HSP 60 and some small molecular weight proteins was detected in SDS-PAGE gels.

The sizes of the bands that were induced (70, 65, 40, 25 and 19 kDa) also corresponded to the size of proteins induced in other lactic acid bacteria during heat shock. Labelling of heat shock proteins in *Leuconostoc oenos*, using the more sensitive <sup>35</sup>S technique, has demonstrated induction of proteins of approximately 75 kDa, 64 kDa, 24 kDa, 18 kDa and 14.5 kDa (Guzzo *et al.*, 1994).

Antibody or labelling techniques, using <sup>35</sup>S labelling for example, may have detected minor changes in regulation of proteins related to HSP 90 and HSP 70, but were not performed in the present study. Western blotting techniques were also not used to confirm that the 65 kDa protein reacted with antibodies produced against HSP 60.

Only a few reports have been made of heat shock in lactic acid bacteria including (Whitaker and Batt, 1991; Arnau *et al.*, 1996; Hartke *et al.*, 1997; Carey Walker *et al.*, 1999). In *Lb. johsonii* the groESL operon has been sequenced and transcription activity studied (Carey Walker *et al.*, 1999). Maximum transcription activity in this strain, VPI 11088, was found to occur at 55°C. The authors attributed the relatively high temperature of expression to the presence of two CIRCE elements, which they felt would contribute a stronger level of negative regulation. It may be possible that dual CIRCE elements are present in heat shock operons of other lactic acid bacteria, including strain VUP Bb 12. If that is so then the most suitable conditions for induction of the high molecular weight heat shock proteins may not have been tested, as the test range in this study was limited to 47°C, a temperature where growth was strongly inhibited. Future experiments would need to look at different temperatures or heat shock regimes and more sensitive techniques to observe any changes. Sequencing of the operons would also need to be performed to confirm the presence of dual CIRCE elements.

In *L. lactis* subsp. *lactis* LM0230 heat shock proteins were observed after a temperature shift from 30°C to 42°C (Whitaker and Batt, 1991). Two of the up-regulated proteins were homologues of HSP 60 and HSP 70. Other up-regulated proteins were found to have molecular weights of 93, 76, 67, 60, 51, 49, 42, 40, 28 and 24 kDa, some of which correspond to proteins detected in this study and the study by Guzzo (1994).

# 4.5.1.4 Observations of protein profiles of strain Bb 12 cultures grown between pH 4 and pH 7

Marked up-regulation of a protein of approximately 33 kDa was a feature of SDS-PAGE gels of whole cells of strain VUP Bb 12 grown under strongly acidic conditions. A 14-fold increase was observed in the relative amount of the protein present in cultures grown at pH 4.0. To my knowledge, the marked up-regulation of this protein has not previously been reported in the literature for this species. The same molecular weight protein was also noted to be faintly up-regulated during growth at pH 5.0 and strongly up-regulated during growth in a fermenter with no pH control. No increase in regulation was observed in cultures grown at pH 6.0 or 7.0.

A second protein of 36 kDa was also up-regulated during growth at low pH, although not as strongly as the 33 kDa protein; however, the relative amount of the two proteins appears linked as they appear to increase at the same rate. The 33 kDa protein was also observed in 2-D SDS-PAGE after solvent extraction using chloroform and diethyl ether. This is discussed in section 4.5.2.2.

### 4.5.1.5 Identification of 33 kDa protein up-regulated during acid stress

N-terminal sequencing was unable to provide data that could assist in the identification of the 33 kDa protein seen during growth at pH 4.3. Observations of 2-D SDS PAGE gels showed that the protein was not present as a discrete spot; it appeared as a large smear, and it was not possible to a have a second attempt at N-terminal sequencing because the protein concentration was not high enough in the smear. The unusual appearance of the protein in these gels may be due to

the presence of a carbohydrate or lipid bound to the protein that interferes with electrophoresis. The presence of carbohydrate or lipid could be detected through modified staining methods; for example glycoproteins can be detected using a modified Periodic Acid-Sciff method. However, these stains were not performed as part of this work and would be included in future studies.

# 4.5.1.6 Observations of protein profiles and growth kinetics of strain Bb 12 cultures grown in a fermenter sparged with instrument air

A study by Beerens (2000) suggested that *Bifidobacterium* spp. derived from animal sources were more likely to be tolerant of exposure to air than *Bifidobacterium* spp. derived from human sources. In this study, while no major changes were observed in protein profiles grown in the presence of instrument air compared to a control, several minor proteins were down-regulated, suggesting that aeration stress resulted in sub-lethal cellular damage only and strain VUP Bb 12 could be considered tolerant to exposures of this type.

While considered obligate anaerobes, *Bifidobacterium* species show considerable variation in the growth response and metabolism in the presence of oxygen (Shimamura *et al.*, 1992). Few studies have been made of oxidative stress in LAB, but these have included *Lactobacillus* and *Lactococcus* species (Duwat *et al.*, 1995; Marty-Teysset *et al.*, 2000; Guerzoni *et al.*, 2001). In a study of oxidative stress in *Lb. delbrueckii*, it was determined that aeration of cultures did not affect the  $\mu_{max}$  values achieved during growth, but aerated cultures entered stationary phase much earlier than non-aerated cultures (Marty-Teysset *et al.*, 2000). The authors attributed this to an increase of H<sub>2</sub>O<sub>2</sub> in the growth medium due to the presence of the enzyme NADH oxidase. This enzyme has also been reported in other lactobacilli and some streptococci, but not in bifidobacteria. In the present study entry into stationary phase was delayed by exposure to instrument air and the  $\mu_{max}$  value was lower than for the control, which suggests that a similar mechanism was not present in strain VUP 12006.

# 4.5.2 Characterisation of *B. animalis* (syn. *lactis*) strain VUP Bb 12 during acid stress

The affect of a sudden decrease in pH or continuous exposure to low pH was determined for strain VUP Bb 12. The use of a citrate phosphate buffer was incorporated into the methodology so that up to seven different environmental conditions could be tested at the same time. Preliminary experiments indicated that the buffer did not affect the growth rate of strain VUP Bb 12. However, some decrease in growth rate did occur, compared to growth rates in fermenters, due probably to mixing on a rotating platform, which was not as effective as agitation achieved through gas sparging and the use of a propeller stirrer.

## 4.5.2.1 Effect of decrease in pH on growth kinetics, viable cell counts and carbohydrate utilisation

Several observations were made of the effect of a sudden decrease in pH or continuous exposure to low pH on growth kinetics, viable cell counts and carbohydrate utilisation and they are discussed in this section.

In terms of an increase in OD<sub>600</sub>, cultures grown continuously at pH 5.3 performed nearly as well as cultures grown at pH 6.3 over the first 8 hours. This was reflected in the carbohydrate utilisation studies, where glucose use and lactate production, were very similar for the two cultures. It was also noted that entry into stationary phase did not occur due to lack of a carbon source, as well over half of the initial glucose supplied, was still present in the medium after twenty-four hours incubation. Growth at pH 5.3 did not appear to have any effect on the relative proportion of acetate and lactate produced, compared to growth at pH 6.3. However, at pH 4.3 production of lactate and acetate did not occur.

Literature reviews had suggested that changes in growth conditions would affect the relative amounts of acetate and lactate production (Sgorbati *et al.*, 1995). The theoretical ratio of acetate to lactate should be 1.5 to 1, based on known bifidobacterial glucose metabolism, and it has been

observed, in *B. lactis* strain UR1, that oxygen stress has resulted in a higher ratio of acetate to lactate with an associated decrease in formate (Meile *et al.*, 1997). Changes of this nature were not observed in this limited trial for either growth at low pH or with aeration.

Cultures grown at pH 4.3 demonstrated a comparatively strong growth rate in the first five hours and the viable cell count increased by nearly one order of magnitude. However, there was little or no growth for the next nineteen hours, as the culture appeared to go into stationary phase. It was not determined exactly at what point during the last nineteen hours of growth that viable cells could not be detected; however the viable cell count dropped over four orders of magnitude in that time. Significantly, during the 24 hours of growth, significant amounts of glucose were not utilised, nor were any significant amounts of lactate produced, despite the observation that viable cells could be detected. These results showed that growth at low pH severely affected the cells and growth was not significant. These cells also failed to survive at pH 4.3 overnight.

The major difference between cultures grown at pH 6.3, 5.3 and 4.3 was seen in the final viable cell counts. The twenty-four hour viable cell count for cultures grown at pH 5.3 and 4.3, showed two-orders and four-orders of magnitude loss of viability, respectively, over the previous sixteen hours, while the viable cell count of the control remained essentially unchanged. The exact mechanism to explain why cells grown at pH 5.3 and 4.3 lose viability so rapidly is not apparent from these studies but is possibly related to proton permeation of the cell membrane. The apparent shift into stationary phase also appears to be part of a protective mechanism for this strain when they are exposed to low pH environments with subsequent induction of the general stationary phase response, changes to fatty acid profiles and changes in cell morphology: all of which would contribute to protection of cells at low pH.

Cultures that were exposed to a sudden decrease in pH during growth did not achieve similar  $OD_{600}$  readings to those of cultures grown continuously at the same pH. Clearly, sudden

decreases in pH had a more detrimental effect on growth rates than continuous exposure to low pH, but the explanation may be that it was not the sudden shift that affected growth as much as the growth phase in which it occurred. The sudden change in pH occurred during the exponential phase of growth, the phase of growth where *Lb. acidophilus* strain 12006 was shown to be more acid-sensitive. Cultures exposed continuously to low pH were exposed from lag phase, and may well have adapted to these conditions, which is consistent with the production of the 33 kDa protein in cultures where pH was uncontrolled and in pH 4.3-grown cultures.

There appeared to be some degree of contradiction in the results observed when determining if a brief exposure to pH 5.3 would make a culture more acid-tolerant to pH 4.3. In terms of final OD<sub>600</sub>, cultures exposed to a brief exposure at pH 5.3 prior to exposure to pH 4.3 did not appear to perform better than cultures shifted directly from pH 6.3 to 4.3. In contrast, after twenty-four hours continuous growth at pH 4.3, no viable cells could be detected; but low numbers of viable cells could be detected after twenty-four hours in cultures that were stepped down through pH 5.3. It can be seen from the results presented in the previous section that growth curves based on OD<sub>600</sub> do not reflect viable cell counts and that viable cell counts better reflect the survival of cells at low pH.

A similar limitation of the experimental protocol was appreciated when trying to interpret the results that would indicate whether pre-acid-adapted starters performed better than control cultures prepared from starters grown at pH 6.3. While changes in OD indicate there was no difference between acid-adapted and non-adapted starter cultures, viable cell counts clearly indicated better growth and survival. This is strongly suggestive that using an acid-adapted culture is protective against subsequent growth at low pH. Clearly more work would be needed to confirm these observations. In the present study no attempt was made to optimise the growth conditions with regard to the growth phase when cultures were stepped down through pH 5.3 and

the cultures were permitted to grow less than one generation before further exposure to pH 4.3. Both of these conditions would have to be considered in future experimental design.

### 4.5.2.2 Effect of growth at low pH on protein profiles

Several difficulties were experienced when using SDS-PAGE profiles to study changes in protein regulation during growth at low pH. Changes in protein profile as determined by 1-D SDS-PAGE failed to show any further protein induction following sudden shift in pH so it was decided to switch to 2-D SDS-PAGE, where changes in lower-abundance proteins could be examined. As noted in the results section, the immediate disadvantage of the switch was that the 33 kDa protein could not be detected in samples prepared by standard preparation of cytosolic proteins by bead-beating and subsequent centrifugation of cellular debris. This suggested that the 33 kDa protein was a cell surface or membrane protein and was among the 30% of proteins that are calculated to be not detected by routine 2-D SDS-PAGE (Santoni et al., 2000). Several attempts were made to improve the representation of the 33 kDa protein in 2-D gels. Detergent extraction, one of the methods commonly suggested in the literature, did solubilise more proteins in the presence of CHAPS and Triton X-100, but did not appear to be a candidate method for use in 2-D SDS-PAGE as the 1-D profile was not changed significantly. More success was obtained using solvents to extract hydrophobic proteins, although the 33 kDa protein was never detected as a discrete spot in 2-D gels and more often appeared as a smudge. However, further sequencing was not pursued because insufficient protein was transferred to PVDF membrane for detection by Amido Black 10B.

Methods for extracting cell-surface proteins were developed. In general, solvent extraction appears to have resulted in two profile types. Octanol and chloroform produced proteins of high molecular weight and profiles with few bands. The remaining solvents studied in this work produced profiles that contained many of the bands seen in the gel when proteins were extracted from whole cells into Tris buffer, but the relative abundance of the bands changed according to

the solvent type. For 2-D SDS-PAGE the best results were obtained from mixing equal volumes of chloroform- and di-ethyl ether-extracted proteins after the extracted proteins had been redissolved in Tris buffer. It was found that mixing the solvents together for use as an extraction solution resulted in very low yields of proteins and although the reason for this is unknown it is possible that mixing the solvents together caused the proteins to precipitate or complex with the cell debris. Whatever the cause, there was insufficient protein to run the sample by 1-D SDS-PAGE or in the first dimension of 2-D SDS-PAGE. This problem was solved by making separate extractions and making up the dried extracts in Tris buffer subsequently.

Results of 2-D SDS-PAGE using solvent-extracted proteins demonstrated the usefulness of the technique for showing changes in hydrophobic proteins during growth at pH 4.3 when compared to pH 6.3. The pI and estimated size of the solvent-extracted proteins up-regulated during growth at pH 4.3 were, with one exception, different to that obtained from cytosolic proteins. However, a protein concentration step would be required if these solvent extracted samples were to be used to determine the N-terminal sequence of up-regulated proteins.

### 4.5.2.3 Function of up-regulated cytosolic proteins

The function of the three proteins up-regulated during growth at pH can, in part, be deduced from the function of closely related proteins. A BLASTP search of Genbank revealed that the closest matches to all three proteins had very similar pl and molecular weight to that observed by 2-D SDS-PAGE.

Protein Bsb-6 was found to have sequence similarity to elongation factor P or EF-P. EF-P is associated with elongation of protein biosynthesis. In particular it serves as a catalyst in the formation of the very first peptide bond of bacterial proteins (Aoki *et al.*, 1997). In *E. coli*, EF-P is thought to affect the affinity of the ribosome for charged tRNA, thereby facilitating peptide synthesis. The protein Bsb-7 was found to have sequence homology to a number of hypothetical

proteins based on database matching to the completed genomes of *B. longum*, *Corynebacterium glutamicum* and *Mycobacterium leprae*. The function of these hypothetical proteins could not be determined.

Bsb-8 was found to have homology to the N-terminal sequence from the 50S ribosomal protein L7/L12. The 50S sub-unit function is related to formation of peptide bonds: in particular, it catalyses the peptidyl transferase reaction of protein synthesis. The ribosomal protein L7/L12 is an important component of this sub-unit and in *E. coli* is present in four copies organized as two dimers. The L7/L12 dimer is thought to interact with elongation factors as part of translation elongation indicating a possible relationship to Bsb-6 (a potential elongation factor).

The possible function of both Bsb-6 and 8 appear to be closely related to the protein synthesis, in particular ribosomal function. The presence of these proteins may be related to a number of factors, including production of novel proteins that are essential for growth during acid stress, or because the growth phases are out of step. In the first case, it is possible the cells are making proteins, like the 33 kDa protein seen during growth at pH 4.3, because they are essential for cell viability. In the second, the increase in ribosomal function could be related to the fact that cells grown at pH 6.3 have reached stationary phase, while those at pH 4.3 are potentially still in early exponential phase, refer to Figure 4.10.

### 4.5.2.4 Areas for further study

There are several areas that could be considered for further study. Firstly, in order to obtain the complete nucleotide sequence of the gene encoding the 33 kDa protein in *B. animalis* (syn. *lactis*), it would be of interest to obtain amino acid sequence data for that protein. While further attempts could be made at obtaining the N-terminal sequence using the traditional Edman degradation reaction, an alternative approach would be to obtain a peptide-mass fingerprint from a proteolytic digest of purified protein. Briefly, a trypsin digest would be performed on a spot cut

from an acrylamide gel (1-D) corresponding to the location of the 33 kDa protein. Then, using MALDI-TOF, an accurate mass of the peptides would be obtained and compared to data in protein databanks obtained from proteins digested by the same enzyme. Where this approach was insufficient further N-terminal or C-terminal data may be able to be obtained using PSD (post-source dissociation) or CID (collision-induced dissociation). Once sequence data had been obtained a similar approach to that used in Chapter 3 would be followed to obtain the sequence of the gene encoding this protein.

Other areas of further study could also include the following:

- Preparation of antibodies against the 33 kDa protein identified during growth at low pH in B. animalis (syn. lactis) strain VUP Bb 12. It would be useful to monitor the regulation of the gene through western blot analysis. These antibodies could also be used to determine the presence of analogues of the gene in related strains, related species and non-related species.
- Determination of the sub-cellular location of the 33 kDa protein. This work would assist in understanding the function of the protein.
- Further studies are also required to identify the proteins up-regulated during heat stress in *B. animalis* (syn. *lactis*) Bb 12. A different approach, using a lower challenge temperature, would be used to see if a stronger heat shock response could be initiated. Although heat shock studies could include western blot analysis, alternative approaches could involve N-terminal sequencing or peptide mass fingerprinting of 2-D SDS-PAGE spots.

### 4.6 Conclusion

- 16S rRNA was able to confirm strain Bb12 as *B. animalis* (syn.) *lactis*.
- *B. animalis* (syn.) *lactis* grew optimally at  $42^{\circ}C \pm 0.2$  and at pH 6.3 ± 0.2.

- When heat stressed at 47°C, *B. animalis* (syn.) *lactis* did not demonstrate a classic heat stress response.
- When acid stressed, *B. animalis* (syn.) *lactis* produced a protein of approximately 33 kDa which was found to be cell-membrane- or cell-surface-associated. The N-terminal sequence of this protein could not be determined.
- Cytosolic proteins Bsb-6 and Bsb-8 were N-terminally sequenced and identified as being functionally related to elongation factor-P and 50S ribosomal protein L7/L12 respectively.

# REFERENCES

- Abdelal, A. (1979). "Arginine catabolism by microorganisms." <u>Annual Review of Microbiology</u> 33: 139-168.
- Adhikari, K., A. Mustapha, I. Grün and L. Fernando (2000). "Viability of microencapsulated bifidobacteria in set yoghurt during refrigerated storage." <u>Journal of Dairy Science</u> 83: 1946-1951.
- Alander, M., J. Mättö, W. Kneifel, M. Johansson, B. Kögler, R. Crittenden, T. Mattila-Sandholm and M. Saarela (2001). "Effect of galacto-oligosaccharide supplementation on human faecal microflora and on survival and persistance of *Bifidobacterium lactis* bb-12 in the gastrointestinal tract." <u>International Dairy Journal</u> 11: 817-825.
- Albright, L., E. Huala and F. Ausbel (1989). "Prokaryotic signal transduction mediated by sensor and regulator protein pairs." <u>Annual Review of Genetics</u> 23: 311-336.
- Amann, R., W. Ludwig and K.-H. Schleifer (1994). "Identification of uncultured bacteria: A challenging task for molecular taxonomists." <u>ASM news</u> 60: 360-365.
- Ananthan, J., A. Goldberg and R. Voellmy (1985). "Abnormal proteins serve as eukaryotic stress signals and trigger the activation of heat shock genes." <u>Science</u> 232: 522-524.
- Aoki, H., K. Dekany, S. Adams and M. Ganoza (1997). "The gene encoding the elongation factor p protein is essential for viability and is required for protein synthesis." <u>Journal of Biological</u> Chemistry 272(51): 32254-32259.
- Arnau, J., K. Sørenson, K. Appel, F. Vogensen and K. Hammer (1996). "Analysis of heat shock gene expression in *Lactococcus lactis* MG1363." <u>Microbiology</u> 142: 1685-1691.
- Auger, A., K. E. Redding, T. Plumb, L. Childs, H.-Y. Meng and G. N. Bennett (1989).
   "Construction of *lac* fusions to the inducible arginine- and lysine decarboxylase genes of *Escherichia coli* K12." <u>Molecular Microbiology</u> 3(5): 609-620.

- Augustyn, O., J. Kock and D. Ferreira (1990). "Differentiation between yeast species, and strains within a species by cellular fatty acid analysis 3. *Saccharomyces sensu lato, arxiozyma* and *pachytichospora.*" <u>Systematic and Applied Microbiology</u> 13: 44-55.
- Axelsson, L. (1993). Lactic acid bacteria: Classification and physiology. <u>Lactic acid bacteria</u>. New York, Marcel Dekker Inc.
- Baik, H., S. Bearson, S. Dunbar and J. Foster (1996). "The acid tolerance response of *Salmonella typhimurium* provides protection against organic acids." <u>Microbiology</u> 142(11): 3195-3200.
- Baldini, R., M. Avedissian and S. Gomes (1998). "The CIRCE element and its putative repressor control cell cycle expression of the Caulobacter crescentus groESL operon." <u>Journal of</u> <u>Bacteriology</u> 180(7): 1632-1641.
- **Baldwin, R.** (1989). "How does protein folding get started?" <u>Trends in Biochemical Sciences</u> 14: 291-294.
- Baumeister, W., J. Walz, F. Zühl and E. Seemüller (1998). "The proteasome: Paradigm of a selfcompartmentalizing protease." <u>Cell</u> 92: 367-380.
- Bearson, S., B. Bearson and J. Foster (1997). "Acid stress responses in enterobacteria." <u>FEMS</u> <u>Microbiology Letters</u> 147: 173-180.
- Becker, J. and E. Craig (1994). "Heat shock proteins as molecular chaperones." <u>European Journal</u> of Biochemistry 219: 11-23.
- Beerens, H., F. Gavini and C. Neut (2000). "Effect of exposure to air on 84 strains of bifidobacteria." <u>Anaerobe</u> 6: 65-67.
- Belli, W. A. and R. E. Marquis (1991). "Adaption of *Streptococcus mutans* and *Ent. hirae* to acid stress in continuous culture." <u>Applied and Environmental Microbiology</u> 57(4): 1134-1138.
- Benkirane, R., M. Gottschalk and D. Dubreuil (1997). "Identification of a *Streptococcus suis* 60-kda heat-shock protein using western blotting." <u>FEMS Microbiology Letters</u> 153: 379-385.

- Blignaut, E., R. Senekal, J. Kock, A. Botha and J. van der Westhuizen (1996). "The value of cellular fatty acid analysis in the identification of oral yeast." <u>Systematic and Applied</u> <u>Microbiology</u> 19: 381-387.
- Boot, H., C. Kolen, F. Andreadaki, R. Leer and P. Pouwels (1996). "The *Lactobacillus acidophilus* S-layer protein gene expression site comprises two consensus promoter sequences, one of which directs transcription of stable mRNA." *Journal of Bacteriology* 178(18): 5388-5394.
- Boot, H., C. Kolen and P. Pouwels (1995). "Identification, cloning and nucleotide sequence of a silent s-layer protein gene of *Lactobacillus acidophilus* ATCC 4356 which has extensive similarity with the s-layer protein gene of this species." <u>Journal of Bacteriology</u> 177(24): 7222-7230.
- Boot, H., C. Kolen, J. van Noort and P. Pouwels (1993). "S-layer protein of Lactobacillus acidophilus ATCC 4356: Purification, expression in Escherichia coli, and nucleotide sequence of the corresponding gene." Journal of Bacteriology 175(19): 6089-6096.
- **Booth, I. R.** (1985). "Regulation of cytoplasmic pH in bacteria." <u>Microbiological Reviews</u> 49(4): 359-378.
- Boston, R., P. Viitanen and E. Vierling (1996). "Molecular chaperones and protein folding in plants." Plant Molecular Biology 32: 191-222.
- Broadbent, J., C. Orberg and L. Wei (1998). "Characterisation of the *Lactobacillus helveticus* groELS operon." <u>Research in Microbiology</u> 149: 247-253.
- Brown, J., T. Ross, T. McMeekin and P. Nichols (1997). "Acid habituation of *Escherichia coli* and the potential role of cyclopropane fatty acids in low pH tolerance." <u>International Journal of</u> Food Microbiology 37: 163-173.
- Bukau, B. and A. Horwich (1998). "The hsp70 and hsp60 chaperone machines." Cell 92: 351-366.

- Cai, Y., M. Matsumoto and Y. Benno (2000). "*Bifidobacterium lactis* Meile *et al.* 1997 is a subjective synonym of *Bifidobacterium animalis* (Mitsuoka 1969) Scardovi and Trovatelli 1974." <u>Microbiology and Immunology</u> 44(10): 815-820.
- Canducci, F., F. Cremonini, A. Armuzzi, S. Di Caro, M. Gabrielli, L. Santarelli, E. Nista, A. Lupascu, D. De Martini and A. Gasbarrini (2002). "Probiotics and *Helicobacter pylori* eradication." <u>Digestive and Liver Disease</u> 34: S81-S83.
- Canganella, F., M. Gasbarri, S. Massa and L. Trovatelli (1996). "A microbiological investigation on probiotic preparations used for animal feeding." <u>Microbiological Research</u> 151: 167-175.
- Carey Walker, D., H. Girgis and T. Klaenhammer (1999). "The *gro*ESL chaperone operon of *Lactobacillus johnsonii*." <u>Applied and Environmental Microbiology</u> 65(7): 3033-3041.
- Castellanos, M., A. Chauvet, A. Deschamps and C. Barreau (1996). "PCR methods for identification and specific detection of probiotic lactic acid bacteria." <u>Current Microbiology</u> 33: 100-103.
- **Chambers, J.** (1993). Buffers, chelating agents and denaturants. <u>Biochemistry labfax</u>. J. Chambers and D. Rickwood. Oxford, BIOS Scientific: 1-35.
- Chia, J., L. Chang, C. Shun, Y. Chang, Y. Tsay and J. Chen (2001). "A 60-kilodalton immunodominant glycoprotein is essential for cell wall integrity and the maintenance of cell shape in *Streptococcus mutans*." <u>Infection and Immunity</u> 69(11): 6987-6998.
- Clark, P., L. Cotton and J. Martin (1993). "Selection of bifidobacteria for use as dietary adjuncts in cultured dairy foods: II-tolerance to simulated pH of human stomachs." <u>Cultured Dairy</u> <u>Products Journal</u> November: 11-14.
- Clark, P. and J. Martin (1994). "Selection of bifidobacteria for use as dietary adjuncts in cultured dairy foods:III-tolerance to simulated bile concentrations of human small intestines." <u>Cultured</u> Dairy Products Journal August: 18-21.

- **Coll, J., M. Heyde and R. Portelier** (1994). "Expression of the *nmpc* gene of *Escherichia coli* K-12 is modulated by external pH. Identification of cis-acting regulatory sequences involved in this region." <u>Molecular Microbiology</u> 12(1): 83-93.
- Collins, M., B. Phillips and P. Zanoni (1989). "Deoxyribonucleic acid homology studies of Lactobacillus paracasei sp. Nov., subsp. paracasei and subsp. tolerans and Lactobacillus rhamnosus sp. Nov comb. Nov." International Journal of Systematic Bacteriology 39: 105-108.
- Collins, M. D., U. Rodrigues, C. Ash, M. Aguirre, J. Farrow, A. Martinez-Murcia, B. Phillips, A. Williams and S. Wallbanks (1991). "Phylogenetic analysis of the genus *Lactobacillus* and related lactic acid bactería as determined by reverse transcriptase sequencing of 16s rRNA." <u>FEMS Microbiology Letters</u> 77: 5-12.
- Condon, S. (1987). "Responses of lactic acid bacteria to oxygen." <u>FEMS Microbiology Reviews</u> 46: 269-280.
- Cook, G. and J. Russell (1994). "The effect of extracellular pH and lactic acid on pH homeostasis in Lactococcus lactis and Streptococcus bovis." <u>Current Microbiology</u> 28: 165-168.
- **Correa, O. S. and A. J. Barneix** (1997). "Cellular mechanisms of pH tolerance in Rhizobium loti." World Journal of Microbiology and Biotechnology 13: 153-157.
- Couto, J., N. Rozes and T. Hogg (1996). "Ethanol-induced changes in the fatty acid composition of *Lactobacillus hilgardii*, its effects on plasma membrane fluidity and relationship with ethanol tolerance." Journal of Applied Bacteriology 81(2): 126-132.
- Craig, E. (1995). Chaperones, molecular. Molecular biology dictionary: 162-165.
- Craig, E., B. Gambill and R. Nelson (1993). "Heat shock proteins: Molecular chaperones of protein biogenesis." <u>Microbiological Reviews</u> 57: 402-414.
- Craig, E. and C. Gross (1991). "Is HSP 70 the cellular thermometer?" <u>Trends in Biochemical</u> <u>Sciences</u> 16: 135-140.

- Creighton, T. (1995). Protein folding. <u>Molecular biology and biotechnology: A comprehensive desk</u> reference. R. Meyers. New York, VCH Publishers: 754-757.
- Creighton, T. (1997). "How important is the molten globule for correct protein folding?" <u>Trends in</u> <u>Biochemical Sciences</u> 22: 6-10.
- Crittenden, R., L. Morris, M. Harvey, L. Tran, H. Mitchell and M. Playne (2001). "Selection of a *Bifidobacterium* strain to complement resistant starch in a synbiotic yoghurt." <u>Journal of</u> <u>Applied Microbiology</u> 90: 268-278.
- Crociani, F., B. Biavati, A. Alessandri, C. Chiarini and V. Scardovi (1996). "*Bifidobacterium inopinatum* sp. *nov.* and *Bifidobacterium denticolens* sp. nov., two new species isolated from human dental caries." <u>International Journal of Systematic Bacteriology</u> 46(2): 654-571.
- Curran, T., J. Lieou and R. Marquis (1995). "Arginine deiminase system and acid adaptation of oral streptococci." <u>Applied and Environmental Microbiology</u> 61(12): 4494-4496.
- **De Angelis, M., L. Bini, V. Pallini, P. Cocconcelli and M. Gobbetti** (2001). "The acid-stress response in *Lactobacillus sanfranciscensis* CB1." <u>Microbiology</u> 147: 1863-1873.
- **de Koning-Ward, T. and R. Robins-Browne** (1995). "Contribution of urease to acid tolerance in *Yersinia enterocolitica.*" Infection and Immunity 63(10): 3790-3795.
- De Man, J., M. Rogosa and M. Sharpe (1960). "A medium for the cultivation of lactobacilli." <u>Journal</u> of Applied Bacteriology 23: 130-136.
- **Decallone, J., M. Delmee, P. Wauthoz, M. El Lioui and R. Lambert** (1991). "A rapid procedure for the identification of lactic acid bacteria based on the gas chromatographic analysis of cellular fatty acids." Journal of Food Protection 54(3): 217-224.
- Dellaglio, F., L. Dicks, M. Du Toit and S. Torriani (1991). "Designation of ATCC 334 in place of ATCC 393 as the neotype strain of *Lactobacillus casei* subsp. *casei* and rejection of the name *Lactobacillus paracasei*." <u>International Journal of Systematic Bacteriology</u> 41: 340-342.

- Dellaglio, F., G. Felis and S. Torriani (2002). "The status of the species Lactobacillus casei (Orla-Jensen1916) Hansen and Lessel 1971 and Lactobacillus paracasei Collins et al. 1989.
   Request for an opinion." International Journal of Systematic and Evolutionary Microbiology 52: 285-287.
- **Demple, B.** (1991). "Regulation of bacterial oxidative stress genes." <u>Annual Review of Genetics</u> 25: 315-337.
- Dicks, L., E. Du Plessis, F. Dellaglio and E. Lauer (1996). "Reclassification of Lactobacillus casei subsp. casei ATCC 393 and Lactobacillus rhamnosus ATCC 15820 as Lactobacillus zeae nom. rev., designation of ATCC 334 as the neotype of L. casei subsp. casei and rejection of the name Lactobacillus paracasei." International Journal of Systematic Bacteriology 46: 337-340.
- **Dicks, L., B. Janssen and F. Dellaglio** (1995). "Differentiation of *Carnobacterium divergens* and *Carnobacterium piscicola* by numerical analysis of total soluble protein patterns and DNA-DNA hybridizations." <u>Current Microbiology</u> 31: 77-79.
- **Djordjevic, G., B. Bojovic, N. Miladinov and L. Topisiric** (1997). "Cloning and molecular analysis of promoter-like sequences isolated from the chromosomal DNA of *Lactobacillus acidophilus* ATCC 4356." Canadian Journal of Microbiology 43: 61-69.
- Drasar, B., M. Shiner and G. McLeod (1969). "Studies on the intestinal flora. 1. The bacterial flora of the gastrointestinal tract in healthy and achlorhydric persons." <u>Gastroenterology</u> 56(1): 71-79.
- Du Plessis, E., L. Dicks, M. Vescovo, S. Torriani and F. Dellaglio (1996). "Lactobacillus acidophilus and related species: A review." <u>Annali di Microbiologica ed Enzimologia</u> 46: 319-340.
- Dubernet, S., N. Desmasures and M. Gueguen (2002). "A PCR-based method for identification of lactobacilli at the genus level." <u>FEMS Microbiology Letters</u> 214: 271-275.

- **Duncan, H. and S. Edberg** (1995). "Host-microbe interaction in the gastrointestinal tract." <u>Critical</u> <u>Reviews in Microbiology</u> 21(2): 85-100.
- Duwat, P., R. de Oliveira, D. Ehrlich and S. Boiteux (1995). "Repair of oxidative DNA damage in gram-positive bacteria: The *Lactococcus lactis fpg* protein." <u>Microbiology</u> 141: 411-417.
- Duwat, P., S. Ehrlich and S. Gruss (1992). "Use of degenerate primers for polymerase chain reaction cloning and sequencing of the *Lactococcus lactis* subsp. *lactis recA* gene." <u>Applied</u> <u>and Environmental Microbiology</u> 58(8): 2674-2678.
- **Dykes, G. and A. von Holy** (1993). "Taxonomy of lactic acid bacteria from spoiled, vacuum packaged Vienna sausages by total soluble protein profiles." <u>Journal of Basic Microbiology</u> 33: 169-177.
- Ebringer, L., M. Ferencik, N. Lahitová, L. Kacáni and D. Michálková (1995). "Anti-mutagenic and immuno-stimulatory properties of lactic acid bacteria." <u>World Journal of Microbiology and Biotechnology</u> 11: 294-298.
- **Eichel, J., Y.-Y. Chang, D. Riesenberg and J. Cronan** (1999). "Effect of ppGpp on *Escherichia coli* cyclopropane fatty acid synthesis is mediated through the rpos sigma factor ( σ<sup>s</sup>)." <u>Journal</u> <u>of Bacteriology</u> 181(2): 572-576.
- Ellis, R. (1993). The general concept of molecular chaperones. <u>Molecular chaperones</u>. R. Ellis, R. Laskey and G. Lorimer. London, Chapman and Hall.
- Ellis, R. (1999). "Principles of protein folding in the cellular environment." <u>Current Opinion in</u> <u>Structural Biology</u> 9: 102-110.
- Ellis, R. (2000). "Chaperone substrates inside the cell." <u>Trends in Biochemical Sciences</u> May: 210-212.
- Ellis, R. (2001). "Macromolecular crowding: An important but neglected aspect of the intracellular environment." <u>Current Opinion in Structural Biology</u> 11: 114-119.
- Ellis, R., J. Martin, M. Mayhew and F. U. Hartl (1996). The chaperonins. London, Academic Press.

- Euzeby, J. (2003). List of bacterial names with standing in nomenclature-genus *Lactobacillus*, J. P. Euzeby. 2003.
- Evans, R., P. McClure, G. Gould and N. Russell (1998). "The effect of growth temperature on the phospholipid and fatty acyl composition of non-proteolytic *Clostridium botulinum*." <u>International Journal of Food Microbiology</u> 40: 159-167.
- Farber, J. and F. Pagotto (1992). "The effect of acid shock on the heat resistance of *Listeria monocytogenes*." Letters in Applied Microbiology 15: 197-201.
- Faust, E. (1910). "Über die verwendbarket der milchsaure als bestandteil von genßmitteln." Cherniker-Zeitung 34(8): 57-60.
- Fayet, O., T. Ziegelhoffer and C. Georgopoulos (1989). "The groES and groELI heat shock gene products are essential for bacterial growth at all temperatures." <u>Journal of Bacteriology</u> 171: 1379-1385.
- Ferianc, P., A. Farewell and T. Nyström (1998). "The cadmium-stress stimulon of *Escherichia coli K*-12." <u>Microbiology</u> 144: 1045-1050.
- Fernandes, C., K. Shahani and M. Amer (1987). "Therapeutic role of dietary lactobacilli and lactobacillic fermented dairy products." <u>FEMS Microbiology Reviews</u> 46: 343-356.
- Ferro, M., D. Seigneurin-Berny, N. Rolland, A. Chapel, D. Salvi, J. Garin and J. Joyard (2000). "Organic solvent extraction as a versatile procedure to identify hydrophobic chloroplast membrane proteins." <u>Electrophoresis</u> 21: 3517-3526.
- Finlay, B. and S. Falkow (1997). "Common themes in microbial pathogenicity revisited." <u>Microbiology and Molecular Biology Reviews</u> 61(2): 136-169.
- Flahaut, S., A. Hartke, J.-C. Giard, A. Benachour, P. Boutibonnes and Y. Auffray (1996).
  "Relationship between stress response towards bile salts, acid and heat treatment in Enterococcus faecalis." <u>FEMS Microbiology Letters</u> 138: 49-54.

- Flahaut, S., J. Laplace, J. Frere and Y. Auffray (1998). "The oxidative stress response in Enterococcus faecalis: Relationship between H<sub>2</sub>O<sub>2</sub> tolerance and H<sub>2</sub>O<sub>2</sub> stress proteins." <u>Letters in Applied Microbiology</u> 26: 259-264.
- Fooks, L., R. Fuller and G. Gibson (1999). "Prebiotics, probiotics and human gut microbiology." International Dairy Journal 9: 53-61.
- Foster, J. (1991). "Salmonella acid shock proteins are required for the adaptive acid tolerance response." Journal of Bacteriology 173: 6896-6902.
- **Foster, J.** (1993). "The acid tolerance response of *Salmonella typhimurium* involves transient synthesis of key acid shock proteins." <u>Journal of Bacteriology</u> 175(7): 1981-1987.
- Foster, J. (1995). "Low pH adaptation and the acid tolerance response of *Salmonella typhimurium*." <u>Critical Reviews in Microbiology</u> 21(4): 215-237.
- Foster, J. and B. Bearson (1994). "Acid-sensitive mutants of *Salmonella typhimurium* identified through a dinitrophenol lethal sceening strategy." <u>Journal of Bacteriology</u> 176(9): 2596-2602.
- Foster, J. and H. Hall (1990). "Adaptive acidification tolerance response of *Salmonella typhimurium*." Journal of Bacteriology 172(2): 771-778.
- Foster, J. and H. Hall (1991). "Inducible pH homeostasis and the acid tolerance response of Salmonella typhimurium." Journal of Bacteriology 173: 5129-5135.
- Foster, J., Y.-K. Park, I. Bang, K. Karem, H. Betts, H. Hall and E. Shaw (1994). "Regulatory circuits involved with the pH-regulated gene expression in *Salmonella typhimurium*." <u>Microbiology</u> 140: 341-352.
- Foster, J. and M. Spector (1995). "How Salmonella survive against the odds." <u>Annual Review of</u> <u>Microbiology</u> 49: 145-174.
- Frydman, J. and F. U. Hartl (1996). "Principles of chaperone-assisted protein folding: Differences between in vitro and in vivo mechanisms." <u>Science</u> 272: 1497-1502.

- Frydman, J. and J. Höhfeld (1997). "Chaperones get in touch: The hip-hop connection." <u>Trends in</u> <u>Biochemical Sciences</u> 22: 87-92.
- Fujisawa, T., Y. Benno, T. Yaeshima and T. Mitsuoka (1992). "Taxonomic study of the Lactobacillus acidophilus group, with recognition of Lactobacillus gallinarum sp. nov. and Lactobacillus johnsonii sp. nov. and synonymy of Lactobacillus acidophilus group A3 (Johnson et al. 1980) with the type strain of Lactobacillus amylovorus (Nakamura 1981)."
   International Journal of Systematic Bacteriology 42(3): 487-491.
- Fuller, R. (1992). History and development of probiotics. <u>Probiotics-the scientific basis</u>. London, Chapman and Hall.
- **Fuller, R.** (1993). "Probiotic foods: Current use and future developments." <u>International Journal of</u> <u>Food Ingredients</u> 3: 23-26.
- Gadgil, C. and K. Venkatesh (1997). "Structured model for batch culture growth of *Lactobacillus bulgaricus*." Journal of Chemical Technology and Biotechnology 68: 89-93.
- Gahan, C., B. O'Driscoll and C. Hill (1996). "Acid adaptation of *Listeria monocytogenes* can enhance survival in acidic foods and during milk fermentation." <u>Applied and Environmental</u> <u>Microbiology</u> 62(9): 3128-3132.
- Garbay, S. and A. Lonvaud-Funel (1996). "Response of *Leuconostoc oenos* to environmental changes." Journal of Applied Bacteriology 81: 619-625.
- Garcio-Del Portillo, F., J. Foster and B. Finlay (1993). "Role of acid tolerance response genes in Salmonella typhimurium virulence." Infection and Immunity 61: 4489-4492.
- Garren, D., M. Harrison and S. Russell (1997). "Retention of acid tolerance and acid shock responses of *Escherichia coli* 0157:H7 and non-0157:H7 isolates." <u>Journal of Food</u> Protection 60(12): 1478-1482.

- Garren, D. M., M. A. Harrison and S. M. Russell (1998). "Acid tolerance and acid shock response of *Escherichia coli* 0157-H7 and non-0157-H7 isolates provide cross protection to sodium lactate and sodium chloride." <u>Journal of Food Protection</u> 61(2): 158-161.
- **Gasser, F.** (1970). "Electrophoretic characterization of lactic dehydrogenases in the genus Lactobacillus." Journal of General Microbiology 62: 223-239.
- Gething, M.-J. and J. Sambrook (1992). "Protein folding in the cell." Nature 355: 33-45.
- Gibson, G. and A. Roberfroid (1995). "Bifidogenic properties of different types of fructooligosaccharides." <u>Food Microbiology</u> 11: 491-498.
- Gilliland, S. and M. Speck (1977). "Instability of *Lactobacillus acidophilus* in yogurt." <u>Journal of Dairy</u> <u>Science</u> 60: 1394-1398.
- Gomez, A., P. de Urraza and G. De Antoni (2000). "Characterization of *Bifidobacterium* strains using box primers." <u>Anaerobe</u> 6: 169-177.
- Gomez, A. and F. Malcata (1999). "*Bifidobacterium* spp. and *Lactobacillus acidophilus*: Biological, biochemical, technological and therapeutical properties relevant for use as probiotics." <u>Trends in Food Science and Technology</u>(10): 139-157.
- **Goodson, M. and R. Rowbury** (1989). "Resistance of acid-habituated *Escherichia coli* to organic acids and its medical and applied significance." <u>Letters in Applied Microbiology</u> 8: 211-214.
- Gorden, J. and P. Small (1993). "Acid resistance in enteric bacteria." Infection and Immunity 61: 364-367.
- Goss, T., G. O'Hara, M. Dilworth and A. Glenn (1990). "Cloning, characterisation and complementation of lesions causing acid sensitivity in tn5-induced mutants of *Rhizobium meliloti* wsm419." Journal of Bacteriology 172(9): 5173-5179.
- **Gottesman, S. and M. Maurizi** (1992). "Regulation by proteolysis: Energy-dependent proteases and their targets." Microbiological Reviews 56: 592-621.

- Grogan, D. and J. Cronan (1997). "Cyclopropane ring formation in membrane lipids of bacteria." <u>Microbiology and Molecular Biology Reviews</u> 61(4): 429-441.
- Guerzoni, M., R. Lanciotti and P. Coccoccelli (2001). "Alteration in cellular fatty acid composition as a response to salt, acid, oxidative and thermal stresses in *Lactobacillus helveticus*." <u>Microbiology</u> 147: 2255-2264.
- Guzzo, J., J. Cavin and C. Divies (1994). "Induction of stress proteins in *Leuconostoc oenos* to perform direct inoculation of wine." <u>Biotechnology Letters</u> 16(11): 1189-1194.
- Hammes, W. and P. Tichaczek (1994). "The potential of lactic acid bacteria for the production of safe and wholesome food." <u>Lebens. Unters. Forsch.</u> 198: 193-201.
- Hammes, W. and R. Vogel (1995). The genus *Lactobacillus*. <u>The genera of lactic acid bacteria</u>. <u>The</u> <u>lactic acid bacteria vol. 2</u>. B. J. B. Wood and W. H. Holzapfel. Glagow, Scotland, Blackie Academic & Professional. 2.
- Hansen, P. A. and G. Mocquot (1970). "Lactobacillus acidophilus (moro) comb. nov." International Journal of Systematic Bacteriology 20(3): 325-327.
- Hartke, A., S. Bouche, X. Gansel, P. Boutibonnes and Y. Auffray (1994). "Starvation-induced stress resistance in *Lactococcus lactis* subsp. *lactis* IL1403." <u>Applied and Environmental</u> <u>Microbiology</u> 60(9): 3474-3478.
- Hartke, A., S. Bouche, J.-C. Giard, A. Benachour, P. Boutibonnes and Y. Auffray (1996). "The lactic acid stress response of *Lactococcus lactis* subsp. *lactis*." <u>Current Microbiology</u> 33: 194-199.
- Hartke, A., S. Bouche, J. Laplace, A. Benachour, P. Boutibonnes and Y. Auffray (1995). "UVinducible proteins and UV-induced cross-protection against acid, ethanol, H<sub>2</sub>O<sub>2</sub> or heat treatments in *Lactococcus lactis* subsp. *lactis*." <u>Archives of Microbiology</u> 163: 329-336.
Hartke, A., J. Frere, P. Boutibonnes and Y. Auffray (1997). "Differential induction of the chaperonin groEL and the co-chaperonin groES by heat, acid, and UV-irradiation in *Lactococcus lactis* subsp. *lactis*." <u>Current Microbiology</u> 34: 23-26.

Hartl, F. U. (1994). "Secrets of a double doughnut." Nature 371: 557-558.

- Hartl, F. U. (1996). "Molecular chaperone in the cellular protein folding." Nature 381: 571-580.
- Havenaar, R. and J. Huis in't Veld (1992). Probiotics: A general view. <u>The lactic acid bacteria in health and disease</u>. B. Wood. London, Elsevier Applied Science. 1: 151-170.
- Hecker, M., W. Schumann and U. Völker (1996). "Heat-shock and general stress response in *bacillus subtilis*." <u>Molecular Microbiology</u> 19(3): 417-428.
- Hendrick, J. and F. U. Hartl (1993). "Molecular chaperone functions of heat shock proteins." <u>Annual</u> <u>Review of Biochemistry</u> 62: 349-384.
- Hertel, C., W. Ludwig, B. Pot, K. Kersters and K.-H. Schleifer (1993). "Differentiation of lactobacilli occuring in fermented milk products by using oligonucleotide probes and electrophoretic protein profiles." <u>Systematic and Applied Microbiology</u> 16: 463-467.
- Heyde, M. and R. Portalier (1990). "Acid shock proteins of *Escherichia coli*." <u>FEMS Microbiology</u> Letters 69: 19-26.
- Hickey, E. N. and I. N. Hirshfield (1990). "Low pH induced effects on patterns of protein synthesis and on internal pH in *E. coli* and *Salmonella typhimurium*." <u>Applied and Environmental</u> <u>Microbiology</u> 56(4): 1038-1045.
- Hill, C., B. O'Driscoll and I. Booth (1995). "Acid adaptation and food poisoning microorganisms." International Journal of Food Microbiology 28: 245-254.
- Hochman, A. (1997). "Programmed cell death in prokaryotes." <u>Critical Reviews in Microbiology</u> 23(3): 207-214.

- Hoffmann, A. and P. Parsons (1994). Evolutionary genetics and environmental stress. Oxford, Oxford University Press.
- Hoier, E. (1992). "Use of probiotic starter cultures in dairy products." Food Australia 44(9): 418-420.
- Holt, J., N. Kreig, P. Sneath, J. Staley and S. Williams (1994). <u>Bergey's manual of determinative</u> <u>bacteriology</u>. Baltimore, Williams and Wilkins.
- Hood, S. and E. Zottola (1988). "Effect of low pH on the ability of *Lactobacillus acidophilus* to survive and adhere to human intestinal cells." Journal of Food Science 53(5): 1514-1516.
- Hull, R., P. Conway and A. Evans (1992). "Probiotic foods-a new opportunity." Food Australia 44(3): 112-113.
- Hull, R. R., A. V. Roberts and J. J. Mayes (1984). "Survival of *Lactobacillus acidophilus* in yoghurt." <u>The Australian Journal of Dairy Technology</u> Dec.: 164-166.
- Humphrey, T., N. Richardson, K. Statton and R. Rowbury (1993). "Acid habituation in Salmonella enteritidis PT4: Impact of inhibition of protein synthesis." <u>Letters in Applied Microbiology</u> 16: 228-230.
- Humphrey, T., N. Richardson, K. Statton and R. Rowbury (1993). "Effects of temperature shift on acid and heat tolerance in *Salmonella enteritidis* phage type 4." <u>Applied and Environmental</u> <u>Microbiology</u> 59(9): 3120-3122.
- Hutkins, R. W. and N. L. Nannen (1993). "pH homeostasis in lactic acid bacteria." Journal of Dairy Science 76: 2354-2365.
- Iriarte, M., I. Stainer and G. Cornelis (1995). "The *rpoS* gene from *Yersinia enterocolitica* and its influence on the expression of virulence factors." Infection and Immunity 63(5): 1840-1847.
- Jakob, U. and J. Buchner (1994). "Assisting spontaneity: The role of hsp90 and small hsps as molecular chaperones." <u>Trends in Biochemical Sciences</u> 19: 205-211.

- Janion, C. (2001). "Some aspects of the SOS response system a critical survey." <u>Acta Biochimica</u> <u>Polonica</u> 48(3): 599-610.
- Jian, W., L. Zhu and X. Dong (2001). "New approach to phylogenetic analysis of the genus Bifidobacterium based on partial hsp60 gene sequences." <u>International Journal of Systematic</u> and Evolutionary Microbiology 51: 1633-1638.
- Johnson, J. and J. Craig (1997). "Protein folding *in vivo*: Unraveling complex pathways." <u>Cell</u> 90: 201-204.
- Johnson, J., C. Phelps, C. Cummins, J. London and F. Gasser (1980). "Taxonomy of the Lactobacillus acidophilus group." <u>International Journal of Systematic Bacteriology</u> 30(1): 53-68.
- Johnsson, T., P. Nikkila, L. Toivonen, H. Rosenquist and S. Laako (1995). "Cellular fatty acid profiles of *Lactobacillus* and *Lactococcus* strains in relation to the oleic acid content of the cultivation medium." <u>Applied and Environmental Microbiology</u> 61(12): 4497-4499.
- Kailaspathy, K. and J. Chin (2000). "Survival and therapeutic potential of probiotic organisms with reference to Lactobacillus acidophilus and Bifidobacterium spp." <u>Immunology and Cell</u> <u>Biology</u> 78: 80-88.
- Kailaspathy, K. and S. Rybka (1997). "L. acidophilus and Bifidobacterium spp.- their therapeutic potential and survival in yoghurt." <u>The Australian Journal of Dairy Technology</u> 52(1): 28-35.
- Karem, K., J. Foster and A. Bej (1994). "Adaptive acid tolerance (ATR) in *Aeromonas hydrophila*." <u>Microbiology</u> 140: 1731-1736.
- Kashket, E. (1987). "Bio-energetics of lactic acid bacteria: Cytoplasmic pH and osmotolerance." FEMS Microbiology Reviews 46: 233-244.
- Kaufmann, P., A. Pfefferkorn, M. Teuber and L. Meile (1997). "Identification and quantification of *Bifidobacterium* species isolated from food with genus-specific 16S rRNA-targeted probes by colony hybridisation and pcr." <u>Applied and Environmental Microbiology</u> 63(4): 1268-1273.

- Kelley, W. (1998). "The J-domain family and the recruitment of chaperone power." <u>Trends in</u> <u>Biochemical Sciences</u> 23: 222-227.
- Khuller, G. and H. Goldfine (1974). "Phosolipids of *C. butyricum*. (v) Effects of growth temperature of fatty acid, alk-1-enyl ether group and phosolipid composition." <u>J. Lipid Res.</u> 15(500-507).
- Kim, W. and N. Dunn (1997). "Identification of a cold shock gene in lactic acid bacteria and the effect of cold shock on cryotolerance." <u>Current Microbiology</u> 35: 59-63.
- Kim, W., L. Perl, J. Park, J. Tandianus and N. Dunn (2001). "Assessment of stress response of the probiotic *Lactobacillus acidophilus*." <u>Current Microbiology</u> 43: 346-350.
- Kirchgatterer, A. and P. Knoflach (2004). "Natural therapy instead of chemistry? Probiotics in gastroenterology." <u>Acta Medica Austriaca</u> 31(1): 13-17.
- Klein, G. (2001). "International committee on systematic bacteriology subcommittee on the taxonomy of *Bifidobacterium, Lactobacillus* and related organisms. Minutes of the meetings, 22 and 23 September 1999, Veldhoven, the Netherlands." <u>International Journal of Systematic and Evolutionary Microbiology</u> 51: 259-261.
- Kobayashi, H., M. Yamamoto and R. Aono (1998). "Appearance of a stress-response protein, phage-shock protein a, in *Escherichia coli* exposed to hydrophobic organic solvents." <u>Microbiology</u> 144: 353-359.
- Kroll, R. G. and R. A. Patchett (1992). "Induced acid tolerance in *Listeria monocytogenes*." Letters in Applied Microbiology 14: 224- 227.
- **Krzewinski, F.** (1997). *Bifidobacterium* species, Thesis, Université des Sciences et Techniques de Lille (France).
- Krzewinski, F. (2002). *Bifidobacterium* species. 2002. http://www.inra.fr/bifidproject/Bifidobacteria.htm
- Laemmli, U. (1970). "Cleavage of structural proteins during the assembly of the head of bacteriophage T4." <u>Nature</u> 227: 680-685.

- Langer, T. (2000). "AAA proteases: Cellular machines for degrading membrane proteins." <u>Trends in</u> <u>Biochemical Sciences</u> May: 247-251.
- Lankaputhra, W., N. Shah and M. Britz (1996). "Survival of bifidobacteria during refrigerated storage in the presence of acid and hydrogen peroxide." <u>Milchwissenchaft-Milk Science</u> International 51(2): 65-70.
- Laroia, S. and J. Martin (1991). "Effect of pH on survival of *Bifidobacterium bifidum* and *Lactobacillus acidophilus* in frozen fermented dairy desserts." 25: 13-21.
- Lee, I., J. Lin, H. Hall, B. Bearson and J. Foster (1995). "The stationary-phase sigma factor (*rpos*) is required for a sustained acid tolerance in virulent salmonella typhimurium." <u>Molecular Microbiology</u> 17: 155-167.
- Lee, I., J. Slonczewski and J. Foster (1994). "A low-pH-inducible, stationary-phase acid tolerance response in *Salmonella typhimurium*." Journal of Bacteriology 176(5): 1422-1426.
- Lelivelt, M. and T. Kawala (1995). "Hsc66, an hsp70 homolog in *Escherichia coli,* is induced by cold shock but not by heat shock." Journal of Bacteriology 177(17): 4900-4907.
- Leyer, G. and E. Johnson (1992). "Acid adaptation promotes survival of salmonella spp. in cheese." Applied and Environmental Microbiology 58(6): 2075-2080.
- Leyer, G. and E. Johnson (1997). "Acid adaptation sensitizes *Salmonella typhimurium* to hypochlorous acid." Applied and Environmental Microbiology 63(2): 461.
- Leyer, G., L. Wang and E. Johnson (1995). "Acid adaptation of *Escherichia coli* 0157:H7 increases survival in acidic foods." <u>Applied and Environmental Microbiology</u> 61(10): 3752-3755.
- Leyer, G. J. and E. A. Johnson (1993). "Acid adaptation induces cross-protection against environmental stresses in *Salmonella typhimurium.*" <u>Applied and Environmental Microbiology</u> 59(6): 1842-1847.
- Lilly, D. and R. Stillwell (1965). "Probiotics: Growth promoting factors produced by microorganisms." Science 147: 747-748.

- Lin, J., I.-S. Lee, J. Frey, J. Slonczewski and J. Foster (1995). "Comparative analysis of extreme acid survival in *Salmonella typhimurium, Shigella flexneri,* and *Escherichia coli*." Journal of <u>Bacteriology</u> 177: 4097-4104.
- Lindquist, S. and E. Craig (1988). "The heat shock proteins." <u>Annual Review of Genetics</u> 22: 631-677.
- Little, J. (1996). The SOS regulatory system. <u>Regulation of gene expression in Escherichia coli</u>. E. Lin and A. Lynch. Austin, Chapman and Hall: 453-480.
- Liu, S.-Q., G. Pritchard, M. Hardman and G. Pilone (1996). "Arginine catabolism in wine lactic acid bacteria: Is it via the arginine-deiminase pathway or the arginine-urease pathway." <u>Journal of</u> <u>Applied Bacteriology</u> 81: 486-492.
- Lopez, M., W. Patton, B. Utterback, N. Chung-Welch, P. Barry, W. Skea and R. Cambria (1991). "Effect of various detergents on protein migration in the second dimension of twodimensional gels." <u>Annals of Biochemistry</u> 199: 35-44.
- Lorca, G., R. Raya, M. Taranto and G. de Valdez (1998). "Adaptive acid tolerance response in Lactobacillus acidophillus." <u>Biotechnology Letters</u> 20(3): 239-241.
- Lorimer, G. (1997). "Folding with a two-stroke motor." <u>Nature</u> 388: 720-723.
- Lund, P. (1995). "The roles of molecular chaperones in vivo." Essays in Biochemistry 29: 113-124.
- Marks, J. (1996). Current status and future prospects. <u>Gut flora and health-past, present and future</u>.A. Leeds and I. Rowlands. London, The Royal Society of Medicine Limited: 57-66.
- Marquis, R., G. Bender, D. Murray and A. Wong (1987). "Arginine deiminase system and bacterial adaptation to acid environments." <u>Applied and Environmental Microbiology</u> 53: 198-200.
- Marshall, B., L. Barrett, C. Prakash, R. McCallum and R. Guerrant (1990). "Urea protects Helicobacter (Campylobacter) pylori from the bactericidal effect of acid." <u>Gastroenterology</u> 99: 97-102.

- Marteau, P. and J.-C. Rambaud (1996). Therapeutic applications of probiotics in humans. <u>Gut flora</u> <u>and health-past, present and future</u>. A. Leeds and I. Rowland. London, The Royal Society of Medecine Limited: 47-56.
- Marty-Teysset, C., F. De La Torre and J.-R. Garel (2000). "Increased production of hydrogen peroxide by *Lactobacillus delbrueckii* subsp. b*ulgaricus* upon aeration: Involvement of an NADH oxidase in oxidative stress." <u>Applied and Environmental Microbiology</u> 66(1): 262-267.
- Mattila-Sandholm, T., J. Mättö and M. Saarela (1999). "Lactic acid bacteria with health claimsinteractions and interference with gastrointestinal flora." <u>International Dairy Journal</u> 9: 25-35.
- McDonald, L., H. Fleming and H. Hassan (1990). "Acid tolerance of *Leuconostoc mesenteroides* and *Lactobacillus plantarum*." <u>Applied and Environmental Microbiology</u> 56: 2120-2124.
- McMeekin, T., J. Olley, T. Ross and D. Ratkowsky (1993). <u>Predictive microbiology: Theory and</u> <u>application.</u>. Taunton, Somerset, England, Research Studies Press.
- Meile, L., W. Ludwig, U. Rueger, C. Gut, P. Kauffmann, G. Dasen, S. Wenger and M. Teuber (1997). "*Bifidobacterium lactis* sp. nov., a moderately oxygen tolerant species isolated from fermented milk." <u>Systematic and Applied Microbiology</u> 20: 57-64.
- Melnykovych, G. and E. Snell (1958). "Nutritional requirements for the formation of arginine decarboxylase in *Escherichia coli*." Journal of Bacteriology 76: 518-523.
- Metchnikoff, E. (1907). The prolongation of life. London, Heinemann.
- Miller, G. (1959). "Use of dinitrosalicylic acid reagent for determination of reducing sugar." <u>Analytical</u> <u>Chemistry</u> 31(3): 426-458.
- Miller, R. and B. Britigan (1997). "Role of oxidants in microbial pathophysiology." <u>Clinical</u> <u>Microbiology Reviews</u> 10(1): 1-18.
- Missiakas, D. and S. Raina (1997). "Protein misfolding in the cell envelope of *Escherichia coli*: New signaling pathways." <u>Trends in Biochemical Sciences</u> 22: 59-63.

- Missiakas, D., S. Raina and C. Georgopoulos (1996). Heat shock regulation. <u>Regulation of gene</u> <u>expression in *Escherichia coli*</u>. E. Lin and A. Lynch. Austin, Chapman and Hall: 481-501.
- Miwa, T., H. Esaki, J. Umemori and T. Hino (1997). "Activity of H+-ATPase in ruminal bacteria with special reference to acid tolerance." <u>Applied and Environmental Microbiology</u> 63(6): 2155-2158.
- Miyake, T., K. Watanabe, T. Watanabe and H. Oyaizu (1998). "Phylogenetic analysis of the genus *Bifidobacterium* and related genera based on 16s rDNA sequences." <u>Microbiology and</u> <u>Immunology</u> 42(10): 661-667.
- Mobley, H. and R. Hausinger (1989). "Microbial ureases: Significance, regulation, and molecular characterisation." <u>Microbiological Reviews</u> 53: 85-108.
- Mogensen, G. (2002). "The history, development and probiotic effect of B. *lactis* Bb-12 in AB yoghurt", Korea Society of Food Science and Technology. 2002.
- Morimoto, R., M. Kline, D. Bimston and J. Cotto (1997). "The heat shock response: Regulation and function of heat-shock proteins and molecular chaperones." <u>Essays in Biochemistry</u> 32: 17-28.
- **Muffler, A., M. Barth, C. Marschall and R. Henge-Aronis** (1997). "Heat shock regulation of σ<sup>s</sup> turnover: A role for DnaK and relationship between stress responses mediated by σ<sup>s</sup> and σ<sup>2</sup> in *Escherichia coli.*" Journal of Bacteriology 179(2): 445-452.
- Mukhopadhyay, S. and H. Schellhorn (1997). "Identification and characterization of hydrogen peroxide-sensitive mutants of *Escherichia coli*: Genes that require *oxyR* for expression." Journal of Bacteriology 179(2): 330-338.
- Neely, M., C. Dell and E. Olsen (1994). "Roles of lysP and cadC in mediating the lysine requirement for acid induction of the *Escherichia coli* cad operon." <u>Journal of Bacteriology</u> 176(11): 3278-3285.

- Netzer, W. and F. U. Hartl (1998). "Protein folding in the cytosol: Chaperonin-dependent and independent mechanisms." <u>Trends in Biochemical Sciences</u> 23: 68-73.
- Nilsson, B. (1991). "Proper and improper folding of proteins in the cellular environment." <u>Annual</u> <u>Review of Microbiology</u> 45: 607-635.
- Norris, C., M. Brown, E. Hickey, L. Weber and L. Hightower (1997). "Low-molecular-weight heat shock proteins in a desert fish (*Poeciliopsis lucida*): Homologs of human hsp27 and *xenopus* hsp30." <u>Molecular Biology and Evolution</u> 14(10): 1050-1061.
- **Oberman, H. and Z. Libudzisz** (1998). Fermented milks. <u>Microbiology of fermented foods</u>. B. Wood. London, Blackie Scientific and Professional. 1: 308-350.
- O'Connell-Motherway, M., G. Fitzgerald and D. Van Sinderen (1997). "Cloning and sequence analysis of putative histidine protein kinases isolated from *Lactococcus lactis* MG1363." <u>Applied and Environmental Microbiology</u> 63(6): 2454-4529.
- O'Driscoll, B., G. Cormac, C. Gahan and C. Hill (1997). "Two-dimensional polyacrylamide gel electrophoresis analysis of the acid tolerance response in *Listeria monocytogenes* LO28." <u>Applied and Environmental Microbiology</u> 63(7): 2679-2685.
- O'Driscoll, B., G. Cormac and C. Hill (1996). "Adaptive acid tolerance response in *Listeria monocytogenes*: Isolation of an acid-tolerant mutant which demonstrates increased virulence." Applied and Environmental Microbiology 62(5): 1693-1698.
- Ohara, G. and A. Glenn (1994). "The adaptive acid tolerance response in root nodule bacteria and *Escherichia coli.*" <u>Archives of Microbiology</u> 161: 286-292.
- O'Riordan, K., D. Andrews, K. Buckle and P. Conway (2001). "Evaluation of microencapsulation of a *Bifidobacterium* strain with starch as an approach to prolonging viability during storage." Journal of Applied Microbiology 91: 1059-1066.
- **Ouwehand, A., P. Kirjavainen, C. Shortt and S. Salminen** (1999). "Probiotics: Mechanisms and established effects." International Dairy Journal 9: 43-52.

- Padan, E., D. Zilbertstein and S. Schuldiner (1981). "pH homeostasis in bacteria." <u>Biochim.</u> <u>Biophys. Acta.</u> 650: 151-166.
- Parkinson, J. and E. Kofoid (1992). "Communication modules in bacterial signaling proteins." <u>Annual Review of Genetics</u> 26: 71-112.
- Parsell, D. and S. Lindquist (1993). "The function of heat-shock proteins in stress tolerance: Degradation and reactivation of damaged proteins." <u>Annual Review of Genetics</u> 27: 437-496.
- Patarata, L., M. Pimental, B. Pot, K. Kersters and A. Faia (1994). "Identification of lactic acid bacteria isolated from Portuguese wines and musts by SDS-PAGE." <u>Journal of Applied</u> <u>Bacteriology</u> 76: 288-293.
- Pepper, S., N. Shah and M. Britz (1995). <u>A rapid and simple method for comparitive identification of</u> <u>lactic acid bacteria using microtitre plates</u>. Proceedings of International Dairy Lactic Acid Bacteria Conference (New Zealand). Palmerston North.
- Playne, M. (1995). Probiotic microorganisms. <u>Recent advances in microbiology</u>. G. Gilbert. Melbourne, The Australian Society for Microbiology. 3: 215-254.
- Playne, M., L. Bennet and G. Smithers (2003). "Functional dairy foods and ingredients." <u>The</u> <u>Australian Journal of Dairy Technology</u> 58(3): 242-264.
- Poolman, B., A. Driessen and W. Konings (1987). "Regulation of arginine-ornithine exchange and the arginine deiminase pathway in *Streptococcus lactis*." <u>Journal of Bacteriology</u> 169(12): 5597-5604.
- Pot, B., C. Hertel, W. Ludwig, P. Descheemaeker, K. Kersters and K.-H. Schleifer (1993). "Identification and classification of *Lactobacillus acidophilus*, *L. gasseri* and *L. johnsonii* strains by SDS-PAGE and rRNA-targeted oligonucleotide probe hybridisation." <u>Journal of</u> <u>General Microbiology</u> 139: 513-517.

- Pot, B., W. Ludwig, K. Kersters and K.-H. Schleifer (1994). Taxonomy of lactic acid bacteria. Bacteriocins of lactic acid bacteria. L. De Vuyst and E. Vandamme. London, Blackie Academic & Professional: 14-67.
- Quivey, R., R. Faustoferri, K. Clancy and R. Marquis (1995). "Acid adaptation in Streptococcus mutans UA159 alleviates sensitization to environmental stress due to recA deficiency." <u>FEMS Microbiology Letters</u> 126: 257-262.
- Rallu, F., A. Gruss and E. Maguin (1996). "Lactococcus lactis and stress." Antonie van Leeuwenhoek 70: 243-251.
- Ramakrishnan, V. and S. White (1998). "Ribosomal protein structures: Insights into the architecture, machinery and the evolution of the ribosome." <u>Trends in Biochemical Sciences</u> 23: 208-212.
- Rasic, J. and J. Kurmann (1983). Bifidobacteria and their role. Basel, Birkhauser Verlag.
- **Reid, G.** (1999). "The scientific basis for probiotic strains of *Lactobacillus*." <u>Applied and</u> <u>Environmental Microbiology</u> 65(9): 3763-3766.
- Rementzis, J. and J. Samelis (1996). "Rapid GC analysis of cellular fatty acids for characterizing *Lactobacillus sake* and *Lact. curvatus* strains of meat origin." <u>Letters in Applied Microbiology</u> 23: 379-384.
- Rocha, E., T. Selby, J. Coleman and C. Smith (1996). "Oxidative stress response in an anaerobe, Bacteroides fragilis: A role for catalase in protection against hydrogen peroxide." <u>Journal of</u> <u>Bacteriology</u> 178(23): 6895-6903.
- **Rodtong, S. and G. Tannock** (1993). "Differentiation of lactobacillus strains by ribotyping." <u>Applied</u> <u>and Environmental Microbiology</u> 59(10): 3480-3484.
- Rowbury, R. and M. Goodson (1999). "An extracellular acid stress-sensing protein needed for acid tolerance induction in *Escherichia coli*." <u>FEMS Microbiology Letters</u> 174: 49-55.

- Rowbury, R., M. Goodson and T. Humphrey (1993). "Acid sensitivity induction (ASI) at alkaline pH in *Escherichia coli* involves two major sensitization components, induction of both being switched on by increased internal pH." <u>Letters in Applied Microbiology</u> 17: 272-275.
- Rowbury, R., M. Goodson and T. Humphrey (1993). "Novel acid sensitivity induced in *Escherichia coli* at alkaline pH." Letters in Applied Microbiology 16: 223-227.
- **Rowland, I.** (1996). Gut microflora and cancer. <u>Gut flora and health-past, present and future</u>. A. Leeds and I. Rowland. London, The Royal Society of Medicine Press Limited: 19-26.
- Rowland, I. (1998). Human studies on the effects of food and food components, In Press.
- Ruddon, R. and E. Bedows (1997). "Assisted protein folding." <u>Journal of Biological Chemistry</u> 272(6): 3125-3128.
- Russell, N., R. Evans, P. ter Steeg, J. Hellemons, A. Verheul and T. Abee (1995). "Membranes as a target for stress adaptation." International Journal of Food Microbiology 28: 255-261.
- Saibil, H. (2000). "Molecular chaperones: Containers and surfaces for folding, stabilising or unfolding proteins." <u>Current Opinion in Structural Biology</u> 10: 251-258.
- Sako, T., K. Matsumoto and R. Tanaka (1999). "Recent progress on research and applications of non-digestible galacto-oligosaccharides." International Dairy Journal 9: 69-80.
- Salminen, S., A. Ouwehand, Y. Benno and Y. Lee (1999). "Probiotics: How should they be defined?" <u>Trends in Food Science and Technology</u> 10: 107-110.
- Saloff-Coste, C. (1997). Danone world newsletter. http://www.danonenewsletter.fr.
- Salotra, P., D. Singh, K. Seal, N. Krishna, H. Jaffe and R. Bhatnagar (1995). "Expression of DnaK and groEL homologs in *Leuconostoc mesenteroides* in response to heat shock, cold shock or chemical stress." <u>FEMS Microbiology Letters</u> 131: 57-62.
- Sambrook, J., E. Fritsch and T. Maniatis (1989). <u>Molecular cloning: A laboratory manual</u>. New York, Cold Spring Harbour Laboratory Press.

- Sanchez, Y., J. Taulien, B. K. and S. Lindquist (1992). "HSP 104 is required for tolerance to many forms of stress." <u>The EMBO journal</u> 11(6): 2357-2364.
- Santoni, V., M. Molloy and T. Rabiloud (2000). "Membrane proteins and proteomics: Un amour impossible?" <u>Electrophoresis</u> 21: 1054-1070.
- Schirmer, E., J. Glover, M. Singer and S. Lindquist (1996). "HSP 100/CLP proteins: A common mechanism explains diverse functions." <u>Trends in Biochemical Sciences</u> 21: 289-296.
- Schleifer, K.-H. and W. Ludwig (1994). Molecular taxonomy: Classification and identification. <u>Bacterial diversity and systematics</u>. F. Priest, A. Ramos-Cormenzana and B. Tindall. New York, Plenum Press. 75: 1-16.
- Schleifer, K.-H. and W. Ludwig (1995). "Phylogeny of the genus *Lactobacillus* and related genera." <u>Systematic and Applied Microbiology</u> 18: 461-467.
- Schmidt, T. (1995). "Fingerprinting bacterial genomes using ribosomal RNA genes and operons." <u>Methods in Molecular and Cellular Biology</u> 5(1): 3-12.
- Scott, F. and N. Lee (1996). Recommendations for defining and dealing with functional foods. Report of the bureau of nutritional sciences committee on functional foods, Food Directorate, Health Protection Branch, Health Canada.
- Segal, G. and E. Ron (1996). "Regulation and organization of the groE and dnaK operons in eubacteria." <u>FEMS Microbiology Letters</u> 138: 1-10.
- Semanchek, J. and D. Golden (1998). "Influence of growth temperature on inactivation and injury of *Escherichia coli* 0157:H7 by heat, acid and freezing." <u>Journal of Food Protection</u> 61(4): 395-401.
- Sgorbati, B., B. Biavati and D. Palenzona (1995). The genus *Bifidobacterium*. <u>The genera of lactic</u> <u>acid bacteria</u>. B. Wood and W. Holzapfel. Glasgow, Scotland, Blackie Academic & Professional. 2: 279-306.

- Sgouras, D., P. Maragkoudakis, K. Petraki, B. Martinez-Gonzalez, E. Eriotou, S. Michopoulos,
   G. Kalantzopoulos, E. Tsakalidou and A. Mentis (2004). "In vitro and in vivo inhibition of Helicobacter pylori by Lactobacillus casei strain shirota." <u>Applied and Environmental</u> <u>Microbiology</u> 70(1): 518-526.
- Shah, N., W. Lankaputhra, M. Britz and W. Kyle (1995). "Survival of Lactobacillus acidophilus and Bifidobacterium bifidum in commercial yoghurt during refrigerated storage." <u>International</u> <u>Dairy Journal</u> 5(5): 515-521.
- Shimamura, S., F. Abe, N. Ishibashi, H. Mikayawa, T. Yaeshima, T. Araya and M. Tomita (1992). "Relationship between oxygen sensitivity and oxygen metabolism of *Bifidobacterium* species." <u>Journal of Dairy Science</u> 75: 3296-3306.
- Shin, H.-S., J.-H. Lee, J. Pestka and Z. Ustunol (2000). "Growth and viability of commercial bifidobacterium spp. in skim milk containing oligosaccharides and inulin." Food Microbiology and Safety 65(5): 884-887.
- Slonczewski, J., T. Gonzalez, F. Bartholomew and N. Holt (1987). "Mu D-directed lacZ fusions regulated by low pH in *Escherichia coli*." Journal of Bacteriology 169: 3001-3006.
- Small, P., D. Blankenhorn, D. Welty, E. Zinser and J. Slonczewski (1994). "Acid and base resistance in *Escherichia coli* and *Shigella flexneri*: Role of rpoS and growth pH." <u>Journal of</u> <u>Bacteriology</u> 176: 1729-1737.
- Smith, A., R. Quivey and R. Faustoferri (1996). "Cloning and nucleotide sequence analysis of the Streptococcus mutans membrane-bound, proton-translocating ATPase operon." <u>Gene</u> 183: 87-96.
- Speelmans, G., B. Poolman and W. Konings (1995). "Na+ as coupling ion in energy transduction in extremophilic bacteria and archaea." <u>World Journal of Microbiology and Biotechnology</u> 11: 58-70.
- **Spira, B., N. Silberstein and E. Yagil** (1995). "Guanosine 3',5'-bispyrophosphate (ppGpp) synthesis in cells of *Escherichia coli* starved for P<sub>i</sub>." Journal of Bacteriology 177(14): 4053-4058.

- Squires, C. and C. L. Squires (1992). "The CLP proteins: Proteolysis regulators or molecular chaperones?" Journal of Bacteriology 174(4): 1081-1085.
- Stock, J., A. Ninfa and A. Stock (1989). "Protein phosphorylation and regulation of adaptive responses in bacteria." <u>Microbiological Reviews</u> 53(4): 450-490.
- Sugimura, K., S. Fukuda, Y. Wada, M. Taniai, M. Suzuki, T. Kimura, T. Ohno, K. Yamamoto and
   I. Azuma (1990). "Identification and purification of arginine deiminase that originated from Mycoplasma arginini." <u>Infection and Immunity</u> 58(8): 2510-2515.
- Suzuki, C., M. Rep, J. van Dijl, K. Suda, L. Grivell and G. Schatz (1997). "ATP-dependent proteases that also chaperone protein biogenesis." <u>Trends in Biochemical Sciences</u> 22: 118-123.
- Tagg, J., A. Dajani and L. Wannamaker (1976). "Bacteriocins of Gram-positive-bacteria." Bacteriological Reviews 40: 722-756.
- Tamine, A. (1999). Biology of bifidobacteria. <u>Encyclopedia of food microbiology</u>. R. Robinson. London, Academic Press. 2: 1355-1360.
- Tannock, G. (1995). Chapter One-More than a smell: The complexity of the normal microflora. <u>Normal microflora: An introduction to microbes inhabiting the human body</u>. London, Chapman and Hall: 21-28.
- Tannock, G. (1999). Introduction. <u>Probiotics: A critical review</u>. G. Tannock. Wymondham, Horizon Scientific Press: 1-4.
- Tannock, G., A. Tilsala-Timisjarvi, S. Rodtong, J. Ng, K. Munro and T. Alatossava (1999). "Identification of *Lactobacillus* isolates from the gastrointestinal tract, silage and yoghurt by 16S-23S rRNA gene intergenic spacer region sequence comparisons." <u>Applied and Environmental Microbiology</u> 65(9): 4264-4267.

- Taranto, M., F. Sesma, A. Pesce de Ruiz Holgado and G. de Valdez (1997). "Bile salts hydrolase plays a key role on cholesterol removal by *Lactobacillus reuteri*." <u>Biotechnology Letters</u> 19(9): 845-847.
- Tiwari, R., W. Reeve, M. Dilworth and A. Glenn (1996). "Acid tolerance in *Rhizobium meliloti* strain WSM419 involves a two-component sensor-regulator system." <u>Microbiology</u> 142: 1693-1704.
- Tiwari, R., W. Reeve, M. Dilworth and A. Glenn (1996). "An essential role for *actA* in acid tolerance of *Rhizobium meliloti*." <u>Microbiology</u> 142: 601-610.
- Tiwari, R., W. Reeve and A. Glenn (1992). "Mutations conferring acid sensitivity in the acid-tolerant strains *Rhizobium meliloti* wsm419 and *Rhizobium leguminosarum* biovar viciae WSM710." <u>FEMS Microbiology Letters</u> 100: 107-112.
- Torriani, S., G. Zapparoli and F. Dellglio (1999). "Use of PCR-based methods for rapid differentiation of *Lactobacillus delbrueckii* subsp. *bulgaricus* and *L. delbrueckii* subsp. *lactis.*" <u>Applied and Environmental Microbiology</u> 65(10): 4351-4356.
- Tsai, Y.-W. and S. Ingham (1997). "Survival of *Escherichia coli* 0157:H7 and *Salmonella* spp. in acidic condiments." Journal of Food Protection 60(7): 751-755.
- Tsakalidou, E., E. Manolopoulou, E. Kabaraki, E. Zoidou, B. Pot, K. Kersters and G. Kalantzopoulos (1994). "The combined use of whole-cell protein extracts for the identification (SDS-PAGE) and enzyme activity screening of lactic acid bacteria isolated from traditional Greek dairy products." <u>Systematic and Applied Microbiology</u> 17: 444-458.

Vale, R. (2000). "AAA proteins: Lords of the ring." The Journal of Cell Biology 150(1): F13-F19.

Van de Guchte, M., P. Serror, C. Chervaux, T. Smokvina, S. D. Ehrlich and E. Maguin (2002). "Stress responses in lactic acid bacteria." <u>Antonie Van Leeuwenhoek International Journal of</u> <u>General and Molecular Microbiology</u> 82(1-4): 187-216.

- Vandamme, P., B. Pot, M. Gillis, P. de Vos, K. Kersters and J. Swings (1996). "Polyphasic taxonomy, a consensus approach to bacterial systematics." <u>Microbiological Reviews</u> 60(2): 407-438.
- Varshavsky, A. (1997). "The ubiquitin system." Trends in Biochemical Sciences 22: 383-387.
- Veiga, M. and F. Gutiérrez (1991). "Basic computer program for the graphic representation of microbial growth curves and batch fermentations." <u>Journal of Microbiology Methods</u> 13: 23-38.
- Ventura, M., R. Reniero and R. Zink (2001). "Specific identification and targeted characterization of Bifidobacterium lactis from different environmental isolates by a combined multiplex-PCR approach." <u>Applied & Environmental Microbiology</u> 67(6): 2760-2765.
- Ventura, M. and R. Zink (2002). "Rapid identification, differentiation and proposed new taxonomic classification of *Bifidobacterium lactis*." <u>Applied and Environmental Microbiology</u> 68(12): 6429-6434.
- Walter, J., C. Hertel, G. Tannock, C. Lis, K. Munro and W. Hammes (2001). "Detection of Lactobacillus, Pediococcus, Leuconostoc and Weissella species in human faeces by using group-specific PCR primers and denaturing gradient gel electrophoresis." <u>Applied and Environmental Microbiology</u> 67(6): 2578-2585.
- Ward, L. and M. Timmins (1999). "Differentiation of *Lactobacillus casei*, *Lactobacillus paracasei* and *Lactobacillus rhamnosus* by polymerase chain reaction." <u>Letters in Applied Microbiology</u> 29: 90-92.
- Waterman, S. and P. Small (1996). "Identification of σ<sup>s</sup> -dependent genes associated with the stationary-phase acid-resistance phenotype of *Shigella flexneri*." <u>Molecular Microbiology</u> 21(5): 925-940.
- Wendakoon, C. N. and L. Ozimek (2002). "Inhibition of *Helicobacter pylori* growth *by Lactobacillus casei* in fermented milk." <u>Milchwissenschaft-Milk Science International</u> 57(9-10): 506-509.

- Whitaker, R. and C. Batt (1991). "Characterisation of heat shock response in *Lactococcus lactis* ssp. *lactis*." <u>Applied and Environmental Microbiology</u> 57: 1408-1412.
- Wind, A. (2001). Identification of strain Bb12 used by Chr. Hansen: Personal Communication.
- Wood, B., Ed. (1998). Microbiology of fermented foods. London, Blackie Academic and Professional.
- Wurgler-Murphy, S. and H. Saito (1997). "Two-component signal transducers and mapK cascades." <u>Trends in Biochemical Sciences</u> 22: 172-176.
- Yamamoto, N. and T. Takano (1999). "Antihypertensive peptides derived from milk proteins." <u>Nahrung Food</u> 43(3): 159-164.
- Yeung, P., M. Sanders, C. Kitts, R. Cano and P. Tong (2002). "Species-specific identification of commercial probiotic strains." Journal of Dairy Science 85: 1039-1051.
- Yi, X., E. Kot and A. Bezkorovainy (1998). "Properties of NADH oxidase from *Lactobacillus delbrueckii* ssp *bulgaricus*." Journal of the Science of Food and Agriculture 78: 527-534.
- Yuan, G. and S.-L. Wong (1995). "Isolation and characterisation of *bacillus subtilis groE* regulatory mutants: Evidence for *orf39* in the *DnaK* operon as a repressor gene in regulating the expression of both *groE* and *dnaK*." Journal of Bacteriology 177(22): 6462-6468.
- Yura, T., H. Nagai and H. Mori (1993). "Regulation of the heat-shock response in bacteria." <u>Annual</u> Review of <u>Microbiology</u> 47: 321-350.
- Zeilstra-Ryalls, J., O. Fayet and C. Georgopoulos (1991). "The universally conserved groE (hsp60) chaperonin." <u>Annual Review of Microbiology</u> 45: 301-325.
- Zuber, U. and W. Schumann (1994). "Circe, a novel heat shock element involved in regulation of heat shock operon *DnaK* of *Bacillus subtilis*." Journal of Bacteriology 176(5): 1359-1363.

## APPENDICES

#### Appendix 1

## List of Laboratory Instrumentation, equipment and

#### chemicals

#### Instrumentation and equipment

Anerobic jar and catalyst system (2L) Supplied by Oxoid Australia Pty Ltd

Bioreactor (fermenter) 2L Braun Biostat B Supplied by B. Braun Australia

Bioreactor (fermenter) 1L 1012 motor controller and FC4 Data system Applikon dependable instruments, Schiedam Holland Supplied by Real Time Engineering

Cell Homogenizer. Supplied by B. Braun, Australia

Cell Homogenizer. 50 mL Duran glass or stainless steel sample bottles Supplied by B. Braun, Australia

Centrifuge J2-HS, refrigerated Supplied by Beckman Industries Australia Ltd

Centrifuge, refrigerated Sorvall Model RT7 Supplied by Amrad Pharmacia Biotech

Dry Block Heater Supplied by Ratek Instruments Australia

Eppendorf tubes (1 mL) Elkay Products Inc Supplied by Pathtech Diagnostics

Fermenter (One litre Bio Reactor Under Computer Evaluation) BRUCE system Developed at CBFT, Vic University by Dr. Takashi Ueno

Gas Chromatograph model 3400, including Cx autosampler and Star chromatography software, version 4.0, Saturn 2000 GC/MS/MS and Gas chromatography sample vials Supplied by Varian Australia Pty Ltd

Gas Chromatography column BPX-70 25 metre bonded phase, fused silica, very polar column Supplied by SGE Australia

Gas supplies including high purity Nitrogen, Instrument air and gases for GC Supplied by BOC Gases

GelAir Drying System Protein gel dryer Supplied by Bio-Rad Laboratories Pty Ltd

G100 Electrophoresis tank and fittings Supplied by Amrad Pharmacia Biotech

HPLC column Aminex HPX-87H Supplied by Bio-Rad

Hybond N+ Nucleic acid transfer membrane Supplied by Amersham Australia Pty Ltd

HPLC (including Varian 9100 Autosampler, Varian 9012 Pump, Varian 9050 UV-VIS Detector and Varian Star 4.0 software Supplied by Varian Australia Pty Ltd

Kimwipes Fine Grade Cat # 4103 Supplied by Kimberly Clark

Microfuge Eppendorf Centrifuge 54515C Supplied by Crown Scientific Pty Ltd

Microfuge Microspin 24 Sorvall Instruments Dupont. Supplied by Varian Australia Pty Ltd

Mini-Beadbeater-8 BioSpec products, Inc Supplied by Daintree Scientific

Mini Prep Cell (preparative electrophoresis) Supplied by Bio-Rad Laboratories Pty Ltd

Model 491 Prep Cell (preparative electrophoresis) Supplied by Bio-Rad Laboratories Pty Ltd

Needles hypodermic 18G 1½ TW 1.00 x 38 mm Terumo Corporation Supplied by Pathtech Diagnostics

Nytran-Plus Nylon Membrane Supplied by Progen Industries Ltd Pasteur pipettes Cat# BILB 150 Supplied by Selby Scientific

pH meter model PHM220 Supplied by Radiometer Pacific

Photographics Supplied by Prossper Photographics

Protean II cell Supplied by Bio-Rad Laboratories Pty Ltd

Rotofor Cell Bio-Rad Laboratories Pty Ltd

Screw top open hole vial caps 8 mm Alltech Cat # 16-000698-00 Screw cap vials 12x32 mm (amber) Alltech Cat # 95194 Supplied by Progen Industries Ltd

Syringes, Terumo Corporation 1 mL, 5 mL, 10 mL, 50mL Supplied by Crown Scientific

Trans-Blot SD cell Supplied by Bio-Rad Laboratories Pty Ltd

Triple wide vertical slab electrophoresis unit Supplied by Silenus Labs Pty Ltd

Two-dimensional gel electrophoresis Supplied by Pharmacia

Water baths Supplied by Ratek Instruments Australia

Whatman Chromatography Paper 3MM Chr 46x57 cm Cat # 3030917 Supplied by Lab Supply

#### Chemicals

<sup>32</sup>P Deoxycytidine 5'-Triphosphate Supplied by Bresatec ACN 070 039 913

Acetic acid Glacial AnalaR Cat # 945040 Merck Pty Ltd ACN 005 063 791 Supplied by Crown Scientific

Acrylamide powder Acrylamide 99.9% Supplied by Bio-Rad Laboratories Pty Ltd

ATP Adenosine 5'-Triphosphate Disodium Salt Cat # A 6144 Supplied by Sigma Aldrich Pty Ltd Agar Bacteriological (Agar No 1) Code 11, Cat # LP011F Unipath Ltd, Basingstoke, Hampshire, England Supplied by Oxoid Australia Pty Ltd

Ammonium Acetate Molecular Biology Grade MW=77.08 Cat # A 1542 Supplied by Sigma Aldrich Pty Ltd

Ammonium Persulphate MW=228.20 Supplied by Bio-Rad Laboratories Pty Ltd

Ampicillin D [-]- Aminobenylpenicillin MW=371.4 Cat # A 6144 Supplied by Sigma Aldrich Pty Ltd

Antibodies (GroEL, Dna-K etc.) Supplied by Sigma Immuno Chemicals, BioRad and DAKO

Biolog Lactic Acid Suspension Broth Biolog Inc. Supplied by SPL Diagnostics

Bis (N,N'-methylene-bis-acrylamide) Supplied by Bio-Rad Laboratories Pty Ltd

Bovine Serum Albumin Fraction V Laboratory Reagent Cat # 100 018 Supplied by Boehringer Mannheim Pty Ltd

Butan-2-ol Supplied by Sigma Aldrich Pty Ltd

Coomassie Blue G-250 dye Supplied by Bio-Rad Laboratories Pty Ltd

DEPC Diethyl Pyrocarbonate MW=162.1 Cat # D 5758 Supplied by Sigma Aldrich Pty Ltd

Dithiothreitol Supplied by Sigma Aldrich Pty Ltd

DNA Grade Agarose (Molecular Biology Grade) Cat # 200-0011 Supplied by Progen Industries Ltd

EcoR1 Restriction Nuclease and Buffer Cat # 101S Pst1 Restriction Nuclease and NEBuffer 3 Cat # 140S Other restriction enzymes and buffers Manufactured by New England Biolabs Supplied by Genesearch Pty Ltd

Ethanol Absolute Supplied by CSR Yarraville Distillery

Ethldium Bromide MW=394.3 Cat # E 8751 Supplied by Sigma Aldrich Pty Ltd Ethylenediaminetetra acetic acid Disodium Salt: Dihydrate MW= 372.2 Cat # E 4884 Supplied by Sigma Aldrich Pty Ltd

ExpressHyb Clontech Supplied by Clontech Laboratories

Folin/Ciocalteau Reagent Supplied by Sigma Aldrich Pty Ltd

Formaldehyde MW=30.03 37% Solution ACS Reagent Cat # F 1268 Supplied by Sigma Aldrich Pty Ltd

Formamide MW=45.04 Cat # F 7503 Supplied by Sigma Aldrich Pty Ltd

Gas Chromatography Lipid Standards Product Numbers: 189-1, 189-3, 189-9, 189-4 Supplied by Sigma Aldrich Pty Ltd

D-Glucose BDH MW=180.16 Cat # 28450 Merck Pty Ltd ACN 005 063 791 Supplied by Crown Scientific

Glycerol BDH General Purpose Reagent Product # 28454 Merck Pty Ltd ACN 005 063 791 Supplied by Crown Scientific

Isoamyl Alcohol BDH AnalaR Product # 1003832 Merck Pty Ltd ACN 005 063 791 Supplied by Crown Scientific

Isopropanol (2-Propanol) MW=60.10 Cat # | 9516 Supplied by Sigma Aldrich Pty Ltd

IPTG Isopropyl- -D-Thiogalactopyranosid MW=238.3 Supplied by Boehringer Mannheim Pty Ltd

DL-Lactic acid (Sodium salt) FW=112.1 Supplied by Sigma Aldrich Pty Ltd

Lysozyme Cat # 1585657 Supplied by Boehringer Mannheim Pty Ltd

Magnesium Chloride BDH MW=246.47 Merck Pty Ltd ACN 005 063 791 Supplied by Crown Scientific

Magnesium Sulphate BDH AnalaR MW=246.47 Product # 10151 Merck Pty Ltd ACN 005 063 791 Supplied by Crown Scientific 2-Mercaptoethanol (2-Hydroxyethylmercaptan; -mercaptoethanol) MW=78.13 Cat # M 6250 Supplied by Sigma Aldrich Pty Ltd

Methanol HPLC Grade BDH Merck Pty Ltd ACN 005 063 791 Supplied by Crown Scientific

Molecular weight standards Supplied by Bio-Rad Laboratories Pty Ltd

Petroleum Ether BDH AnalaR, Product # 10179 Merck Pty Ltd ACN 005 0630791 Supplied by Crown Scientific

Phenol Supplied by Novachem Pty Ltd

Polyvinylpirrolidine (Molecular Biology Reagent) MW=360,000 Supplied by Sigma Aldrich Pty Ltd

Potassium Acetate MW=98.24 Cat # P 1147 Supplied by Sigma Aldrich Pty Ltd

Potassium Hydroxide BDH MW=56.11 AnalaR, Product # 10210 Merck Pty Ltd ACN 005 0630791 Supplied by Crown Scientific

Potassium Chloride BDH MW=74.55, Merck (Laboratory Reagent) Product # 29594 Merck Pty Ltd ACN 005 0630791 Supplied by Crown Scientific

RNAse Ribonuclease I Bovine Pancreas Supplied by Amrad Pharmacia Biotech

Rotolytes Supplied by Bio-Rad Laboratories Pty Ltd

Sodium Acetate BDH AnalaR MW=82.03 Product # 10236 Merck Pty Ltd ACN 005 0630791 Supplied by Crown Scientific

Sodium Chloride BDH MW=58.44 AnalaR, Product #10241 Merck Pty Ltd ACN 005 0630791 Supplied by Crown Scientific

Sodium Hydroxide MW=40.00 Cat # 10236 Ajax Chemicals Supplied by Crown Scientific

Sodium Lactate MW=112.07 Supplied by Crown Scientific

Sodium Lauryl Sulphate (SDS) MW=288.4 Cat # L4509 Supplied by Sigma Aldrich Pty Ltd Sodium Citrate BDH MW=294.10 AnalaR, Product # 10242 Merck Pty Ltd ACN 005 0630791 Supplied by Crown Scientific

TEMED (N,N,N',N'-tetramethyl-ethylenediamine) MW=116.21 Supplied by Bio-Rad Laboratories Pty Ltd

Tris chloride MW=121.14 AnalaR, Product # 103157P Merck Pty Ltd ACN 005 0630791 Supplied by Crown Scientific

Tris chloride MW=157.61 Cat # 103130.18 Supplied by ICN Biochemicals Australasia Pty Ltd

Triton X-100 t-octylphenoxypolyethoxyethanol Cat # X 100 Supplied by Sigma Aldrich Pty Ltd

Tryptone Code L42 Oxoid Australia, Cat # LP042B Unipath Ltd, Basingstoke, Hapmshire, England Supplied by Oxoid Australia Pty Ltd

Yeast Extract Code L21, Unipath Ltd, Basingstoke, Hampshire, England Supplied by Oxoid Australia Pty Ltd

X-ray Film X AR 5 (43x35) Cat # 1651512 Supplied by Integrated Sciences

## Appendix 2

### List of Suppliers

Amersham Australia Pty Ltd	PO Box 99 North Ryde NSW 2113
Amrad Pharmacia Biotech	34 Wadhurst Dve Boronia Vic 3155
B. Braun Australia	10 Arcade Rd. Box Hill North, Vic 3129
Beckman Industries Australia Ltd	Unit 16, 170 Forester Rd Mt Waverly Vic 3149
Bio-Rad Laboratories Pty Ltd	PO Box 210, Regents Park NSW 2143
BOC Gases	90 Bell St, PO Box 43 Preston Vic 3072
Boehringer Mannheim Pty Ltd	353 Whitehorse Rd Nunawading Vic 3131
Bresatec ACN 070 039 913	PO Box 11 Rundle Mall Adelaide SA 5000
Chr. Hansen Pty Ltd	49 Barry St. Bayswater, Vic 3153
Clontech Laboratories	4 Research Park Drive, Macquarie University
	Research Park North Ryde NSW 2113
Crown Scientific Pty Ltd	PO Box 101 Burwood Vic 3125
CSR Yarraville Distillery	PO Box 83 Yarraville Vic 3013
Daintree Scientific	20 Kiama Pde. St Helens Tasmania
DAKO Australia Pty Ltd	Lord St Botany New South Wales
Genesearch Pty Ltd	14 Technology Drive Arundel, Qld 4214
ICN Biochemicals Australasia Pty Ltd	PO Box 187, Unit 12 167 Prospect Highway
	Seven Hills NSW 2147
Integrated Sciences	1401 Burke Rd East Kew Vic 3102
Kimberly Clark	PO Box 343 Milsons Point NSW 2061
Lab Supply	PO Box 688 Mulgrave Nth Vic 3170
Novachem Pty Ltd	50 Garden St South Yarra Vic 3141
Oxoid Australia Pty Ltd	104 Northern Rd Western Heidelberg Vic 3081
Pathtech Diagnostics	PO Box 175 Balwyn Vic 3103

D	
Progen industries Lta	2806 lpswich Rd Darra Qld 4076
Prossper Photographics	149 Nelson Place Williamstown Vic 3016
Ratek Instruments Australia	PO Box 131 Mitcham Vic 3155
Real Time Engineering	Level 1 15-19 Boundary St. Rushcutters Bay NSW
	2011
Selby Scientific	PO Box 276 Acacia Ridge Qld 4110
SGE Australia	7 Argent Place Ringwood Vic, 3134
Sigma Aldrich Pty Ltd and Sigma Immuno	Unit 2, 14 Anella Ave Castle Hill NSW 2154
Chemicals	
Silenus Labs Pty Ltd	34 Wadhurst Drive Boronia Vic 3155
SPL Diagnostics	1408 Centre Rd. Clayton, Vic 3168
Varian Australia Pty Ltd	679 Springvale Rd Mulgrave Vic 3170

Appendix 3 Pun File : c:\star\maria\maria042.run Method File : C:\STAR\MARIA.MTH Sample ID : Sue 6b 71514 Injection Date: 21-JAN-95 8:34 PM Calculation Date: 21-JAN-96 8:50 PM Operator : Maria Detector Type: ADCB (1 Volt) Bus Address : 16 Samplo Rate : 10.00 Hz Run Time : 24.002 min Whickstation: MS-DOS\_6 . Instrument : Varian Star #1 Channel : A = RI-4 1 volts \*\*\*\*\*\*\*\*\* Star Chromatography Software \*\*\*\*\*\*\*\*\* Version 4.01 \*\*\*\*\*\*\*\*\*\*\*\* 0.87 cm/min 0.000 min Attenuation = 800 End Time = 24.000 min Min / Tick = 2.00 .025 0.050 0.075 0.100 0.125 0.150 Chart Speed = cm/min c... Start Time = Г 10.000 valite - [ ] maltose 7.089  $\leq$ glucose <del>8:863</del> <W[=32.0 -11 lactate 12.471 formate 13.308 acetate 14.842 <W[=16.0 - [ ] ethanol 22.200

~

۰.

367

•

Appendix 4

INL Microbial Biast Results

http://genome.oml.gov/Blast/cgi-bin/blastrun.c

•	Computational Biology at ORIVI		_	
	Channel • Generation • Grail • GrailEXP • Pipeline • Parser • PF	OSPECT		
	ORNL Microbial Blast Results			
	TBLASTN 2.2.3 [May-13-2002]			
	Reference: Altschul, Stephen F., Thomas L. Madden, Alejandro A. S Jinghui Zhang, Zheng Zhang, Webb Miller, and David J. Lipman (199 "Gapped BLAST and PSI-BLAST: a new generation of protein database programs", Nucleic Acids Res. 25:3389-3402.	chaffer, 7), search		
	Query= (20 lotters)			
(	Jatabase: /auto/microbe/lcas/chromosome/1/data/lcas.fna 199 sequences; 2,618,854 total letters			
	Searching.done	Score	F	
	Sequences producing significant alignments:	(bits) Va	lue	
	Scaffold3 Scaffold7	41 21	1e-05 9.9	
	>Scaffold3 Length = 82578			
	Score = 40.8 bits (94), Expect = 1e-05 Identities = 20/20 (100%), Positives = 20/20 (100%) Frame = -2			
	Query: 1 DTSDSIA3NKSETNALLKQI 20 DTSDSIASNKSETNALLKQI Sbjct: 9560 DTSDSIASNKSETNALLKQI 9501			
(	Scaffold7 Length = 75568			
	Score = 21.2 bits (43), Expect = 9.9 Identities = 10/16 (62%), Positives = 12/16 (74%) Frame = -2			
	Query: 2 TSDSIASNKSETNALL 17 +SD I +NKS T ALL			
	Sbjct: 24543 SSDMIRTNKSATYALL 24496			
	Database: /auto/microbe/lcas/chromosome/1/data/lcas.fna Posted date: Oct 7, 2002 5:30 PM Number of letters in database: 2,618,854 Number of sequences in database: 199			
	Lambda K H 0.293 0.110 0.252			
	Gapped			

INL Microbial Blast Results		http://genome.ornl.gov/Blast/	'egi-bin/blastrun.e
Matrix: BLOSUM62 Gap Penalties: Existence: Number of Hits to DB: 21, Number of Sequences: 199 Number of sequences: 199 Number of successful exter Number of sequences bette: Number of HSP's better the Number of HSP's better the Number of HSP's successful Number of HSP's successful Number of HSP's gapped (no length of query: 20 length of database: 872,9 effective HSP length: 0 effective length of query effective length of datab effective search space: 2 effective search space: 2 effective search space us frameshift window, decay T: 13 A: 40 X1: 17 (7.2 bits) X2: 38 (14.6 bits) X3: 64 (24.7 bits) S1: 43 (21.3 bits) (-52: 43 (21.2 bits)	<pre>11, Extension: 1 771 nsions: 2 r than 10.0: 4 an 10.0 without gapping: 2 lly gapped in prelim test: 0 mpted gapping in prelim test on-prelim): 2 51 : 24 ase: 872,951 0950824 ed: 20950824 const: 50, 0.1</pre>	) z: 0	

.

ו• - -

17/09/02 (1:26

...

<u>Site Map</u> • <u>Feedback</u> • <u>Life Sciences Division</u> • <u>ORNL</u> • <u>Disclaimer</u> • <u>Webmaster</u>

?of2

Part of Lb paracsei 12006.2

# **Analysis Summary**

Nucleotide Similarities: blastn Result
 Function: blastx Result
 Gene Prediction: GenScan Result

## **Graphic Summary**

370

6/06/03

Appendix 5

Page 1 of 6

http://hiomanaper.angis.org.au/zope1/get/57600/1177474/mainFrame?view=graphic&frames=1&resetview=0





2011

: ) .'

http://biomanaget.angis.org.au/zope1/get/57600/1177474/mainFrame?view=graphic&frames=1&resetview=0

6/06/03

Part of Lb paracse: 12006.2

Details

blastn Result

4

Page 3 of 6

::9

Matching Entry	Boats Call			
	nuz-unao	Uescription	Score	IE Value
<u>AC122028</u>	[24-45]	Mus musculus chromosome 2 clone RP24-394D4, complete sequence.	44.1	0.47
AJ414145	[645-666]	Yersinia pestis strain CO92 complete genome; segment 5/20.	44.1	0.47
<u>AE013935</u>	[645-666]	Yersinia pestis KIM section 335 of 415 of the complete genome.	44.1	0.47
<u>BC031528</u>	[502-526]	Mus musculus, Similar to nuclear receptor coactivator 4, clone MGC:28270 IMAGE:4008986, mRNA, complete cds.	42.1	6.
UHO240996	[774-798]	uncultured Holophaga/Acidobacterium Sva0777 16S rRNA gene, partial.	42.1	1.9
AC016724	[151-171]	Homo sapiens BAC clone RP11-339F22 from 2, complete sequence.	42.1	6
AC016724	[502-526]	Homo sapiens BAC clone RP11-339F22 from 2, complete sequence.	42.1	1.9
AY075354	[858-877]	Drosophila melanogaster GH28809 full length cDNA.	40.1	7.4
AL831719	[24-43]	Mouse DNA sequence from clone RP23-16618 on chromosome 4, complete sequence.	40.1	7.4
<b>CNS07EFT</b>	[182-205]	Human DNA sequence chromosome 8 of Homo sapiens (human).	40.1	7.4
<u>AP003781</u>	[22-41]	Homo sapiens genomic DNA, chromosome 11q clone:RP11-49K4, complete sequences.	40.1	7.4
AP001699	[25-48]	Homo sapiens genomic DNA, chromosome 21q, section 43/105.	40.1	7.4
AP001604	[25-48]	Homo saplens genomic DNA, chromosome 21, clone:KB1648B8, APP-D21S292 Iregion, complete sequence.	40.1	7.4
AP000929	[22-41]	Homo saplens genomic DNA, chromosome 11q, clone:RP11-778B17, complete sequence.	40.1	7.4
DMC114E2	[858-877]	Drosophila melanogaster cosmid 114E2.	40.1	7.4
DMC114E2	[858-877]	Drosophila melanogaster cosmid 114E2.	40.1	7.4
<u>AE003436</u>	[182-201]	Drosophila melanogaster chromosome X, section 20 of 74 of the complete sequence.	40,1	7.4
<u>AE003428</u>	[858-877]	Drosophila melanogaster chromosome X, section 12 of 74 of the complete sequence.	40.1	4.7
<u>AC023720</u>	[182-201]	Drosophila melanogaster X BAC RP98-7F15 (Roswell Park Cancer Institute Drosophila BAC Library) complete sequence.	40.1	4
_				

:

http://biomanager.angis.org.au/zope1/get/57600/1177474/mainFrame?view=graphic&frames=1&resetview=0

;

6/06/03

Part of Lb paracsei 12006.2

	4 of 6
	Page
-	
-	
•	

- PULLE -

•

Contraction of the second s

----

Section 1

4.	4	4.	4	4	4	4	4	4	4	4	4	4	4	4	4
1	1	1	1	4		2		2	<u>L</u>	<u>}</u>			7		8 <u>-</u>
40.	40	40.	40.	40.	40.	40	40.1	40.1	40.1	40.1	40.1	40.1	40.1	40.1	40.1
Bacillus subtilis complete genome (section 12 of 21): from 2195541 to 2409220.	Bacillus subtilis phage SPB ATTL site.	B.subtills phage SPB ATT site.	Bifidobacterium longum NCC2705 section 187 of 202 of the complete genome.	Xanthomonas axonopodis pv. citri str. 306, section 354 of 469 of the complete genome.	Homo saplens chromosome 11p15.4 clone RPC11-610120 P2-containing olfactory receptor gene cluster, complete sequence.	Homo sapiens jerky gene product homolog mRNA, complete cds.	Homo sapiens BAC clone RP11-12K22 from 4, complete sequence.	Homo sapiens chromosome 8, clone RP11-25K19, complete sequence.	Homo sapiens chromosome 11, clone RP11-560B16, complete sequence.	Homo sapiens chromosome 8, clone RP11-108L23, complete sequence.	Homo sapiens 12 BAC RP11-316O1 (Roswell Park Cancer Institute Human BAC Library) complete sequence.	Homo sapiens 12 BAC RP11-230G5 (Roswell Park Cancer Institute Human BAC Library) complete sequence.	Homo sapiens PAC clone RP5-887P4 from 7p21-p22, complete sequence.	Bacteriophage SPBc2 complete genome.	Mus musculus, clone RP23-243H19, complete sequence.
[[16-35]	[16-35]	[16-35]	[928-955]	[384-403]	[714-733]	[22-41]	[25-44]	[182-205]	[714-733]	[182-205]	[24-43]	[24-43]	[665-684]	[16-35]	[893-912]
BSUB0012	BACPATTL	BACPATT	<u>AE014800</u>	<u>AE011976</u>	<u>AF321237</u>	<u>AF004715</u>	AC093730	AC087698	AC087280	AC083934	AC079948	AC025419	0 AC004538	AF020713	<u>AC119855</u>

+
Ξ
0
ã
×
17

blastx Result				·
Matching Entry	Begin-End	Description	Score	E Value
Q9AG98	[28-1251]	General stress protein GSP-781 (Glucan-binding protein B).	98.3	2e-19
Q938V0	[28-1251]	Glucan-binding protein B.	97.9	13e-19
Q938V3	[28-1251]	Glucan-binding protein B.	97.9	3e-19
Q938V2	[28-1251]	Glucan-binding protein B.	95.6	1e-18
Q938V1	[28-1251]	Glucan-binding protein B.	95.6	1e-18
Q97N55	[40-1251]	Secreted 45 kDa protein.	81.5	2e-14
Q9K2J9	[28-678]	Secreted antigen SagA.	81.5	2e-14
2			÷ .	

5

http://biomanager.angis.org.au/zope1/get/57600/1177474/mainFrame?view=graphic&frames=1&resetview=0

6/06/03

	•	
0.00	;	
	i	
		1000
		į

Part of Lb paracsei 12006.2 14 14

1 States

A Ballen ;

Ţ Page 5 of 6

. . . . .

1 19
ll 🖬
ŝB
В В
e'.
4
Ital
tat
0d
× III
terr
5 D
اچ ا
pot
ptid
IA.
ğ
op:
aA
ŝ
Š
106
22
od/
9
-
÷

httn://hiณฑภทตตะr สมขis ณฑ สแ/วณคะ1/get/57600/1177474/mainFrame?view≖ณฑphic&frames=1&resetview≖0

374

.

e " . .

**\***.. • .. .

6/06/03